COMPUTATIONAL TECHNIQUES IN FINITE SEMIGROUP THEORY

Wilf A. Wilson

A Thesis Submitted for the Degree of PhD at the University of St Andrews



2018

Full metadata for this item is available in St Andrews Research Repository at: <u>http://research-repository.st-andrews.ac.uk/</u>

Please use this identifier to cite or link to this item: <u>http://hdl.handle.net/10023/16521</u>

This item is protected by original copyright

Computational techniques in finite semigroup theory

WILF A. WILSON



This thesis is submitted in partial fulfilment for the degree of Doctor of Philosophy (PhD) at the University of St Andrews

November 2018

Declarations

Candidate's declarations

I, Wilf A. Wilson, do hereby certify that this thesis, submitted for the degree of PhD, which is approximately 64500 words in length, has been written by me, and that it is the record of work carried out by me, or principally by myself in collaboration with others as acknowledged, and that it has not been submitted in any previous application for any degree. I was admitted as a research student at the University of St Andrews in September 2014. I received funding from an organisation or institution and have acknowledged the funders in the full text of my thesis.

Supervisor's declaration

I hereby certify that the candidate has fulfilled the conditions of the Resolution and Regulations appropriate for the degree of PhD in the University of St Andrews and that the candidate is qualified to submit this thesis in application for that degree.

Permission for publication

In submitting this thesis to the University of St Andrews we understand that we are giving permission for it to be made available for use in accordance with the regulations of the University Library for the time being in force, subject to any copyright vested in the work not being affected thereby. We also understand, unless exempt by an award of an embargo as requested below, that the title and the abstract will be published, and that a copy of the work may be made and supplied to any bona fide library or research worker, that this thesis will be electronically accessible for personal or research use and that the library has the right to migrate this thesis into new electronic forms as required to ensure continued access to the thesis.

I, Wilf A. Wilson, confirm that my thesis does not contain any third-party material that requires copyright clearance.

The following is an agreed request by candidate and supervisor regarding the publication of this thesis: no embargo on print copy; no embargo on electronic copy.

Date: Signature of candidate:

Underpinning research data or digital outputs Candidate's declaration

I, Wilf A. Wilson, hereby certify that no requirements to deposit original research data or digital outputs apply to this thesis and that, where appropriate, secondary data used have been referenced in the full text of my thesis.

Acknowledgements

I looked forward to many aspects of life as a PhD student: the opportunities to travel to new places, the freedom to set my own schedule, and the independence to develop my mathematical interests. On these points, my expectations matched reality. There were also less-positive things that I did not expect, such as how my motivation could not be taken for granted. I can confirm that doing a PhD is not easy, and I would like to thank those of you who have helped me along the way: my friends, family, colleagues, and funders.

In my first year as a PhD student, my mother became ill, and died. That time, and the subsequent time for adjusting, has been very difficult. I regret that my PhD, ostensibly an exciting stage in my life, has coincided with such sadness. More than anything, I am grateful to have had, and to continue to have, the love and support of so many people, particularly my five siblings: Laura, Will, Poppy, Holly, and Alan. I'm so lucky to have you.

Thank you James D. Mitchell for supervising me, and for introducing me to computational semigroup theory back in the summer of 2012. It was the right topic for me at the right time, and I'm glad we've been able to work together since then. Thank you for your openness, for treating me with respect, and for your help along the way, especially with this thesis.

I'm also grateful to my examiners, Nik Ruškuc and Florent Hivert.

I'm very lucky to have shared my office with Michael Torpey over these years. There are few people with whom I've spent as much time as I have now spent with you, and so it's fortunate that you've been relentlessly funny, eccentric, friendly, and generous.

I'd also like to thank Daniel Billinge, for welcoming me into your home, for being my friend, and for being so kind and patient with me. You've left a big impression on me.

And thank you Nina da Rocha, for your support, and for reminding me how to say yes!

Funding

This work was supported by the Carnegie Trust for the Universities of Scotland, via a Carnegie PhD scholarship, and by the University of St Andrews (School of Mathematics and Statistics).

Abstract

A semigroup is simply a set with an associative binary operation; computational semigroup theory is the branch of mathematics concerned with developing techniques for computing with semigroups, as well as investigating semigroups with the help of computers. This thesis explores both sides of computational semigroup theory, across several topics, especially in the finite case.

The central focus of this thesis is computing and describing maximal subsemigroups of finite semigroups. A maximal subsemigroup of a semigroup is a proper subsemigroup that is contained in no other proper subsemigroup. We present novel and useful algorithms for computing the maximal subsemigroups of an arbitrary finite semigroup, building on the paper of Graham, Graham, and Rhodes from 1968. In certain cases, the algorithms reduce to computing maximal subgroups of finite groups, and analysing graphs that capture information about the regular \mathscr{J} -classes of a semigroup. We use the framework underpinning these algorithms to describe the maximal subsemigroups of many families of finite transformation and diagram monoids. This reproduces and greatly extends a large amount of existing work in the literature, and allows us to easily see the common features between these maximal subsemigroups.

This thesis is also concerned with direct products of semigroups, and with a special class of semigroups known as Rees 0-matrix semigroups. We extend known results concerning the generating sets of direct products of semigroups; in doing so, we propose techniques for computing relatively small generating sets for certain kinds of direct products. Additionally, we characterise several features of Rees 0-matrix semigroups in terms of their underlying semigroups and matrices, such as their Green's relations and generating sets, and whether they are inverse. In doing so, we suggest new methods for computing Rees 0-matrix semigroups.

Contents

Preface 1			15
1	Intr	oduction	19
	1.1	Sets, relations, and functions	19
	1.2	Graphs and digraphs	20
	1.3	Groups and semigroups	22
		1.3.1 Subsemigroups, ideals, and generation	24
		1.3.2 Rees 0-matrix semigroups and completely 0-simple semigroups	25
		1.3.3 Green's relations and regularity	26
		1.3.4 Group theory and group actions	29
		1.3.5 Semigroups of partial transformations	30
	1.4	Computational semigroup theory	30
		1.4.1 Algorithms and complexity	31
		1.4.2 Data structures for semigroups	33
		1.4.3 Finite semigroups specified by generating sets	34
2	Gen	nerating sets for direct products of semigroups	39
	2.1	Introduction	39
	2.2	Decomposable and indecomposable elements	42
	2.3	Computing indecomposable elements and non-trivial factorizations	43
		2.3.1 Corresponding features in the SEMIGROUPS package for GAP	47
	2.4	Arbitrary direct products of semigroups	47
	2.5	Two finitely generated surjective semigroups	54
3	Ree	s 0-matrix semigroups over a semigroup	59
	3.1	Introduction	59
	3.2	Preliminaries	61
		3.2.1 Isomorphisms and normalizations of Rees 0-matrix semigroups	61
		3.2.2 The idempotent generated subsemigroup of a Rees 0-matrix semigroup	64
	3.3	The Green's structure of a Rees 0-matrix semigroup	68
		3.3.1 Rees 0-matrix semigroups with row- or column-regular matrices	70
	3.4	Generating sets for Rees 0-matrix semigroups	76
		3.4.1 Decomposable and indecomposable elements	78
		3.4.2 Rees 0-matrix semigroups over monoids	79
		3.4.3 Rees 0-matrix semigroups with row- or column-regular matrices	80
	3.5	Special kinds of Rees 0-matrix semigroups	81
		3.5.1 Regular Rees 0-matrix semigroups	81
		3.5.2 Rees 0-matrix monoids and inverse monoids	83
		3.5.3 Inverse Rees 0-matrix semigroups	84
4	Con	nputing maximal subsemigroups of a finite semigroup	91
	4.1	Introduction	91
	4.2	The form of a maximal subsemigroup	93

	4.3	Finite regular Rees 0-matrix semigroups over groups	. 96
		4.3.1 Maximal subsemigroups of types (R1) and (R2) $\ldots \ldots \ldots \ldots$. 98
		4.3.2 Maximal subsemigroups of types $(R3)$ and $(R4)$. 98
		4.3.3 Maximal subsemigroups of type $(R5)$. 100
		4.3.4 Maximal subsemigroups of type $(R6) \ldots \ldots \ldots \ldots \ldots \ldots$. 101
		4.3.5 An example	. 108
		4.3.6 Maximal subsemigroups of type (R6) that contain a given set	. 110
	4.4	Arbitrary finite semigroups	. 115
		4.4.1 Maximal subsemigroups that intersect every \mathscr{H} -class: (M1)	. 115
		4.4.2 Graphs and digraphs for regular \mathscr{J} -classes	. 118
		4.4.3 Maximal subsemigroups that are unions of \mathscr{L} - and \mathscr{R} -classes: (M2) \ldots	. 122
		4.4.4 Maximal subsemigroups that are unions of \mathscr{L} - or \mathscr{R} -classes: (M3)–(M4)	. 125
		4.4.5 Maximal subsemigroups by removing a \mathcal{J} -class: (M5)	. 127
	4.5	Arbitrary finite monoids	. 129
		4.5.1 Maximal subsemigroups from the group of units	. 129
		4.5.2 Maximal subsemigroups from a \mathcal{J} -class covered by the group of units .	. 130
	4.6	The algorithm	. 135
		4.6.1 Relevant functionality in the SEMIGROUPS package for GAP	. 137
		4.6.2 Performance measurements	. 138
5	Max	ximal subsemigroups of finite transformation and diagram monoids	145
	5.1	Introduction	. 145
		5.1.1 Maximal subgroups of cyclic, dihedral, and symmetric groups	. 146
	5.2	Partial transformation monoids	. 147
		5.2.1 Definitions	. 147
		5.2.2 Preliminaries	. 149
		5.2.3 $\mathcal{PT}_n, \mathcal{T}_n, \text{ and } \mathcal{I}_n \dots \dots$. 151
		5.2.4 \mathcal{PO}_n and \mathcal{POD}_n	. 153
		5.2.5 \mathcal{O}_n and \mathcal{OD}_n	. 156
		5.2.6 \mathcal{POI}_n and \mathcal{PODI}_n	. 159
		5.2.7 $\mathcal{POP}_n, \mathcal{POR}_n, \mathcal{OP}_n, \text{ and } \mathcal{OR}_n \dots \dots \dots \dots \dots \dots \dots \dots$. 161
		5.2.8 \mathcal{POPI}_n and \mathcal{PORI}_n	. 163
	5.3	Diagram monoids	. 165
		5.3.1 Definitions \ldots	. 165
		5.3.2 Preliminaries	. 167
		5.3.3 The partition monoid \mathcal{P}_n	. 169
		5.3.4 The Brauer monoid \mathcal{B}_n and the uniform block bijection monoid \mathfrak{F}_n	. 170
		5.3.5 The partial Brauer monoid \mathcal{PB}_n	. 171
		5.3.6 The dual symmetric inverse monoid \mathcal{I}_n^*	. 172
		5.3.7 The Jones monoid \mathcal{J}_n and the annular Jones monoid $\mathcal{A}\mathcal{J}_n$. 173
		5.3.8 The Motzkin monoid \mathcal{M}_n	. 176
	5.4	Table of results	. 179
A	Оре	en problems	181
Bi	ibliog	graphy	183
Ta	able o	of notation	191
In	dex		195

List of algorithms

1.17	Test membership in a list 31
1.21	Naively compute the right Cayley digraph of a finite semigroup
2.11	Indecomposable elements in a finite semigroup via the Green's structure 44
2.12	Non-trivial factorizations and indecomposable elements from a Cayley digraph . 45
2.13	Non-trivial factorizations and indecomposable elements using techniques from [37] 46
3 1 2	Compute the idempotent generated subsemigroup of $\mathscr{M}^0[C: I \land P]$ 67
0.12	$ \begin{array}{c} \begin{array}{c} compute the identifyed by compared by the identifyed by the identified by the identifyed by the identified by the identifie$
3.31	Compute whether a finite Rees 0-matrix semigroup has a row-regular matrix \therefore 74
3.67	Compute whether a finite Rees 0-matrix semigroup is inverse
4.16	Maximal subsemigroups of a regular Rees 0-matrix semigroup that have type $(R_3)100$
4 29	Maximal subcomissions of a normal P_{opt} of matrix comissions that have two $(P6)107$
4.32	Maximal subsemigroups of a regular field of matrix semigroup that have type (K0)107
4.44	Maximal subsemigroups of type (R6) that contain a given set
4.86	Maximal subsemigroups of an arbitrary finite semigroup

List of figures

1.1 1.2 1.9 1.20	An example of a graph with 7 vertices and 6 edges21An example of a digraph with 7 vertices and 10 edges21Containment of the Green's relations on a semigroup27A right Cayley digraph of the semigroup with four elements defined in Table 1.1935
3.10	The Graham-Houghton graph and spanning forest from Example 3.9
4.15	The Graham-Houghton graph from Example 4.14
4.37	The Graham-Houghton graph of the semigroup from Section 4.3.5
4.53	The digraphs $\Gamma_{\mathscr{L}}(S, J)$ and $\Gamma_{\mathscr{R}}(S, J)$ from Example 4.52
4.54	The graph $\Delta(S, J)$ from Example 4.52
4.55	The graph $\Theta(S, J)$ from Example 4.52
4.89	Experiment: computing the maximal subsemigroups of \mathcal{T}_n
4.90	Experiment: computing the maximal subsemigroups of \mathcal{PORI}_n
4.91	Experiment: computing the maximal subsemigroups of \mathcal{J}_n
4.92	Experiment: computing maximal subsemigroups of random subsemigroups of \mathcal{T}_9 143
4.93	Experiment: computing maximal subsemigroups of Rees 0-matrix semigroups $\ . \ . \ 144$
5.9	The graph $\Delta(\mathcal{PT}_n)$
5.10	The graph $\Delta(\mathcal{T}_n)$
5.11	The graph $\Delta(\mathcal{I}_n)$
5.15	The graph $\Delta(\mathcal{PO}_n)$
5.16	The graph $\Delta(\mathcal{POD}_n)$, when n is odd
5.17	The graph $\Delta(\mathcal{POD}_n)$, when <i>n</i> is even
5.20	The graph $\Delta(\mathcal{O}_n)$
5.21	The graph $\Delta(\mathcal{OD}_n)$, when <i>n</i> is odd
5.22	The graph $\Delta(\mathcal{OD}_n)$, when <i>n</i> is even
5.41	Containment of the diagram monoids from Section 5.3 $\ldots \ldots \ldots$
5.46	The graph $\Delta(\mathcal{P}_n)$
5.52	The graph $\Delta(\mathcal{J}_n)$

List of tables

1.19	The multiplication table of a semigroup with four elements
4.88	Information about 100 subsemigroups of \mathcal{T}_9 analysed in Figure 4.92
5.3	Composition of order/orientation-preserving/reversing partial transformations . . 147
5.8	The numbers of maximal subsemigroups of \mathcal{PT}_n , \mathcal{T}_n , and \mathcal{I}_n
5.14	The numbers of maximal subsemigroups of \mathcal{PO}_n and \mathcal{POD}_n
5.28	The numbers of maximal subsemigroups of \mathcal{O}_n and $\mathcal{OD}_n \dots \dots$
5.31	The numbers of maximal subsemigroups of \mathcal{POI}_n and \mathcal{PODI}_n
5.34	The numbers of maximal subsemigroups of \mathcal{POP}_n , \mathcal{OP}_n , \mathcal{POR}_n , and \mathcal{OR}_n 163
5.37	The numbers of maximal subsemigroups of \mathcal{POPI}_n and \mathcal{PORI}_n
5.51	The numbers of maximal subsemigroups of \mathcal{P}_n , \mathcal{PB}_n , \mathcal{B}_n , \mathfrak{F}_n , and \mathcal{I}_n^*
5.59	The numbers of maximal subsemigroups of \mathcal{J}_n and $\mathcal{A}\mathcal{J}_n$
5.61	The number of maximal subsemigroups of \mathcal{M}_n
5.62	A selection of the results of Chapter 5

Preface

The overarching theme of this thesis is *computational semigroup theory*, which includes both the study of semigroups via computers, and the development of the theoretical results and tools that facilitate this study. These aspects reinforce each other: the development of better tools enables a more diverse range of computations, whilst experimentation with computers assists the identification of further advances in semigroup theory. This thesis contains elements of both of these aspects of computational semigroup theory.

The use of computers is well-established in many branches of mathematics, especially in applied mathematics and statistics. Software can be employed to solve equations, model systems, perform simulations, and analyse data. This should not be surprising, since computers are inherently mathematical machines. The fundamental utility of computers is that they perform calculations that are far too difficult or time-consuming for a human to perform. Moreover, given correct programming, computers are essentially guaranteed to give correct results.

In addition, computers are increasingly being used for pure mathematical research, in areas such as semigroup theory. For example, the SEMIGROUPS [101] and DIGRAPHS [10] packages for GAP [58] can be used to visualise particular features of a finite semigroup, such as the egg-box diagram of a \mathscr{D} -class, the Hasse diagram of its partial order of \mathscr{J} -classes, or a picture of its Cayley digraphs. For those researchers, teachers, or students who 'think in pictures', as in [64, p. xii], these kinds of visualisations are especially valuable.

Crucially, mathematical software for semigroups allows a researcher to construct and analyse examples of semigroups in far greater number, and of far greater complexity, than would be possible to consider otherwise. Software can be used in this way to produce instructive examples to include in a paper or talk, for instance, or to demonstrate concepts to a collaborator or student. By performing computational experiments on a wide range of examples, a mathematician can learn about the properties of various semigroups, and perhaps spot patterns and develop conjectures, or find counterexamples to pre-existing conjectures. Several results in Chapter 5 were first observed in this way. This kind of computational experimentation is particularly powerful when conducted in combination with pre-computed libraries of data. For example, the SMALLSEMI [32] package for GAP contains information about every semigroup (up to isomorphism and anti-isomorphism) containing at most 8 elements, and can be used to exhaustively search through these semigroups. Similar applications exist in many other branches of pure mathematics.

When computing with semigroups, there is often an 'obvious' way of computing a property, but this is rarely the most appropriate way. Roughly speaking, one of the main aims of theoretical research in computational semigroup theory is to exploit the structural properties of a semigroup to reformulate a certain problem in terms of features that are more feasible to compute. As a very straightforward example, consider the problem of deciding whether a finite semigroup is commutative. By definition, a semigroup S is commutative if and only if xy = yx for all $x, y \in S$. It follows that an algorithm that directly applies the definition to test the commutativity of a finite semigroup S would perform |S|(|S| - 1) multiplications in the worst case. However, a semigroup is commutative if and only if its generators commute, and a semigroup S is often defined by a generating set X that contains far fewer elements than S. In such cases, the commutativity of S can be tested by performing at most |X|(|X| - 1) multiplications, which can be accomplished much more quickly.

Computational semigroup theory is closely related to *computational group theory*, which is defined analogously. This is a much more mature area of research: many books have been published on the topic of computational group theory in the last few decades [73, 118, 119], whereas there are none, as yet, on the topic of computational semigroup theory. Since every group is a semigroup, it might appear on first glance that computational semigroup theory encompasses computational group theory. However, as with the more general branches of group theory and semigroup theory themselves, these topics are distinct, although linked. In semigroup theory, we consider a particular problem to be solved if it can be reduced to a problem in group theory. However, many other problems in semigroup theory turn out to be solved via a reduction to a combinatorial problem, or to a graph-theoretic problem. We take the same approach when computing with semigroups: for example, the techniques that we present in Chapter 4 for computing maximal subsemigroups reduce, in some cases, to computing maximal subgroups in groups, or to computing maximal cliques in graphs. In this thesis, we regard algorithms in computational group theory as being given, and we do not concern ourselves with their inner workings.

Some of the earliest research on the topic of computing with semigroups was that of For sythe [53], who enumerated the 126 semigroups of order 4 with the help of a computer, followed by that of Cannon [18, 19] and Perrot [106]. In the subsequent decades, researchers have continued to study semigroups with the use of computers and study techniques for computing with semigroups. This has led to the development and publication of a range of software, with diverse approaches and foci. There are numerous stand-alone pieces of software for computing with semigroups, such as AUTOMATE [21]; SEMIGROUPE [107] (which implements the techniques described in [54]); LIBSEMIGROUPS [102]; and SGPWIN [99]. The major computer algebra systems, like GAP [58], MAGMA [13], and SAGEMATH [122], include some functionality for semigroups, which is supplemented by specialised packages such as the NU-MERICALSGPS [24], KBMAG [72], SMALLSEMI [32], and SEMIGROUPS [101] packages for GAP, for instance. Computer code implementing some of the ideas of the second, third, and fourth chapters of this thesis is available in the current version, or in an upcoming version, of the SEMIGROUPS package for GAP; most of the examples contained in this thesis were constructed by the use of the SEMIGROUPS package; and many of the ideas presented herein arose through experimentation with this software.

We briefly summarise the contents of this thesis. In Chapter 1, we give the necessary definitions, notions, and preliminary results that are required in the thesis.

In Chapter 2, we consider generating sets for direct products of semigroups from the perspective of computation, building on several results from the literature. More specifically, we describe generating sets, for arbitrary direct products of semigroups, that do not necessarily contain every element of the direct product. We also discuss techniques for directly computing generating sets of this kind in the case that the direct product is finitely generated and the factors are either finite or finitely presented. The crucial step in this procedure is an algorithm to find a non-trivial factorization of an element over some generating set, or to prove that none exists. The main ideas presented in this chapter, in the case of direct products of finite semigroups, have been implemented in the development version of the SEMIGROUPS [101] package for GAP [58], and will be included in an upcoming released version.

In Chapter 3, we present results relating to the computation of Rees 0-matrix semigroups, particularly those that are finite. There are well-established techniques for computing with finite regular Rees 0-matrix semigroups over groups. However, a Rees 0-matrix semigroup may be defined over an arbitrary semigroup, and there is far less research on the topic of computing with Rees 0-matrix semigroups over semigroups that are not groups. We characterise certain

features and properties Rees 0-matrix semigroups in terms of their matrices and underlying semigroups. Given the ability to compute with the underlying semigroups, we discuss how these characterisations may be used to compute whether a Rees 0-matrix semigroup is inverse or regular, or to determine generating sets or the Green's structure in certain cases.

We focus on maximal subsemigroups of finite semigroups in the remainder of the thesis. Much of the research detailed in these chapters has been published in two research papers [35,45] in *Journal of Algebra*, in collaboration with several co-authors, but is more fully exposited here.

In Chapter 4, we put forth an algorithm for computing maximal subsemigroups in an arbitrary finite semigroup, building on the description of maximal subsemigroups given by Graham, Graham, and Rhodes in [61]. In order to do so, we first describe algorithms for computing maximal subsemigroups of an arbitrary finite regular Rees 0-matrix semigroup over a group. The algorithms described in this chapter are fully implemented in the SEMIGROUPS package for GAP. We also deduce theoretical results from the general solution that allow us to describe certain maximal subsemigroups of finite monoids.

In Chapter 5, we exploit the techniques described in Chapter 4 in order to describe and count the maximal subsemigroups of a wide range of families of finite transformation and diagram monoids. In particular, we describe the maximal subsemigroups of many monoids of order- or orientation-preserving or -reversing partial transformations, along with the maximal subsemigroups of the Motzkin, Brauer, Jones and partition monoids, and several further monoids. This work unifies and greatly extends many results in the literature concerning the maximal subsemigroups of particular transformation semigroups. Many of the results presented in Chapter 5 arose through experimentation with the SEMIGROUPS package for GAP.

Several open problems are posed throughout the main text of this thesis; they are collected in Appendix A for easy reference. In order to aid the reader, a table of notation is included at the end of this thesis on page 191, along with an index on page 195. In the original digital version of this document, many instances of notation serve as hyperlinks to their corresponding entries in the table of notation.

Chapter 1

Introduction

The purpose of this chapter is to introduce the standard ideas, definitions, and results that are used throughout this thesis. We introduce additional and more specialised notions in situ when it is more appropriate to do so.

1.1 Sets, relations, and functions

A partition \mathcal{P} of a set X is a set of non-empty pairwise-disjoint subsets of X, called parts, whose union is X. If \mathcal{P} is a partition of a set X, then a *transversal* of \mathcal{P} is a subset of X that contains exactly one element from each part of \mathcal{P} .

The Cartesian product of an ordered list of sets X_1, X_2, \ldots, X_n is the set of n-tuples

 $X_1 \times X_2 \times \dots \times X_n = \{ (x_1, x_2, \dots, x_n) : x_i \in X_i \text{ for all } i \in \{1, \dots, n\} \}.$

A relation is just a subset of the Cartesian product $X \times Y$, for some sets X and Y. Let $\rho \subseteq X \times Y$ be a relation. If $x \in X$ and $y \in Y$, then we often write $x \rho y$ to denote that $(x, y) \in \rho$; this is particularly common for partial orders, and for the Green's relations on a semigroup, which are defined later. The *inverse* of ρ , denoted ρ^{-1} , is the relation $\{(y, x) : (x, y) \in \rho\} \subseteq Y \times X$. If $\tau \subseteq Y \times Z$ is a relation, then the *composition* of ρ and τ is the relation

 $\rho \circ \tau = \{ (x, z) \in X \times Z : (x, y) \in \rho \text{ and } (y, z) \in \tau \text{ for some } y \in Y \}.$

A relation on a set X is a subset of $X \times X$. Let X be a set and let ρ be a relation on X. Then ρ is reflexive if $\{(x, x) : x \in X\} \subseteq \rho$, ρ is symmetric if $\rho = \rho^{-1}$, ρ is antisymmetric if $\rho \cap \rho^{-1} \subseteq \{(x, x) : x \in X\}$, and ρ is transitive if $\rho \circ \rho \subseteq \rho$.

An equivalence relation, or an equivalence, is a relation on a set that is reflexive, symmetric, and transitive. If ρ is an equivalence on the set X, then the equivalence class of a point $x \in X$ is the set $\{y \in X : (x, y) \in \rho\}$. The equivalence classes of an equivalence relation on a set form a partition of that set. If $Y \subseteq X$ is a union of equivalence classes of some equivalence ρ on X, then we denote the set of these classes by Y/ρ .

A partial order on a set is a relation on the set that is reflexive, antisymmetric, and transitive. A partially-ordered set is a set with a partial order; unless otherwise specified, the partial order is denoted by \leq . When using the symbol \leq for a partial order, as usual we write x < y to denote that $x \leq y$ and $x \neq y$. Let X be a partially-ordered set, and let $x, y \in X$. We say that x is less than y, and that y is greater than x, if x < y. A point $x \in X$ is maximal if $x \leq y$ implies that x = y, for all $y \in X$, and x is minimal if $y \leq x$ implies that x = y, for all $y \in X$. A partially-ordered set may contain multiple maximal or minimal elements, or, when the set is infinite, possibly none. If x < y and there is no element $z \in X$ such that x < z < y, then we say that x is covered by y in the partial order. In this thesis, |X| denotes the cardinality of the set X, and \emptyset denotes the empty set; \mathbb{Z} denotes the integers, $\mathbb{N} = \{1, 2, 3, \ldots\}$ is the set of natural numbers, and \mathbb{N}_0 is the set of non-negative integers $\mathbb{N} \cup \{0\} = \{0, 1, 2, 3, \ldots\}$. For a real number x, $\lceil x \rceil$ denotes the least integer greater than or equal to x, and |x| denotes the greatest integer less than or equal to x.

Let X and Y be sets. A partial function $f: X \longrightarrow Y$ is a subset of the Cartesian product $X \times Y$ where, for any $x \in X$, there is at most one $y \in Y$ such that $(x, y) \in f$. In particular, a partial function is a relation. The *domain* of a partial function $f: X \longrightarrow Y$ is the set

$$\operatorname{dom}(f) = \left\{ x \in X : (x, y) \in f \text{ for some } y \in Y \right\} \subseteq X.$$

If $x \in \text{dom}(f)$, then the *image of* x under f is the unique $y \in Y$ such that $(x, y) \in f$. We most commonly write a partial function to the right of its arguments, so that (x)f, or simply xf, denotes the image of x under f. The *image* of f is the set

$$\operatorname{im}(f) = \{xf : x \in \operatorname{dom}(f)\} \subseteq Y,$$

and the *kernel* of f is the equivalence on dom(f) given by

$$\ker(f) = \{ (x_1, x_2) \in \operatorname{dom}(f) \times \operatorname{dom}(f) : x_1 f = x_2 f \}.$$

A partial function $f : X \longrightarrow Y$ is *injective* if its inverse relation f^{-1} defines a partial function $Y \longrightarrow X$, which we call the *inverse* of f. A partial function $f : X \longrightarrow Y$ is *surjective* if im(f) = Y, and f is a *function* if dom(f) = X. Another name for a function is an *operation*. A function is *bijective*, and is called a *bijection*, when it is both injective and surjective.

Let $f: X \longrightarrow Y$ and $g: Y \longrightarrow Z$ be partial functions. The *composition* of f and g, denoted $f \circ g$, or more usually fg, is the partial function $X \longrightarrow Z$ given by the composition of f and g as relations. In particular, if f and g are functions and $x \in X$, then (x)fg = (xf)g. Note that we compose functions from left to right.

A partial transformation is a partial function $\{1, \ldots, n\} \longrightarrow \{1, \ldots, n\}$, for some $n \in \mathbb{N}$. The degree of a partial transformation $\{1, \ldots, n\} \longrightarrow \{1, \ldots, n\}$ is n. If α is a partial transformation of degree n, then we define rank (α) , the rank of α , to be $|im(\alpha)|$. A transformation is a partial transformation, and a permutation is a bijective transformation. A partial permutation is an injective partial transformation.

We may write a partial transformation of degree n in *two-line notation*. This uses a $2 \times n$ matrix, where the i^{th} entry in the first row contains the number i, and the i^{th} entry in the second row contains if when $i \in \text{dom}(f)$, otherwise it contains a dash. For example, if f is the partial transformation of degree 8 whose domain is $\{2, 4, 6, 8\}$, and where if = i/2 for any $i \in \text{dom}(f)$, then f is written in two-line notation as

$$f = \begin{pmatrix} 1 & 2 & 3 & 4 & 5 & 6 & 7 & 8 \\ - & 1 & - & 2 & - & 3 & - & 4 \end{pmatrix}.$$

Furthermore, we often write permutations in *disjoint cycle notation*; see [77, Chapter 1] for a definition of this notation, and for the standard terminology relating to permutations.

1.2 Graphs and digraphs

A graph is a pair (V, E), where V is any set of vertices, and E is a set of 2-subsets of V, called edges. Let $\Gamma = (V, E)$ be a graph, and let $u, v \in V$ be distinct vertices of Γ . We say that u is adjacent to v, and vice versa, and that u and v are adjacent, when $\{u, v\} \in E$. A vertex u is a *neighbour* of a vertex v if u and v are adjacent. The *degree* of a vertex is the number of neighbours that it has; a vertex in a graph with degree 0 is *isolated*.

A clique of a graph is a set of mutually adjacent vertices, and an *independent subset* is a set of mutually non-adjacent vertices. The cliques of a graph are partially-ordered by inclusion, as are the independent subsets. Thus a clique of a graph is *maximal* if it is properly contained in no clique of the graph and, similarly, an independent subset is *maximal* if it is properly contained in no independent subset of the graph. A graph is *bipartite* if its vertices can be partitioned into two independent subsets. A graph (V, E) is a *complete bipartite graph* if $\{U', V'\}$ is a partition of V, and $E = \{\{u, v\} : u \in U' \text{ and } v \in V'\}$.



Figure 1.1: Let $V = \{1, \ldots, 7\}$, and let $E = \{\{1, 2\}, \{2, 7\}, \{3, 7\}, \{5, 6\}, \{5, 7\}, \{6, 7\}\}$. This figure gives a visualisation of the graph (V, E), which has seven vertices and six edges. The set of neighbours of the vertex 7 is $\{2, 3, 5, 6\}$, and so it has degree 4. The vertex 4 is isolated. The pair $\{5, 6\}$ is a clique, however it is not a maximal clique, since it is properly contained in the maximal clique $\{5, 6, 7\}$. The set $\{1, 3, 4, 5\}$ is a maximal independent subset.

A digraph is a pair (V, E), consisting of a set of vertices V, and a set of edges $E \subseteq V \times V$. In other words, the set of edges of a digraph is a relation on the set of vertices. Let (u, v) be an edge of a digraph. Then v is an *out-neighbour* of u, and u is an *in-neighbour* of v, and the edge is a *loop* if u = v. A sink is a vertex in a digraph with no out-neighbours, and a source is a vertex in a digraph with no in-neighbours. A colouring of a graph or digraph (V, E) is simply a function $V \longrightarrow \mathbb{Z}$.



Figure 1.2: Let $E = \{(1, 6), (2, 2), (2, 7), (2, 4), (4, 7), (5, 6), (5, 7), (6, 1), (6, 7), (7, 1)\}$, and let $\Gamma = (\{1, \ldots, 7\}, E)$. Then Γ is a digraph with seven vertices and ten edges. The in-neighbours of the vertex 6 are 1 and 5, and the out-neighbours are 1 and 7. The vertices 3, 4, and 5 are sources of Γ , and 3 is the only sink. There is a single loop in Γ , at vertex 2.

A path in a graph (V, E) is a finite non-trivial sequence (v_0, \ldots, v_n) , for some $n \in \mathbb{N}$, where $v_i \in V$ for all $i \in \{0, 1, \ldots, n\}$, and $\{v_{i-1}, v_i\} \in E$ for all $i \in \{1, \ldots, n\}$. A path in a digraph is defined analogously. If (v_0, \ldots, v_n) is a path in a graph or digraph, then the path is said

to be a path from v_0 to v_n . A vertex v is reachable from a vertex u in a graph or digraph Γ if either u = v, or there exists a path from u to v in Γ . A cycle is a path from a vertex to itself. The length of the path or cycle (v_0, \ldots, v_n) is n.

Two vertices of a graph are contained in the same *connected component* of the graph if one is reachable from the other, and two vertices of a digraph are contained in the same *strongly connected component* if one is reachable from the other, and vice versa. The vertex set of the graph or digraph is partitioned by its connected components or strongly connected components, respectively. A graph is *connected* if its vertices form a single connected component, and a digraph is *strongly connected* if it has a single strongly connected component. A graph without cycles is called a *forest*, and a connected forest is called a *tree*. A digraph without cycles is *acyclic*. Let $\Gamma = (V, E)$ be a graph. A *spanning forest* of Γ is any forest $\Gamma' = (V, E')$, such that $E' \subseteq E$ and such that the connected components of Γ' coincide with those of Γ . Any spanning forest of Γ has |V| - n edges, where n is the number of connected components of Γ . A *spanning tree* is a spanning forest of a connected graph.

Example 1.3. Let $\Gamma = (V, E)$ be the graph defined in Figure 1.1. Then (1, 2, 7, 5, 6) is a path from 1 to 6 in Γ , and the path (5, 6, 7, 5) is a cycle in Γ . No vertex other than 4 itself is reachable from the vertex 4. The two connected components of Γ are $\{4\}$ and $V \setminus \{4\}$. If $E' = E \setminus \{\{5, 6\}\}$, then the graph (V, E') is a spanning forest of Γ .

Example 1.4. Let $\Delta = (V, E)$ be the digraph defined in Figure 1.2. Then (4, 2, 7, 1, 6) is a path in Δ from 4 to 6, and so 6 is reachable from 4 in Δ . However, 4 is not reachable from 6, and so 4 and 6 are contained in different strongly connected components of Δ . The strongly connected components of Δ are $\{1, 6, 7\}$, and the singletons $\{2\}, \{3\}, \{4\}, \text{ and } \{5\}$. The paths (2, 2) and (1, 6, 1) are cycles, and so Δ is not acyclic.

Let $\Gamma = (V, E)$ be a graph. The *complement* of Γ is the graph with vertices V and edges

$$\{\{u, v\} : u, v \in V, u \neq v, \text{ and } \{u, v\} \notin E\}.$$

Thus a clique of a graph is an independent subset of the complement, and an independent subset is a clique of the complement. If $V' \subseteq V$, then the *induced subgraph* of Γ on V' is the graph $(V', \{\{u, v\} \in E : u, v \in V'\})$. If \mathcal{P} is a partition of V, then the *quotient of* Γ by \mathcal{P} is the graph with vertex set \mathcal{P} , where two distinct parts $A, B \in \mathcal{P}$ are adjacent in the quotient if and only if there exists $u \in A$ and $v \in B$ such that $\{u, v\} \in E$. An *isomorphism* from a graph (V, E) to a graph (V', E') is a bijection $\phi : V \longrightarrow V'$, where $\{u, v\} \in E$ if and only if $\{u\phi, v\phi\} \in E'$, for all $u, v \in V$. The *complement*, *induced subdigraph*, and *quotient* of a digraph, and isomorphisms between digraphs, are defined analogously.

1.3 Groups and semigroups

In this section, we introduce the notation, definitions, and terminology from semigroup theory and group theory that we require in this thesis. We predominantly follow the notation of [76], which is prevalent in the literature. Throughout this thesis, we tend to refer the reader to [76, 110] for proofs of standard results, although there are many other well-regarded introductory books on semigroup theory, such as [22, 67, 71].

A semigroup is a set S with an associative operation $S \times S \longrightarrow S$. The order of a semigroup is its cardinality. Examples of semigroups include the real numbers \mathbb{R} with their usual multiplication, the set of all $n \times n$ (for some $n \in \mathbb{N}$) matrices over a semiring with matrix multiplication, and the set of all relations on a set with composition of relations. Unless otherwise stated, the associative operation is denoted by juxtaposition; more precisely, for elements $x, y \in S$, the juxtaposition xy of x and y denotes the image of the pair (x, y) under the operation. We often use the terminology of multiplication for such images. For example, xy is often referred to as the *product* of the elements x and y, and a subset A of S is closed under multiplication if $ab \in A$ for all $a, b \in A$. A semigroup S is commutative if xy = yxfor all $x, y \in S$. Associativity is the property that

$$(xy)z = x(yz)$$
 for all $x, y, z \in S$

Because a semigroup is associative, we may unambiguously form the product of any finite number of elements in a semigroup. If $x \in S$ and $k \in \mathbb{N}$, then x^k denotes the product $\underline{x \cdots x}$.

A homomorphism between semigroups S and T is a function $S \longrightarrow T$ that preserves the multiplication of S. In other words, a homomorphism is a function $\phi: S \longrightarrow T$ that satisfies $(xy)\phi = (x\phi)(y\phi)$ for all $x, y \in S$. An anti-homomorphism is a function that reverses the multiplication of S. An isomorphism of semigroups is a bijective homomorphism, and two semigroups S and T are isomorphic if there is an isomorphism between them; in this case, we write $S \cong T$. An anti-isomorphism is a bijective anti-homomorphism. Isomorphic semigroups have the same semigroup-theoretic properties; anti-isomorphic semigroups are left-right duals of each other. An injective homomorphism is called an embedding; if S and T and semigroups and $\phi: S \longrightarrow T$ is an embedding, then $S \cong \operatorname{im}\phi$.

An equivalence ρ on a semigroup S is a *left congruence* if it is compatible with left multiplication (i.e. $(sx, sy) \in \rho$ for all $(x, y) \in \rho$ and $s \in S$), a *right congruence* if it is compatible with right multiplication, and a *congruence* if it is both a left congruence and a right congruence. Given a semigroup S and a congruence ρ on S, we may form the *quotient semigroup* S/ρ of S by ρ . This consists of the equivalence classes of ρ , where the product of $A, B \in S/\rho$ is the equivalence class that contains ab, where $a \in A$ and $b \in B$ are arbitrary.

Let X be any set. The *free semigroup* over X, denoted X^+ , is the semigroup consisting of all non-empty sequences over X with the operation of concatenation. A *semigroup presentation* is a pair $\langle X | R \rangle$, where X is a set of *generators*, and R is a subset of $X^+ \times X^+$, called *relations*. The semigroup defined by a presentation $\langle X | R \rangle$ is the quotient X^+/ρ , where ρ is the least congruence on X^+ (with respect to containment) that contains R. When X and R are both finite, the semigroup defined by $\langle X | R \rangle$ is called a *finitely presented semigroup*.

We name many kinds of semigroups and semigroup elements; see [104] for a comprehensive description of special kinds of semigroup. An element x of a semigroup is *idempotent*, and is called *an idempotent*, if $x^2 = x$ (that is, *idempotent* is both an adjective and a noun). Every finite semigroup contains an idempotent, but the natural numbers with addition is an example of an infinite semigroup that contains no idempotents. The set of idempotents contained in a subset X of a semigroup is denoted by E(X). A semigroup in which every element is idempotent is called a *band*.

A multiplicative zero of a semigroup S, often just called a zero of S, is an element $0 \in S$ such that 0x = 0 = x0 for all $x \in S$. We use the notation 0_S to denote the multiplicative zero of the semigroup S, provided that it has one. A zero semigroup is a semigroup S with zero such that $xy = 0_S$ for all $x, y \in S$. A left identity of a semigroup S is an element $e \in S$ such that ex = x for all $x \in S$, and a right identity of S is defined analogously. A semigroup in which every element is a left identity is called a right zero semigroup, and a semigroup in which every element is a right identity. Note that multiplicative zeroes, and left and right identities, are idempotent, and that a semigroup contains at most one zero and at most one identity. Let S be a semigroup and let $x \in S$. If $s \in S$ satisfies sx = x, then s is a relative left identity for x,

and if xs = x, then s is a relative right identity for x. Such an element need not be idempotent.

If S is a semigroup, then we denote by S^0 the semigroup $S \cup \{0\}$, where 0 is a new element not contained in S, and the multiplication on S is extended to a multiplication on S^0 by defining 0x = 0 = x0 for all $x \in S^0$. In other words, 0 is a multiplicative zero for S^0 . We call S^0 the semigroup formed by adjoining a zero to S.

A semigroup that contains an identity is called a *monoid*. To any semigroup S we associate the monoid S^1 . If S is itself a monoid, then we define $S^1 = S$. On the other hand, if S does not contain an identity, then we adjoin a new element 1_S , and we extend the multiplication on S to a multiplication on $S^1 = S \cup \{1_S\}$ by defining $1_S x = x = x 1_S$ for all $x \in S^1$. For any semigroup S, we write 1_S to denote the identity of the monoid S^1 . Let S be a monoid. An element $u \in S$ is a *unit* of S if there exists some (necessarily unique) *inverse* element $u^{-1} \in S$ such that $uu^{-1} = 1_S = u^{-1}u$. A monoid in which every element is a unit is called a group. The subset of units in a monoid is a group, and is called the group of units of the monoid.

1.3.1 Subsemigroups, ideals, and generation

Let S be a semigroup, and let A and B be subsets of S. We denote the set product of A and B by $AB = \{ab : a \in A \text{ and } b \in B\}$. If $x \in S$, then xA denotes $\{x\}A$ and Ax denotes $A\{x\}$. This notation can be extended to the product of an arbitrary finite number of sets and elements in an obvious way. For example, if $A, B \subseteq S$ and $x \in S$, then AxB denotes the set $\{axb : a \in A, b \in B\}$. In this thesis, we say that a semigroup is surjective if $S = S^2$.

A subsemigroup of a semigroup is a subset that is closed under the same multiplication, i.e. a subset T such that $T^2 \subseteq T$. If U and V are subsets of a semigroup, then we write $U \leq V$ to denote that U is a subsemigroup of V. A submonoid of a semigroup is a subsemigroup that is itself a monoid, and a subgroup of a semigroup is any subsemigroup that is itself a group.

Let X be any subset of a semigroup S. The subsemigroup of S generated by X, denoted $\langle X \rangle$, is the intersection of all subsemigroups of S that contain X. Equivalently, $\langle X \rangle$ is the least subsemigroup of S, with respect to the partial order defined by containment, that contains X, and it consists of all finite products of elements in X. The set X is said to be a generating set for $\langle X \rangle$. If X_1, X_2, \ldots, X_m is any collection of subsets of S, and x_1, x_2, \ldots, x_n is any collection of elements of S, then we use the notation $\langle X_1, X_2, \ldots, X_m, x_1, x_2, \ldots, x_n \rangle$, or some rearrangement of this, for the subsemigroup of S generated by the union

$$X_1 \cup X_2 \cup \cdots \cup X_m \cup \{x_1, x_2, \ldots, x_n\}.$$

The generating sets of a semigroup are partially ordered by containment, and the cardinalities of the generating sets are also naturally ordered. A generating set is minimal with respect to containment if it does not properly contain another generating set, and minimal with respect to cardinality if its cardinality is least possible. The rank of a semigroup S, denoted rank(S), is the least cardinality of a generating set for S, and a semigroup S is finitely generated if it has a finite generating set, or equivalently, if rank $(S) \in \mathbb{N}_0$. A semigroup is monogenic if it has rank 1.

Let S be a semigroup and let $X \subseteq S$. We write F(X) for the subsemigroup $\langle E(X) \rangle$ generated by the idempotents in X. In particular, F(S) is the *idempotent generated subsemigroup* of S.

Let I be a subset of a semigroup S. Then I is a left ideal if $SI \subseteq I$, a right ideal if $IS \subseteq I$, and a two-sided ideal, more usually called simply an ideal, if it is both a left ideal and a right ideal. Certainly any kind of ideal is a subsemigroup. Let $x \in S$ be arbitrary. The principal left ideal generated by x is the left ideal $S^1x = Sx \cup \{x\}$, and the principal right ideal generated by x is the right ideal $xS^1 = xS \cup \{x\}$. The principal ideal generated by x is the ideal S^1xS^1 . If $X \subseteq S$, then the ideal of S generated by X is the intersection of all ideals of S that contain X; it is the union of the principal ideals $\{S^1xS^1 : x \in X\}$. The left and right ideals of a semigroup generated by a subset are defined analogously.

A subsemigroup or ideal of a semigroup is *proper* if it is not equal to the semigroup.

At various points throughout this thesis, we are concerned with describing generating sets for certain semigroups, and finding upper bounds on their ranks. The following lemma, and its subsequent corollary, will be useful in some of these cases.

Lemma 1.5. Let S be a semigroup and let $X \subseteq S^0$. The following hold:

(i) X generates S^0 if and only if $X = X' \cup \{0\}$, for some generating set X' of S.

Moreover, if X generates S^0 , then

- (ii) with respect to cardinality, X generates S^0 minimally if and only if $X \setminus \{0\}$ generates S minimally; and
- (iii) with respect to containment, X generates S^0 minimally if and only if $X \setminus \{0\}$ generates S minimally.

Proof. Suppose that X generates S^0 , and let $x \in S$. Then $x \in S^0 = \langle X \rangle$, and since $x \neq 0$, it follows that $x \in \langle X \setminus \{0\} \rangle$. Therefore $S = \langle X \setminus \{0\} \rangle$, and certainly $0 \in X$, since a product of non-zero elements of S^0 is contained in the subsemigroup S. Conversely, if $S = \langle X' \rangle$, then

$$S \lneq \langle X' \cup \{0\} \rangle \leq S^0 = S \cup \{0\},$$

and so $X' \cup \{0\}$ generates S^0 . Therefore (i) holds.

By (i), the generating sets of S are in an inclusion-preserving one-to-one correspondence with the generating sets of S^0 , via the addition or removal of 0. Therefore, it is clear that the minimality of generating sets is preserved under this correspondence, and so (ii) and (iii) hold.

Corollary 1.6. Let S be any semigroup. Then $rank(S^0) = rank(S) + 1$.

1.3.2 Rees 0-matrix semigroups and completely 0-simple semigroups

A semigroup with no proper ideals is called *simple*, and a semigroup S is completely simple if it is simple and there exists an idempotent $e \in E(S)$ such that $ef = fe = f \Rightarrow e = f$ for all $f \in E(S)$. A non-trivial semigroup with zero S is θ -simple if it is not a zero semigroup of order 2, and its only ideals are $\{0_S\}$ and S. A semigroup is completely θ -simple if it is 0-simple, and there exists a non-zero idempotent $e \in S$ such that, for all $f \in E(S)$,

$$ef = fe = f \neq 0 \Rightarrow e = f.$$

Every finite simple semigroup is completely simple, and every finite 0-simple semigroup is completely 0-simple, but in general these concepts are distinct.

Completely simple and completely 0-simple semigroups can be constructed as Rees matrix and Rees 0-matrix semigroups, respectively. Let T be any semigroup, let I and Λ be nonempty sets, and let $P = (p_{\lambda,i})_{\lambda \in \Lambda, i \in I}$ be a $\Lambda \times I$ matrix with entries in T. Then the $I \times \Lambda$ Rees matrix semigroup over T with matrix P is the semigroup $\mathscr{M}[T; I, \Lambda; P] = (I \times T \times \Lambda)$, with multiplication defined by

$$(i, t, \lambda) (j, u, \mu) = (i, tp_{\lambda,j}u, \mu) \quad \text{for all} \quad (i, t, \lambda), (j, u, \mu) \in I \times T \times \Lambda.$$
(1.7)

If we instead permit the matrix P to contain entries of T^0 (i.e. P can also contain 0 as an entry), then the $I \times \Lambda$ Rees 0-matrix semigroup over T with matrix P is the semigroup $\mathscr{M}[T; I, \Lambda; P] = (I \times T \times \Lambda) \cup \{0\}$, with multiplication defined by

$$(i, t, \lambda) (j, u, \mu) = \begin{cases} 0 & \text{if } p_{\lambda,j} = 0, \\ (i, tp_{\lambda,j}u, \mu) & \text{if } p_{\lambda,j} \neq 0, \end{cases}$$
(1.8)
and $(i, t, \lambda) 0 = 0 (i, t, \lambda) = 0^2 = 0,$

for all $(i, t, \lambda), (j, u, \mu) \in I \times T \times \Lambda$. Clearly, Rees matrix and Rees 0-matrix semigroups are closely related; indeed, one may regard a Rees matrix semigroup $\mathscr{M}[T; I, \Lambda; P]$ as the subsemigroup $(I \times T \times \Lambda)$ of the Rees 0-matrix semigroup $\mathscr{M}^0[T; I, \Lambda; P]$. This similarity allows many results concerning Rees matrix semigroups to be deduced from corresponding results for Rees 0-matrix semigroups.

A Rees matrix or 0-matrix semigroup $\mathscr{M}[T; I, \Lambda; P]$ or $\mathscr{M}^0[T; I, \Lambda; P]$ is finite if and only if the underlying semigroup T and the sets I and Λ are finite. Note that the nature of the elements of I and Λ is not relevant to the multiplication of the semigroup; I and Λ merely act as indexing sets. Therefore we may assume without loss of generality that $I \cap \Lambda = \emptyset$.

Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup. We define the *Graham-Houghton* graph of S, introduced in [62, 75], to be the bipartite graph $(I \cup \Lambda, \{\{i, \lambda\} : p_{\lambda,i} \neq 0\})$.

The original interest in Rees matrix and 0-matrix semigroups stems from their connection with completely simple and completely 0-simple semigroups. By the Rees-Suschkewitsch Theorem [76, Theorem 3.3.1], a semigroup is completely simple if and only if it is isomorphic to a Rees matrix semigroup $\mathscr{M}[G; I, \Lambda; P]$, where G is a group, and by the Rees Theorem [76, Theorem 3.2.3], a semigroup is completely 0-simple if and only if it is isomorphic to a Rees 0-matrix semigroup $\mathscr{M}^0[G; I, \Lambda; P]$, where G is a group and P contains a non-zero entry in each row and each column. In this thesis, when we define $\mathscr{M}^0[G; I, \Lambda; P]$ to be a regular Rees 0-matrix semigroup over a group, we require that G be a group and that P be a $\Lambda \times I$ matrix over G^0 that contains at least one non-zero entry in each of its rows and columns. We study Rees 0-matrix semigroups over arbitrary semigroups in Chapter 3.

1.3.3 Green's relations and regularity

One of the most fundamental ways of understanding a semigroup is by analysing its *Green's* relations [66]. These are five equivalences that can be defined on any semigroup. Let S be a semigroup. Green's \mathscr{L} -relation on S is defined by $x \mathscr{L} y$ if and only if $S^1x = S^1y$ for all $x, y \in S$. Green's \mathscr{L} -relation is a right congruence. Green's \mathscr{R} -relation is the dual of \mathscr{L} , and is a left congruence. Green's \mathscr{H} -relation is the intersection $\mathscr{L} \cap \mathscr{R}$, and Green's \mathscr{D} -relation is the composition $\mathscr{L} \circ \mathscr{R}$, which is equal to the composition $\mathscr{R} \circ \mathscr{L}$ [76, Proposition 2.1.3]. Green's \mathscr{J} -relation is defined by $x \mathscr{J} y$ if and only if $S^1xS^1 = S^1yS^1$ for all $x, y \in S$. On many kinds of semigroups, including finite semigroups, Green's \mathscr{D} - and \mathscr{J} -relations are equal; see for instance [110, Corollary A.2.5] and [76, Proposition 2.1.4]. Indeed, the Green's relations on any group G are all equal to $G \times G$, and all of the Green's relations on any commutative semigroup coincide. Figure 1.9 shows the containment of the Green's relations on a semigroup.

The following lemma, concerning finite semigroups, is used repeatedly in this thesis.

Lemma 1.10 ([110, Theorem A.2.4]). Let S be a finite semigroup and let $x, y \in S^1$. Then

 $x \not j yx \text{ if and only if } x \not L yx, \quad \text{ and } \quad x \not j xy \text{ if and only if } x \not R xy.$

In other words, every finite semigroup is stable, in the sense of [110, Definition A.2.1].



Figure 1.9: A Hasse diagram displaying the containment of the Green's relations on any semigroup. The inclusions shown here are not necessarily strict: for example, $\mathcal{D} = \mathcal{J}$ on any finite semigroup, and all of the Green's relations on any commutative semigroup are equal.

Certainly $\mathscr{L} \subseteq \mathscr{J}$ for any semigroup; by Lemma 1.10, in a finite semigroup, when a left multiple of an element is \mathscr{J} -related to that element, then it is also \mathscr{L} -related. The analogous statement holds for right multiples and Green's \mathscr{R} -relation.

Let S be a semigroup and let $\mathscr{H} \in \{\mathscr{H}, \mathscr{L}, \mathscr{R}, \mathscr{D}, \mathscr{J}\}$ be one of Green's relations on S. When it is necessary to emphasise that \mathscr{H} is defined the semigroup S, we write \mathscr{H}^S instead of \mathscr{H} . The semigroup S is said to be \mathscr{H} -trivial if \mathscr{H} is the equality relation on S, i.e. if $\mathscr{H} = \{(x, x) : x \in S\}$. Let $x, y \in S$. We say that x and y are \mathscr{H} -related if $x \mathscr{H} y$. The \mathscr{H} -class of x is the equivalence class of x in \mathscr{H} . This is denoted by K_x , or occasionally by K_x^S in order to emphasise the semigroup S on which \mathscr{H} is defined. We refer to equivalence classes of Green's relations as *Green's classes*.

The following result, known as Green's Lemma, is fundamental to the study of semigroups.

Lemma 1.11 (Green's Lemma [76, Lemmas 2.2.1, 2.2.2]). Let S be a semigroup, and $x, y \in S$.

- (i) If x L y, and u, v ∈ S¹ satisfy ux = y and vy = x, then the functions λ_u : R_x → S and λ_v : R_y → S, defined by aλ_u = ua and bλ_v = vb, give mutually inverse bijections from R_x to R_y, and R_y to R_x, respectively, that preserve Green's L-relation.
- (ii) If x R y, and s,t ∈ S¹ satisfy xs = y and yt = x, then the functions ρ_s : L_x → S and ρ_t : L_y → S, defined by aρ_s = as and bρ_t = bt, give mutually inverse bijections from L_x to L_y, and L_y to L_x, respectively, that preserve Green's *R*-relation.

Let S be a semigroup. We define partial orders on the \mathscr{L} -, \mathscr{R} -, or \mathscr{J} -classes of S, in terms of containment of the corresponding left, right, or two-sided ideals. More specifically, for any $x, y \in S, L_x \leq L_y$ if and only if $S^1x \subseteq S^1y$; $R_x \leq R_y$ if and only if $xS^1 \subseteq yS^1$; and $J_x \leq J_y$ if and only if $S^1xS^1 \subseteq S^1yS^1$. The group of units of a monoid is an \mathscr{H} -class, and in a finite monoid the group of units is also the unique maximal \mathscr{J} -class in the partial order of \mathscr{J} -classes of the monoid. We require the following lemma, which relates the partial orders of Green's classes of a semigroup with multiplication.

Lemma 1.12. Let S be a semigroup, and let $x_1, x_2, \ldots, x_n \in S$ for some $n \in \mathbb{N}$.

- (i) If $i \in \{1, ..., n\}$, then $L_{x_1 \cdots x_n} \leq L_{x_i \cdots x_n}$.
- (ii) If $i \in \{1, ..., n\}$, then $R_{x_1 \cdots x_n} \leq R_{x_1 \cdots x_i}$.
- (iii) If $i, j \in \{1, \ldots, n\}$ and $i \leq j$, then $J_{x_1 \cdots x_n} \leq J_{x_i \cdots x_j}$.

Proof. The statements (i) and (ii) are dual, so we prove only (i) and (iii).

(i). Let $i \in \{1, ..., n\}$. Then

$$S^{1}(x_{1}\cdots x_{n}) = S^{1}(x_{1}\cdots x_{i-1})(x_{i}\cdots x_{n})$$
$$= \left(S^{1}(x_{1}\cdots x_{i-1})\right)(x_{i}\cdots x_{n})$$
$$\subseteq S^{1}(x_{i}\cdots x_{n}),$$

and so $L_{x_1\cdots x_n} \leq L_{x_i\cdots x_n}$ by definition of the partial order of \mathscr{L} -classes of S. (iii). Let $i, j \in \{1, \ldots, n\}$ with $i \leq j$. Then

$$S^{1}(x_{1}\cdots x_{n})S^{1} = S^{1}(x_{1}\cdots x_{i-1})(x_{i}\cdots x_{j})(x_{j+1}\cdots x_{n})S^{1}$$

= $\left(S^{1}(x_{1}\cdots x_{i-1})\right)(x_{i}\cdots x_{j})\left((x_{j+1}\cdots x_{n})S^{1}\right)$
 $\subseteq S^{1}(x_{i}\cdots x_{j})S^{1},$

and the result follows by the definition of the partial order of \mathcal{J} -classes of S.

A notion intimately connected to Green's relations is that of regularity. An element x in a semigroup S is regular if there exists an element $y \in S$ such that xyx = x. If additionally yxy = y, then x and y are mutually *inverse*. A semigroup that contains only regular elements is called a *regular semigroup*. Every regular element has an inverse [76, Section 2.3]; a regular element is \mathscr{D} -related to all of its inverses. Any \mathscr{D} -class, and hence any \mathscr{L} -, \mathscr{R} -, or \mathscr{H} -class, of a semigroup S contains either no regular elements, and is called *non-regular*, or contains only regular elements, and is said to be *regular* [76, Proposition 2.3.1].

If J is a \mathcal{J} -class of a semigroup, then the *principal factor* of J is the semigroup $J^* = J \cup \{0\}$, where $0 \notin J$ and the multiplication * on J^* is given by

$$x * y = \begin{cases} xy & \text{if } xy \in J, \\ 0 & \text{if } xy \notin J, \end{cases}$$

and $0 * x = x * 0 = 0 * 0 = 0,$ (1.13)

for all $x, y \in J$. By [76, Theorem 3.1.6], the principal factor of a \mathscr{J} -class either is a zero semigroup, or is 0-simple. In particular, if S is a finite semigroup and J is a \mathscr{J} -class of S, then either J is a non-regular \mathscr{D} -class and J^* is a zero semigroup, or J is a regular \mathscr{D} -class and J^* is isomorphic to a regular Rees 0-matrix semigroup over a group.

A regular \mathscr{L} -class contains some idempotents [76, Propositions 2.3.1], which are relative right identities for any of its elements, and similarly a regular \mathscr{R} -class contains some idempotents, which are relative left identities for the element of the \mathscr{R} -class [76, Propositions 2.3.2 & 2.3.3]. An \mathscr{H} -class that contains an idempotent is a subgroup [76, Corollary 2.2.6]; otherwise, an \mathscr{H} -class H satisfies $H^2 \cap H = \varnothing$ [76, Theorem 2.2.5]. In semigroup theory, a maximal subgroup of a semigroup is an \mathscr{H} -class of the semigroup that is a subgroup. However, this is inconsistent with the notion of a maximal subgroup from group theory: in this context, a maximal subgroup of a group is a proper subgroup that is contained in no other proper subgroup of the group. In this thesis, we exclusively use the term maximal subgroup in the group-theoretic sense; we call an \mathscr{H} -class of a semigroup that is a subgroup a group \mathscr{H} -class.

We use the following lemma repeatedly in this thesis.

Lemma 1.14. Let T be a finite subsemigroup of a semigroup S that has non-empty intersection with each \mathcal{H} -class of S. Then T contains every idempotent of S.

Proof. Let $e \in E(S)$, and let $x \in T \cap H_e^S$ be arbitrary. Since H_e^S is a subgroup of S and T is finite, it follows that x has finite order in H_e^S , or in other words, $e = x^k \in T$ for some $k \in \mathbb{N}$. \Box

Multiplication within a \mathscr{D} -class is particularly systematic, as shown in the following results.

Lemma 1.15 ([76, Proposition 2.3.7]). Let S be a semigroup and let $(x, y) \in \mathscr{D}^S$. Then

 $xy \in R_x \cap L_y$ if and only if $L_x \cap R_y$ is a group.

Corollary 1.16. Let x and y be elements of a finite semigroup.

- (i) Suppose that $x \mathscr{R} y$. Then $xy \in R_x = R_y$ if and only if H_x is a group.
- (ii) Suppose that $x \mathscr{L} y$. Then $xy \in L_x = L_y$ if and only if H_y is a group.

Proof. If $x \mathscr{R} y$, then $R_x = R_y$, and so Lemma 1.15 becomes $xy \in R_y \cap L_y$ if and only if H_x is a group. Thus, it remains to prove that $xy \in L_y$ whenever $xy \in R_y$. If $xy \in R_y$, then $xy \not \not y$, and so $xy \mathscr{L} y$ by Lemma 1.10. This proves (i); the proof of (ii) is dual.

Whenever an element $x \in S$ has a unique inverse y, we define $x^{-1} = y$ to be the *inverse* of x. This notation is consistent with the notation u^{-1} introduced earlier for a unit u of a monoid. An *inverse semigroup* is a semigroup in which each element has a unique inverse. In particular, any group is an inverse semigroup. There are many other characterisations of inverse semigroups; for instance, a semigroup is inverse if and only if it is regular and its idempotents commute; which occurs if and only if each of its \mathscr{L} -classes and each of its \mathscr{R} -classes contains exactly one idempotent [76, Theorem 5.1.1]. See [85] for a thorough treatment of inverse semigroups.

A regular *-semigroup, as introduced in [105], is a semigroup S with an operation $*: S \longrightarrow S$ that satisfies $(x^*)^* = x$, $(xy)^* = y^*x^*$, and $x = xx^*x$ for all $x, y \in S$. In particular, the *operation is an anti-isomorphism. A regular *-monoid is simply a regular *-semigroup that is a monoid. In Chapter 5, we describe the maximal subsemigroups of several families of finite regular *-monoids. Any regular *-semigroup is a regular semigroup, since x^* is an inverse of the element x. An idempotent x of a regular *-semigroup is called a *projection* if $x^* = x$. In a regular *-semigroup, the \mathscr{R} -class of an element x contains the unique projection xx^* , and the \mathscr{L} -class of x contains the unique projection x^*x . Since a regular *-semigroup is anti-isomorphic to itself via the * operation, this gives a correspondence between left multiplication and right multiplication. In particular, if S is a regular *-semigroup, and $x, y \in S$, then $x \mathscr{L} y$ if and only if $x^* \mathscr{R} y^*$. Since $x \mathscr{D} x^*$, the number of \mathscr{L} -classes is equal to the number of \mathscr{R} -classes in any \mathscr{D} -class of a regular *-semigroup. Any inverse semigroup S can be thought of as a regular *-semigroup in which every idempotent is a projection, by defining $x^* = x^{-1}$ for all $x \in S$. Conversely, any regular *-semigroup in which every idempotent is a projection is inverse.

1.3.4 Group theory and group actions

A group is a monoid consisting of units. In this thesis we require several group-theoretic notions. If G is a group, and V is a subgroup of G, and $g \in G$, then the right coset of V in G defined by g is the set Vg. Two right cosets Vg and Vh of V in G are equal if and only if $gh^{-1} \in V$. The right cosets of V in G, $\{Vg : g \in G\}$, form a partition of G, and the *index* of V in G, denoted [G:V], is the number of cosets of V in G, if there are finitely many, or ∞ .

Another important concept in group theory is that of *conjugation*. Two subgroups U and V of a group G are *conjugate* if there exists $g \in G$ such that $g^{-1}Ug = V$. Conjugate subgroups are isomorphic, and the collection of subgroups of a group is partitioned by its conjugacy classes. A normal subgroup of a group is a subgroup that is closed under conjugation, i.e. a subgroup whose conjugacy class is trivial. In any group G, the group G itself and the trivial subgroup $\{1_G\}$ are normal subgroups. The normalizer of a subgroup V in a group G, denoted $N_G(V)$, is the least normal subgroup of G, with respect to containment, that contains V.

If G is a group and X is a set, then a right action of G on X is a function $\psi : X \times G \longrightarrow X$ where $(x, 1_G) \psi = x$ and $((x, g) \psi, h) \psi = (x, gh) \psi$ for all $x \in X$ and $g, h \in G$. We usually write $x \cdot g$ instead of $(x, g) \psi$, when the right action ψ is clear from the context. The relation \sim on X defined by $x \sim y$ if and only if $x \cdot g = y$ for some $g \in G$ is an equivalence, since $x \cdot 1_G = x$, $x \cdot g = y$ if and only if $y \cdot g^{-1} = x$, and $x \cdot g = y$ and $y \cdot h = z$ implies that $x \cdot gh = z$, for all $x, y, z \in X$ and $g, h \in G$. The equivalence classes of this equivalence are called *orbits*, and so a right action of a group on a set partitions that set into orbits. When a right action has only one orbit, we say that the action is *transitive*, and we say that the group acts *transitively*. A left action of a group on a set, and its orbits, are defined analogously.

1.3.5 Semigroups of partial transformations

Composition of relations is associative, and therefore, so is composition of partial functions. It follows that we may create semigroups from relations or partial functions on a set. Let $n \in \mathbb{N}$. We define \mathcal{PT}_n , the *partial transformation monoid of degree n*, to be the monoid consisting of all partial transformations of degree *n*, with composition of partial functions. We also define the following submonoids of \mathcal{PT}_n :

- $\mathcal{T}_n = \{ \alpha \in \mathcal{PT}_n : \operatorname{dom}(\alpha) = \{1, \ldots, n\} \}$, the full transformation monoid of degree n;
- $\mathcal{I}_n = \{ \alpha \in \mathcal{PT}_n : |im(\alpha)| = |dom(\alpha)| \}$, the symmetric inverse monoid of degree n; and
- $S_n = \{ \alpha \in \mathcal{PT}_n : im(\alpha) = \{1, \ldots, n\} \}$, the symmetric group of degree n.

In other words, \mathcal{T}_n consists of all transformations of degree n, \mathcal{I}_n consists of all partial permutations of degree n, and \mathcal{S}_n consists of all permutations of degree n. Note that the symmetric group \mathcal{S}_n is the group of units of \mathcal{PT}_n , \mathcal{T}_n , and \mathcal{I}_n . We define id_n , the *identity permutation of degree* n, to be the identity of each of these monoids; id_n fixes each point $i \in \{1, \ldots, n\}$. Full transformation monoids, symmetric inverse monoids, and symmetric groups can also be defined on arbitrary sets, including on infinite sets, however these will not be required in this thesis.

By Cayley's theorem, every group is isomorphic to a subgroup of some symmetric group; moreover, every finite group G is isomorphic to a subgroup of $S_{|G|}$. Thus, in a sense, the study of permutation groups is the study of all groups. Semigroups of partial transformations, especially semigroups of transformations, are similarly foundational in semigroup theory. By an analogue of Cayley's theorem for groups, every semigroup is isomorphic to a subsemigroup of some full transformation monoid [76, Theorem 1.1.2]; indeed, any finite semigroup S is isomorphic to a subsemigroup of $\mathcal{T}_{|S|+1}$. Symmetric inverse monoids play the analogous role for inverse semigroups, by the Vagner-Preston theorem [76, Theorem 5.1.7]. In particular, any finite semigroup can be realised by transformations of degree n, for some $n \in \mathbb{N}$, and similarly any finite inverse semigroup can be realised by partial permutations of some degree m. See [57] for a detailed study of finite semigroups of partial transformations.

1.4 Computational semigroup theory

We end this chapter by introducing several additional concepts from computational semigroup theory, in order to provide the context in which the mathematical results of this thesis are presented. To reiterate, computational semigroup theory is both the investigation of semigroups with the help of computers, and the development of the tools that facilitate this study. In this section, we informally discuss the notions of algorithms, which we use to define and communicate techniques in computation, and computational complexity, which we use to classify the efficiency of an algorithm. We also briefly explain some of the main paradigms used for computing with semigroups, their aims, and their relative advantages and disadvantages.

1.4.1 Algorithms and complexity

In a casual sense, an *algorithm* is a method, or procedure, that describes how to solve any instance of a particular collection of problems in a finite number of steps. In other words, an algorithm describes a solution to a computational problem. See, for example, Algorithm 1.17, which uses *pseudocode* to define an algorithm for determining whether a given number is contained in a given list of numbers. A computer program is a specific implementation of an algorithm in some programming language.

A Turing machine [126] is one of the earliest abstract models of computation. Although the definition of a Turing machine is not important for the purposes of this thesis, this abstraction is crucial to the foundations of computer science. According to the widely-accepted *Church-Turing thesis*, Turing machines are 'universal' for computation: a problem that can be solved by some computer can also be solved by a Turing machine. Viewed in this context, an algorithm is a specification for defining a Turing machine that solves a particular problem, in such a way that given any instance of the problem, the Turing machine will terminate with the correct solution after a finite number of steps.

Algorithm 1.17 A simple algorithm for testing membership in a list of natural numbers. Input: A finite list *L* consisting of natural numbers, and a natural number $x \in \mathbb{N}$. Output: TRUE if *x* is an entry of *L*; else FALSE. 1: for $n \in L$ do 2: if n = x then

if n = x then
return TRUE.
return FALSE.

For some computational problems, numerous diverse algorithms have been developed to solve the problem. For instance, bubble sort, insertion sort, quick sort, and merge sort are well-known and significantly different algorithms for sorting a list into ascending order.

The execution of a computer program requires time and space (the memory used to store intermediate results). The resources of any given computer are finite, and different algorithms typically require differing amounts of these resources when implemented and executed: the amount of time required by an algorithm is called its *time complexity*, and the amount of space required is the *space complexity*. We therefore require strategies for comparing algorithms and analysing their resource requirements. This helps us to choose the most suitable algorithm for a particular computation, and to determine whether a new algorithm is better than an existing one. In this thesis, we focus on the time complexity of algorithms. See [59, 128] for a much more thorough discussion of algorithms and their complexities than the one that follows.

An obvious way to study the performance of an implementation of an algorithm is to execute the algorithm on a carefully chosen range on inputs, and to measure the resources required to produce the output. There are good reasons to study algorithms in this way: it is easy to do, and it often demonstrates the real-life behaviour that a user might find in practice. Indeed, this is the approach we take in Chapter 4 for analysing the performance of algorithms for computing maximal subsemigroups of finite semigroups. Furthermore, by executing the same examples with competing algorithms, it is possible to fairly compare the performance of these algorithms on the same inputs.

However, there are problems with this approach. Perhaps, by some fluke, the algorithms

demonstrate atypical performance on the range of examples that we choose to examine. For many computational tasks, there are an infinite number of possible inputs, and so it is impossible to test them all. Additionally, the specific implementation of the algorithm being tested may be poor, leading to unsatisfactory performance that obscures the potential of the algorithm.

We therefore desire a more systematic and mathematical way of measuring and comparing the resource requirements of algorithms. An algorithm defines a sequence of steps, each of which has some basic time cost. Therefore, if we can calculate the number of steps that are required for a given input, then we can use the known costs to describe the time requirements of the algorithm in terms of its input. However, this is hard to do exactly, and so we generally attempt to give upper bounds on the resource requirements of an algorithm, given the input.

We use big O notation [128, Chapter 1] to relate the resource requirements of an algorithm to its inputs, and thereby to describe its time complexity. Let X be a set of elements with some notion of size, and let f and g be functions $X \longrightarrow \mathbb{R}$. We say that the function f is O(g) if there exists a positive constant c such that $0 \le (x)f \le c \cdot (x)g$ for all sufficiently large $x \in X$. If X is the set of possible inputs to some algorithm, and $g: X \longrightarrow \mathbb{R}$ is a function, then we say that the time complexity of the algorithm is O(g) if we can show that the theoretical function $f: X \longrightarrow \mathbb{R}$, describing the time required to run the algorithm on the input X, on some computer and in some unit of measurement, is O(g). Note that we may ignore the details of the specific computer used and the specific units of time, since big O notation disregards constant factors. See Example 1.18 for an analysis of the time complexity of Algorithm 1.17.

Example 1.18. Let L be a finite list of natural numbers, and let $x \in \mathbb{N}$. Suppose that it takes some constant length of time c (in some units) to test the equality of two natural numbers. Consider using the algorithm described in Algorithm 1.17 to test whether x is an entry of L. In the best case, the first element of L is equal to x, and the algorithm terminates with the correct answer after only one comparison. However, if x does not appear until the final entry of L, or if x is not an entry of L, then the algorithm does not determine this until every entry of Lhas been compared against x. It follows that, in the worst, case Algorithm 1.17 requires $c \cdot |L|$ units of time in order to produce its output. We can therefore say that the time complexity of Algorithm 1.17 is O(|L|), or O(n), where n = |L|.

Analysing time complexity in this way is often difficult, but it provides a useful way of comparing the requirements of different algorithms, and of understanding how the requirements of an algorithm grow as the input grows. If the time complexity of one algorithm is O(n), and the time complexity of a competing algorithm is O(1), then roughly speaking, we can expect that doubling the size of the input will double the length of time taken to compute with the first algorithm. We may therefore consider the second algorithm to be more time-efficient. More concretely, sorting a finite list L using the merge sort algorithm has time complexity $O(|L| \log |L|)$, whereas sorting L using the insertion sort algorithm has time complexity $O(|L|^2)$, which suggests that the merge sort algorithm is more time-efficient.

In this thesis, we do not formally describe the algorithms that we present with big O notation, however we use this notion of complexity in order to provide some background information and context to the problems that we approach. Algorithms in computational semigroup theory tend to have high time complexity, at least in comparison with algorithms in computational group theory. Despite this, we still find that it is possible to execute algorithms in computational semigroup theory on a wide range of interesting and useful examples.

1.4.2 Data structures for semigroups

In order to compute with a given semigroup, it is first necessary to store the definition of the semigroup on a computer in some way. In other words, we must choose a structure for defining the semigroup to a computer. A semigroup is a set with an associative binary operation, and so to specify a semigroup, it is necessary to unambiguously and clearly define its set of elements and a way of calculating the product of any two elements in the set.

One of the simplest ways of specifying a finite semigroup to a computer is by a *multiplication* table. If S is a finite semigroup of order n, then the multiplication table of S is an $n \times n$ array whose rows and columns are labelled by the elements of S; if $x, y \in S$, then the entry in the row labelled by x and the column labelled by y is the product of x and y in S. Thus the multiplication table encodes both the elements and the operation of the semigroup. See Table 1.19 for an example of a multiplication table that defines a semigroup with four elements.

0	x	y	z	t
x	x	y	z	t
y	y	x	z	t
z	z	t	z	t
t	t	z	z	t

Table 1.19: The multiplication table of a semigroup isomorphic to the full transformation monoid of degree 2 (see Section 1.3.5), which contains four elements, including an identity.

Multiplication tables provide a useful data structure in many respects. Given sufficient space, any finite semigroup can be given by a multiplication table. Furthermore, the multiplication table of a semigroup contains the elements of the semigroup and complete information about their products. Therefore, producing the set of elements and multiplying elements is simply a case of looking up the information in whatever data structure it is stored. The SMALLSEMI [32] package for GAP [58], which is a library containing every semigroup of order 8 or less, stores its semigroups via multiplication tables.

The most significant downside of giving a semigroup by its multiplication table is that storing a multiplication table requires a relatively large amount of memory, which can easily exceed the capabilities of a computer. Let S be a finite semigroup with n elements. In order to be able to distinguish between the elements of S on a computer, it is necessary to use at least k bits to identify each element, where $2^k \ge n$. In other words, we must choose $k \ge \log n$. It follows that storing the multiplication table of S on a computer, without using compression techniques, requires at least $n^2 \log n$ bits. At present, the memory on a personal computer is typically about 16 gigabytes, which is exceeded by the amount of memory required to store the multiplication table of a semigroup with just 100,000 elements. We wish to compute with finite semigroups that contain far more elements than this, and sometimes even with infinite semigroups, and a multiplication table is not appropriate for these cases.

All finite semigroups and many infinite semigroups may be given as the semigroup defined by a finite presentation $\langle X | R \rangle$. In certain cases, it is possible to develop techniques that directly manipulate the generators X and the relations R in order to compute properties of the semigroup that the presentation defines. We briefly describe a method for computing nontrivial factorizations of elements in a finitely presented semigroup in Chapter 2, however finitely presented semigroups are not the focus of this thesis. Computing with these kinds of semigroups involves many difficulties, not least because it has been shown that for many semigroup-theoretic properties, there exists no algorithm for determining whether that property holds in an arbitrary finitely presented semigroup; see [95, 108].

We can often define a semigroup by the way in which it relates to some other known semigroup. For example, a finite Rees 0-matrix semigroup can be specified to a computer by its underlying semigroup, in some form, and its matrix. A more general-purpose technique along these lines is to specify a semigroup as the subsemigroup of some other semigroup defined by a finite set of generators: the elements are the products of the generators, and the product of two elements is the same as the product in the parent semigroup. The parent semigroup may be given explicitly: for example, a semigroup could be defined as the subsemigroup of some Rees 0-matrix semigroup generated by a particular set of elements. On the other hand, the parent semigroup may be given implicitly: a semigroup could be given as the subsemigroup generated by a particular set of transformations of degree n, and the semigroup may be understood to be a subsemigroup of the full transformation monoid of degree n. The algorithms that we present in Chapter 4 for computing the maximal subsemigroups of an arbitrary finite semigroup require a semigroup defined by a generating set.

For some semigroups, such as left- or right-zero semigroups, the set of all elements of the semigroup is the only generating set. For some other semigroups, such as direct products, it is not immediately apparent how to specify any generating set other than the set of all elements. On the other hand, a generating set for a semigroup may be readily available and may be much smaller than the order of the semigroup, and thereby permit a compact way of storing the semigroup. For example, the full transformation monoid of degree n is generated by two permutations and any transformation of rank n - 1, and many infinite semigroups are finitely generated, such as the natural numbers \mathbb{N} under addition. The NUMERICALSGPS [24] package for GAP [58] enables the creation and computation of finitely generated subsemigroups of \mathbb{N} , known as numerical semigroups.

1.4.3 Finite semigroups specified by generating sets

In this thesis, we predominantly focus on the computation of finite semigroups specified by generating sets. For a typical algorithm that requires a semigroup given in this fashion, the time complexity of the algorithm is given in terms of the number of generators that define the semigroup, amongst other factors, such as, perhaps, the order of the semigroup and the specific type of elements by which it is given. Therefore, from a computational perspective, we desire a generating set for any given semigroup that is as small as is reasonably practicable. Driven by this motivation, a significant portion of the research in this thesis is dedicated to the problem of describing relatively small generating sets for certain kinds of semigroups.

A fundamental prerequisite of many algorithms in computational semigroup theory, including the algorithms presented in Chapter 4, is the ability to compute the Green's structure of a finite semigroup from its generating set. Broadly speaking, there are two main approaches to computing with a finite semigroup defined by generating set, and thereby describing its Green's structure. There are those techniques that involve exhaustively enumerating the semigroup, i.e. producing and storing every element of the semigroup in some useful data structure, and there are those that produce a data structure describing the Green's structure of the semigroup without necessarily enumerating the semigroup. We briefly describe these approaches below.

Exhaustive enumeration

The most well-known techniques for exhaustively enumerating a semigroup are those that construct the left or right Cayley digraphs of the semigroup. Let S be a finite semigroup with a generating set X. The left Cayley digraph of S with respect to X is the digraph (S, E), where

$$E = \{ (s, xs) : s \in S, x \in X \},\$$

and the right Cayley digraph of S with respect to X is the digraph $(S, \{(s, sx) : s \in S, x \in X\})$. When constructing a Cayley digraph, we label an edge with the set of generators corresponding to that edge. In this way, the left and right Cayley digraphs of a semigroup encode left and right multiplication in that semigroup. See Figure 1.20 for an example.



Figure 1.20: A picture of the right Cayley digraph of the semigroup defined by the multiplication table in Table 1.19, with respect to its generating set $\{y, z\}$. The dashed edges correspond to the generator y; the solid edges correspond to the generator z; and the dotted edge corresponds to both generators. The strongly connected components of this right Cayley digraph are $\{x, y\}$ and $\{z, t\}$, and so these are the \mathscr{R} -classes of the associated semigroup.

It is straightforward to see that the strongly connected components of a left Cayley digraph of S are the \mathscr{L} -classes of S, and the strongly connected components of a right Cayley digraph of S are its \mathscr{R} -classes. Furthermore, if (S, E) and (S, E') are left and right Cayley digraphs of S with respect to some generating set, then the strongly connected components of the digraph $(S, E \cup E')$ are the \mathscr{J} -classes of S, and the partial orders of the \mathscr{L} -, \mathscr{R} -, and \mathscr{J} -classes of S are given by the quotients of these digraphs by their strongly connected components, respectively. There are several well-known algorithms, such as those of Tarjan [121] and Gabow [55], for finding the strongly connected components of a digraph. See [80] and the references therein for more information about Cayley digraphs of semigroups.

A naive algorithm for computing the right Cayley digraph of a semigroup S with respect to a generating set X is given in Algorithm 1.21. This algorithm has time complexity O(|S||X|). Note that the left Cayley digraph of S with respect to X can be deduced from the right Cayley digraph of S with respect to X, and vice versa: if $x \in X$ and $s \in S$, then there exist generators $y_1, \ldots, y_n \in X$ such that $s = y_1 \cdots y_n$, and $xs = (xy_1 \cdots y_{n-1}) \cdot y_n$, turning left multiplication by a generator into right multiplication.

The most famous and sophisticated algorithm for exhaustively enumerating a semigroup S defined by a generating set X, and obtaining the left and right Cayley digraphs of S with respect to X, is the Froidure-Pin Algorithm [54], which has time complexity O(|S||X|). Although the Froidure-Pin Algorithm has the same time complexity as Algorithm 1.21, in practice it is much quicker to use the Froidure-Pin Algorithm to compute a Cayley digraph, since it deduces many of the edges of the digraph from previous calculations, and thereby avoids many unnecessary multiplications. The Froidure-Pin Algorithm has been extended in various ways, including by the development of a parallel version [79], which is implemented in LIBSEMIGROUPS [102]. The methods in LIBSEMIGROUPS [102] can be applied to a wide range of semigroups defined generating sets, including semigroups of transformations or partial permutations, semigroups of partitioned binary relations [96], and semigroups of boolean matrices.

Algorithm 1.21 A basic algorithm to compute the right Cayley digraph of a finite semigroup. Input: A finite generating set X for a finite semigroup.

Output: The right Cayley digraph (V, E) of that semigroup with respect to X. 1: $V \leftarrow X$ 2: $E \leftarrow \emptyset$ 3: for $s \in V, x \in X$ do 4: if $sx \notin V$ then 5: $V \leftarrow V \cup \{sx\}$ 6: $E \leftarrow E \cup \{(s, sx)\}$ 7: Add x to the label of the edge (s, sx)8: return (V, E).

A major limitation to algorithms that involve exhaustively enumerating a semigroup is memory. Consider a semigroup that consists of transformations of degree n. A straightforward approach to storing a transformation, as a list of images of each point, requires at least nk bits, where $k \ge \log n$. Therefore, fully enumerating and storing the elements of a transformation semigroup S of degree n in this simple way therefore requires at least $|S| \cdot n \log n$ bits: to illustrate, the full transformation monoid of degree 10 requires at least 38 gigabytes of memory with this approach. In order to compute large semigroups, we therefore require different techniques.

Techniques that are not necessarily exhaustive

There are a number of algorithms given in the literature, and implemented in software, that produce a data structure describing the Green's relations of a finite semigroup defined by a generating set without necessarily enumerating the whole semigroup. Lallement and McFadden [83] developed such an algorithm for computing with transformation semigroups; this was adapted by Konieczny [81] for semigroups of boolean matrices. Further techniques for computing with transformation semigroups were developed and implemented in the MONOID [89] package by Linton and co-authors [90, 91]. More recently, these approaches have been significantly extended and generalised by Mitchell and co-authors in [37]; this work applies to subsemigroups of arbitrary finite regular semigroups. The techniques in [37] have been fully implemented in the SEMIGROUPS [101] package for GAP [58], which allows for the non-exhaustive computation of semigroups given by a wide range of generating sets, including semigroups of matrices over finite fields, subsemigroups of regular Rees 0-matrix semigroups over groups, and semigroups given by generating sets consisting of transformations, partial permutations, or partitions (in the sense of Section 5.3.1), amongst other types.

The unifying approach of the above techniques is that they exploit certain actions of a semigroup (defined analogously to group actions) that are inherent when the semigroup is given in a particular form; these actions are connected to the Green's relations of the semigroup. These techniques construct groups from these actions, related to the group \mathscr{H} -classes of the semigroup, that allow certain computations about the semigroup to be reduced to computations about these groups. In doing so, these algorithms can take advantage of mature and efficient non-exhaustive algorithms from computational group theory, such as the Schreier-Sims algorithm [73, Section 4.4.2].

The above algorithms describe the Green's structure of a semigroup, in part, by enumerating representatives of the \mathscr{L} - and \mathscr{R} -classes of a semigroup, rather than by enumerating the \mathscr{L} - and \mathscr{R} -classes themselves. Therefore, for a semigroup with relatively large \mathscr{J} -classes (and with small numbers of \mathscr{L} - and \mathscr{R} -classes in comparison to the number of elements), the Green's
example, it takes 15 seconds to compute the number of \mathscr{R} -classes of the full transformation monoid of degree 11 using the SEMIGROUPS package for GAP on a 2.66 GHz Intel Core i7 processor with 8GB of RAM, even though it would be difficult to store all 11¹¹ transformations of degree 11 on a computer with such little memory. On the other hand, this means that computing \mathscr{J} -trivial semigroups with these methods involves exhaustive enumeration, in a fashion that is far less efficient than the Froidure-Pin Algorithm.

While there are non-exhaustive techniques for computing with many types of finite semigroups given by generating sets, they do not currently apply to all such semigroups.

At various points in this thesis, we require the ability to compute the Green's structure of a given finite semigroup. In these instances, the details of how this is achieved, whether exhaustively or otherwise, are not important, and we do not concern ourselves with this problem further.

Chapter 2

Generating sets for direct products of semigroups

2.1 Introduction

The direct product of a finite list of semigroups S_1, \ldots, S_n is the semigroup consisting of the Cartesian product $S_1 \times \cdots \times S_n$ with component-wise multiplication. In other words, the multiplication on $S_1 \times \cdots \times S_n$ is given by

$$(s_1, \ldots, s_n)(t_1, \ldots, t_n) = (s_1 t_1, \ldots, s_n t_n)$$
 (2.1)

for all (s_1, \ldots, s_n) , $(t_1, \ldots, t_n) \in S_1 \times \cdots \times S_n$. The semigroups S_1, \ldots, S_n are called the *factors* of the direct product $S_1 \times \cdots \times S_n$. Direct products of many other kinds of algebraic structures are defined analogously.

In this chapter, we build on the results of [2, Chapter 3] and [112,113] to describe generating sets for direct products of arbitrary semigroups which do not necessarily contain every element of the semigroup. These works predominantly concern the finite generation and presentability of direct products of semigroups, and describing generating sets for (and thereby finding upper bounds on the ranks of) certain kinds of direct products. Here, we particularly focus on computing relatively small generating sets for direct products of finite semigroups. The research detailed in this chapter was motivated by a desire to improve the performance of the SEMI-GROUPS package [101] for GAP [58]. In version 3.0.0 of this software, creating a direct product of semigroups that are not all monoids required a brute-force search for a generating set, which had disappointing performance. For example, constructing a generating set for the direct product $(\mathcal{T}_4 \setminus S_4) \times (\mathcal{T}_4 \setminus S_4)$ took roughly 6 seconds on a 2.66 GHz Intel Core i7 processor with 8GB of RAM, while constructing a generating set for the direct product $(\mathcal{T}_5 \setminus S_5) \times (\mathcal{T}_5 \setminus S_5)$ did not terminate after several minutes. In the current development version of the SEMIGROUPS package, which implements some of the techniques described in this chapter, this latter computation takes roughly 10 milliseconds on the same hardware.

In certain instances, computing with a direct product can be reduced to performing independent computations in each of the factors. For example, the order of the direct product $S_1 \times \cdots \times S_n$ is equal to $|S_1| \cdots |S_n|$, and the set of idempotents of $S_1 \times \cdots \times S_n$ is $E(S_1) \times \cdots \times E(S_n)$. Furthermore, a direct product is regular, or inverse, or commutative if and only if each of its factors is regular, or inverse, or commutative, respectively. Similar statements hold for many other semigroup-theoretic properties.

Computing independently in the factors rather than in the direct product itself is advantageous for several reasons. For finite semigroups, the factors of a direct product are often much smaller than the direct product itself, and consequently they are much cheaper to compute with. To illustrate this, note that it is typically easier to enumerate three semigroups of order 1000 than it is to enumerate their direct product, which contains a billion elements. Additionally, the independence of the computations in the factors allows them to be carried out in parallel, which potentially saves time.

On the other hand, for some other questions, it is not immediately apparent how or whether any reduction to the factors can be achieved. For example, it is not necessarily the case that the Green's classes of a direct product are products of the Green's classes of the factors; any direct product involving \mathbb{N} is \mathscr{J} -trivial, for instance.

Open Problem 2.2. Develop techniques for computing the Green's structure of a direct product of finite semigroups in terms of the Green's structures and other semigroup-theoretic properties of the factors.

Similarly, the maximal subsemigroups of a direct product do not necessarily correspond in an obvious way to maximal subsemigroups of the factors; see Example 2.4 for a demonstration of this. The maximal subgroups of direct products of groups can be given in terms of the maximal subgroups and the maximal normal subgroups of the factors (see for example [123, Lemma 1.3]) but an analogous characterisation for semigroups has not yet been found.

Open Problem 2.3. Investigate how the maximal subsemigroups of a direct product of finite semigroups relate to the maximal subsemigroups of the factors.

Example 2.4. Let \mathcal{T}_2 denote the full transformation monoid of degree 2. The maximal subsemigroups of \mathcal{T}_2 are its group of units \mathcal{S}_2 and the set $\mathcal{T}_2 \setminus \{(1 \ 2)\}$, where (1 2) is the non-identity permutation of degree 2; see Theorem 5.7. However, there are five maximal subsemigroups of the direct product $\mathcal{T}_2 \times \mathcal{T}_2$, three of which correspond to maximal subgroups of its group of units, which is the Klein four-group $\mathcal{S}_2 \times \mathcal{S}_2$, along with the sets

$$(\mathcal{T}_2 \times \mathcal{T}_2) \setminus ((\mathcal{T}_2 \setminus \mathcal{S}_2) \times \mathcal{S}_2) \text{ and } (\mathcal{T}_2 \times \mathcal{T}_2) \setminus (\mathcal{S}_2 \times (\mathcal{T}_2 \setminus \mathcal{S}_2))$$

These computations were performed with the SEMIGROUPS [101] package, which uses the algorithms presented in Chapter 4. Note that the maximal subsemigroups of $\mathcal{T}_2 \times \mathcal{T}_2$ are not direct products of maximal subsemigroups of \mathcal{T}_2 . A correspondence between the maximal subsemigroups of \mathcal{T}_2 and the maximal subsemigroups of $\mathcal{T}_2 \times \mathcal{T}_2$ is not immediately apparent.

When it is possible to compute independently in the factors, this fact still has to be observed, and proven, and the corresponding algorithm has to be implemented, before it can be exploited in a computational algebra system. Where there do not exist specialised computational methods for direct products of semigroups, either because such algorithms are not known, or are simply not implemented, we instead rely on more general-purpose methods.

As discussed in Section 1.4, many algorithms for computing with semigroups require a semigroup defined by a finite generating set. Therefore, in order to compute effectively with finitely generated direct products, we require the ability to construct finite generating sets for them, which is anyway inherently interesting.

The only generating set given by the definition of a direct product is the set of all elements. This presents a problem, especially when the direct product is infinite, because it is impossible to store an infinite set of elements. When the direct product is finite, using the set of all elements is still undesirable: as discussed in Section 1.4.3, the time complexities of algorithms that require a generating set are typically given in terms of the size of the generating set, amongst other factors. For example, using the Froidure-Pin algorithm to compute the right Cayley digraph of a finite semigroup S with respect to a generating set X has time complexity O(|S||X|). Roughly speaking, this means that the larger the generating set for a given semigroup, the more time that is required to compute with the semigroup. Furthermore, the cardinality of a

direct product is the product of the cardinalities of the factors. Even if the factors themselves are finite and relatively small, storing in memory the set of all elements in the product can easily exceed the capabilities of a computer. In the finite case, when possible, we aim to compute generating sets that are significantly smaller than the set of all elements.

For monoids defined by generating sets, it requires no calculation to specify a generating set for the direct product that is typically much smaller than the order of the direct product. In more detail, let $n \in \mathbb{N}$, and for each $i \in \{1, \ldots, n\}$, let S_i be a monoid with generating set X_i . It is straightforward to see that the direct product $S_1 \times \cdots \times S_n$ is generated by the set

$$\bigcup_{i=1}^{n} \left\{ \left(1_{S_1}, \dots, 1_{S_{i-1}}, x, 1_{S_{i+1}}, \dots, 1_{S_n} \right) : x \in X_i \right\}$$

which contains at most $|X_1| + \cdots + |X_n|$ elements. By choosing the generating sets X_i to have minimal cardinality amongst the generating sets for S_i , and noting that for each *i*, the monoid S_i is a homomorphic image of $S_1 \times \cdots \times S_n$, it follows that

$$\max\{\operatorname{rank}(S_1), \dots, \operatorname{rank}(S_n)\} \le \operatorname{rank}(S_1 \times \dots \times S_n) \le \operatorname{rank}(S_1) + \dots + \operatorname{rank}(S_n).$$
(2.5)

The bounds given in (2.5) are tight. If p_1, \ldots, p_n are distinct primes and S_i is a cyclic group of order p_i for each *i*, then the direct product $S_1 \times \cdots \times S_n$ is cyclic. Therefore, the lower bound in (2.5) can be obtained. On the other hand, if each S_i is a cyclic group of some common order k > 1, then rank $(S_1 \times \cdots \times S_n) = n$, and so the upper bound can also be obtained. It follows by (2.5) that a direct product of monoids is finitely generated if and only if each of its factors is finitely generated.

The same is not true for semigroups in general, however. Let \mathbb{N} denote the semigroup consisting of the natural numbers with addition. This semigroup is monogenic, since $\mathbb{N} = \langle 1 \rangle$; indeed, \mathbb{N} is the unique infinite monogenic semigroup, up to isomorphism. Despite this, the direct product $\mathbb{N} \times \mathbb{N}$ is not finitely generated. To see this, let X be an arbitrary generating set for $\mathbb{N} \times \mathbb{N}$, and let $n \in \mathbb{N}$ be arbitrary. By definition, there exists a sequence of generators $(x_1, y_1), \ldots, (x_m, y_m) \in X$ such that

$$(1, n) = (x_1, y_1) + \dots + (x_m, y_m) = (x_1 + \dots + x_m, y_1 + \dots + y_m).$$

In particular, $1 = x_1 + \cdots + x_m$. But 1 cannot be written as the sum of two or more natural numbers. It follows that m = 1, and so $(1, n) \in X$. Since X and n were arbitrary, we conclude that any generating set for $\mathbb{N} \times \mathbb{N}$ contains an infinite number of elements. Indeed, Corollary 2.20 implies that $\{(1, n), (n, 1) : n \in \mathbb{N}\}$ is the unique minimal generating set for $\mathbb{N} \times \mathbb{N}$, with respect to containment.

The lower bound in (2.5) holds for direct products of semigroups in general, since each factor is a homomorphic image of the direct product. Therefore, if a direct product is finitely generated, then its factors are finitely generated. However, we have seen that the converse does not necessarily hold. The problem of determining when the direct product of two semigroups is finitely generated was solved in [113]. This work was extended to the direct product of any finite number of semigroups by Araújo [2, Theorem 3.39]; we restate this in the following proposition. Recall that a semigroup S is surjective if $S^2 = S$.

Proposition 2.6 ([2, Theorem 3.39]; see [113, Remark 2.6]). Let S_1, \ldots, S_n be semigroups. The direct product $S_1 \times \cdots \times S_n$ is finitely generated if and only if each semigroup S_i is finitely generated, and either:

(i) S_i is surjective for all *i*; or

- (ii) S_i is finite for all *i*; or
- (iii) S_i is infinite for some *i*, and S_j is finite and surjective for all $j \neq i$.

In [113, Proposition 2.5] and [2, Chapter 3.1], the authors describe generating sets for the direct product of two surjective semigroups; in doing so, they obtain upper bounds on the ranks of direct products of this kind. Moreover, the bound $\operatorname{rank}(S \times T) \leq 2\operatorname{rank}(S)\operatorname{rank}(T)$ for arbitrary surjective semigroups S and T, given in [2, Corollary 3.6], is tight. As mentioned in [113, Remark 2.6], and as we discuss later, these results can be used to construct finite generating sets of the first kind described in Proposition 2.6. However, the authors do not directly address the problem of finding generating sets for direct products of the second or third kinds given in Proposition 2.6.

The main result of this chapter is Theorem 2.17, which describes a generating set for an arbitrary direct product of semigroups which does not, in general, contain every element of the direct product. Moreover, when the factors are defined by finite generating sets and the direct product is finitely generated, then the generating set given in Theorem 2.17 is finite. We discuss techniques for constructing the generating set described by Theorem 2.17 in certain cases.

This chapter is organised as follows. In Section 2.2, we introduce the concepts of decomposable and indecomposable elements in a semigroup. These notions prove to be important when constructing generating sets for direct products. In Section 2.3, we discuss methods for computing the indecomposable elements in certain kinds of semigroup, and for obtaining nontrivial factorizations of decomposable elements. We present the main results of this chapter in Section 2.4. In Section 2.5, we restrict our attention to describing generating sets for direct products of two finitely generated surjective semigroups. In some cases, these generating sets can be smaller than the generating sets described by the more general results of Section 2.4.

2.2 Decomposable and indecomposable elements

In this section, we introduce the concepts of decomposable and indecomposable elements in a semigroup, as in [113], which are closely linked to the generation of direct products. We present several results concerning these kinds of elements which are important for the later parts of this chapter. In Section 2.3, we discuss how to compute the indecomposable elements in a finite or finitely presented semigroup, and we discuss how to obtain non-trivial factorizations of the decomposable elements in such semigroups.

Let S be a semigroup, and let $s \in S$ be arbitrary. We say that the element s is *decomposable* if there exist elements $u, v \in S$ such that s = uv, or equivalently, if $s \in S^2$. The element s is *indecomposable* if it is not decomposable. Therefore, an indecomposable element of a semigroup is one that cannot be written as a product of two elements in the semigroup. The set of decomposable elements in S is S^2 , and so the set of indecomposable elements in S is $S \setminus S^2$. Note that a surjective semigroup is one in which each of its elements is decomposable.

Any monoid S is surjective, since $S = 1_S S \subseteq S^2$, as is any regular or idempotent generated semigroup. Conversely, non-trivial zero semigroups are not surjective: if x is a non-zero element in a zero semigroup S, then $uv = 0_S \neq x$ for all $u, v \in S$. Other examples of semigroups that are not surjective are free semigroups, and monogenic semigroups that are not groups.

As discussed in [78, Section 6] and [113, Section 2], any generating set for a semigroup S contains the set of indecomposable elements $S \setminus S^2$, since an indecomposable element cannot be written as a product of other elements. It follows that the set of indecomposable elements in a finitely generated semigroup is finite. By the same token, if the set of indecomposable elements in a semigroup is infinite, then the semigroup is not finitely generated. It is for this reason that the conditions relating to surjective semigroups appear in Proposition 2.6.

Let S_1, \ldots, S_n be semigroups. If an element $(s_1, \ldots, s_n) \in S_1 \times \cdots \times S_n$ is decomposable, then, by definition, there exist elements $(x_1, \ldots, x_n), (y_1, \ldots, y_n)$ in the direct product such that

$$(s_1, \ldots, s_n) = (x_1, \ldots, x_n)(y_1, \ldots, y_n).$$

In particular, $s_i = x_i y_i$ for each *i*, and so each s_i is decomposable. It is clear that the converse also holds. To summarise, an element of a direct product is decomposable if and only if each of its components is decomposable, and so a direct product is surjective if and only if each of its factors is surjective. Therefore, computing whether a direct product is surjective, and computing the decomposable elements in a direct product, can be reduced to computing independently in the factors.

It follows that an element (s_1, \ldots, s_n) is indecomposable if and only if s_i is indecomposable for some *i*. Since a generating set for a semigroup contains the indecomposable elements, we conclude that every generating set for the direct product $S_1 \times \cdots \times S_n$ contains the set of all indecomposable elements

$$\bigcup_{i=1}^{n} \left(S_1 \times \dots \times S_{i-1} \times \left(S_i \setminus S_i^2 \right) \times S_{i+1} \times \dots \times S_n \right).$$
(2.7)

The cardinality of this set gives a lower bound on the rank of the direct product. If the sets in this union are finite, then the cardinality of the union may be found with the principle of inclusion-exclusion. In $\mathbb{N} \times \mathbb{N}$, the set of indecomposable elements $(\{1\} \times \mathbb{N}) \cup (\mathbb{N} \times \{1\})$ is infinite, and so the semigroup is not finitely generated, as shown above.

The notions of decomposable and indecomposable semigroup elements are closely related to factorizations. A factorization of an element s over a set A is a sequence of elements of A whose product is s, and a factorization is non-trivial if the sequence consists of two or more elements. Let S be a semigroup with a generating set X, and let $s \in S$ be arbitrary. If s = uvfor some $u, v \in S$, then a non-trivial factorization of s over X can be obtained by concatenating a factorization of u over X with a factorization of v over X. On the other hand, if s has a non-trivial factorization over X, i.e. if $s = x_1 \cdots x_m$ for some $x_1, \ldots, x_m \in X$ with $m \ge 2$, then $s = x_1(x_2 \cdots x_m) \in S^2$. Therefore, an element $s \in S$ is decomposable in S if and only if there exists a non-trivial factorization of s over X.

Another connection between factorizations and decomposable elements is given by the following lemma, which we use repeatedly in later results.

Lemma 2.8. Let S be a semigroup generated by a set X, and let $s \in S$. Then there exist arbitrary long factorizations of s over $S^2 \cap X$, or s can be expressed as a product in X that includes an indecomposable element.

Proof. Assume that if $s = x_1 \cdots x_m$ for any $x_1, \ldots, x_m \in X$, then $x_1, \ldots, x_m \in S^2 \cap X$. Let $x_1, \ldots, x_m \in X$ be such that $s = x_1 \cdots x_m$. Then since x_m is decomposable, there exist elements $y_1, \ldots, y_n \in X$, $n \ge 2$, such that $x_m = y_1 \cdots y_n$, and hence $s = x_1 \cdots x_{m-1} y_1 \cdots y_n$ is a strictly longer factorization of s over $S^2 \cap X$.

2.3 Computing indecomposable elements and non-trivial factorizations

In this section, we discuss techniques for computing the indecomposable elements of a semigroup, and for computing non-trivial factorizations of decomposable elements, in finite and finitely presented semigroups. These steps are required to construct the generating sets described in the main theorem of this chapter. In particular, to apply the results of Theorem 2.17 to direct products involving a semigroup $S = \langle X \rangle$, we require the ability, for each decomposable generator $s \in S^2 \cap X$, to construct elements $a, b \in S$ such that $s \in aX \cap Xb$. Such elements can be constructed from any non-trivial factorization $x_1 \cdots x_m$ of s over X, by defining $a = x_1 \cdots x_{m-1}$ and $b = x_2 \cdots x_m$. The indecomposable elements of S are those elements in Xfor which there does not exist a non-trivial factorization over X.

The indecomposable elements of a semigroup can also be described in terms of the partial order of \mathcal{J} -classes of the semigroup, as is done in the following lemma.

Lemma 2.9. Let S be a semigroup and let $s \in S$ be arbitrary. Then s is indecomposable if and only if $\{s\}$ is a maximal \mathcal{J} -class of S and $s^2 \neq s$.

Proof. (\Rightarrow) Suppose that s is indecomposable. Certainly $s^2 \neq s$. Suppose that $J_s \leq J_t$ for some $t \in S$. Then s = utv for some $u, v \in S^1$, and since s is indecomposable, it follows that $u = v = 1_S$, and s = t. Thus $J_s = \{s\}$ is a maximal \mathscr{J} -class of S.

(\Leftarrow) Suppose that s is decomposable and that J_s is a maximal \mathscr{J} -class of S. Then s = uv for some $u, v \in S$, and $J_s \leq J_u$ and $J_s \leq J_v$ by Lemma 1.12. By the maximality of $J_s = \{s\}$, it follows that $J_u = J_v = \{s\}$. In particular, s = u = v, and s is idempotent.

In a finite semigroup, Green's \mathcal{D} - and \mathcal{J} -relations coincide [76, Proposition 2.1.4]. A maximal \mathcal{J} -class in a finite semigroup is hence either a regular \mathcal{D} -class or a non-regular \mathcal{D} -class consisting of a single non-idempotent element. Thus we obtain the following corollary to Lemma 2.9.

Corollary 2.10 (cf. Lemma 2.2 in [112]). Let S be a finite semigroup. Then S is surjective if and only if every maximal \mathcal{J} -class of S is regular.

Hence, to determine whether a finite semigroup is surjective, or to find the indecomposable elements in a finite semigroup, it suffices to find its partial order of \mathscr{J} -classes, to find the maximal elements in this partial order, and to test which of them consists of a single non-idempotent; see Algorithm 2.11. If S is a finite semigroup to which the techniques of [37] apply, then the partial order of \mathscr{J} -classes of S can be found by the algorithm described in [37, Algorithm 14], which does not necessarily exhaustively enumerate the elements of the semigroup. The partial order of \mathscr{J} -classes of a finite semigroup can also be constructed from the quotient of the union of the left and right Cayley digraphs by its strongly connected components.

Algorithm 2.11 Compute the indecomposable elements in a finite semigroup using the partial order of \mathcal{J} -classes.

Input: A finite semigroup S with a generating set. **Output:** The set A of indecomposable elements in S. 1: $A \leftarrow \emptyset$ 2: $\mathfrak{J} \leftarrow$ the maximal \mathscr{J} -classes of $S \Rightarrow [37, \text{Algorithm 14}]$, or via the Cayley digraphs of S3: for $J_x \in \mathfrak{J}$ do 4: if $|J_x| = 1$ and $x^2 \neq x$ then 5: $A \leftarrow A \cup \{x\}$ 6: return A.

However, when computing with the Cayley digraphs of a semigroup, there are more straightforward ways of finding indecomposable elements; see Algorithm 2.12. We formulate the following in terms of right Cayley digraphs, however analogous statements hold concerning left Cayley digraphs. Let S be a finite semigroup with a generating set X, and let Γ be the right Cayley digraph of S with respect to X. If $s = x_1 \cdots x_n$ is a non-trivial factorization of s over X, then $(x_1, x_1x_2, \cdots, x_1 \cdots x_n = s)$ is a path in Γ , of length at least 1, from a generator to s. In particular, s has an in-neighbour in Γ . Therefore, if s has no in-neighbour in Γ , then there exists no non-trivial factorization of s, and s is indecomposable. Conversely, if s has an in-neighbour u in Γ , then there exists a generator $x \in X$ that corresponds to the edge (u, s), i.e. s = ux. Therefore, a non-trivial factorization of s over X can be found by appending x to any factorization of u over X. It follows that the indecomposable elements of a semigroup are the sources in its right Cayley digraph, with respect to any generating set.

A factorization of an element $u \in S$ over X can be calculated from Γ by searching for a path in Γ from a generator to u. Finding such a path can be achieved by performing a breadth- or depth-first search in Γ , starting at the vertex of each generator. The factorization is then given by the relevant generator, followed by the generators corresponding to the edges in the path. Note that a path may have length 0, which gives a trivial factorization.

Algorithm 2.12 Compute a non-trivial factorization for an element	in a finite semigroup, or
determine that the element is indecomposable, by using the right Cay	vley digraph.
Input: A finite semigroup S with a generating set X , and an elemen	t $s \in S$.
Output: A non-trivial factorization of s over X , if one exists, else FA	AIL.
1: $\Gamma \leftarrow$ the right Cayley digraph of S with respect to X	\triangleright [54]; Algorithm 1.21
2: if s is a source of Γ then	
3: return FAIL.	
4: $u \leftarrow $ an in-neighbour of s in Γ	
5: $x \leftarrow$ a generator in X corresponding to the edge (u, s)	
6: $[x_1, \ldots, x_n] \leftarrow a$ (possibly trivial) factorization of u over X	\triangleright See main text
7: return $[x_1, \ldots, x_n, x]$.	

For a finite semigroup $S = \langle X \rangle$ to which the techniques of [37] apply, we can obtain nontrivial factorizations of decomposable elements over X without constructing either the left or right Cayley digraphs of S, and without necessarily enumerating the semigroup. We briefly outline a method for doing so below; see Algorithm 2.13 for pseudocode describing this method.

Algorithm 13 in [37] gives a method for factorizing an arbitrary element of S over X. However, when $x \in X$, this algorithm may return the trivial factorization.

Let $x \in X$ be arbitrary. If x has a relative right identity, then a relative right identity for x can be constructed from the data structure used in [37]. This step is complicated to describe, and we do not include details here, however we note that this step has been implemented in an upcoming version of the SEMIGROUPS package [101] for GAP [58], by the author. Given a relative right identity f for x, it follows that $x = xx_1 \cdots x_n$, where $f = x_1 \cdots x_n$ is any factorization of f over X. If x has a relative left identity, then a non-trivial factorization of x over X may be obtained analogously.

Suppose that x has neither a relative left identity nor a relative right identity in S. If $y \in R_x \setminus \{x\}$, then there exist elements $u, v \in S$ such that y = xu and x = yv; in particular, x = yv = x(uv), and x has a relative right identity. Therefore $R_x = \{x\}$, and similarly $L_x = \{x\}$. We conclude that $R_x = L_x = J_x = \{x\}$. It follows that x is contained in any transversal of the set of \mathscr{R} -classes of S. Therefore, in the data structure returned by [37, Algorithm 11], either we find a generator $x' \in X$ and a representative r of an \mathscr{R} -class of S such that x = x'r, or we determine that R_x is a maximal \mathscr{R} -class of S. In the first case, we obtain a non-trivial factorization of x over X by first factorizing r over X. In the second case, it is straightforward to show that $\{x\}$ is also a maximal \mathscr{L} -class of S, and therefore a maximal \mathscr{J} -class of S, too. It follows by Lemma 2.9 that x is indecomposable, or equivalently, that

there exists no non-trivial factorization of x over X.

0 10 0

Algorithm 2.13 Compute a non-trivial factorization for an element in a finite semigroup, or	
determine that the element is indecomposable, by using the data st	ructure described in $[37]$.
Input: A finite semigroup S with a generating set X , and an elem	ent $s \in S$.
Output: A non-trivial factorization of s over X , if one exists, else	FAIL.
1: $[x_1, \ldots, x_n] \leftarrow a$ factorization of s over X	\triangleright [37, Algorithm 13]
2: if $n > 1$ then	\triangleright If $n = 1$, then $s \in X$
3: return $[x_1,, x_n]$.	
4: else if s has a relative left identity then	
5: $e \leftarrow \text{a relative left identity for } s$	
6: $[y_1, \ldots, y_m] \leftarrow a$ factorization of e over X	
7: return $[y_1,, y_m, s].$	
8: else if s has a relative right identity then	
9: $f \leftarrow$ a relative right identity for s	
10: $[y_1, \ldots, y_m] \leftarrow a$ factorization of f over X	
11: return $[s, y_1,, y_m]$.	
12: $\mathfrak{R} \leftarrow a$ set of \mathscr{R} -class representatives for S	\triangleright [37, Algorithm 11]
13: for $x \in X, r \in \mathfrak{R}$ do	
14: if $s = xr$ then	
15: $[y_1, \ldots, y_m] \leftarrow \text{a factorization of } r \text{ over } X$	
16: return $[x, y_1, \ldots, y_m]$.	
17: return FAIL.	

Of course, it is not possible to construct an algorithm to enumerate the indecomposable elements of an arbitrary semigroup, since an infinite semigroup (for example, an infinite zero semigroup) may contain an infinite number of indecomposable elements. However, as mentioned above, the set of indecomposable elements of a semigroup is contained in any generating set. It follows that the problem of enumerating the indecomposable elements in a semigroup defined by a finite generating set is a more appropriate problem to attempt to solve.

One way of specifying an infinite semigroup is by a semigroup presentation. The elements of a semigroup defined by a presentation are equivalence classes of finite sequences over the generators. The sequences in an equivalence class give the factorizations of the element over the generators. Therefore, an element has a non-trivial factorization if and only if its equivalence class contains a non-trivial sequence. It follows that a generator has a non-trivial factorization in the semigroup if and only if the presentation relates that generator to a non-trivial sequence. We clarify this observation in the following lemma.

Lemma 2.14. Let S be the semigroup defined by the presentation $\langle X | R \rangle$, and suppose that R contains no relations of the form x = y, for any $x, y \in X$. Then for each $x \in X$, the generator of S corresponding to x is decomposable if and only if x = w or w = x belongs to R, for some $w \in X^+$. Moreover, w gives a non-trivial factorization of x over X.

It follows that if S is a semigroup defined by a finite presentation of the kind described in Lemma 2.14, then the indecomposable generators, and non-trivial factorizations of the decomposable generators, can be obtained by searching the relations according to Lemma 2.14. Note that relations of the form x = y, for generators x and y, can be easily removed from any finite presentation to yield an equivalent presentation.

2.3.1 Corresponding features in the Semigroups package for GAP

The techniques described in this section have been implemented by the author in the development version of the SEMIGROUPS package [101] for GAP [58].

If S is a finite semigroup, or a semigroup defined by a finite presentation, then the command IndecomposableElements(S); returns a list of the indecomposable elements in S. If S is already known to be a surjective semigroup (because it is known to be a monoid or a regular semigroup, for example), then IndecomposableElements returns an empty list. Otherwise, if S is defined by a finite presentation, then the function IndecomposableElements uses Lemma 2.14 to compute the indecomposable elements. Otherwise, IndecomposableElements uses Lemma 2.9 to compute the indecomposable elements via the partial order of \mathcal{J} -classes of S. The command IsSurjectiveSemigroup(S); returns true if IndecomposableElements(S); returns an empty list, and false otherwise.

If S is a finite semigroup, X is the generating set of S given by GeneratorsOfSemigroup(S); and $s \in S$, then NonTrivialFactorization(S, s); returns a list defining a non-trivial factorization of s over X, if one exists, else it returns fail. If $s \in S \setminus X$, then any factorization of s over X is non-trivial. In this case, the function NonTrivialFactorization delegates to the pre-existing SEMIGROUPS package function Factorization. Otherwise, the function NonTrivialFactorization executes a version of either Algorithm 2.12 or Algorithm 2.13, as appropriate. Suppose that NonTrivialFactorization(S, s); returns a list L of length n > 1. Then the non-trivial factorization of s over X is given by

$$s = X[L[1]] \cdot X[L[2]] \cdots X[L[n]].$$

2.4 Arbitrary direct products of semigroups

As proved by Araújo [2, Theorem 3.39] and repeated in Proposition 2.6, the direct product of semigroups $S_1 \times \cdots \times S_n$ is finitely generated if and only if each semigroup S_i is finitely generated, and either:

- (i) S_i is surjective for all i; or
- (ii) S_i is finite for all i; or
- (iii) S_i is infinite for some *i*, and S_j is finite and surjective for all $j \neq i$.

Given an arbitrary collection of semigroups whose direct product is finitely generated, the proof of this result could, in principle, be used to obtain a finite generating set for the direct product. However, it appears that this result was not proven with the aim of computing such finite generating sets. In this section, we build on this work by directly describing finite generating sets for any finitely generated direct product, with the aim of computing generating sets that are, in some sense, relatively small.

The following result of Araújo, which was an improvement to [113, Proposition 2.5 and Corollary 2.7], describes a generating set for the direct product of two surjective semigroups. The main theorem of this section was inspired by this result and its proof.

Proposition 2.15 ([2, Proposition 3.5 and Corollary 3.6]). Let $S = \langle X \rangle$ and $T = \langle Y \rangle$ be surjective semigroups. For each $x \in X$, choose $b_x \in S$ such that $x \in Xb_x$, and for each $y \in Y$, choose $a_y \in T$ such that $y \in a_yY$. Define $A = \{a_y : y \in Y\}$ and $B = \{b_x : x \in X\}$. Then the set $(X \times A) \cup (B \times Y)$ generates $S \times T$. In particular, rank $(S \times T) \leq 2$ rank(S)rank(T). *Proof.* Let $(s, t) \in S \times T$ be arbitrary, and let $s = x_1 \cdots x_{m-1} x_m$ and $t = y_1 y_2 \cdots y_n$ be nontrivial factorizations of s and t over X and Y, respectively. We may iteratively replace the generator that follows x_{m-1} in the factorization of s by a product xb, for some $x \in X$ and $b \in B$. Similarly, we may replace the generator in $y_1 y_2 \cdots y_n$ that precedes y_2 by a product ay, for some $a \in A$ and $y \in Y$. In particular, if we repeat this process n and m times, respectively, we conclude that

$$s = x_1 \cdots x_{m-1} x' b_1 b_2 \cdots b_n$$
 and $t = a_1 \cdots a_{m-1} a_m y' y_2 \cdots y_n$

for some $x' \in X$ and $b_1, \ldots, b_n \in B$, and for some $y' \in Y$ and $a_1, \ldots, a_m \in A$. Therefore

$$(s, t) = (x_1, a_1) \cdots (x_{m-1}, a_{m-1}) (x', a_m) (b_1, y') (b_2, y_2) \cdots (b_n, y_n)$$

$$\in (X \times A)^m (B \times Y)^n,$$

and so $(X \times A) \cup (B \times Y)$ generates $S \times T$. Since $|A| \leq |Y|$ and $|B| \leq |X|$ by construction, it follows that $|(X \times A) \cup (B \times Y)| \leq 2|X||Y|$. The upper bound on rank $(S \times T)$ is obtained by choosing generating sets X and Y of minimal cardinality.

The example in [113, Example 2.8] presents a pair of surjective semigroups, of arbitrary finite ranks, whose direct product reaches the upper bound given in Proposition 2.15. In this sense, the bound given in Proposition 2.15 is optimal.

As noted in [113, Remark 2.6], since a direct product is surjective if and only if each of its factors is surjective, we may use Proposition 2.15 to describe generating sets for arbitrary direct products of surjective semigroups, i.e. for any direct product of type (i) from Proposition 2.6. More specifically, given surjective semigroups S_1, \ldots, S_n defined by finite generating sets, we may find a finite generating set for $S_1 \times \cdots \times S_n$ by first using Proposition 2.15 to construct a finite generating set for $S_1 \times S_2$, which is itself surjective. We may then use Proposition 2.15 again to construct a finite generating set for $(S_1 \times S_2) \times S_3$, and so on, iterating in this way until complete. We may use this observation in an inductive argument to obtain $2^{n-1} \cdot \operatorname{rank}(S_1) \cdots \operatorname{rank}(S_n)$ as an upper bound for the rank of the direct product of any collection of surjective semigroups S_1, \ldots, S_n . However, for n > 2, this bound is not tight; we improve on this bound in Corollary 2.18.

Any direct product of type (ii) is finite, and so from the point of view of finite generation there is nothing to be solved, since the semigroup itself defines a finite generating set. However, for the purposes of computation, smaller generating sets are usually better. Using the set of all elements for computation is not satisfactory, in general. Indeed, the rank of a direct product of finite semigroups may be much smaller than its cardinality, even when the semigroups are not all surjective; see Example 2.16.

Example 2.16. Let $n \in \mathbb{N}$, $n \geq 2$, let $S = \langle x \rangle$ be a monogenic semigroup of order n that is not a group, and let \mathcal{I}_1 be the symmetric inverse monoid of degree 1. Note that S is not surjective, and that \mathcal{I}_1 is an inverse monoid of order 2 with identity element id_1. Therefore, the order of $S \times \mathcal{I}_1$ is 2n. If $(x^m, y) \in S \times \mathcal{I}_1$, then $(x^m, y) = (x, y) (x, id_1)^{m-1}$. It follows that $\{x\} \times \mathcal{I}_1$ is a generating set for $S \times \mathcal{I}_1$ that contains only two elements.

As far as we are aware, there are no results in the literature directly describing finite generating sets for direct products of type (iii).

The main result of this section is the following, which describes a generating set for an arbitrary direct product of semigroups. This theorem can be used, in combination with the techniques described in Section 2.3, to construct a generating set for a finitely generated direct product of finite or finitely presented semigroups. The proof of this theorem, which appears towards the end of this section, relies on several forthcoming lemmas.

Theorem 2.17. Let $n \in \mathbb{N}$. For each $i \in \{1, \ldots, n\}$, let S_i be a semigroup generated by a set X_i , and for each $x \in S_i^2 \cap X_i$, choose $a_x, b_x \in S_i$ such that $x \in a_x X_i \cap X_i b_x$, and define $A_i = \{a_x : x \in S_i^2 \cap X_i\}$ and $B_i = \{b_x : x \in S_i^2 \cap X_i\}$. Let

$$\Gamma = \bigcup_{i=1}^{n} \Big((B_1 \times \dots \times B_{i-1} \times (S_i^2 \cap X_i) \times A_{i+1} \times \dots \times A_n) \\ \cup (S_1 \times \dots \times S_{i-1} \times (S_i \setminus S_i^2) \times S_{i+1} \times \dots \times S_n) \Big).$$

Then the direct product $S_1 \times \cdots \times S_n$ is generated by Γ .

Suppose that the generating sets X_i in the statement of Theorem 2.17 are finite. For each $i \in \{1, \ldots, n\}$, the sets

$$B_1 \times \cdots \times B_{i-1} \times (S_i^2 \cap X_i) \times A_{i+1} \times \cdots \times A_n$$

contain at most $|X_1| \cdots |X_n|$ elements, since $|A_i| \le |X_i|$ and $|B_i| \le |X_i|$ by construction. In particular, these sets are finite when the generating sets X_i are finite. Furthermore, the sets

$$S_1 \times \cdots \times S_{i-1} \times (S_i \setminus S_i^2) \times S_{i+1} \times \cdots \times S_n$$

comprise the indecomposable elements of the direct product, and are therefore contained in any generating set for the direct product. Hence these sets are finite when the direct product is finitely generated. In particular, Γ is finite when the generating sets X_i are finite and the direct product is finitely generated. In the worst case, the generating set Γ from the statement of Theorem 2.17 contains at most $n \cdot |X_1| \cdots |X_n|$ redundant elements; see Corollary 2.18.

When n = 2 and S_1 and S_2 are surjective, Theorem 2.17 implies that $S_1 \times S_2$ is generated by $(X_1 \times A_2) \cup (B_1 \times X_2)$. Therefore, Proposition 2.15 is a special case of Theorem 2.17.

Let S_1, \ldots, S_n be a collection of finitely generated semigroups whose direct product is finitely generated, and suppose that each semigroup S_i is defined by a finite generating set X_i . By Theorem 2.17, in order to construct a generating for the direct product $S_1 \times \cdots \times S_n$, it suffices to determine, for each $i \in \{1, \ldots, n\}$, which generators in X_i are decomposable and which are indecomposable, and for each decomposable generator x, to find elements $a_x, b_x \in S_i$ such that $x \in a_x X_i \cap X_i b_x$. In order to reduce the size of the sets $\{a_x : x \in S_i^2 \cap X_i\}$ and $\{b_x : x \in S_i^2 \cap X_i\}$, and hence to reduce the size of the corresponding generating set, the elements a_x and b_x should be chosen carefully. For example, if S_i has a left identity e, then it would be sensible to choose $a_x = e$ for all $x \in S_i^2 \cap X_i$. More generally, the elements a_x and b_x can be constructed from non-trivial factorizations, as discussed in Section 2.3.

Note that the sets A_1 and B_n are not required by the generating set described in Theorem 2.17, and therefore they are not required to be constructed in any algorithm which implements the result given in Theorem 2.17.

Methods for computing a generating set for the direct product of an arbitrary collection of finite semigroups, based on Theorem 2.17, have been implemented by the author in the development version of the SEMIGROUPS package [101] for GAP [58]. These methods will be included in a future release of this software. With these new methods, the function DirectProduct may be applied to any collection of finite semigroups to produce a semigroup that is isomorphic to their direct product. If the arguments are all semigroups of partial permutations, then their direct product is returned as a semigroup of partial permutation; similarly, if the factors are all semigroups of partitioned binary relations (defined in [96]), then their direct product is returned as a semigroup of the same kind. Otherwise, the direct product is returned as a transformation semigroup. Given a direct product P of the factors S_1, \ldots, S_n and some $i \in \{1, \ldots, n\}$,

the command Embedding(P, i); returns an embedding of the i^{th} factor S_i into P, and the command Projection(P, i); returns a projection of P onto its i^{th} factor S_i .

We may use Theorem 2.17 to describe bounds on the ranks of arbitrary direct products.

Corollary 2.18. Let S_1, \ldots, S_n be semigroups, and let m denote the cardinality of

$$\bigcup_{i=1}^{n} \left(S_1 \times \cdots \times S_{i-1} \times \left(S_i \setminus S_i^2 \right) \times S_{i+1} \times \cdots \times S_n \right)$$

Then

$$m \leq \operatorname{rank}(S_1 \times \cdots \times S_n) \leq m + n \cdot \operatorname{rank}(S_1) \cdots \operatorname{rank}(S_n)$$

In particular, if each semigroup S_i is surjective, then

$$\operatorname{rank}(S_1 \times \cdots \times S_n) \le n \cdot \operatorname{rank}(S_1) \cdots \operatorname{rank}(S_n).$$

Proof. Note that m is the number of indecomposable elements of the direct product $S_1 \times \cdots \times S_n$. Therefore, the first inequality holds. The second inequality can be shown to hold by counting the size of the generating set given in Theorem 2.17. For each $i \in \{1, \ldots, n\}$, choose X_i to be a generating set for S_i of minimal cardinality, and define sets A_i and B_i as in Theorem 2.17. Then the generating set is the union of (2.7), which contains the m indecomposable elements, along with the set

$$\bigcup_{i=1}^{n} (B_1 \times \cdots \times B_{i-1} \times X_i \times A_{i+1} \times \cdots \times A_n).$$

Since $|A_i| \leq |X_i|$ and $|B_i| \leq |X_i|$ for each *i* by construction, it follows that

 $|B_1 \times \cdots \times B_{i-1} \times X_i \times A_{i+1} \times \cdots \times A_n| \le |X_1| \cdots |X_n| = \operatorname{rank}(S_1) \cdots \operatorname{rank}(S_n),$

for all $i \in \{1, ..., n\}$. Therefore, the second inequality holds. If each semigroup S_i is surjective, then m = 0, and so the final inequality follows from the second.

By adapting the example used in [113, Example 2.8], we may show that the upper bound given by Corollary 2.18 for the rank of a direct product of surjective semigroups is tight.

Example 2.19 (cf. [113, Example 2.8]). Let $n \in \mathbb{N}$. For each $i \in \{1, \ldots, n\}$, let $m_i \in \mathbb{N}$, and define a semigroup

$$S_{i} = \begin{cases} C_{i,1} & \text{if } m_{i} = 1, \\ C_{i,1} \cup \dots \cup C_{i,m_{i}} \cup \{0\} & \text{if } m_{i} > 1, \end{cases}$$

where each $C_{i,j}$ is a cyclic group of order 2, and the product of elements within distinct groups, or involving 0, is defined to be 0. It is straightforward to see that S_i is a commutative inverse semigroup, and that the unique minimal generating set of S_i consists of the non-identity element of each $C_{i,j}$. In particular, S_i is surjective and rank $(S_i) = m_i$. It follows by Corollary 2.18 that

$$\operatorname{rank}(S_1 \times \cdots \times S_n) \le n \cdot \operatorname{rank}(S_1) \cdots \operatorname{rank}(S_n) = n \cdot m_1 \cdots m_n.$$

For each $i \in \{1, \ldots, n\}$, choose $j_i \in \{1, \ldots, m_i\}$ arbitrarily. Note that $C_{1,j_1} \times \cdots \times C_{n,j_n}$ is a direct product of n cyclic groups of order 2, and so

$$\operatorname{rank}(C_{1,j_1} \times \cdots \times C_{n,j_n}) = n.$$

But $(S_1 \times \cdots \times S_n) \setminus (C_{1,j_1} \times \cdots \times C_{n,j_n})$ is an ideal of $S_1 \times \cdots \times S_n$, which implies that any generating set for $S_1 \times \cdots \times S_n$ contains a generating set for $C_{1,j_1} \times \cdots \times C_{n,j_n}$. Since there are m_i choices for each j_i , it follows that any generating set for $S_1 \times \cdots \times S_n$ contains at least $n \cdot m_1 \cdots m_n$ elements. Therefore

$$\operatorname{rank}(S_1 \times \cdots \times S_n) = n \cdot \operatorname{rank}(S_1) \cdots \operatorname{rank}(S_n).$$

We also deduce the following corollary to Theorem 2.17.

Corollary 2.20. Let S be a semigroup and suppose that $S = \langle S \setminus S^2 \rangle$. Then, with respect to containment, any direct product involving S has a unique minimal generating set, which consists of the indecomposable elements in the direct product. In particular, $S \setminus S^2$ is the unique minimal generating set for S, with respect to containment.

Proof. By applying Theorem 2.17 to a collection of semigroups involving S with its generating set $S \setminus S^2$, we deduce that any direct product involving S is generated by its indecomposable elements. On the other hand, any generating set contains the indecomposable elements. \Box

It follows by Corollary 2.20 that the generating set for the semigroup $S \times \mathcal{I}_1$ described in Example 2.16 is the unique minimal generating set of $S \times \mathcal{I}_1$, with respect to containment.

In order to prove Theorem 2.17, we first prove the following two technical lemmas. We note that the set Γ from the statement of the theorem contains $B_1 \times \cdots \times B_{i-1} \times X_i \times A_{i+1} \times \cdots \times A_n$ for each *i*, since $X_i = (S_i^2 \cap X_i) \cup (S_i \setminus S_i^2)$. We use this observation in the following proofs.

Lemma 2.21. For all $i \in \{1, ..., n-1\}$ and $k \in \mathbb{N}_0$,

$$S_1 \times \dots \times S_{i-1} \times \left(\left(S_i \setminus S_i^2 \right) \underbrace{B_i \cdots B_i}_{k \ times} \right) \times S_{i+1} \times \dots \times S_n \subseteq \langle \Gamma \rangle.$$

$$(2.22)$$

Proof. Let $i \in \{1, \ldots, n-1\}$ be arbitrary. Note that if S_i is surjective, then the set $S_i \setminus S_i^2$, and consequently the product of sets in (2.22), is empty and so there is nothing to prove. Otherwise, we prove by induction that (2.22) holds for all $k \in \mathbb{N}_0$. Certainly (2.22) holds when k = 0, since Γ contains the indecomposable elements of $S_1 \times \cdots \times S_n$. Suppose that (2.22) holds for some k = l, where $l \in \mathbb{N}_0$, and let

$$\mathbf{s} = (s_1, \dots, s_{i-1}, s_i b_{i,1} \cdots b_{i,l+1}, s_{i+1}, \dots, s_n)$$

$$\in S_1 \times \cdots \times S_{i-1} \times \left(\left(S_i \setminus S_i^2 \right) \underbrace{B_i \cdots B_i}_{l+1 \text{ times}} \right) \times S_{i+1} \times \cdots \times S_n$$

be arbitrary. We aim to show that $\mathbf{s} \in \langle \Gamma \rangle$.

If s_j is indecomposable for any $j \neq i$, then **s** is indecomposable, and $\mathbf{s} \in \Gamma$. Suppose instead that s_j is decomposable for each $j \neq i$. Thus we may decompose s_j as a product $u_j x_j$, for some $u_j \in S_j$ and $x_j \in X_j$. Since we have assumed that (2.22) holds for k = l, it follows that

$$\mathbf{u} = (u_1, \ldots, u_{i-1}, s_i b_{i,1} \cdots b_{i,l}, u_{i+1}, \ldots, u_n) \subseteq \langle \Gamma \rangle.$$

If x_j is indecomposable for any $j \neq i$, then $\mathbf{x} = (x_1, \ldots, x_{i-1}, b_{i,l+1}, x_{i+1}, \ldots, x_n)$ is indecomposable. Therefore $\mathbf{x} \in \Gamma$, and so $\mathbf{s} = \mathbf{u}\mathbf{x} \in \langle \Gamma \rangle$, as required. Otherwise, x_j is decomposable, and is therefore contained in $A_j X_j \cap X_j B_j$, for each $j \neq i$. Hence for each $j \in \{1, \ldots, n-1\} \setminus \{i\}$, we may express x_j as a product $x'_j b_j$, for some $x'_j \in X_j$ and $b_j \in B_j$, and we may express x_n as a product $a_n x'_n$, for some $x'_n \in X_n$ and $a_n \in A_n$. Since we have assumed that (2.22) holds for k = l, it follows that

$$\mathbf{v} = (u_1 x'_1, \dots, u_{i-1} x'_{i-1}, s_i b_{i,1} \cdots b_{i,l}, u_{i+1} x'_{i+1}, \dots, u_{n-1} x'_{n-1}, u_n a_n) \subseteq \langle \Gamma \rangle,$$

and by definition,

$$\mathbf{b} = (b_1, \ldots, b_{i-1}, b_{i,l+1}, b_{i+1}, \ldots, b_{n-1}, x'_n) \in B_1 \times \cdots \times B_{n-1} \times X_n \subseteq \Gamma$$

Therefore $\mathbf{s} = \mathbf{vb} \in \langle \Gamma \rangle$. Since \mathbf{s} was arbitrary, it follows that (2.22) holds for k = l + 1. By induction, (2.22) holds for all $k \in \mathbb{N}_0$.

Lemma 2.23. $X_1 \times \cdots \times X_n \subseteq \langle \Gamma \rangle$.

Proof. If n = 1, then $X_1 = \Gamma$, and so we assume for the remainder of this proof that n > 1. Let $\mathbf{x} = (x_1, \ldots, x_n) \in X_1 \times \cdots \times X_n$ be arbitrary. It suffices to show that $\mathbf{x} \in \langle \Gamma \rangle$. If x_i is indecomposable for some *i*, then \mathbf{x} is itself indecomposable, and we are done. Otherwise, x_i is decomposable for each *i*, i.e. $x_i \in S_i^2 \cap X_i$. In order to show that $\mathbf{x} \in \langle \Gamma \rangle$ in this case, we introduce the following sets, which are defined in terms of the sets X_i , A_i , and B_i from the statement of Theorem 2.17. For each $i \in \{1, \ldots, n\}$ and $k \geq 2$, we define

$$U_{i,k} = \begin{cases} X_1 B_1^{k-1} & \text{if } i = 1, \\ A_i^{k-1} X_i & \text{if } k \le i, \\ A_i^{i-1} X_i B_i^{k-i} & \text{if } 1 < i < k, \end{cases}$$

and
$$\mathbf{C}_i = B_1 \times \dots \times B_{i-1} \times X_i \times A_{i+1} \times \dots \times A_n \subseteq \Gamma$$

Note that $U_{1,k} \times \cdots \times U_{n,k} = \mathbf{C}_1 \cdots \mathbf{C}_{k-1} \cdot (B_1 \times \cdots \times B_{k-1} \times X_k \times \cdots \times X_n)$ for $2 \le k < n$, and that $U_{1,n} \times \cdots \times U_{n,n} = \mathbf{C}_1 \cdots \mathbf{C}_n$. By finding values of k for which $x_i \in U_{i,k}$ for all i, we aim to show that \mathbf{x} is contained in a product of subsets of Γ , including those of the form \mathbf{C}_i .

Note that $U_{1,2} = X_1B_1$ and $U_{i,2} = A_iX_i$ for all $i \ge 2$. Since we have assumed that x_i is decomposable for each i, and since $S_i^2 \cap X_i \subseteq (A_iX_i \cap X_iB_i)$ by construction of the sets A_i and B_i , it follows that $x_i \in U_{i,2}$ for all i. However, it is not necessary that $x_i \in U_{i,k}$ for all k > 2.

Let *i* be arbitrary. By definition, $x_i \in U_{i,k}$ for some *k* if and only if x_i can be expressed as a product of length *k* consisting of min $\{i, k\} - 1$ elements of A_i , followed by a generator in X_i , followed by $k - \min\{i, k\}$ elements of B_i . Suppose that $x_i \in U_{i,k}$ for some *k*, and let $a_1 \cdots a_m x' b_1 \cdots b_q$ be a corresponding expression of length *k*. If x' is decomposable, then $x' \in A_i X_i \cap X_i B_i$. In this case, by expressing x' as a product of length 2 in either $A_i X_i$ or $X_i B_i$, as appropriate, it follows that x_i can be expressed as a product of length k + 1 consisting of min $\{i, k+1\}-1$ elements of A_i , followed by a generator in X_i , followed by $(k+1)-\min\{i, k+1\}$ elements of B_i . In other words, $x_i \in U_{i,k+1}$. Hence if $x_i \in U_{i,k} \setminus U_{i,k+1}$ for some *k*, and $a_1 \cdots a_m x' b_1 \cdots b_q$ is a corresponding expression for x_i of length *k*, then $x' \in S_i \setminus S_i^2$.

We define a function $f: \{1, \ldots, n\} \longrightarrow \{1, \ldots, n\}$ by

$$f(i) = \min(\{n\} \cup \{k \in \mathbb{N} : x_i \notin U_{i,k+1}\}).$$

Note that $f(i) \in \{2, ..., n\}$ for all *i*. The above argument implies that, for any $i \in \{1, ..., n\}$ with f(i) < n,

$$x_{i} \in \begin{cases} (S_{1} \setminus S_{1}^{2})B_{1}^{f(1)-1} & \text{if } i = 1, \\ A_{i}^{f(i)-1}(S_{i} \setminus S_{i}^{2}) & \text{if } f(i) \leq i, \\ A_{i}^{i-1}(S_{i} \setminus S_{i}^{2})B_{i}^{f(i)-i} & \text{if } 1 < i < f(i). \end{cases}$$
(2.24)

Fix $j \in \{1, ..., n\}$ such that $f(j) = \min\{f(i) : i \in \{1, ..., n\}\}$. It follows that

$$x_i \in U_{i,k}$$
 for all $i \in \{1, ..., n\}$ and $2 \le k \le f(j)$. (2.25)

The remainder of the proof is split into four cases, according to the values of j and f(j).

Case 1: f(j) = n. By (2.25), $x_i \in U_{i,n}$ for all *i*. It follows that

$$\mathbf{x} \in U_{1,n} \times \cdots \times U_{n,n} = \mathbf{C}_1 \cdots \mathbf{C}_n \subseteq \langle \Gamma \rangle.$$

Case 2: $f(j) \leq j$ and f(j) < n. By (2.24) and (2.25), it follows that $x_j \in A_j^{f(j)-1}(S_j \setminus S_j^2)$ and $x_i \in U_{i,f(j)}$ for all $i \in \{1, \ldots, n\}$. Therefore

$$\mathbf{x} \in U_{1,f(j)} \times \cdots \times U_{j-1,f(j)} \times \left(A_j^{f(j)-1}(S_j \setminus S_j^2)\right) \times U_{j+1,f(j)} \times \cdots \times U_{n,f(j)}$$
$$\subseteq \mathbf{C}_1 \cdots \mathbf{C}_{f(j)-1} \cdot \left(S_1 \times \cdots \times S_{j-1} \times (S_j \setminus S_j^2) \times S_{j+1} \times \cdots \times S_n\right) \subseteq \langle \Gamma \rangle.$$

Case 3: j = 1 and f(j) < n. By (2.24), $x_1 \in (S_1 \setminus S_1^2)B_1^{f(1)-1}$. Therefore, by Lemma 2.21, $\mathbf{x} \in ((S_1 \setminus S_1^2)B_1^{f(1)-1}) \times S_2 \times \cdots \times S_n \subseteq \langle \Gamma \rangle.$

Case 4: 1 < j < f(j) < n. By (2.24) and (2.25), it follows that $x_j \in A_j^{j-1}(S_j \setminus S_j^2)B_j^{f(j)-j}$ and $x_i \in U_{i,j}$ for all $i \in \{1, ..., n\}$. Therefore

$$\begin{aligned} \mathbf{x} &\in U_{1,j} \times \cdots \times U_{j-1,j} \times \left(A_j^{j-1} (S_j \setminus S_j^2) B_j^{f(j)-j} \right) \times U_{j+1,j} \times \cdots \times U_{n,j} \\ &= \mathbf{C}_1 \cdots \mathbf{C}_{j-1} \cdot \left(B_1 \times \cdots \times B_{j-1} \times (S_j \setminus S_j^2) B_j^{f(j)-j} \times X_{j+1} \times \cdots \times X_n \right) \\ &\subseteq \mathbf{C}_1 \cdots \mathbf{C}_{j-1} \cdot \left(S_1 \times \cdots \times S_{j-1} \times (S_j \setminus S_j^2) B_j^{f(j)-j} \times S_{j+1} \times \cdots \times S_n \right) \subseteq \langle \Gamma \rangle, \end{aligned}$$

by Lemma 2.21.

In all cases, $\mathbf{x} \in \langle \Gamma \rangle$. Since \mathbf{x} was arbitrary, it follows that $X_1 \times \cdots \times X_n \subseteq \langle \Gamma \rangle$.

We may now prove Theorem 2.17, using Lemmas 2.21 and 2.23.

Proof of Theorem 2.17. The statement is clearly true when n = 1, so suppose that n > 1. Certainly $\langle \Gamma \rangle \leq S_1 \times \cdots \times S_n$. Therefore it remains to prove that the reverse inclusion holds, so let $(s_1, \ldots, s_n) \in S_1 \times \cdots \times S_n$ be arbitrary. We split into two cases.

In the first case, assume that, for all $i \in \{1, \ldots, n\}$, every expression for s_i in S_i includes only decomposable elements. For each index $i \in \{1, \ldots, n\}$, by Lemma 2.8, s_i can be expressed as an arbitrarily long product in $S_i^2 \cap X_i$. Choose any such expression that is non-trivial, and define l(i) > 1 to be the length of this expression. Let $i \in \{1, \ldots, n\}$ be arbitrary. Then $s_i = x_1 x_2 \cdots x_{l(i)}$ for some decomposable generators $x_1, x_2, \ldots, x_{l(i)} \in S_i^2 \cap X_i$. Since $(S_i^2 \cap X_i) \subseteq A_i X_i$, we may replace x_1 by ay, for some $a \in A_i$ and $y \in X_i$, which gives a new expression $ayx_2 \ldots x_{l(i)}$ for s_i . It follows by assumption that y is decomposable. We may then replace y by a'y', for some $a' \in A_i$ and $y' \in S_i^2 \cap X_i$, and we may continue in this fashion indefinitely. In this way, we can obtain an expression for s_i consisting of any number of elements of A_i , followed by l(i) elements of $S_i^2 \cap X_i$. Additionally or alternatively, we may replace $x_{l(i)}$ by zb, for some $z \in S_i^2 \cap X_i$ and $b \in B_i$, and we may then replace z by z'b', for some $z' \in S_i^2 \cap X_i$ and $b' \in B_i$, and so on. It follows that, for any $i \in \{1, \ldots, n\}$ and $k, m \in \mathbb{N}$, we may express s_i as an element of the sets

$$A_{i}^{k}(S_{i}^{2} \cap X_{i})^{l(i)}, \quad (S_{i}^{2} \cap X_{i})^{l(i)}B_{i}^{m}, \quad \text{and} \quad A_{i}^{k}(S_{i}^{2} \cap X_{i})^{l(i)}B_{i}^{m}.$$

In particular, if $\mathbf{C}_i = (B_1 \times \cdots \times B_{i-1} \times (S_i^2 \cap X_i) \times A_{i+1} \times \cdots \times A_n)^{l(i)}$ for each *i*, then

$$(s_1,\ldots,s_n)\in \mathbf{C}_1\cdots\mathbf{C}_n\subseteq \langle\Gamma\rangle.$$

For the final case, assume that, for some $i \in \{1, \ldots, n\}$, there exists an expression for s_i that includes an indecomposable element. For each index $i \in \{1, \ldots, n\}$ where this is the case, we may choose an expression for s_i in X_i that includes an element of $(S_i \setminus S_i^2)$. Choose j from these indices such that the expression for s_j has minimal length amongst these expressions. By Lemma 2.8, for the indices $i \in \{1, \ldots, n\}$ for which we have not already chosen an expression, we may express s_i as a product in $S_i^2 \cap X_i$ that is strictly longer than the expression for s_j . For $i \in \{1, \ldots, n\}$, let l(i) be the length of the expression for s_i . Note that, by construction, $l(j) = \min\{l(i) : i \in \{1, \ldots, n\}\}$. For each $i \in \{1, \ldots, n\}$ and $p \in \{1, \ldots, l(i)\}$, let $x_i^p \in X_i$ denote the p^{th} term in the expression for s_i , i.e. $s_i = x_i^1 x_i^2 \cdots x_i^{l(i)}$. Choose any $k \in \{1, \ldots, l(j)\}$

such that x_i^k is indecomposable, and define r = l(j) - k. Then for any $i \in \{1, \ldots, n\}$,

$$s_i = x_i^1 \cdots x_i^{k-1} \left(\prod_{\nu=k}^{l(i)-r} x_i^{\nu} \right) x_i^{l(i)-r+1} \cdots x_i^{l(i)}.$$

In particular, this gives an expression for s_i as a product of l(j) elements of S_i , where each term, except for possibly the k^{th} term, is contained in X_i . We use these expressions to give (s_1, \ldots, s_n) as a product of length l(j) in $\langle \Gamma \rangle$:

$$(s_{1}, \ldots, s_{n}) = (x_{1}^{1}, \ldots, x_{j-1}^{1}, x_{j}^{1}, x_{j+1}^{1}, \ldots, x_{n}^{1})$$

$$(x_{1}^{k-1}, \ldots, x_{j-1}^{k-1}, x_{j}^{k-1}, x_{j+1}^{k-1}, \ldots, x_{n}^{k-1})$$

$$\begin{pmatrix} l(1)-r & l(j-1)-r & l(j+1)-r \\ \prod_{\nu=k} x_{1}^{\nu}, \ldots, \prod_{\nu=k} x_{j-1}^{\nu}, x_{j}^{k}, \prod_{\nu=k} x_{j+1}^{\nu}, \ldots, \prod_{\nu=k} x_{n}^{\nu} \end{pmatrix}$$

$$(x_{1}^{l(1)-r+1}, \ldots, x_{j-1}^{l(j-1)-r+1}, x_{j}^{k+1}, x_{j+1}^{l(j+1)-r+1}, \ldots, x_{n}^{l(n)-r+1})$$

$$\dots$$

$$(x_{1}^{l(1)}, \ldots, x_{j-1}^{l(j-1)}, x_{j}^{l(j)}, x_{j+1}^{l(j+1)}, \ldots, x_{n}^{l(n)}).$$

Each term in this product, except for possibly the k^{th} term, is contained in $X_1 \times \cdots \times X_n$, which is a subset of $\langle \Gamma \rangle$ by Lemma 2.23. Finally, the k^{th} term

$$\left(\prod_{\nu=k}^{l(1)-r} x_1^{\nu}, \dots, \prod_{\nu=k}^{l(j-1)-r} x_{j-1}^{\nu}, x_j^{k}, \prod_{\nu=k}^{l(j+1)-r} x_{j+1}^{\nu}, \dots, \prod_{\nu=k}^{l(n)-r} x_n^{\nu}\right)$$

is contained in $S_1 \times \cdots \times S_{j-1} \times (S_j \setminus S_j^2) \times S_{j+1} \times \cdots \times S_n \subseteq \Gamma$, since we have chosen k so that x_j^k is indecomposable. Therefore $(s_1, s_2, \ldots, s_n) \in \langle \Gamma \rangle$, as required. \Box

2.5 Two finitely generated surjective semigroups

In the final section of this chapter, we restrict our attention to describing generating sets for direct products of pairs of finitely generated surjective semigroups. Generating sets for direct products of this kind are given in Proposition 2.15, and generating sets for direct products involving an arbitrary finite number of surjective semigroups are given in Theorem 2.17. Moreover, the generating sets given by these results have minimal cardinality in certain instances; see Example 2.19. Nonetheless, a generating set for a direct product that is specified by Proposition 2.15 or Theorem 2.17 may be significantly larger than necessary; see Example 2.26. This suggests that it may be possible to improve upon Proposition 2.15 and Theorem 2.17 in some cases.

Example 2.26. Let $m, n \in \mathbb{N}$ be arbitrary, let S be a right zero semigroup of order m, and let T be a left zero semigroup of order n. Note that S and T are finite surjective semigroups that have no proper generating subsets. Using the notation of Proposition 2.15, we find that the only choices for the sets B and A are the semigroups S and T themselves, respectively. Therefore, the only generating set for $S \times T$ that can be directly obtained from Proposition 2.15 is set

of all elements, which has cardinality mn. However, $S \times T$ is an $n \times m$ rectangular band (see [76, Theorem 1.1.3]), and so the minimal cardinality of a generating set for $S \times T$ is $\max\{m, n\}$. Note that $\max\{m, n\} \leq mn$, and indeed $\max\{m, n\} < mn$ when m, n > 1.

Let $S = \langle X \rangle$ and $T = \langle Y \rangle$ be surjective semigroups. Roughly speaking, the fundamental idea behind Proposition 2.15 is to extend the generating sets X and Y to generating sets $X \cup B$ and $Y \cup A$ in such a way that, given any pair $(s, t) \in S \times T$, s and t can be systematically expressed as equal-length products over $X \cup B$ and $A \cup Y$, respectively. By doing so, it is possible to describe a generating set for $S \times T$ in terms of the sets $X \times A$ and $B \times Y$.

In the main result of this section, Theorem 2.31, we present an alternative approach (in the case that the generating sets X and Y are finite) to extending X and Y. We do so by taking into account the Green's structure of the semigroups. More specifically, the set A requires either one or two elements per maximal \mathscr{R} -class of T, and the set B requires requires either one or two elements per maximal \mathscr{R} -class of S. This allows us to give an upper bound on rank $(S \times T)$ that involves the ranks of S and T as well as the Green's structure of S and T; see Corollary 2.33. For certain semigroups, a generating set described by Theorem 2.31 may be smaller than a generating set described by Proposition 2.15; see Example 2.32.

Our approach also suggests a direction for further improvements to Theorem 2.17, and, consequently, improved methods for computing direct products. However, in this section, we do not address the problem of constructing generating sets computationally.

In order to prove the main result of this section, we require the following lemmas, which we use when describing the Green's structure of a finitely generated surjective semigroup.

Lemma 2.27. Let S be a semigroup generated by a set X, and let $\mathcal{K} \in {\mathcal{L}, \mathcal{R}, \mathcal{J}}$. If K is a maximal \mathcal{K} -class of S, then $K \cap X \neq \emptyset$. In particular, if S is finitely generated, then the number of maximal \mathcal{K} -classes of S is finite.

Proof. We prove the result for \mathscr{L} ; the proof for \mathscr{R} is dual, and the proof for \mathscr{J} is similar. Suppose that L is a maximal \mathscr{L} -class of S, let $x \in L$ be arbitrary, and let $x = x_1 \cdots x_n$ be a factorization of x over X. It follows that $L \leq L_{x_n}$ by Lemma 1.12(i), and so $x_n \in L$ by the maximality of L.

Lemma 2.28. Let S be a semigroup with a finite generating set X, and let $\mathcal{K} \in \{\mathcal{L}, \mathcal{R}, \mathcal{J}\}$. For every \mathcal{K} -class K of S, there exists a maximal \mathcal{K} -class K' of S such that $K \leq K'$.

Proof. Again, we prove only the result concerning \mathscr{L} . Let $L \in S/\mathscr{L}$ and $x \in L$ be arbitrary, and let $x = x_1 \cdots x_n$ be a factorization of x over X. It follows that $L \leq L_{x_n}$ by Lemma 1.12(i). If L_{x_n} is maximal, then we are done. Otherwise, there exists an \mathscr{L} -class $L' \in S/\mathscr{L}$ such that $L_{x_n} < L'$, and again, $L' \leq L_{x'}$ for some generator $x' \in X$. Therefore $L_{x_n} < L_{x'}$, which implies that $x \neq x_n$. Continuing in this way, either we find a maximal \mathscr{L} -class L_y for some $y \in X$ such that $L \leq L_y$, or we obtain an arbitrarily long strictly ascending chain of \mathscr{L} -classes of S, each of which contains a distinct element of X. Since X is finite, the latter is not possible.

In the following lemma, we show that every element in a finitely generated surjective semigroup can be written as a left multiple of an element of a maximal \mathscr{L} -class, and as a right multiple of an element of a maximal \mathscr{R} -class.

Lemma 2.29. Let S be a surjective semigroup with finite generating set X, and let Λ and Ω be transversals of the sets

 $\{L \cap X : L \text{ is a maximal } \mathscr{L}\text{-class of } S\}$ and $\{R \cap X : R \text{ is a maximal } \mathscr{R}\text{-class of } S\}$, respectively. Then $S = S\Lambda = \Omega S$.

Proof. Certainly $S\Lambda \subseteq S$ and $\Omega S \subseteq S$. We prove that $S \subseteq S\Lambda$; the proof that $S \subseteq \Omega S$ is dual. Let $s \in S$ be arbitrary. Then s = ux for some $u \in S$ and $x \in X$, since s is decomposable. By Lemmas 2.27 and 2.28, there exists a maximal \mathscr{L} -class $L_{x'}$ of S, for some $x' \in X$, such that $L_x \leq L_{x'}$. By construction, there exists some $y \in L_{x'} \cap \Lambda$. Therefore $L_x \leq L_y$, i.e. $x \in S^1 y$. In particular, $s = ux \in SS^1\Lambda = S\Lambda$.

In the main theorem of this section, we require relative left or right identities of the elements of the sets Λ and Ω from Lemma 2.29. Certainly any regular element x has both a relative left identity and a relative right identity, since if x' is an inverse of x, then x = (xx')x = x(x'x). In the following lemma, we show that elements of maximal \mathscr{L} -classes and elements of maximal \mathscr{R} -classes of a surjective semigroup have relative left and right identities, respectively, regardless of regularity.

Lemma 2.30. Let S be a surjective semigroup. Then every element in a maximal \mathscr{L} -class of S has a relative left identity, and every element in a maximal \mathscr{R} -class of S has a relative right identity.

Proof. Let L be a maximal \mathscr{L} -class of S, and let $l \in L$. Since S is surjective, there exist elements $a, b \in S$ such that l = ab. It follows by Lemma 1.12(i) that $L \leq L_b$, and so the maximality of L implies that $b \in L$. Therefore b = ul for some $u \in S^1$. It follows that l = ab = (au)l, i.e. au is a relative left identity for l. The statement concerning \mathscr{R} is dual.

We may now state and prove the main result of this section.

Theorem 2.31. Let S and T be surjective semigroups with finite generating sets X and Y, respectively. Let Λ be a transversal of $\{L \cap X : L \text{ is a maximal } \mathcal{L}\text{-class of } S\}$, and let Ω be a transversal of $\{R \cap Y : R \text{ is a maximal } \mathcal{R}\text{-class of } T\}$. For each regular $l \in \Lambda$, choose a relative right identity e_l for l, and for each non-regular $l \in \Lambda$, choose a relative left identity f_l for l. Similarly, for each regular $r \in \Omega$, choose a relative left identity u_r for r, and for each non-regular $r \in \Omega$, choose a relative right identity v_r for r. Define

$$B = \{e_l : l \in \Lambda \text{ and } l \text{ is regular}\} \cup \{f_l, l : l \in \Lambda \text{ and } l \text{ is non-regular}\}$$

and

$$A = \{u_r : r \in \Omega \text{ and } r \text{ is regular}\} \cup \{v_r, r : r \in \Omega \text{ and } r \text{ is non-regular}\}$$

Then $S \times T$ is generated by the set $\Psi = (X \times A) \cup (B \times Y)$.

Proof. That the sets $\Lambda \subseteq X$ and $\Omega \subseteq Y$ are finite and non-empty follows by Lemmas 2.27 and 2.28; the relative left and right identities exist by Lemma 2.30 and the preceding discussion.

It suffices to show that $S \times T \subseteq \langle \Psi \rangle$. Let $(s, t) \in S \times T$. By Lemma 2.29, $s \in S\Lambda$ and $t \in T\Omega$. Therefore, $s = x_1 \cdots x_m l$ for some $x_1, \ldots, x_m \in X$ and $l \in \Lambda$, and similarly $t = ry_1 \cdots y_n$ for some $y_1, \ldots, y_n \in Y$ and $r \in \Omega$. Since S and T are surjective semigroups, we may assume without loss of generality that m, n > 1. The proof that $(s, t) \in \langle \Psi \rangle$ concludes with four cases, according to the regularity of l and r.

Case 1: l and r are regular.

$$(s, t) = (x_1 \cdots x_m l e_l^{n+1}, u_r^{m+1} r y_1 \cdots y_n) = (x_1, u_r) \cdots (x_m, u_r) (l, u_r) (e_l, r) (e_l, y_1) \cdots (e_l, y_n) \in \langle \Psi \rangle.$$

Case 2: l is regular but r is not.

$$(s, t) = (x_1 \cdots x_m l e_l^n, \ r v_r^m y_1 \cdots y_n) = (x_1, r) (x_2, v_r) \cdots (x_m, v_r) (l, v_r) (e_l, y_1) \cdots (e_1, y_n) \in \langle \Psi \rangle.$$

Case 3: r is regular but l is not.

$$(s, t) = (x_1 \cdots x_m f_l^n l, u_r^m r y_1 \cdots y_n) = (x_1, u_r) \cdots (x_m, u_r) (f_l, r) (f_l, y_1) \cdots (f_l, y_{n-1}) (l, y_n) \in \langle \Psi \rangle.$$

Case 4: neither l nor r is regular.

$$(s, t) = (x_1 \cdots x_m f_l^{n-1} l, r v_r^{m-1} y_1 \cdots y_n) = (x_1, r) (x_2, v_r) \cdots (x_m, v_r) (f_l, y_1) \cdots (f_l, y_{n-1}) (l, y_n) \in \langle \Psi \rangle.$$

Example 2.32. Let $n \in \mathbb{N}$, $n \geq 2$, be arbitrary. We define $S = \mathcal{T}_n \setminus \mathcal{S}_n$ to be the semigroup consisting of all transformations of degree n that are not permutations, and let X be the generating set for S consisting of all n(n-1) idempotents of rank n-1. We also define T to be a $2 \times n$ rectangular band, generated by a set Y consisting of n elements. Note that S and T are finite regular semigroups; S has n maximal \mathscr{L} -classes, and T has 2 maximal \mathscr{R} -classes. Therefore, a generating set for $S \times T$ specified by Theorem 2.31 contains at most $2|X| + n|Y| = 2n(n-1) + n^2 = n(3n-2)$ elements. On the other hand, choosing A = Y and B = X, it follows by Proposition 2.15 that $(X \times A) \cup (B \times Y) = X \times Y$ is a generating set for $S \times T$; this set contains $n^2(n-1)$ elements. It follows that, at least for $n \geq 4$, a generating set given by Proposition 2.15 can be larger than the largest generating set given by Theorem 2.31. Although it is possible to construct smaller generating sets for $S \times T$ with Proposition 2.15, it is not clear from the statement of the proposition that this is the case.

By applying Theorem 2.31 to finitely generated surjective semigroups defined by generating sets of minimal cardinality, we obtain the following corollary.

Corollary 2.33. Let S and T be finitely generated surjective semigroups. Define l to be the number of maximal \mathscr{L} -classes of S that are regular, and l' to be the number of maximal \mathscr{L} -classes of S that are not. Similarly, define r to be the number of maximal \mathscr{R} -classes of T that are regular, and r' to be the number of maximal \mathscr{R} -classes of T that are not. Then

$$\operatorname{rank}(S \times T) \le \operatorname{rank}(S) \cdot (r + 2r') + (l + 2l') \cdot \operatorname{rank}(T).$$

In particular, if every maximal \mathcal{L} -class of S and every maximal \mathcal{R} -class of T is regular, then

$$\operatorname{rank}(S \times T) \le \operatorname{rank}(S) \cdot r + l \cdot \operatorname{rank}(T).$$

A finite semigroup contains at least one maximal \mathscr{J} -class, and a maximal \mathscr{J} -class in a finite surjective semigroup is a regular \mathscr{D} -class by Corollary 2.10; such a \mathscr{D} -class consists of regular maximal \mathscr{L} - and \mathscr{R} -classes. It follows that, in a finite surjective semigroup, there exists at least one regular maximal \mathscr{L} -class and at least one regular maximal \mathscr{R} -class.

Let S be a finitely generated semigroup. By Lemma 2.27, the number of maximal \mathscr{L} -, \mathscr{R} -, or \mathscr{J} -classes of S is at most rank(S). Therefore, if S is a finitely generated semigroup each of whose l maximal \mathscr{L} -classes is regular, and T is a finitely generated semigroup each of whose r maximal \mathscr{R} -classes is regular, then (since S and T are surjective) the upper bound on the rank of $S \times T$ given by Corollary 2.33 is no worse than the bound given by Corollary 2.18 and [2, Corollary 3.6]. More specifically,

$$\operatorname{rank}(S \times T) \le \operatorname{rank}(S) \cdot r + l \cdot \operatorname{rank}(T) \le 2\operatorname{rank}(S)\operatorname{rank}(T).$$

There are obvious analogues to Theorem 2.31 and Corollary 2.33 that involve the maximal \mathscr{R} -classes of S and the maximal \mathscr{L} -classes of T, rather than the maximal \mathscr{L} -classes of S and the maximal \mathscr{R} -classes of T, where S and T are finitely generated surjective semigroups.

Open Problem 2.34. Generalise Theorem 2.31 and Corollary 2.33 for direct products of an arbitrary number of finitely generated surjective semigroups. In particular, develop an upper bound for the rank of such a direct product that involves the ranks and the numbers of maximal \mathcal{L} - and \mathcal{R} -classes of the factors.

We end this chapter by posing the following easy-to-state problem concerning direct products of semigroups. As far as we are aware, this problem has not yet been considered in the literature.

Open Problem 2.35. Let S be an arbitrary finite semigroup. Develop practical methods for computing whether there exist non-trivial semigroups T and U such that $S \cong T \times U$; given this, develop methods for finding such semigroups T and U when they exist.

Chapter 3

Rees 0-matrix semigroups over a semigroup

3.1 Introduction

Given a semigroup T, non-empty sets I and Λ , and a $\Lambda \times I$ matrix P with entries in $T^0 = T \cup \{0\}$, the Rees 0-matrix semigroup $\mathscr{M}^0[T; I, \Lambda; P]$ is the set $(I \times T \times \Lambda) \cup \{0\}$ with multiplication defined, as in (1.8), by

$$(i, t, \lambda) (j, u, \mu) = \begin{cases} 0 & \text{if } p_{\lambda,j} = 0, \\ (i, tp_{\lambda,j}u, \mu) & \text{if } p_{\lambda,j} \neq 0, \end{cases}$$

and $(i, t, \lambda) 0 = 0 (i, t, \lambda) = 0^2 = 0,$

for all (i, t, λ) , $(j, u, \mu) \in I \times T \times \Lambda$. In this chapter, we investigate various semigroup-theoretic properties of Rees 0-matrix semigroups defined over arbitrary semigroups, with a particular emphasis on developing methods for computing with these kinds of semigroups.

The original interest in Rees 0-matrix semigroups stems from their deep association with the completely 0-simple semigroups. By the Rees Theorem [76, Theorem 3.2.3], a semigroup is completely 0-simple if and only if it is isomorphic to a regular Rees 0-matrix semigroup $\mathscr{M}^0[G; I, \Lambda; P]$, where G is a group. The principal factor of a \mathscr{J} -class in any finite semigroup is either a zero semigroup, or is a completely 0-simple semigroup [76, Theorem 3.1.6]. In essence, therefore, completely 0-simple semigroups, and correspondingly regular Rees 0-matrix semigroups over groups, can be thought of as the building blocks of finite semigroups. Because of this connection, many problems in finite semigroup theory reduce to, or at least involve, problems with Rees 0-matrix semigroups. For example, as noted in [65], any minimal generating set for a finite semigroup necessarily contains minimal generating sets for the principal factors of each of its maximal *I*-classes. Therefore, in order to develop techniques for describing minimal generating sets of an arbitrary finite semigroup, we first require this problem to be solved for finite regular Rees 0-matrix semigroups over groups; this topic has been extensively researched, principally by Gray and Ruškuc, see [63,65,114] for more information. Rees 0-matrix semigroups are also central to the problem of computing maximal subsemigroups of finite semigroups, which is the topic of the next chapter. More precisely, the algorithm presented in the next chapter for computing maximal subsemigroups of an arbitrary semigroup relies, amongst other things, on an algorithm for computing the maximal subsemigroups of a finite regular Rees 0-matrix semigroup over a group. Indeed, the main purpose of Section 3.2 is to provide some necessary background information and results about these kinds of Rees 0-matrix semigroups.

Since regular Rees 0-matrix semigroups over groups are so well-understood, and of such importance to finite semigroup theory, there are well-developed methods for computing with these kinds of semigroups in the computational algebra system GAP [58]. Moreover, in GAP, if it is possible to construct a semigroup T and a $\Lambda \times I$ matrix P with entries in T^0 , then it is possible the construct the Rees 0-matrix semigroup $\mathscr{M}^0[T; I, \Lambda; P]$ in GAP, even when T is not a group. This naturally leads one to experiment computationally with Rees 0-matrix semigroups over semigroups that are not groups, many of which turn out to exhibit a rich structure. However, since there is very little research on the topic of computing with Rees 0-matrix semigroups over semigroups that are not groups, there are few specialised methods for computing with them. Instead, GAP uses generic methods from computational semigroup theory, which often perform much more slowly than one might expect. The research in this chapter arose from performing experiments with Rees 0-matrix semigroups in GAP, and identifying areas where the theory, and associated computational methods, could be improved.

As with a direct product of semigroups, a Rees 0-matrix semigroup is defined in terms of its elements. This brings certain advantages for computation: for example, it allows us to easily perform membership checking by computing in the underlying semigroup and index sets, since an arbitrary triple (a, b, c) is contained in the Rees 0-matrix semigroup $\mathscr{M}^0[T; I, \Lambda; P]$ if and only if $a \in I$, $b \in T$, and $c \in \Lambda$. Moreover, the order of a Rees 0-matrix semigroup $\mathscr{M}^0[T; I, \Lambda; P]$ is $|I| \cdot |T| \cdot |\Lambda| + 1$. On the other hand, a Rees 0-matrix semigroup does not necessarily come with a proper generating subset. In the current version of GAP, unless the underlying semigroup is a group, the default method for computing a generating set of a Rees 0matrix semigroup returns the set of all elements or all non-zero elements, depending on whether the matrix contains zero. This is obviously undesirable. In Section 3.4, we address this difficulty by describing relatively small generating sets for certain Rees 0-matrix semigroups.

When computing with a Rees 0-matrix semigroup $\mathscr{M}^0[T; I, \Lambda; P]$, we aim to formulate the relevant problem in terms of the underlying semigroup T, the index sets I and Λ , and the matrix P. In other words, wherever possible, we wish to make use of the representation of the Rees 0-matrix semigroup in question. Since all of the information about the Rees 0-matrix semigroup $\mathscr{M}^0[T; I, \Lambda; P]$ is encoded in T, I, Λ , and P, this goal seems sensible, provided that it is feasible to compute with these components. Perhaps most importantly, we wish to avoid having to exhaustively enumerate every element of a Rees 0-matrix semigroup, especially since we can represent and compute with Rees 0-matrix semigroups that are far too large to exhaustively enumerate and store in memory, such as those defined over large groups. This is because representing a Rees 0-matrix semigroup $\mathscr{M}^0[T; I, \Lambda; P]$ on a computer only requires representing the underlying semigroup T, and storing a matrix that contains $|I| \cdot |\Lambda|$ elements. In particular, we can represent certain infinite Rees 0-matrix semigroups, whereas the elements of an infinite semigroup can never be enumerated by a computer.

As a general rule of thumb, if it is not feasible to compute with a semigroup T, then it is not reasonable to expect to perform a related computation with a Rees 0-matrix semigroup whose underlying semigroup is T. Therefore, throughout this chapter, when we consider a computation with a Rees 0-matrix semigroup, we implicitly assume that its components are such that they may be readily computed.

This chapter is structured as follows.

In Section 3.2, we introduce several concepts and results that are required in later parts of this chapter, and the next. In particular, we discuss isomorphisms and normalizations of Rees 0-matrix semigroups, and we describe an algorithm for computing a generating set for the idempotent generated subsemigroup of a finite Rees 0-matrix semigroup over a group.

In Section 3.3, we consider the Green's structure of Rees 0-matrix semigroups over arbitrary semigroups. In our attempt to classify those Rees 0-matrix semigroups whose Green's structure can be deduced from the index sets and the Green's structure of the underlying semigroup, we introduce the notions of row-regular and column-regular matrices. These notions prove to be

important for the whole of this chapter.

In Section 3.4, we prove results concerning generating sets of Rees 0-matrix semigroups. We give relatively small generating sets for certain kinds of Rees 0-matrix semigroups over monoids, and we give generating sets for Rees 0-matrix semigroups with row-regular or column-regular matrices. We also characterise some of the indecomposable elements of a Rees 0-matrix semigroup, in terms of the matrix and the indecomposable elements of the underlying semigroup.

Finally, in Section 3.5, we completely classify when an arbitrary Rees 0-matrix semigroup is regular, inverse, or a monoid, and we discuss the properties of such Rees 0-matrix semigroups.

Although the topic of this chapter is Rees 0-matrix semigroups, Rees matrix semigroups (without the zero) are not forgotten. Many results concerning Rees matrix semigroups can be deduced from results concerning Rees 0-matrix semigroups. For example, the obvious analogues of the results of Section 3.3 hold for Rees matrix semigroups. Furthermore, if $S = \mathcal{M}[T; I, \Lambda; P]$ is a Rees matrix semigroup, then the corresponding Rees 0-matrix semigroup $\mathcal{M}^0[T; I, \Lambda; P]$ is isomorphic to S^0 , and so by Lemma 1.5, the generating sets of S are in an inclusion-preserving one-to-one correspondence with the generating sets of $\mathcal{M}^0[T; I, \Lambda; P]$. This observation allows the results of Section 3.4 to be adapted for Rees matrix semigroups, for instance.

3.2 Preliminaries

In this section, we present some results about Rees 0-matrix semigroups that will be useful in later parts of this thesis. In Section 3.2.1, we discuss isomorphisms of Rees 0-matrix semigroups over groups and monoids. This topic allows us to introduce the notion of *normalizations* of such Rees 0-matrix semigroups. This is a vital concept required for the algorithms presented in Section 4.3, which concern the maximal subsemigroups of finite regular Rees 0-matrix semigroups over groups. In Section 3.2.2, we describe an algorithm for computing a small generating set for the idempotent generated subsemigroup a finite Rees 0-matrix semigroup over a group. This has inherent interest, and is again useful for the algorithms presented in Section 4.3.

3.2.1 Isomorphisms and normalizations of Rees 0-matrix semigroups

In general, it is not straightforward to determine whether two arbitrary Rees 0-matrix semigroups are isomorphic, or to construct isomorphisms between isomorphic Rees 0-matrix semigroups. For example, consider the Rees 0-matrix semigroups

$$S_1 = \mathscr{M}^0[\mathcal{T}_1; \{1', 2'\}, \{1, 2\}; P]$$
 and $S_2 = \mathscr{M}^0[\mathcal{T}_2; \{1'\}, \{1\}; Q],$

where \mathcal{T}_n denotes the full transformation monoid of degree n (see Section 1.3.5), and P and Q are 2×2 and 1×1 matrices, respectively, that consist entirely of zeroes. Then S_1 and S_2 are zero semigroups of order 5, and are therefore isomorphic. However, this is perhaps counter-intuitive, since the underlying semigroups \mathcal{T}_1 and \mathcal{T}_2 are non-isomorphic, and their index sets have different cardinalities. Describing isomorphisms between arbitrary Rees 0-matrix semigroups is beyond the scope of this thesis; in this section, we restrict our attention to isomorphisms of Rees 0-matrix semigroups over groups or monoids.

Rees 0-matrix semigroups over groups

As shown in the following result, the difference between isomorphic regular Rees 0-matrix semigroups over groups is much smaller than the general case: their underlying groups are required to be isomorphic, and their respective index sets are necessarily in bijective correspondence. There is, however, significant possible variation between their matrices. **Proposition 3.1** ([76, Theorem 3.4.1]). Let $S = \mathscr{M}^0[G; I, \Lambda; P]$ and $S' = \mathscr{M}^0[H; J, M; Q]$ be regular Rees 0-matrix semigroups over groups. Then $S \cong S'$ if and only if there exists an isomorphism of groups $\theta : G \longrightarrow H$, bijections $\psi : I \longrightarrow J$ and $\chi : \Lambda \longrightarrow M$ such that $p_{\lambda,i} \neq 0$ whenever $q_{\lambda\chi,i\psi} \neq 0$, and elements $u_i, v_\lambda \in H$ such that

$$p_{\lambda,i} \neq 0 \quad \Rightarrow \quad p_{\lambda,i}\theta = v_{\lambda} \cdot q_{\lambda\chi,i\psi} \cdot u_i$$

for all $i \in I$ and $\lambda \in \Lambda$.

This characterisation can be used, for instance, to help enumerate the completely 0-simple semigroups of a given order up to isomorphism. For our purposes, this classification is particularly useful since it suggests the possibility of *normalizing* a regular Rees 0-matrix semigroup over a group. In essence, every such Rees 0-matrix semigroup is isomorphic to a Rees 0-matrix semigroup where the entries of the matrix satisfy certain convenient properties. This idea was first explored by Graham; indeed, some authors say that a Rees 0-matrix semigroup has *Graham normal form* if its matrix satisfies the properties described by Graham in [62, Section 4]. The topic has subsequently been considered by several authors, including in [76, Section 3.4], [64, Theorem 2.58], and [110, Theorem 4.13.34]; we refer the interested reader to these works for a more thorough treatment of the topic than the one given here. The results given in the following proposition are sufficient for our purposes. Recall from Section 1.3.2 that the Graham-Houghton graph of the Rees 0-matrix semigroup $\mathscr{M}^0[T; I, \Lambda; P]$, where $(p_{\lambda,i})_{\lambda \in \Lambda, i \in I}$, is the graph $(I \cup \Lambda, \{\{i, \lambda\} : p_{\lambda,i} \neq 0\})$; we assume that $I \cap \Lambda = \emptyset$.

Proposition 3.2 ([62, Section 4]). Let $S' = \mathscr{M}^0[G; I, \Lambda; Q]$ be a finite regular Rees 0-matrix semigroup over a group G, and let $\{I_1, \ldots, I_n\}$ and $\{\Lambda_1, \ldots, \Lambda_n\}$ be partitions of I and Λ , respectively, such that $I_k \cup \Lambda_k$ is a connected component of the Graham-Houghton graph of S'for each k. Then there exists a matrix $P : \Lambda \times I \longrightarrow G^0$ such that $S' \cong S = \mathscr{M}^0[G; I, \Lambda; P]$ and the following hold for each k:

- (i) there exist indices $i_k \in I_k$ and $\lambda_k \in \Lambda_k$ such that $p_{\lambda_k, i_k} = 1_G$;
- (ii) the non-zero entries of the sub-matrix $P_k : \Lambda_k \times I_k \longrightarrow G^0$ of P generate the group $G_k = \{g \in G : (i_k, g, \lambda_k) \in F(S)\}.$

We refer to a finite regular Rees 0-matrix semigroup $S = \mathscr{M}^0[G; I, \Lambda; P]$, over a group G, whose matrix that satisfies properties (i) and (ii) in Proposition 3.2 as being *normalized*, and a *normalization* of a Rees 0-matrix semigroup is an isomorphism to a normalized Rees 0-matrix semigroup. If S is normalized, then by [62, Theorem 2], each connected component of the Graham-Houghton graph of S corresponds to a regular Rees 0-matrix semigroup

$$\mathscr{M}^{0}[G_{k}; I_{k}, \Lambda_{k}; P_{k}] = (I_{k} \times G_{k} \times \Lambda_{k}) \cup \{0\},$$
(3.3)

and the idempotent generated subsemigroup of S is the union of these subsemigroups

$$F(S) = \{0\} \cup \bigcup_{k=1}^{n} (I_k \times G_k \times \Lambda_k).$$
(3.4)

In Section 4.3, we simplify the task of describing the maximal subsemigroups of a finite regular Rees 0-matrix semigroup over a group by assuming (without loss of generality, according to Proposition 3.2) that the semigroup is normalized. Therefore, in order to implement the algorithms described in Chapter 4, we require the ability to compute a normalization of an arbitrary finite regular Rees 0-matrix semigroup $S = \mathcal{M}^0[G; I, \Lambda; P]$, where G is a group. This can be done in a straightforward manner; we briefly and roughly describe the method

that has been implemented in the SEMIGROUPS package [101] for GAP [58], by the author. A normalization of the Rees 0-matrix semigroup S can be found with the SEMIGROUPS package by applying the function Normalization to S.

First, it is necessary to calculate the connected components of the Graham-Houghton graph of S; in doing so, we obtain the sets I_k and Λ_k from the statement of Proposition 3.2. Finding the connected components of a graph is a classical problem in graph theory, and can be solved with a standard depth- or breath-first search algorithm; see [117, Section 4.1]. For a graph (V, E), an algorithm of this kind typically has time complexity O(|V| + |E|), and so for the Graham-Houghton of S, this is at most $O(|I| \cdot |\Lambda|)$.

The remaining step consists of multiplying the entries in each row of P by the inverse of the first non-zero entry in that row, and multiplying the entries in each column of P by the inverse of the first non-zero entry in that column; this step can be accomplished with time complexity $O(|I| \cdot |\Lambda|)$. It is not especially difficult to see that, by choosing the rows and columns in a suitable order, the first non-zero entry in each row and column of the resulting normalized matrix is 1_G . In particular, Proposition 3.2(i) holds. Furthermore, when the first non-zero entry in each row and column of the matrix is 1_G , it can be shown that Proposition 3.2 also holds; proving this is difficult, and is the main result of [62, Section 4].

Rees 0-matrix semigroups over monoids

The classification of isomorphisms between regular Rees 0-matrix semigroups over groups given in Proposition 3.1 exploits the fact that the elements of the matrix are units. Since a monoid contains units, this suggests the possibility of partially generalising this proposition to include Rees 0-matrix semigroups over monoids that are not necessarily groups. We present and prove a generalisation of the converse implication of Proposition 3.1 below. In later parts of this chapter, we use Proposition 3.5 to introduce the notion of normalizing certain Rees 0-matrix semigroups over monoids; see Sections 3.4.2 and 3.5.3.

Proposition 3.5 (cf. [76, Theorem 3.4.1]). Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ and $S' = \mathscr{M}^0[U; J, M; Q]$ be Rees 0-matrix semigroups, where T and U are monoids, and $P = (p_{\lambda,i})_{\lambda \in \Lambda, i \in I}$ and $Q = (q_{\mu,j})_{\mu \in M, j \in J}$. If there exists an isomorphism $\theta : T \longrightarrow U$, bijections $\psi : I \longrightarrow J$ and $\chi : \Lambda \longrightarrow M$ such that $p_{\lambda,i} \neq 0$ whenever $q_{\lambda\chi,i\psi} \neq 0$, and units $u_i, v_\lambda \in U$ such that

$$p_{\lambda,i} \neq 0 \quad \Rightarrow \quad p_{\lambda,i}\theta = v_{\lambda} \cdot q_{\lambda\chi,i\psi} \cdot u_i$$

for all $i \in I$ and $\lambda \in \Lambda$, then the function $\phi : S \longrightarrow S'$ defined by

$$0\phi = 0$$
 and $(i, t, \lambda)\phi = (i\psi, u_i(t\theta)v_\lambda, \lambda\chi)$

for all $(i, t, \lambda) \in I \times T \times \Lambda$ is an isomorphism, and in particular, $S \cong S'$.

Proof. To show that ϕ is an homomorphism, let $x, y \in S$ be arbitrary. If x = 0 or y = 0, then $(xy)\phi = 0\phi = 0 = (x\phi)(y\phi)$. Suppose that $x = (i, t, \lambda), y = (j, s, \mu) \in S \setminus \{0\}$. If $p_{\lambda,j} = 0$, then also $q_{\lambda\chi,j\psi} = 0$, and so again $(xy)\phi = 0\phi = 0 = (x\phi)(y\phi)$. Otherwise,

$$\begin{aligned} (x\phi)(y\phi) &= (i\psi, u_i(t\theta)v_\lambda, \lambda\chi) (j\psi, u_j(s\theta)v_\mu, \mu\chi) \\ &= (i\psi, u_i(t\theta)v_\lambda q_{\lambda\chi,j\psi}u_j(s\theta)v_\mu, \mu\chi) \\ &= (i\psi, u_i(t\theta)(p_{\lambda,j}\theta)(s\theta)v_\mu, \mu\chi) \\ &= (i\psi, u_i(tp_{\lambda,j}s)\thetav_\mu, \mu\chi) \\ &= (i, tp_{\lambda,j}s, \mu)\phi \\ &= (xy)\phi, \end{aligned}$$

and so ϕ is a homomorphism.

If $(i, t, \lambda) \phi = (j, s, \mu) \phi$, then in particular $i\psi = j\psi$ and $\lambda \chi = \mu \chi$. Since the functions ψ and χ are bijective, it follows that i = j and $\lambda = \mu$. In addition, $u_i(t\theta)v_\lambda = u_i(s\theta)v_\lambda$. Using the fact that u_i and v_λ are units in U, we deduce that $t\theta = s\theta$. But θ is also bijective, which implies that t = s, and so ϕ is injective. Let $(i, t, \lambda) \in S' \setminus \{0\}$ be an arbitrary non-zero element, and define $j = i\psi^{-1}$ and $\mu = \lambda\chi^{-1}$. Then

$$(j, u_j^{-1}(t\theta^{-1})v_{\mu}^{-1}, \mu)\phi = (i, t, \lambda)$$

and so ϕ is surjective, and therefore bijective. In conclusion, ϕ is an isomorphism.

In modern computer algebra systems such as GAP [58] and MAGMA [13], there are welldeveloped and practical methods for testing whether arbitrary finite groups are isomorphic, and for finding isomorphisms between isomorphic finite groups, see [20], for example. We do not concern ourselves with the details of how these tools function; however, because these tools exist, it is feasible to implement methods that build on Proposition 3.1 to test whether arbitrary finite regular Rees 0-matrix semigroups over groups are isomorphic, and to find isomorphisms between those semigroups that are isomorphic. Indeed, such functionality is available in the SEMIGROUPS [101] package for GAP. However, there do not currently exist practical methods for testing whether arbitrary finite monoids are isomorphic, or for finding isomorphisms between them. Because of this, Proposition 3.5 does not currently suggest a practical approach to computing isomorphisms between Rees 0-matrix semigroups over monoids.

3.2.2 The idempotent generated subsemigroup of a Rees 0-matrix semigroup

In this section, we consider the problem of quickly computing a small generating set for the idempotent generated subsemigroup of a finite Rees 0-matrix semigroup over a group. It appears to be difficult to find generating sets of minimal cardinality, see [64, Theorem 3.27], but doing so is not our goal. The generating set that we describe is small, in the sense that it does not necessarily include every idempotent of the semigroup, and it contains at most twice the minimal number of generators. The main result of this section is Proposition 3.7.

The work in this section was motivated by the algorithms presented in Chapter 4. In particular, in Algorithm 4.32, we describe a procedure for computing the maximal subsemigroups of an arbitrary finite regular Rees 0-matrix semigroup S over a group that intersect every \mathcal{H} -class of S non-trivially. The maximal subsemigroups that are returned by this algorithm are defined by generating sets. Crucially, for the purposes of this section, each of these sets includes a generating set for the idempotent generated subsemigroup of S.

As discussed in Section 1.4, the time complexities of many algorithms in computational semigroup theory are given in terms of the size of the generating set of the semigroup in question. Roughly speaking, this means that, given a semigroup S defined by a generating set, the smaller the generating set for S, the faster it is to compute with S. Therefore, in order to facilitate speedy computation with the maximal subsemigroups returned by Algorithm 4.32, it is desirable for these maximal subsemigroups to be defined by generating sets that are as small as is reasonably possible. One way to address this problem is to endeavour to produce a small generating set for the corresponding idempotent generated subsemigroup.

An obvious choice of a generating set for the idempotent generated subsemigroup of a finite Rees 0-matrix semigroup over a group is the set of all idempotents in the semigroup. However, the number of generators that are actually required can be smaller than this; see Example 3.6 for a very simple demonstration of this fact. **Example 3.6.** Let $S = \mathscr{M}^0[G; I, \Lambda; P]$ be a Rees 0-matrix semigroup, where G is the trivial group, $I = \{1', 2'\}, \Lambda = \{1, 2\}$, and

$$P = \begin{pmatrix} 1_G & 1_G \\ 1_G & 0 \end{pmatrix}.$$

There are four idempotents in S, and $(2', 1_G, 2)$ is the unique non-idempotent element. Since $(2', 1_G, 2) = (2', 1_G, 1) (1', 1_G, 2)$, it is clear that the idempotent generated subsemigroup of S is S itself. However, the pair of idempotents $\{(1', 1_G, 2), (2', 1_G, 1)\}$ generates S.

In the proof of [64, Proposition 3.13], Gray observes that, for a finite idempotent generated Rees 0-matrix semigroup S over a group, a spanning tree of the Graham-Houghton graph of Scorresponds to a generating set for S that consists of idempotents. Given (3.4), it is clear that a normalized finite Rees 0-matrix semigroup over a group G is idempotent generated if and only if its Graham-Houghton graph is connected, and its non-zero matrix entries generate the group G. Indeed, this result is stated in [64, Theorem 3.1]. Furthermore, in [64, Theorem 2.13], Gray proves that the minimal cardinality of a generating set for a finite idempotent generated Rees 0-matrix semigroup $\mathcal{M}^0[G; I, \Lambda; P]$, where G is a group, is max{ $|I|, |\Lambda|$ }.

However, in Algorithm 4.32, we require a generating set for the idempotent generated subsemigroup of *any* finite regular Rees 0-matrix semigroup over a group, i.e. not only those that are idempotent generated. In Proposition 3.7, we generalise Gray's ideas concerning spanning trees in order to solve this problem.

Proposition 3.7. Let $S = \mathscr{M}^0[G; I, \Lambda; P]$ be a finite Rees 0-matrix semigroup over a group G, and let $(I \cup \Lambda, E)$ be a spanning forest for the Graham-Houghton graph of S. Then the set

$$X = \left\{ \left(i, \, p_{\lambda,i}^{-1}, \, \lambda \right) \, : \, \{i, \, \lambda\} \in E \right\} \cup \{0\}.$$

generates F(S), the idempotent generated subsemigroup of S. In particular, the smallest cardinality of a generating set of idempotents for F(S) is at most $|I| + |\Lambda| - n + 1$, where n is the number of connected components of the Graham-Houghton graph of S.

Proof. Note that, since E is a subset of the edges of the Graham-Houghton graph of S, if $\{i, \lambda\} \in E$, then $p_{\lambda,i} \neq 0$. Hence the set X is well-defined. Furthermore, since X consists of idempotents, it follows that $\langle X \rangle \leq F(S)$. Therefore, to show that X generates F(S), it suffices to show that $\langle X \rangle$ contains every idempotent of S.

Certainly X contains 0. The non-zero idempotents of S are the elements $(i, p_{\lambda,i}^{-1}, \lambda)$, for each non-zero matrix entry $p_{\lambda,i}$, so let $p_{\lambda,i}$ be any such entry. If $\{i, \lambda\} \in E$, then $(i, p_{\lambda,i}^{-1}, \lambda) \in X$ and we are done, so suppose otherwise. Since $\{i, \lambda\}$ is an edge in the Graham-Houghton graph of S, in particular the vertices i and λ are in the same connected component. Therefore, they are connected in the spanning forest, and so there exists a path

$$(i, \mu_1, j_1, \mu_2, \ldots, j_n, \lambda)$$

in the spanning forest from i to λ , for some $n \in \mathbb{N}, \mu_1, \ldots, \mu_n \in \Lambda$, and $j_1, \ldots, j_n \in I$. Consider the product of elements

$$x = (i, p_{\mu_1, i}^{-1}, \mu_1) (j_1, p_{\mu_2, j_1}^{-1}, \mu_2) \cdots (j_n, p_{\lambda, j_n}^{-1}, \lambda).$$
(3.8)

Since $(i, \mu_1, j_1, \ldots, j_n, \lambda)$ is a path in the spanning forest, it follows that each factor in (3.8) is contained in X, and the product itself is non-zero. In particular,

$$x = \left(i, \, p_{\mu_1, i}^{-1} p_{\mu_1, j_1} p_{\mu_2, j_2}^{-1} \cdots p_{\mu_n, \lambda_n} p_{\lambda, j_n}^{-1}, \, \lambda\right) \in \langle X \rangle.$$

Each power of x has the form (i, g, λ) , for some $g \in G$. However, S is finite, and so some power x^k of x is idempotent. In particular, $(i, p_{\lambda,i}^{-1}, \lambda) = x^k \in \langle X \rangle$, and so X generates F(S).

If n is the number of connected components of the Graham-Houghton graph of S, then any spanning tree for the Graham-Houghton graph contains $|I| + |\Lambda| - n$ edges. Therefore the generating set X consists of $|I| + |\Lambda| - n + 1$ idempotents. It follows that a minimal generating set of idempotents for F(S) contains at most this many elements.

Note that Proposition 3.7 applies to any finite Rees 0-matrix semigroup over a group, including those that are not regular, and to those where the Graham-Houghton graph is not connected, as can be seen in the following example.

Example 3.9. Let $S = \mathscr{M}^0[\mathcal{S}_3; I, \Lambda; P]$ be a Rees 0-matrix semigroup over the symmetric group of degree 3, where $I = \{1', 2', 3', 4'\}$ and $\Lambda = \{1, 2, 3, 4, 5\}$, and P is given by

$$P = \begin{pmatrix} (1\ 3) & 0 & 0 & 0\\ 0 & 0 & 0 & 0\\ 0 & (2\ 3) & (1\ 2) & (1\ 2)\\ 0 & (1\ 3\ 2) & 0 & (2\ 3)\\ 0 & (1\ 3) & (1\ 2) & (1\ 2\ 3) \end{pmatrix}.$$

The Graham-Houghton graph of S, and a spanning forest, is shown in Figure 3.10. Since the Graham-Houghton graph has three connected components, there are $|I| + |\Lambda| - 3 = 6$ edges in any spanning forest, and so any generating set for F(S) obtained with Proposition 3.7 contains seven elements. The generating set corresponding to the spanning forest Figure 3.10 is

$$\{0, (1', (1 3), 1), (2', (2 3), 3), (3', (1 2), 3), (4', (1 2), 3), (4', (2 3), 4), (4', (1 3 2), 5)\}$$

In total, there are ten idempotents in S. It is easy to see that, in this case, 0 can be expressed as a product of the non-zero generators, and is therefore redundant.



Figure 3.10: The graphs of the Rees 0-matrix semigroup S described in Example 3.9. On the left is the Graham-Houghton graph of S, with its vertices corresponding to I on the top row, and its vertices corresponding to Λ on the bottom. On the right is a spanning forest of the Graham-Houghton graph.

Let $S = \mathscr{M}^0[G; I, \Lambda; P]$ be a finite Rees 0-matrix semigroup over a group G. We may compare the generating set X for F(S) described by Proposition 3.7 with the set of all idempotents of S. Let k and n be the number of edges and the number of connected components of the Graham-Houghton graph of S, respectively. As noted in the proof of Proposition 3.7, any spanning tree for the Graham-Houghton graph of S contains $|I| + |\Lambda| - n$ edges, which, by definition, is less than or equal to k. Therefore $|X| = |I| + |\Lambda| - n + 1 \le k + 1$, and k + 1is the number of idempotents of S. Note that the maximum possible value of k is $|I| \cdot |\Lambda|$, which occurs when P contains only non-zero entries. It follows that k + 1 can be much larger than |X|, and so in some cases, the generating set X may contain relatively few idempotents in comparison to the total number of idempotents of S. In [64, Chapter 3], Gray is principally concerned with describing idempotent generating sets of minimal cardinality for certain kinds of finite regular Rees 0-matrix semigroups over groups; see especially [64, Theorem 3.27]. However, the relevant criteria for computing such minimal generating sets do not lend themselves well to computation. As is often the case, finding an optimal solution is significantly more difficult than finding a solution that is nearly optimal.

The generating set given by Proposition 3.7 does not necessarily give a generating set of smallest cardinality; nor does it necessarily even give an *idempotent* generating set of smallest cardinality; see Example 3.11, for instance. However, in a certain sense, the generating set described by Proposition 3.7 is close to being optimal, which we now discuss.

Let $S = \mathscr{M}^0[G; I, \Lambda; P]$ be a finite Rees 0-matrix semigroup where G is a group, let I' be the subset of indices $i \in I$ such that $p_{\lambda,i} \neq 0$ for some $\lambda \in \Lambda$, and let $\Lambda' \subseteq \Lambda$ be defined analogously. Clearly, any generating set for F(S) contains at least one generator for each $i \in I'$, and at least one generator for each $\lambda \in \Lambda'$. In particular, any generating set for F(S) contains at least max{ $|I'|, |\Lambda'|$ elements (see also [114, Lemma 3.1]). On the other hand, the number of vertices of the Graham-Houghton graph that are contained in non-trivial connected components is $|I'| + |\Lambda'|$, and so the number of edges in any spanning forest is $|I'| + |\Lambda'| - m$, where m is the number of non-trivial connected components. If m = 0, then the only idempotent of S is 0, and the generating set {0} given in Proposition 3.7 is minimal. Otherwise m > 0; in this case the number of elements in the generating set given in Proposition 3.7 is $|I'| + |\Lambda'| - m + 1$, and

$$|I'| + |\Lambda'| - m + 1 \le |I'| + |\Lambda'| \le 2 \cdot \max\{|I'|, |\Lambda'|\}.$$

In particular, the generating set contains at most two times as many elements as is necessary.

Example 3.11. Let S be the Rees 0-matrix semigroup defined in Example 3.6. The generating set given in Example 3.6 contains two elements, which are idempotents. However, the Graham-Houghton graph of S is itself a tree, and so the generating set described in Proposition 3.7 contains four elements. Therefore, this generating set is not minimal.

Importantly, as well as being reasonably small, the generating set described in Proposition 3.7 can be computed quickly; a simple description of an algorithm that implements this result is given in Algorithm 3.12.

Algorithm 3.12 Compute a generating set for the idempotent generated subsemigroup of a finite Rees 0-matrix semigroup over a group.

Input: A finite Rees 0-matrix semigroup $S = \mathscr{M}^0[G; I, \Lambda; P]$ over a group G. **Output:** A generating set for the idempotent generated subsemigroup of S.

1: $X \leftarrow \emptyset$

2: $\Gamma \leftarrow$ the Graham-Houghton graph of S

3: $E \leftarrow$ the edges of a spanning forest of Γ

4: for each edge $\{i, \lambda\} \in E$ do

5: $X \leftarrow X \cup \left\{ \left(i, p_{\lambda,i}^{-1}, \lambda\right) \right\}$

6: if the induced subgraph of Γ on its non-isolated vertices is empty or complete bipartite then

 $7: \qquad X \leftarrow X \cup \{0\}$

8: return X.

Computationally, the only non-trivial step required by Algorithm 3.12 is the construction of a spanning forest for the Graham-Houghton graph of S. As with finding the connected components of a graph, which was discussed in Section 3.2.1, this can be done by using a standard depth- or breath-first search. Such algorithms have worst-case time complexity $O(|I| \cdot |\Lambda|)$, since $|I| \cdot |\Lambda|$ is the maximum number of edges in the Graham-Houghton graph of S.

The final part of Algorithm 3.12, on line 6, tests whether the induced subgraph of the Graham-Houghton graph of S on its non-isolated vertices is empty or is a complete bipartite graph. If this induced subgraph is empty, then S has no non-zero idempotents, and so certainly 0 is required in any generating set. Similarly, if the induced subgraph is complete bipartite, then the product of any number of non-zero idempotents is non-zero, and again 0 is required to be an element of the generating set. On the other hand, suppose that the induced subgraph is neither empty nor a complete bipartite graph. In this case, there exist non-isolated vertices $i \in I$ and $\lambda \in \Lambda$ such that $\{i, \lambda\}$ is not an edge of the Graham-Houghton graph, and there exist non-zero idempotents $(j, p_{\lambda,j}^{-1}, \lambda)$ and $(i, p_{\mu,i}^{-1}, \mu)$ in S whose product is 0. In particular, 0 is generated by the non-zero idempotents of S, and so 0 need not be a member of the generating set created by Algorithm 3.12. For example, given the Rees 0-matrix semigroup from Example 3.9, Algorithm 3.12 returns a generating set with six elements that does not contain 0.

Algorithm 3.12 has been implemented in the SEMIGROUPS package [101] for GAP [58] by the author. The idempotent generated subsemigroup of a finite Rees matrix or 0-matrix semigroup over a group can be computed with the function IdempotentGeneratedSubsemigroup.

For a finite Rees 0-matrix semigroup over an arbitrary semigroup, it is not clear how to develop algorithms for computing the idempotents, or a relatively small generating set for the idempotent generated subsemigroup, in terms of the underlying semigroup and the Graham-Houghton graph. We therefore end this section by posing the following problems.

Open Problem 3.13. Let S be a finite Rees 0-matrix semigroup over an arbitrary finite semigroup T, and assume that any necessary semigroup-theoretic properties of T are known *a priori*. Give algorithms for counting and listing the idempotents of S, using the properties of T and the Graham-Houghton graph of S.

Open Problem 3.14. Let S be a finite Rees 0-matrix semigroup over an arbitrary finite semigroup T, and assume that any necessary semigroup-theoretic properties of T are known a priori. Give an algorithm for constructing a generating set for F(S) that contains at most a constant multiple of rank(F(S)) elements.

3.3 The Green's structure of a Rees 0-matrix semigroup

As discussed in Chapter 1, one of the most important strategies for studying a semigroup is to analyse its Green's relations. Moreover, computing the Green's structure of a semigroup is a prerequisite for many further algorithms in computational semigroup theory, such as computing its maximal subsemigroups; see Chapter 4. Therefore, we wish to develop a framework for computing the Green's structure of an arbitrary Rees 0-matrix semigroup.

Of course, for any finite Rees 0-matrix semigroup $S = \mathscr{M}^0[T; I, \Lambda; P]$, the Green's structure on S can be computed by constructing and analysing the left and right Cayley digraphs of S, given sufficient time and memory. However, the order of S is $|I| \cdot |T| \cdot |\Lambda| + 1$, which may be significantly larger than the cardinality of any of I, T, or Λ . In particular, even when I, T, and Λ are relatively small, it may not be possible to store the elements of S in memory, which is required when constructing Cayley digraphs.

Furthermore, since S is defined by its underlying semigroup, index sets, and matrix, the structure of S is surely encoded in these components. Computing the Cayley digraphs of S ignores this; it would be sensible to attempt to take as much of this information into account as possible. In particular, we aim to be able to test for the Green's equivalence of elements

of a Rees 0-matrix semigroup by analysing the underlying semigroup, the index sets, and the matrix, and without having to enumerate every element of the semigroup.

In any semigroup-with-zero S, the multiplicative zero 0_S forms a \mathscr{J} -class of S; it is the unique minimal element in the partial order of \mathscr{J} -classes of S. In particular, the Green's classes of 0 in a Rees 0-matrix semigroup are given immediately, and require no further computation. Therefore, it remains to describe the Green's relations on the non-zero elements of a Rees 0-matrix semigroup. For a regular Rees 0-matrix semigroup over a group, this is especially straightforward, and is stated in the following result.

Lemma 3.15. Let $S = \mathscr{M}^0[G; I, \Lambda; P]$ be a regular Rees 0-matrix semigroup over a group G, and let $x = (i, g, \lambda), y = (j, h, \mu) \in S \setminus \{0\}$ be arbitrary non-zero elements. The following hold:

- (i) $x \mathscr{L}^S y$ if and only if $\lambda = \mu$;
- (ii) $x \mathscr{R}^S y$ if and only if i = j;
- (iii) $x \mathscr{H}^S y$ if and only if i = j and $\lambda = \mu$; and
- (iv) $x \mathscr{D}^S y$.

Thus the Green's relations of a regular Rees 0-matrix semigroup $S = \mathscr{M}^0[G; I, \Lambda; P]$, where G is a group, are defined very simply in terms of its index sets. The non-zero \mathscr{L} -classes of S are the sets $I \times G \times \{\lambda\}$ for each $\lambda \in \Lambda$, the non-zero \mathscr{R} -classes of S are the sets $\{i\} \times G \times \Lambda$ for each $i \in I$, the non-zero \mathscr{H} -classes of S are the sets $\{i\} \times G \times \{\lambda\}$ for each $i \in I$ and $\lambda \in \Lambda$, and the \mathscr{D} -classes of S are $\{0\}$ and $S \setminus \{0\}$. However, this is not true in general for an arbitrary Rees 0-matrix semigroup over a group, let alone for an arbitrary Rees 0-matrix semigroup, as demonstrated by the following example.

Example 3.16. Let $S = \mathscr{M}^0[G; I, \Lambda; P]$ be a Rees 0-matrix semigroup over a group G, and suppose there exists $i \in I$ such that $p_{\mu,i} = 0$ for all $\mu \in \Lambda$. Then $x \cdot (i, g, \lambda) = 0$ for all $x \in S$, $g \in G$, and $\lambda \in \Lambda$. In particular, the \mathscr{L} -class of any element of $\{i\} \times G \times \Lambda$ is trivial.

The following lemma and its immediate corollary show a broad connection between the Green's relations on a Rees 0-matrix semigroup, its index sets, and the Green's relations on its underlying semigroup.

Lemma 3.17. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup, and let $x = (i, t, \lambda), y = (j, u, \mu) \in S \setminus \{0\}$ be arbitrary non-zero elements. The following hold;

- (i) if $x \mathscr{L}^S y$, then $t \mathscr{L}^T u$ and $\lambda = \mu$;
- (ii) if $x \mathscr{R}^S y$, then $t \mathscr{R}^T u$ and i = j;
- (iii) if $x \mathscr{H}^S y$, then $t \mathscr{H}^T u$, i = j, and $\lambda = \mu$; and
- (iv) if $x \mathscr{D}^S y$, then $t \mathscr{D}^T u$.

Proof. If x = y, then certainly $t \mathscr{L}^T u$ and $\lambda = \mu$. Therefore, to prove (i), suppose that $x \mathscr{L}^S y$ but that $x \neq y$. By definition, there exist $\alpha = (k, v, \nu), \ \beta = (l, w, \eta) \in S$ such that $x = \beta y$ and $y = \alpha x$. Thus $(i, t, \lambda) = (l, wp_{\eta,j}u, \mu)$ and $(j, u, \mu) = (k, vp_{\nu,i}t, \lambda)$. In particular, $\lambda = \mu$, and $t = (wp_{\eta,j})u$ and $u = (vp_{\nu,i})t$ – i.e. $t \mathscr{L}^T u$.

Note that (ii) is the dual of (i), and (iii) follows from (i) and (ii) since $\mathscr{H} = \mathscr{L} \cap \mathscr{R}$.

To prove (iv), recall that $\mathscr{D} = \mathscr{L} \circ \mathscr{R}$. Therefore, if $x \mathscr{D}^S y$, then there exists an element $z = (k, v, \nu) \in S$ such that $x \mathscr{L}^S z \mathscr{R}^S y$. Then $t \mathscr{L}^T v$ by (i), and $v \mathscr{R}^T u$ by (ii). Thus $t \mathscr{D}^T u$, as required.

Corollary 3.18. Let $S = \mathcal{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup. The following hold:

- (i) $|S/\mathscr{L}| \ge |T/\mathscr{L}| \cdot |\Lambda| + 1;$
- (ii) $|S/\mathscr{R}| \ge |T/\mathscr{R}| \cdot |I| + 1;$
- (iii) $|S/\mathscr{H}| \ge |T/\mathscr{H}| \cdot |I| \cdot |\Lambda| + 1$; and
- (iv) $|S/\mathscr{D}| \ge |T/\mathscr{D}| + 1.$

Note that the additional Green's class counted in Corollary 3.18 in each instance is $\{0\}$.

Given two elements of an arbitrary Rees 0-matrix semigroup, Lemma 3.17 can be used to rule out the Green's equivalence of these elements, in certain cases. For example, a pair of elements with different first components are not \mathscr{R} -related, and a pair of elements are not \mathscr{D} -related if their middle components are not \mathscr{D} -related.

Ideally, we wish to produce an analogue of Lemma 3.15 that holds for an arbitrary finite semigroup, and which classifies the Green's equivalence of elements precisely in terms of separate conditions on the components of the elements. In full generality, the converse statements of Lemma 3.15 do not necessarily hold; see Example 3.19. In general, therefore, such an approach cannot simply involve only the underlying semigroup and the index sets; consideration must also be taken of the matrix.

Example 3.19. Let \mathcal{T}_2 denote the full transformation monoid of degree 2 (see Section 1.3.5), let $f \in \mathcal{T}_2$ be the constant transformation with $\operatorname{im}(f) = \{1\}$, and let P be a matrix whose sole entry is f. Define $S = \mathscr{M}^0[\mathcal{T}_2; \{i\}, \{\lambda\}; P]$. Since $1_{\mathcal{T}_2}$ and f are not \mathscr{D} -related in \mathcal{T}_2 , it follows by Lemma 3.17 that $(i, 1_{\mathcal{T}_2}, \lambda)$ and (i, f, λ) are not \mathscr{D} -related in S. On the other hand, $1_{\mathcal{T}_2}$ and $(1 \ 2)$ are \mathscr{R} -related in \mathcal{T}_2 , where $(1 \ 2)$ is the non-identity permutation of degree 2 written in disjoint cycle notation, but $(i, 1_{\mathcal{T}_2}, \lambda)$ and $(i, (1 \ 2), \lambda)$ are not \mathscr{R} -related in S. This is because any product involving f is a constant transformation, and so if $t \in \mathcal{T}_2$ is arbitrary, then

$$(i, 1_{\mathcal{T}_2}, \lambda)(i, t, \lambda) = (i, ft, \lambda) \neq (i, (1 2), \lambda).$$

Therefore, any characterisation of the Green's relations on S requires detailed reference to P.

The characterisation of the Green's relations on a regular Rees 0-matrix semigroup over a group stated in Lemma 3.15, which is given in terms of its index sets, relies on the fact that the matrix of the Rees 0-matrix semigroup contains a non-zero entry in each row and column. As shown in Example 3.16, without this assumption, the characterisation does not necessarily hold. The matrix of the Rees 0-matrix semigroup considered in Example 3.19 contains a non-zero entry in each row and column, but nevertheless, Green's \mathscr{R} -relation on the semigroup is not classified by Lemma 3.17(ii) and its converse. In order to obtain a more general classification of Green's relations on certain Rees 0-matrix semigroups, that is analogous to Lemma 3.15, we need to impose more complicated conditions on the matrix than simply requiring the existence of non-zero entries. Finding appropriate conditions is the topic of the next section.

3.3.1 Rees 0-matrix semigroups with row- or column-regular matrices

In this section, we aim to develop conditions to impose on the matrix of an arbitrary Rees 0-matrix semigroup, so that its Green's relations may be simply characterised in terms of its index sets and in terms of the Green's relations on its underlying semigroup. Moreover, these conditions on the matrix should be readily computable, as should the resultant characterisation of the Green's relations. With this in mind, we introduce the concept of row-regular and column-regular matrices. The reason for this choice of terminology will become apparent in Section 3.5.1, where we prove that a Rees 0-matrix semigroup over an arbitrary semigroup is regular if and only if its underlying semigroup is regular, and its matrix is both row-regular and column-regular. Let s be an element of a semigroup S. Recall that a relative left identity for s is an element $e \in S$ such that es = s; a relative right identity for s is defined analogously.

Definition 3.20 (Row-regular matrix). Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup. The matrix P is *row-regular* if, for all $\lambda \in \Lambda$ and $t \in T$, there exists $i \in I$ such that $p_{\lambda,i}T$ contains a relative right identity for t.

Definition 3.21 (Column-regular matrix). Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup. The matrix P is *column-regular* if, for all $i \in I$ and $t \in T$, there exists $\lambda \in \Lambda$ such that $Tp_{\lambda,i}$ contains a relative left identity for t.

In order for the notions of row- and column-regularity to be useful for our purposes, it must be practical to check whether or not a given matrix is row- or column-regular. Given only the definition, it is not immediately clear how to do this – on the face of it, testing whether the matrix P of a Rees 0-matrix semigroup $\mathscr{M}^0[T; I, \Lambda; P]$ is row-regular entails, for each $t \in T$ and $\lambda \in \Lambda$, searching for a relative right identity for t in as many as |I| right ideals of T.

In order to provide more useful reformulations of row-regularity, we prove the following lemma, which holds for any Rees 0-matrix semigroup over a finite semigroup.

Lemma 3.22. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup, where T is finite. The following are equivalent:

- (i) The matrix P is row-regular.
- (ii) If $\lambda \in \Lambda$ and L is a maximal \mathscr{L} -class of T, then there exists $i \in I$ such that $E(L \cap p_{\lambda,i}T) \neq \emptyset$.
- (iii) If $\lambda \in \Lambda$ and L is any maximal \mathscr{L} -class of T, then the right ideal of T generated by the non-zero entries of P in row λ contains an idempotent of L.

Proof. (i) \Rightarrow (ii). Let $\lambda \in \Lambda$, and let L_x be a maximal \mathscr{L} -class of T with representative $x \in L$. By assumption, there exists $i \in I$ such that $p_{\lambda,i}T$ contains a relative right identity s for x. In other words, xs = x, and so $xs^k = x$ for all $k \in \mathbb{N}$. Note that $s^k \in p_{\lambda,i}T$, since $p_{\lambda,i}T$ is a right ideal of T and is therefore a subsemigroup.

Since T is finite, there exists $n \in \mathbb{N}$ such that s^n is idempotent. It follows by Lemma 1.12(i) that $L_x = L_{xs^n} \leq L_{s^n}$. Since L_x is a maximal \mathscr{L} -class of T, we conclude that $L_x = L_{s^n}$, and hence that $s^n \in E(L_x \cap p_{\lambda,i}T)$, as required.

(ii) \Rightarrow (iii). Let $\lambda \in \Lambda$, let L be a maximal \mathscr{L} -class of T, and let K be the right ideal of T generated by the non-zero entries of P in row λ . By assumption, there exists $i \in I$ such that $E(L \cap p_{\lambda,i}T) \neq \emptyset$. Therefore $p_{\lambda,i} \neq 0$, which implies that $p_{\lambda,i} \in K$. Since K is a right ideal, $p_{\lambda,i}T \subseteq K$, and so $E(L \cap K) \neq \emptyset$, i.e. K contains an idempotent from L, as required.

(iii) \Rightarrow (i). Let $\lambda \in \Lambda$ and let $t \in T$. Since T is finite, there exists a maximal \mathscr{L} -class Lof T such that $L_t \leq L$. By assumption, the right ideal K of T generated by the set $\{p_{\lambda,i} : i \in I, p_{\lambda,i} \neq 0\}$ contains an idempotent e of L. Since $L_t \leq L = L_e$, it follows that t = uefor some $u \in T^1$. Thus $te = ue^2 = ue$, and e is a relative right identity for t. It remains to prove that there exists $i \in I$ such that $e \in p_{\lambda,i}T$. The ideal K is equal to the union of the principal right ideals $p_{\lambda,i}T^1$, for each $i \in I$ with $p_{\lambda,i} \neq 0$. Thus there exists $i \in I$ such that $e \in p_{\lambda,i}T^1 = p_{\lambda,i}T \cup \{p_{\lambda,i}\}$. If $e = p_{\lambda,i}$, then $e = e^2 = p_{\lambda,i}^2 \in p_{\lambda,i}T$, as required. Otherwise, if $e \neq p_{\lambda,i}$, then there is nothing to prove.

The following lemma is a direct analogue of Lemma 3.22, and is stated without proof.

Lemma 3.23. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup, where T is finite. The following are equivalent.

- (i) The matrix P is column-regular.
- (ii) If $i \in I$ and R is a maximal \mathscr{R} -class of T, then there exists $\lambda \in \Lambda$ such that $E(R \cap Tp_{\lambda,i}) \neq \emptyset$.
- (iii) If $i \in I$ and R is any maximal \mathscr{R} -class of T, then the left ideal of T generated by the non-zero entries of P in column i contains an idempotent of R.

For a Rees 0-matrix semigroup over a monoid, the conditions for row-regularity and column-regularity can be reformulated as follows. Note that, unlike Lemmas 3.22 and 3.23, the following result applies to infinite Rees 0-matrix semigroups as well as those that are finite.

Lemma 3.24. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup over a monoid T. Then:

- (i) the matrix P is row-regular if and only if each row of P contains an element from the *R*^T-class of 1_T; and
- (ii) the matrix P is column-regular if and only if each column of P contains an element from the L^T-class of 1_T.

Proof. We prove only (i), since (ii) is dual.

 (\Rightarrow) Let $\lambda \in \Lambda$. Since P is row-regular and $1_T \in T$, there exist $i \in I$ and $u \in T$ such that $1_T p_{\lambda,i} u = 1_T$. But 1_T is the identity of T, which implies that $1_T p_{\lambda,i} u = p_{\lambda,i} u$. Therefore $p_{\lambda,i} \cdot u = 1_T$ and $1_T \cdot p_{\lambda,i} = p_{\lambda,i}$, i.e. $p_{\lambda,i} \mathscr{R}^T 1_T$.

 (\Leftarrow) Let $\lambda \in \Lambda$ and $t \in T$ be arbitrary. By assumption, there exists $i \in I$ such that $p_{\lambda,i} \mathscr{R}^T \mathbf{1}_T$. Thus there exists $u \in T$ such that $p_{\lambda,i}u = \mathbf{1}_T$. In particular, $p_{\lambda,i}u \in p_{\lambda,i}T$ is a relative right identity for t.

Corollary 3.25. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup, where T is a finite monoid. Then:

- (i) the matrix P is row-regular if and only if each row of P contains a unit in T; and
- (ii) the matrix P is column-regular if and only if each column of P contains a unit in T.

Proof. In a finite monoid T, the group of units is the unique maximal \mathscr{J} -class in the partial order of \mathscr{J} -classes of T. Therefore, an element is \mathscr{R} -related to 1_T if and only if it is \mathscr{L} -related to 1_T if and only if it is a unit.

If a matrix is both row-regular and column-regular, then we simply say that it is *regular*. Traditionally, the matrix of a Rees 0-matrix semigroup over a group is said to be *regular* if it contains a non-zero entry in every row and every column; a Rees 0-matrix semigroup over a group is regular if and only if its matrix is regular. By Lemma 3.24, the notion of regularity given in Definition 3.26 extends this traditional meaning. More specifically, the matrix of a Rees 0-matrix semigroup over a group is row-regular and column-regular if and only if it contains a non-zero entry in every row and every column.

Definition 3.26 (Regular matrix). Let $S = \mathcal{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup. The matrix P is *regular* if it is both row-regular and column-regular.

Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup, and let $x = (i, t, \lambda) \in S \setminus \{0\}$ be arbitrary. By definition, if P is row-regular, then t has a relative right identity $p_{\lambda,j}u$ for some $j \in I$ and $u \in T$; thus the element (j, u, λ) is a relative right identity for x. The analogous statements hold when P is column-regular. In other words, the following lemma holds.
Lemma 3.27. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup. The following hold:

- (i) if P is row-regular, then each element of S and T has a relative right identity; and
- (ii) if P is column-regular, then each element of S and T has a relative left identity.

In particular, if P is row- or column-regular, then S and T are surjective semigroups.

A particular consequence of the previous lemma is that if T is an arbitrary semigroup, then it is not necessarily possible to construct a row- or column-regular matrix over T^0 . For example, if T is not surjective, then it is not possible to construct a row- or column-regular matrix over T^0 .

In Algorithm 3.31, we make use of the following observation. Let $\mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup, where T is finite. If P is row-regular, then Lemma 3.22(ii) implies that each maximal \mathscr{L} -class of T contains an idempotent; similarly, if P is column-regular, then Lemma 3.23(ii) implies that each maximal \mathscr{R} -class of T contains an idempotent. Since an \mathscr{L} or \mathscr{R} -class of a semigroup contains an idempotent if and only if it is regular, we obtain the following corollary to Lemmas 3.22 and 3.23.

Corollary 3.28. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup, where T is finite.

- (i) If the matrix P is row-regular, then every maximal \mathscr{L} -class of T is regular.
- (ii) If the matrix P is column-regular, then every maximal \mathscr{R} -class of T is regular.

On the other hand, the underlying semigroup of a Rees 0-matrix semigroup $\mathscr{M}^0[T; I, \Lambda; P]$ is not itself required to be regular in order for the matrix to be row- or column-regular: if T is a non-regular monoid and every row or column of P contains a unit of T, then P is row- or column-regular, respectively, by Lemma 3.24.

In Algorithm 3.31, we use the formulation of row-regularity given by Lemma 3.22(iii) to present an algorithm for determining whether the matrix of a finite Rees 0-matrix semigroup is row-regular. This algorithm assumes the ability to compute the maximal \mathscr{L} -classes of a finite semigroup (the sources in the quotient of the left Cayley digraph by its strongly connected components), the ability to test these \mathscr{L} -classes for regularity and to find their idempotents [37, Section 5.4]; and the ability to compute a right ideal of a finite semigroup. The right ideal generated by a set consists of the vertices in the right Cayley digraph of the semigroup that are reachable from the generators of the ideal. Therefore, the right ideal can be constructed by performing a depth- or breadth-first search in the Cayley digraph from each of these generators. There do not currently exist well-developed methods for computing left or right ideals, or the partial orders of \mathscr{L} - and \mathscr{R} -classes, of finite semigroups without computing Cayley digraphs.

Open Problem 3.29. Let S be a finite semigroup to which the techniques of [37] apply. Building on the results of [37], develop methods for computing the Green's structure of an arbitrary left or right ideal of S that do not necessarily exhaustively enumerate the ideal.

Open Problem 3.30. Let S be a finite semigroup to which the techniques of [37] apply. Building on the results of [37], develop methods for computing the partial orders of \mathscr{L} - and \mathscr{R} -classes of S without necessarily exhaustively enumerating S.

In the following proposition, we characterise the Green's relations of a Rees 0-matrix semigroup with a row- or column-regular matrix in terms of its index sets, and in terms of the Green's relations on its underlying semigroup. In particular, Proposition 3.32 states that the Green's relations on a Rees 0-matrix semigroup with a regular matrix are characterised by the statements of Lemma 3.17 and their converses. Corollary 3.33 follows immediately from this. Algorithm 3.31 Compute whether a finite Rees 0-matrix semigroup has a row-regular matrix. **Input:** A finite Rees 0-matrix semigroup $\mathscr{M}^0[T; I, \Lambda; P]$, where $P = (p_{\lambda,i})_{\lambda \in \Lambda, i \in I}$. **Output:** TRUE if P is row-regular; else FALSE. 1: $\mathfrak{L} \leftarrow$ the maximal \mathscr{L} -classes of T2: if some \mathscr{L} -class in \mathfrak{L} is not regular then return FALSE \triangleright Corollary 3.28. 3: 4: for $\lambda \in \Lambda$ do $A \leftarrow \{p_{\lambda,i} : i \in I, \ p_{\lambda,i} \neq 0\}$ \triangleright The non-zero entries of P in row λ . 5: $U \leftarrow$ the right ideal of T generated by A 6: for $L \in \mathfrak{L}$ do 7:if $U \cap E(L) = \emptyset$ then 8: return FALSE 9: 10: return TRUE

Proposition 3.32. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup, and let $x = (i, t, \lambda)$ and $y = (j, u, \mu)$ be arbitrary non-zero elements of S. The following hold:

- (i) if P is column-regular, then $x \mathscr{L}^S y$ if and only if $t \mathscr{L}^T u$ and $\lambda = \mu$;
- (ii) if P is row-regular, then $x \mathscr{R}^S y$ if and only if $t \mathscr{R}^T u$ and i = j;
- (iii) if P is regular, then $x \mathscr{H}^S y$ if and only if $t \mathscr{H}^T u$, i = j, and $\lambda = \mu$; and $x \mathscr{D}^S y$ if and only if $t \mathscr{D}^T u$.

Proof. By Lemma 3.17, it remains to prove the converse implication of each part.

Suppose that the matrix P is column-regular, and that $\lambda = \mu$ and $t \mathscr{L}^T u$. By the definition of Green's \mathscr{L} -relation, there exists $a \in S^1$ such that at = u. Since P is column-regular, there exists $\nu \in \Lambda$ such that $Tp_{\nu,i}$ contains a relative left identity for t. In particular, there exists $v \in T$ such that $vp_{\nu,i}t = t$. Therefore $(j, av, \nu) \in S$, and

$$(j, av, \nu) x = (j, avp_{\nu,i}t, \lambda)$$
$$= (j, at, \lambda)$$
$$= (j, u, \lambda) = y.$$

A symmetric argument shows that zy = x for some $z \in S$. Therefore $x \mathscr{L}^S y$, and (i) holds.

Note (ii) is dual to (i), and since $\mathscr{H} = \mathscr{L} \cap \mathscr{R}$, the statement concerning \mathscr{H} in (iii) holds by (i) and (ii). Suppose that P is regular, and that $t \mathscr{D}^T u$. Since $\mathscr{D} = \mathscr{L} \circ \mathscr{R}$, there exists an element $v \in T$ such that $t \mathscr{L}^T v \mathscr{R}^T u$. Thus, by (i) and (ii), $x = (i, t, \lambda) \mathscr{L}^S (j, v, \lambda) \mathscr{R}^S (j, u, \mu) = y$. That is, $x \mathscr{D}^S y$, and (iii) holds.

Corollary 3.33. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup. The following hold:

- (i) if P is column-regular, then $|S/\mathscr{L}| = |T/\mathscr{L}| \cdot |\Lambda| + 1$;
- (ii) if P is row-regular, then $|S/\mathscr{R}| = |T/\mathscr{R}| \cdot |I| + 1$; and
- (iii) if P is regular, then $|S/\mathscr{H}| = |T/\mathscr{H}| \cdot |I| \cdot |\Lambda| + 1$ and $|S/\mathscr{D}| = |T/\mathscr{D}| + 1$.

Let S be a Rees 0-matrix semigroup with a row-regular matrix. By using Proposition 3.32 and Corollary 3.33, if we may easily compute Green's \mathscr{R} -relation on T, then we may easily test for \mathscr{R} -equivalence of elements of S, and count the number of \mathscr{R} -classes of S. If S instead has a

column-regular matrix, then this is true for Green's \mathscr{L} -relation; and if S has a regular matrix, then this is also true for Green's \mathscr{H} - and \mathscr{D} -relations.

It is natural to consider whether there are Rees 0-matrix semigroups that have neither rowregular nor column-regular matrices, but whose Green's relations are nevertheless characterised by the statements of Lemma 3.17 and their converses. In Example 3.35, we provide such a semigroup. Furthermore, by Lemma 3.34, Green's \mathcal{H} -relation on any Rees 0-matrix semigroup over an \mathcal{H} -trivial semigroup is characterised as in the statement of Proposition 3.32(iii). Therefore, the most general converse of Proposition 3.32 does not hold.

Lemma 3.34. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup, where T is \mathscr{H} -trivial, and let $x = (i, t, \lambda)$ and $y = (j, u, \mu)$ be arbitrary non-zero elements of S. Then $x \mathscr{H}^S y$ if and only if $t \mathscr{H}^T u$ and i = j and $\lambda = \mu$.

Proof. The direct implication holds by Lemma 3.17. On the other hand, since T is \mathscr{H} -trivial, if $t \mathscr{H}^T u$ and i = j and $\lambda = \mu$, then in fact x = y, so certainly $x \mathscr{H}^S y$.

Example 3.35. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$, where T is any \mathscr{J} -trivial semigroup, $|I| = |\Lambda| = 1$, and the sole entry of P is 0. Then S is a zero semigroup, since the product of any two elements in S is 0. In particular, S is \mathscr{J} -trivial. Let $\mathscr{K} \in \{\mathscr{H}, \mathscr{L}, \mathscr{R}, \mathscr{D}, \mathscr{J}\}$ be any Green's relation, and let $x = (i, t, \lambda), y = (j, u, \mu) \in S \setminus \{0\}$ be arbitrary non-zero elements. Then

In particular, since $|I| = |\Lambda| = 1$, it follows that $x \mathscr{L}^S y$ if and only if $t \mathscr{L}^T u$ and $\lambda = \mu$; that $x \mathscr{R}^S y$ if and only if $t \mathscr{R}^T u$ and i = j; that $x \mathscr{H}^S y$ if and only if $t \mathscr{H}^T u$, i = j, and $\lambda = \mu$; and that $x \mathscr{D}^S y$ if and only if $t \mathscr{D}^T u$. Therefore, the Green's relations on S are characterised by the statements of Lemma 3.17 and their converses, but the matrix P is neither row-regular nor column-regular.

Rees 0-matrix semigroups over \mathscr{H} -trivial semigroups, and the Rees 0-matrix semigroups that were considered in Example 3.35, are somewhat atypical. Indeed, in Proposition 3.36, we provide a partial converse to Proposition 3.32, which applies to Rees 0-matrix semigroups with non-trivial index sets. Since one could argue that a 'typical' Rees 0-matrix semigroup has non-trivial index sets, Proposition 3.36 suggests that row-regularity and column-regularity are suitable conditions to require, if one desires to consider Rees 0-matrix semigroups whose Green's structure may be readily computed. This opinion is further reinforced in Section 3.5.1, where we prove that a Rees 0-matrix semigroup is regular if and only if its underlying semigroup is regular, and its matrix is row-regular and column-regular.

Proposition 3.36. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup. The following hold:

(i) if |I| > 1, and Green's \mathcal{L} -relation on S is characterised by

 $x \mathscr{L}^S y$ if and only if $t \mathscr{L}^T u$ and $\lambda = \mu$,

for all non-zero elements $x = (i, t, \lambda)$ and $y = (j, u, \mu)$ in S, then P is column-regular;

(ii) if $|\Lambda| > 1$, and Green's \mathscr{R} -relation on S is characterised by

 $x \mathscr{R}^S y$ if and only if $t \mathscr{R}^T u$ and i = j,

for all non-zero elements $x = (i, t, \lambda)$ and $y = (j, u, \mu)$ in S, then P is row-regular; and

(iii) if |I| > 1 and $|\Lambda| > 1$, and Green's \mathscr{D} -relation on S is characterised by

 $x \mathscr{D}^S y$ if and only if $t \mathscr{D}^T u$,

for all non-zero elements $x = (i, t, \lambda)$ and $y = (j, u, \mu)$ in S, then P is regular.

Proof. (i). Let (i, t, λ) be an arbitrary non-zero element of S. Since |I| > 1, there exists an index $j \in I \setminus \{i\}$ that is different from i. By assumption $(i, t, \lambda) \mathscr{L}^S(j, t, \lambda)$. Therefore, there exist $\mu \in \Lambda$ and $u \in T$ such that

$$(j, u, \mu)(i, t, \lambda) = (j, t, \lambda).$$

In particular, $up_{\mu,i}t = t$, and so $Tp_{\mu,i}$ contains a relative left identity for t. Since $i \in I$ and $t \in T$ were chosen arbitrarily, it follows that P is column-regular.

The proof of (ii) is dual to the proof of (i), and is therefore omitted.

(iii). Let (i, t, λ) be an arbitrary non-zero element of S. Since |I| > 1 and $|\Lambda| > 1$, there exist indices $j \in I \setminus \{i\}$ and $\mu \in \Lambda \setminus \{\lambda\}$. By assumption, $(i, t, \lambda) \mathscr{D}^S$ (j, t, μ) , and so there exists an element $x \in S \setminus \{0\}$ such that $(i, t, \lambda) \mathscr{L}^S x \mathscr{R}^S$ (j, t, μ) . By Lemma 3.17, $x = (j, u, \lambda)$ for some $u \in T$. Since $(i, t, \lambda) \mathscr{L}^S$ (j, u, λ) , there exist indices $\gamma, \delta \in \Lambda$ and elements $v, w \in T$ such that

$$(j, v, \gamma)(i, t, \lambda) = (j, u, \lambda)$$
 and $(i, w, \delta)(j, u, \lambda) = (i, t, \lambda)$.

By multiplying, we find that $vp_{\gamma,i}t = u$ and $wp_{\delta,j}u = t$, and so

$$(wp_{\delta,i}v)p_{\gamma,i}t = t.$$

In particular, $Tp_{\gamma,i}$ contains a relative left identity for t. It follows that P is column-regular. The proof that P is row-regular follows similarly from the fact that $(j, t, \mu) \mathscr{R}^S$ (j, u, λ) . Therefore P is regular.

3.4 Generating sets for Rees 0-matrix semigroups

As with direct products of semigroups, which were the topic of Chapter 2, a Rees 0-matrix semigroup is defined only in terms of its elements, and so it does not naturally come with a proper generating subset. Of course, if the matrix of a Rees 0-matrix semigroup S contains 0 as an entry, then it is obvious that $S \setminus \{0\}$ is a proper generating subset of S. Beyond this, however, it is not immediately apparent how to describe any smaller generating set for S. Sometimes this is not possible: any Rees 0-matrix semigroup with a matrix containing only 0 is a zero semigroup, and so for any such semigroup, the set of all non-zero elements is the only proper generating subset. However, for many Rees 0-matrix semigroups, it is possible to find generating sets that contain relatively few elements in comparison with the order of the semigroup: see Example 3.37 for an example of a Rees 0-matrix semigroup over a non-group, with thirteen elements, that has a minimal-cardinality generating set with only five elements.

Example 3.37. Let T be the semigroup $\{1, 2, 3\}$, with the operation of greatest common divisor. Clearly T is a commutative band. Let $I = \{1', 2'\}$, $\Lambda = \{1, 2\}$, and

$$P = \begin{pmatrix} 2 & 3\\ 3 & 2 \end{pmatrix},$$

and define S to be the Rees 0-matrix semigroup $\mathscr{M}^0[T; I, \Lambda; P]$. Then S contains 13 elements. It may be verified computationally that the set $X = \{0, (1', 2, 2), (1', 3, 1), (2', 2, 1), (2', 3, 2)\}$ is the unique minimal generating set of S. In particular, rank(S) = 5.

In this section, we describe relatively small generating sets for several kinds of Rees 0-matrix semigroups. The motivation for proving such results is the same as that behind Chapter 2. To repeat: many, perhaps most, algorithms in computational semigroup theory require a generating set, and so to compute most efficiently with a Rees 0-matrix semigroup, we require a generating set. As a rule of thumb, the smaller the generating set for a given semigroup, the faster it is to compute with that semigroup. Therefore, when it is possible and reasonably practical, we would like to describe and compute generating sets that contain relatively few elements in comparison with the order of the semigroup.

The finitely generated and presented Rees 0-matrix semigroups were classified by Ayik and Ruškuc in [5]: the Rees 0-matrix semigroup $\mathscr{M}^0[T; I, \Lambda; P]$ is finitely generated if and only if T is finitely generated and the sets I, Λ , and $T \setminus K$ are all finite, where K is the ideal of T generated by the non-zero entries of P.

Significant attention has been given to the problem of describing small and minimal generating sets of Rees 0-matrix semigroups over groups, particularly by Gray and Ruškuc; indeed, this topic featured in each of their PhD theses [64,115]. In [115, Chapter 4, Theorem 2.1], given a minimal generating set for the group G, Ruškuc describes a generating set for a regular Rees 0-matrix semigroup $\mathscr{M}^0[G; I, \Lambda; P]$, where G is a group, that contains $\operatorname{rank}(G) + |I| + |\Lambda| - 1$ elements. In [114, Theorems 3.9 and 4.6] Ruškuc gives a formula for the rank of a finite regular Rees 0-matrix semigroup over a group whose Graham-Houghton graph is connected. Gray and Ruškuc describe a formula for the rank of an arbitrary regular Rees 0-matrix semigroup over a group in [63, Theorems 7.1], and in [64, Thereom 2.68], Gray gives the rank of an arbitrary finite (not necessarily regular) Rees 0-matrix semigroup over a group.

Our focus is not on finding generating sets that are minimal. Describing generating sets for Rees 0-matrix semigroups whose underlying semigroups are not groups has not been considered in the literature. Currently, with the computational algebra system GAP [58], it is possible to construct the Rees 0-matrix semigroup $S = \mathcal{M}^0[T; I, \Lambda; P]$, given a semigroup T and a matrix $P : \Lambda \times I \longrightarrow T^0$. However, unless T is a group, GAP uses the set of all non-zero elements of S as the generating set of S whenever one is required, or the set of all elements, when the matrix P contains only non-zero entries. For an infinite Rees 0-matrix semigroup, therefore, it is not currently possible to compute a generating set in GAP, even though the semigroup may be finitely generated. When S is finite, it is theoretically possible to enumerate the set of all elements, given sufficient time and space, but doing so is clearly undesirable in general.

In Section 3.4.1, we present some results concerning the decomposable and indecomposable elements of Rees 0-matrix semigroups. Since a generating set of a semigroup contains its indecomposable elements, this allows us to provide lower bounds on the ranks of certain Rees 0-matrix semigroups. In Section 3.4.2, we provide a generating set for any Rees 0-matrix semigroup $\mathscr{M}^0[T; I, \Lambda; P]$, where T is a monoid and P contains a unit. This generating set contains $|I| + |\Lambda| - 1$ more elements than a given generating set for T. In Section 3.4.3, we present results concerning generating sets of Rees 0-matrix semigroups with row-regular or column-regular matrices.

Before we proceed, we first note that the following inequality holds for any Rees 0-matrix semigroup $\mathscr{M}^0[T; I, \Lambda; P]$. This is because, by the definition of multiplication in a Rees 0matrix semigroup, any generating set contains at least one member of $\{i\} \times T \times \Lambda$, for each $i \in I$, and at least one member of $I \times T \times \{\lambda\}$, for each $\lambda \in \Lambda$.

Lemma 3.38. For any Rees 0-matrix semigroup $S = \mathcal{M}^0[T; I, \Lambda; P]$, rank $(S) \ge \max\{|I|, |\Lambda|\}$.

3.4.1 Decomposable and indecomposable elements

As discussed in Section 2.2, an element s in a semigroup S is decomposable if $s \in S^2$, and indecomposable if $s \notin S^2$. By Lemma 2.9, the indecomposable elements of a semigroup are those elements of maximal \mathscr{J} -classes (in the partial order of \mathscr{J} -classes) that consist of a single non-idempotent element. The relevance of decomposable and indecomposable elements to this section is due to the observation that any generating set for a semigroup S contains its set of indecomposable elements $S \setminus S^2$. Therefore, by describing the indecomposable elements of a Rees 0-matrix semigroup S, we find elements that are contained in any generating set for S; in doing so, we deduce a lower bound on rank(S).

By Lemma 3.27, a Rees 0-matrix semigroup with a row- or column-regular matrix is surjective. However, as explored in the following lemma, an arbitrary Rees 0-matrix semigroup may contain a large number of indecomposable elements.

Lemma 3.39. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup, let K be the ideal of T generated by the non-zero elements of P, and let $(i, t, \lambda) \in S \setminus \{0\}$ be an arbitrary non-zero element. The following hold:

- (i) if (i, t, λ) is a decomposable element of S, then $t \in K$;
- (ii) if (i, t, λ) is a decomposable element of S, then t is a decomposable element of T;
- (iii) if every non-zero entry of P has a relative left identity and a relative right identity in T, then (i, t, λ) is decomposable if and only if t ∈ K.

Proof. (i). If $x = (i, t, \lambda)$ is decomposable, then there exist $\alpha = (j, u, \mu)$, $\beta = (k, v, \nu) \in S$ such that $x = \alpha\beta$. In particular, $p_{\mu,k} \neq 0$, and by equating the middle components, we find that $t = up_{\mu,k}v \in Tp_{\mu,k}T \subseteq K$.

(ii). As shown in the proof of (i), if $x = (i, t, \lambda)$ is decomposable, then there exist $u, v \in T$ and a non-zero matrix entry $p_{\mu,k} \in P$ such that $t = up_{\mu,k}v$. In particular, t is decomposable.

(iii). By (i), it remains to prove the converse implication. Suppose that $t \in K$. Since K is generated as an ideal by the non-zero elements of P, there exist $j \in I$ and $\mu \in \Lambda$ such that $p_{\mu,j} \neq 0$ and $t \in T^1 p_{\mu,j}T^1$. Thus $t = ap_{\mu,j}b$ for some $a, b \in T^1 = T \cup \{1_T\}$. By assumption, $p_{\mu,j}$ has a relative left identity $l \in T$ and a relative right identity $r \in T$. Thus $t = (al)p_{\mu,j}(rb) \in Tp_{\mu,j}T$. Define $\alpha = (i, al, \mu) \in S$ and $\beta = (j, rb, \lambda) \in S$. Then $x = \alpha\beta$, and x is decomposable.

Note that Lemma 3.39(i) implies that the number of elements in $T \setminus K$ is finite when S is finitely generated (using the notation of the lemma). This was previously shown in [5, Main Theorem], which classifies the finitely generated Rees 0-matrix semigroups.

We obtain the following immediate corollary from the contrapositive of Lemma 3.39(ii). This gives a lower bound on the number of indecomposable elements, and hence the rank, of any Rees 0-matrix semigroup.

Corollary 3.40. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup. The number of indecomposable elements of S, and also rank(S), is at least $|I| \cdot |T \setminus T^2| \cdot |\Lambda|$.

Example 3.41. Let $S = \mathscr{M}^0[\mathbb{N}; I, \Lambda; P]$ be any Rees 0-matrix semigroup over the natural numbers with addition. Since 1 is an indecomposable element of \mathbb{N} , it follows by Lemma 3.39(ii) that $I \times \{1\} \times \Lambda$ is a set of indecomposable elements of S.

Lemma 3.39(iii) classifies the decomposable and indecomposable elements of any Rees 0matrix semigroup where each non-zero entry of the matrix has a relative left identity and a relative right identity. Therefore, for any such Rees 0-matrix semigroup, we may provide a formula for the number of indecomposable elements. This applies, for instance, to Rees 0-matrix semigroups over regular semigroups, and Rees 0-matrix semigroups over monoids.

Corollary 3.42. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup where every non-zero entry of P has a relative left identity and a relative right identity in T, let A be the set of non-zero entries of P, and define

$$\mathfrak{J} = \{ J \in S / \mathscr{J} : J \not\leq J_x \text{ for any } x \in A \}.$$

Then the number of indecomposable elements of S is

$$|I|\cdot|\Lambda|\cdot\sum_{J\in\mathfrak{J}}|J|.$$

In particular, rank(S) is at least this number.

Proof. Let K be the ideal of T generated by A. By Lemma 3.39(iii), an element (i, t, λ) is indecomposable if and only if $t \notin K$. Since K is generated as an ideal by $A, t \notin K$ if and only if $t \notin T^1xT^1$ for any $x \in A$, which, by definition of the partial order of \mathscr{J} -classes of a semigroup, is true if and only if $J_t \nleq J_x$ for any $x \in A$. To summarise, (i, t, λ) is indecomposable if and only if $J_t \in \mathfrak{J}$. Note that the multiplicative zero of a Rees 0-matrix semigroup is decomposable. \Box

Example 3.43. Let $n \in \mathbb{N}$, let $S = \mathscr{M}^0[\mathcal{T}_n; I, \Lambda; P]$ be a Rees 0-matrix semigroup over the full transformation monoid \mathcal{T}_n of degree n, and suppose that P contains a transformation of rank n-1, but no permutations. Since \mathcal{T}_n is a monoid, every non-zero entry of P has a relative left and right identity in \mathcal{T}_n . Every ideal of \mathcal{T}_n is principal, and the ideals form a chain, described by $\mathcal{T}_n \ x \ \mathcal{T}_n \subseteq \mathcal{T}_n \ y \ \mathcal{T}_n$ if and only if $\operatorname{rank}(x) \leq \operatorname{rank}(y)$, for all $x, y \in \mathcal{T}_n$. Therefore the ideal of \mathcal{T}_n generated by the non-zero elements of P is the set $\mathcal{T}_n \setminus \mathcal{S}_n$ of all transformations with rank at most n-1. By Lemma 3.39(iii), the set of indecomposable elements of S is given by

$$S \setminus S^2 = I \times \mathcal{S}_n \times \Lambda.$$

3.4.2 Rees 0-matrix semigroups over monoids

Given that the ranks of Rees 0-matrix semigroups over groups have been described by Gray and Ruškuc, a natural extension is to consider the ranks, and minimal generating sets, in the case that the underlying semigroup is a monoid. In this short section, we present results relating the generating sets, and thereby the ranks, of certain Rees 0-matrix semigroups over monoids to those of the monoids themselves.

Lemma 3.44 (cf. [115, Chapter 4, Theorem 2.1]). Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup, where T is a monoid and P contains a unit in T. Let X be any generating set for T, and fix $i \in I$ and $\lambda \in \Lambda$ such that $p_{\lambda,i}$ is a unit in T. Then S is generated by the set

$$Y = \left\{ \left(i, x p_{\lambda,i}^{-1}, \lambda\right) : x \in X \right\} \cup \left\{ \left(i, 1_T, \mu\right) : \mu \in \Lambda \setminus \{\lambda\} \right\} \cup \left\{ \left(j, 1_T, \lambda\right) : j \in I \setminus \{i\} \right\} \cup \{0\}.$$

Proof. Certainly $\langle Y \rangle \subseteq S$, so it remains to prove the converse inclusion. Let $t \in T$ be arbitrary. Since X generates T and $tp_{\lambda,i} \in T$, there exists a sequence x_1, x_2, \ldots, x_n of elements of X such that $x_1x_2 \cdots x_n = tp_{\lambda,i}$. Therefore

$$(i, x_1 p_{\lambda,i}^{-1}, \lambda) (i, x_2 p_{\lambda,i}^{-1}, \lambda) \cdots (i, x_n p_{\lambda,i}^{-1}, \lambda) = (i, x_1 (p_{\lambda,i}^{-1} p_{\lambda,i}) x_2 (p_{\lambda,i}^{-1} p_{\lambda,i}) \cdots x_n p_{\lambda,i}^{-1}, \lambda)$$

$$= (i, x_1 x_2 \cdots x_n p_{\lambda,i}^{-1}, \lambda)$$

$$= (i, t p_{\lambda,i} p_{\lambda,i}^{-1}, \lambda)$$

$$= (i, t, \lambda).$$

In particular, $(i, t, \lambda) \in \langle Y \rangle$, and since $t \in T$ was arbitrary, it follows that $\{i\} \times T \times \{\lambda\} \subseteq \langle Y \rangle$. In particular, note that $(i, 1_T, \lambda) \in \langle Y \rangle$. Let $(j, t, \mu) \in S \setminus \{0\}$ be an arbitrary non-zero element of S. By the previous arguments, $(i, p_{\lambda,i}^{-1}tp_{\lambda,i}^{-1}, \lambda) \in \langle Y \rangle$, and so

$$(j, 1_T, \lambda)(i, p_{\lambda,i}^{-1}tp_{\lambda,i}^{-1}, \lambda)(i, 1_T, \mu) = (j, 1_T(p_{\lambda,i}p_{\lambda,i}^{-1})t(p_{\lambda,i}^{-1}p_{\lambda,i})1_T, \mu) = (j, t, \mu).$$

In particular, $(j, t, \mu) \in \langle Y \rangle$. Since $0 \in Y$ by definition, the result follows.

By choosing the generating set X in the statement of Lemma 3.44 to be a generating set of least cardinality, and by using Lemma 3.38, we deduce the following corollary to Lemma 3.44.

Corollary 3.45 (cf. [115, Chapter 4, Corollary 2.3]). Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0matrix semigroup, where T is a monoid and P contains a unit. Then

$$\max\{|I|, |\Lambda|\} \le \operatorname{rank}(S) \le \operatorname{rank}(T) + |I| + |\Lambda| - 1.$$

Although the problem of providing a formula for the rank of an arbitrary Rees 0-matrix semigroup seems rather intractable, we pose the following problem, whose solution could be a next step in the generalisation of the work of Gray and Ruškuc.

Open Problem 3.46. Let $S = \mathcal{M}^0[T; I, \Lambda; P]$ be an arbitrary finite Rees 0-matrix semigroup where T is a monoid and P contains a unit. Give a formula that describes rank(S) in terms of rank(T), the semigroup-theoretic properties of T, and the matrix P.

3.4.3 Rees 0-matrix semigroups with row- or column-regular matrices

In Section 3.3.1, we defined row- and column-regular matrices of Rees 0-matrix semigroups. We showed that, in essence, the Green's structure of a Rees 0-matrix semigroup with a row-regular or column-regular matrix can be calculated in terms of the index sets of the matrix and the Green's structure on the underlying semigroup. This suggests that such semigroups are an important class of Rees 0-matrix semigroups, a viewpoint that will be bolstered by Section 3.5.1, which concerns regular Rees 0-matrix semigroups. In the following results, we describe generating sets for these kinds of semigroups.

Proposition 3.47. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup where P is rowregular, and let X be a generating set for T. Fix $\mu \in \Lambda$, and for each $x \in X$, choose $i_x \in I$ and $t_x \in T$ such that $p_{\mu,i_x}t_x$ is a relative right identity for x. Then S is generated by the set:

$$\Gamma = \{(i, x, \mu) : i \in I, x \in X\} \cup \{(i_x, t_x x, \lambda) : x \in X, \lambda \in \Lambda\} \cup \{0\}$$

Proof. Note that, for each $x \in X$, the index $i_x \in I$ and the element $t_x \in T$ exist since P is assumed to be row-regular; to reiterate, $xp_{\mu,i_x}t_x = x$ for each $x \in X$. Certainly $\Gamma \subseteq S$ and $0 \in \Gamma$, so it remains to prove that $S \setminus \{0\} \subseteq \langle \Gamma \rangle$.

Let $s = (i, t, \lambda) \in S \setminus \{0\}$ be an arbitrary non-zero element of S. Since $t \in T$ and T is generated by X, we can write t as a product in the generators X. Indeed, since S is surjective by Lemma 3.27, it follows by Lemma 2.8 that there exist generators $x_1, x_2, \ldots, x_n \in X$, with $n \geq 4$, such that $t = x_1 \cdots x_n$. Thus $x_j p_{\mu, i_{x_j}} t_{x_j} = x_j$ for all $j \in \{1, \ldots, n\}$. It follows that

$$s = (i, t, \lambda) = (i, x_1 x_2 \cdots x_{n-1} x_n, \lambda)$$

= $(i, x_1 (p_{\mu, i_{x_1}} t_{x_1}) \cdot x_2 (p_{\mu, i_{x_2}} t_{x_2}) \cdots x_{n-1} (p_{\mu, i_{x_{n-1}}} t_{x_{n-1}}) \cdot x_n, \lambda)$
= $(i, x_1, \mu) (i_{x_1}, t_{x_1} x_2, \mu) \cdots (i_{x_{n-2}}, t_{x_{n-2}} x_{n-1}, \mu) (i_{x_{n-1}}, t_{x_{n-1}} x_n, \lambda)$
 $\in \langle \Gamma \rangle.$

Example 3.48. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be the Rees 0-matrix semigroup defined in Example 3.37, where T is the set $\{1, 2, 3\}$ with the operation of greatest common divisor, $I = \{1', 2'\}$, $\Lambda = \{1, 2\}$, and

$$P = \begin{pmatrix} 2 & 3 \\ 3 & 2 \end{pmatrix}.$$

The right ideal of T generated by the non-zero entries $\{2, 3\}$ of each row of P is T itself. Since every element of T is idempotent, it follows that this ideal contains an idempotent of each maximal \mathscr{L} -class of T. In particular, P is row-regular by Lemma 3.22(iii). Therefore, we may use Proposition 3.47 to describe a generating set for S. Let $X = \{2, 3\}$. Then X is a generating set for T. Fix $\mu = 1$. Since each element of T is idempotent, it follows that $p_{\mu,1'}2$ is a relative right identity for 2, and $p_{\mu,2'}3$ is a relative right identity for 3. In other words, if we define $i_2 = 1'$ and $t_2 = 2$, and $i_3 = 2'$ and $t_3 = 3$, it follows by Proposition 3.47 that S is generated by

$$\{0, (1', 2, 1), (1', 3, 1), (2', 2, 1), (2', 3, 1), (1', 2, 2), (2', 3, 2)\},\$$

which contains seven elements. As noted in Example 3.37, rank(S) = 5.

There is an obvious result that is dual to Proposition 3.47, which concerns generating sets of Rees 0-matrix semigroups with column-regular matrices. We state this result for completeness.

Proposition 3.49. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup where P is columnregular, and let X be a generating set for T. Fix $j \in I$, and for each $x \in X$, choose $\lambda_x \in \Lambda$ and $t_x \in T$ such that $t_x p_{\lambda_x, j}$ is a relative left identity for x. Then S is generated by the set:

 $\Gamma = \{(j, x, \lambda) : x \in X, \lambda \in \Lambda\} \cup \{(i, xt_x, \lambda_x) : i \in I, x \in X\} \cup \{0\}.$

By choosing a generating set X for T of minimal cardinality, we may use Propositions 3.47 and 3.49 to obtain an upper bound on the rank of a Rees 0-matrix semigroup with a row- or column-regular matrix. In combination with Lemma 3.38, we obtain the following corollary.

Corollary 3.50. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup where P is row- or column-regular. Then

$$\max\{|I|, |\Lambda|\} \le \operatorname{rank}(S) \le \operatorname{rank}(T)(|\Lambda| + |I|) + 1.$$

3.5 Special kinds of Rees 0-matrix semigroups

In the final section of this chapter, we present results that allow us to characterise several properties of Rees 0-matrix semigroups. In particular, we classify the Rees 0-matrix semigroups that are regular, are inverse, or are monoids. Moreover, these classifications are given in terms of the underlying semigroup and the matrix of the Rees 0-matrix semigroup, which may be readily computed.

3.5.1 Regular Rees 0-matrix semigroups

In this section, we show that the regular Rees 0-matrix semigroups are those with regular matrices constructed from regular semigroups. As stated in the following lemma, the underlying semigroup of a regular Rees 0-matrix semigroup is regular.

Lemma 3.51. Let $S = \mathcal{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup. If S is a regular semigroup, then T is a regular semigroup.

Proof. Let $t \in T$, $i \in I$, and $\lambda \in \Lambda$ be arbitrary, and define $x = (i, t, \lambda)$. Since S is a regular semigroup, there exists a non-zero element $y = (j, u, \mu) \in S \setminus \{0\}$ such that x = xyx. By equating the middle components of x and xyx, we find that $t = t(p_{\lambda,j}up_{\mu,i})t$, and so t is a regular element of T. Since t was arbitrary, the semigroup T is regular.

However, the converse of Lemma 3.51 does not hold, in general; for example, any Rees 0-matrix semigroup whose matrix consists of zeroes is a non-trivial zero semigroup, and is not regular, irrespective of the regularity of the underlying semigroup.

In Section 3.3.1, we introduced the notions of row-regular and column-regular matrices, in an attempt to classify those Rees 0-matrix semigroups whose Green's structure can be 'read off' from the index sets and the Green's structure of the underlying semigroup. In the following theorem, which is the main result of this section, we show that the regularity of a Rees 0-matrix semigroup is closely related to the row- and column-regularity of its matrix.

Theorem 3.52. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup. Then S is a regular semigroup if and only if T is a regular semigroup and the matrix P is regular.

Proof. (\Rightarrow) By Lemma 3.51, it remains to prove that the matrix P is both row-regular and column-regular. We prove that P is row-regular; the proof that P is column-regular is dual. Let $\lambda \in \Lambda$ and $t \in T$ be arbitrary. Fix $j \in I$, and define $x = (j, t, \lambda) \in S$. Since S is regular, there exists $y = (i, u, \mu) \in S$ such that x = xyx, and so $t = tp_{\lambda,i}up_{\mu,j}t$. Thus $p_{\lambda,i}up_{\mu,j}t$ is a relative right identity for t, and $p_{\lambda,i}(up_{\mu,j}t) \in p_{\lambda,i}T$, as required.

(\Leftarrow) Certainly 0 is a regular element, so let $x = (i, t, \lambda) \in S \setminus \{0\}$ be an arbitrary non-zero element of S. Since P is row-regular, there exist $j \in I$ and $a \in T$ such that $p_{\lambda,j}a$ is a relative right identity for t; since P is column-regular, there exist $\mu \in \Lambda$ and $b \in T$ such that $bp_{\mu,i}$ is a relative left identity for t. By assumption, T is a regular semigroup, and so there exists $u \in T$ such that t = tut. Define $y = (j, aub, \mu) \in S$. Then

$$xyx = (i, t, \lambda)(j, aub, \mu)(i, t, \lambda)$$

= $(i, (tp_{\lambda,j}a)u(bp_{\mu,i}t), \lambda)$
= (i, tut, λ)
= $(i, t, \lambda) = x.$

In other words, x is a regular element of S, and so S is regular.

Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup. By Theorem 3.52, we can determine whether S is regular solely by determining whether its underlying semigroup and matrix are regular. For a finite Rees 0-matrix semigroup, row-regularity of the matrix can be tested with Algorithm 3.31, and an analogous algorithm can be used to test for column regularity.

Throughout this chapter, we have often referred to the fact that a Rees 0-matrix semigroup over a group is regular if and only if each row and column of its matrix contains a non-zero element. This is easy to verify directly; it also follows from Theorem 3.52 and Lemma 3.24.

When the underlying semigroup of a Rees 0-matrix semigroup is a finite monoid, we may reformulate Theorem 3.52 in the following way.

Corollary 3.53. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup, where T is a finite monoid. Then S is regular if and only if T is regular and every row and every column of P contains a unit of T.

Proof. By Corollary 3.25, the matrix P is regular if and only if each row and each column of P contains a unit in T. The result follows by Theorem 3.52.

Example 3.54. Let T be the semigroup $\{-1, 1\}$ with the operation of min. Then T is a monoid with identity element 1 and multiplicative zero -1. Let I and Λ be index sets with |I| = 3 and $|\Lambda| = 2$, and let

$$P = \begin{pmatrix} 1 & -1 & -1 \\ -1 & 0 & 1 \end{pmatrix} \quad \text{and} \quad Q = \begin{pmatrix} -1 & 1 & 0 \\ 1 & 0 & 1 \end{pmatrix}.$$

By Corollary 3.53, the Rees 0-matrix semigroup $\mathscr{M}^0[T; I, \Lambda; P]$ is not a regular semigroup, since the second column of P does not contain a unit. On the other hand, $\mathscr{M}^0[T; I, \Lambda; Q]$ is a regular semigroup, since T is regular and each row and column of Q contains the identity element 1.

3.5.2 Rees 0-matrix monoids and inverse monoids

In this section, we prove that a Rees 0-matrix semigroup S over a semigroup T is a monoid if and only if T is a monoid and S is isomorphic to T^0 . Since not every monoid has a multiplicative zero, and certainly not every monoid is isomorphic to a monoid with zero adjoined, it follows that not all monoids can be represented as Rees 0-matrix semigroups. Moreover, we find that the only possible way to construct a Rees 0-matrix monoid is rather trivial. We also find that analogous statements hold for inverse Rees 0-matrix monoids.

We first prove the main result concerning Rees 0-matrix monoids.

Proposition 3.55. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup. Then S is a monoid if and only if $|I| = |\Lambda| = 1$, T is a monoid, and the sole entry of P is a unit in T.

Proof. (\Rightarrow) Any Rees 0-matrix semigroup has a non-zero element by definition, and so

$$0 \neq 1_S = (i, t, \lambda) \in (I \times T \times \Lambda).$$

Let $x = (j, u, \mu) \in (I \times T \times \Lambda)$ be arbitrary. Then

$$(j, u, \mu) = 1_S x$$

= $(i, t, \lambda)(j, u, \mu)$
= $(i, tp_{\lambda,j}u, \mu)$,

and so i = j. Since j was arbitrary, it follows that $I = \{i\}$. Similarly, the equation $x1_S = x$ implies that $\Lambda = \{\lambda\}$. The sole matrix entry $p_{\lambda,i}$ is non-zero, since 1_S is idempotent. In particular, $p_{\lambda,i} \in T$. Let $u \in T$ be arbitrary. Then $1_S(i, u, \lambda) = (i, u, \lambda) = (i, u, \lambda) 1_S$, which implies that $(tp_{\lambda,i})u = u = u(p_{\lambda,i}t)$. Therefore $tp_{\lambda,i}$ is a left identity for T, and $p_{\lambda,i}t$ is a right identity for T, and so T is a monoid with identity $tp_{\lambda,i} = p_{\lambda,i}t = 1_T$. Furthermore, $p_{\lambda,i}$ is a unit, with inverse $p_{\lambda,i}^{-1} = t$.

(\Leftarrow) Suppose that $I = \{i\}$ and $\Lambda = \{\lambda\}$. Define $e = (i, p_{\lambda,i}^{-1}, \lambda) \in S$, and let $x \in S$ be arbitrary. If x = 0, then certainly ex = x = xe. Otherwise, $x = (i, t, \lambda)$ for some $t \in T$, and

,

1) (

$$ex = (i, p_{\lambda,i}^{-1}, \lambda)(i, t, \lambda)$$
$$= (i, p_{\lambda,i}^{-1}p_{\lambda,i}t, \lambda)$$
$$= (i, 1_T t, \lambda) = x.$$

Similarly, xe = x. It follows that e is the identity element of S.

Note that by Lemma 3.24, a Rees 0-matrix monoid has a regular matrix.

According to the following lemma, we may normalize a Rees 0-matrix semigroup over a monoid; that is, we may assume up to isomorphism that its matrix entry is the identity element of its underlying monoid.

Corollary 3.56. Let $S = \mathscr{M}^0[T; \{i\}, \{\lambda\}; P]$ be a Rees 0-matrix monoid, and let Q be the $\{\lambda\} \times \{i\}$ matrix with unique entry 1_T . Then $S \cong \mathscr{M}^0[T; \{i\}, \{\lambda\}; Q]$.

Proof. In Proposition 3.5, choose θ , ψ , and χ to be identity functions, and choose $u_i = p_{\lambda,i}$ and $v_{\lambda} = 1_T$. The result follows.

For a normalized Rees 0-matrix monoid $S = \mathscr{M}^0[T; I, \Lambda; P]$, the function $\phi : S \longrightarrow T^0$ given by $0\phi = 0$ and $(i, t, \lambda)\phi = t$ for all $(i, t, \lambda) \in S \setminus \{0\}$ is clearly an isomorphism. Therefore, we deduce the following result from Corollary 3.56.

Corollary 3.57. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup. Then S is a monoid if and only if T is a monoid and $S \cong T^0$.

Given the preceding results concerning Rees 0-matrix monoids, it should perhaps not be surprising that the analogous results hold for inverse Rees 0-matrix monoids. Thus we present the following proposition and corollary, which are analogous to Proposition 3.55 and Corollary 3.57.

Proposition 3.58. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup. Then S is an inverse monoid if and only if $|I| = |\Lambda| = 1$, T is an inverse monoid, and the entry of P is a unit in T.

Proof. (\Rightarrow) By Proposition 3.55, it remains to prove that T is inverse. By Corollary 3.57, $S \cong T^0$, and so T^0 is an inverse monoid, i.e. every element of T^0 has a unique inverse. The sole inverse of any multiplicative zero is itself, and so every element of T has a unique inverse in T. Thus T is an inverse subsemigroup of T^0 .

 (\Leftarrow) By Corollary 3.57, $S \cong T^0$. Since T is an inverse monoid, so too are T^0 and S.

Corollary 3.59. A Rees 0-matrix semigroup $S = \mathscr{M}^0[T; I, \Lambda; P]$ is an inverse monoid if and only if T is an inverse monoid and $S \cong T^0$.

We end this section about monoids by considering the rank of a Rees 0-matrix monoid, which represents an initial step towards addressing Open Problem 3.46.

Lemma 3.60. Let $S = \mathscr{M}^0[T; \{i\}, \{\lambda\}; P]$ be a Rees 0-matrix monoid, and let X be a generating set for T. Then the set

$$Y = \left\{ \left(i, \, x p_{\lambda,i}^{-1}, \, \lambda \right) \, : \, x \in X \right\} \cup \{ 0 \}.$$

is a generating set for S. Furthermore, with respect to cardinality or containment, Y is a minimal generating set for S if and only if X is a minimal semigroup generating set for T. In particular, $\operatorname{rank}(S) = \operatorname{rank}(T) + 1$.

Proof. The fact that Y generates S follows by Lemma 3.44. Since $S \cong T^0$, the statements concerning minimality follows by Lemma 1.5, and the statement concerning rank(S) holds by Corollary 1.6.

3.5.3 Inverse Rees 0-matrix semigroups

In Section 3.5.2, we learned that constructing an inverse Rees 0-matrix monoid is essentially the same as adjoining a zero to an inverse monoid. In this section, we consider inverse Rees 0matrix semigroups more generally, and find that their classification is somewhat richer than the classification of inverse Rees 0-matrix monoids. We also describe an algorithm for computing whether a finite Rees 0-matrix semigroup is inverse, and we describe and count the idempotents of an inverse Rees 0-matrix semigroup. First we give the well-known classification of inverse Rees 0-matrix semigroups over groups. Let G be a group, and let I be any non-empty set. The $I \times I$ Brandt semigroup over G, denoted B(G, I), is the Rees 0-matrix semigroup $\mathscr{M}^0[G; I, I; P]$, where $P = (p_{i,j})_{i,j \in I}$ is the $I \times I$ matrix such that $p_{i,i} = 1_G$ and $p_{i,j} = 0$ when $i \neq j$. When $I = \{1, \ldots, n\}$ for some $n \in \mathbb{N}$, we denote the Brandt semigroup B(G, I) by B(G, n), which we call the $n \times n$ Brandt semigroup over G. By Proposition 3.1, two Brandt semigroups B(G, I) and B(H, J) are isomorphic is and only if $G \cong H$ and |I| = |J|. In particular, if I is a finite set, then $B(G, I) \cong B(G, |I|)$.

The importance of Brandt semigroups is that they characterise the inverse completely 0simple semigroups, as shown in the following result.

Proposition 3.61 ([76, Theorem 5.1.8]). Let S be any semigroup. Then S is completely 0simple and inverse if and only if S is isomorphic to the Brandt semigroup B(G, I), for some group G and some non-empty set I.

In particular, the principal factor of a \mathscr{J} -class of a finite inverse semigroup is isomorphic to the Brandt semigroup B(G, n), for some group G and some $n \in \mathbb{N}$.

Given the description of isomorphisms of regular Rees 0-matrix semigroups over groups from Proposition 3.1, we may reformulate Proposition 3.61 in the following way.

Corollary 3.62. Let $S = \mathscr{M}^0[G; I, \Lambda; P]$ be a Rees 0-matrix semigroup over a group G. Then S is inverse if and only if the matrix P contains exactly one non-zero entry in each row and each column.

The following theorem, which is the main result of this section, generalises Corollary 3.62 to Rees 0-matrix semigroups over arbitrary semigroups, rather than just groups. The proof of the converse implication of Theorem 3.63 relies on the following property of inverse semigroups: if T is an inverse semigroup, and $x, y \in T$ are elements such that x = xyx, then $x = ey^{-1}$ for some idempotent $e \in E(T)$ [76, Proposition 5.2.1]. Indeed, if x = xyx, then $x^{-1}x$ and yx are clearly \mathscr{L} -related (and hence equal) idempotents. Idempotents are self-inverse, and so $x^{-1}x = x^{-1}y^{-1}$, which implies that

$$x = x(x^{-1}x) = x(x^{-1}y^{-1}) = ey^{-1}$$

where $e = xx^{-1}$ is an idempotent.

Theorem 3.63. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a Rees 0-matrix semigroup. Then S is inverse if and only if T is an inverse monoid, the matrix P contains exactly one non-zero entry in each row and each column, and the non-zero matrix entries are units in T.

Proof. (\Rightarrow) Since an inverse semigroup is regular, it follows by Theorem 3.52 that T is regular, and that the matrix P is regular. In particular, each row and column of P contains at least one non-zero entry. Let $i \in I$ be arbitrary, and let $\lambda, \mu \in \Lambda$ be indices such that $p_{\lambda,i}$ and $p_{\mu,i}$ are non-zero. By the regularity of T, there exist inverses $u, v \in T$ of $p_{\lambda,i}$ and $p_{\mu,i}$ respectively. The elements (i, u, λ) and (i, v, μ) are idempotent, and so they commute, because S is inverse. In particular, $\lambda = \mu$, and so there exists a unique index $\lambda \in \Lambda$ such that $p_{\lambda,i} \neq 0$. An analogous argument shows that any row of P contains exactly one non-zero entry. Therefore, without loss of generality, we may assume that $I = \Lambda$, and that the matrix P is diagonal, i.e. that the non-zero entries of P are the entries $p_{i,i}$ for each $i \in I$.

We aim to show that T is a monoid, and that the non-zero entries of P are units in T; to do this, we first show that these entries are \mathscr{D}^T -related, and then that they are \mathscr{H}^T -related.

Let $i \in I$ and $t \in T$ be arbitrary. Since P is row-regular, $p_{i,i}T$ contains a relative right identity for t, and since P is column-regular, $Tp_{i,i}$ contains a relative left identity for t. In other

words, there exist elements $a, b \in T$ such that $ap_{i,i}t = t$ and $tp_{i,i}b = t$. In particular, $t \mathscr{L}^T p_{i,i}t$ and $t \mathscr{R}^T tp_{i,i}$. Since this holds for all $i \in I$ and $t \in T$, it follows that

$$p_{i,i} \mathscr{R}^T p_{i,i} p_{j,j} \mathscr{L}^T p_{j,j} \mathscr{R}^T p_{j,j} p_{i,i} \mathscr{L}^T p_{i,i}$$
(3.64)

for all $i, j \in I$. In particular, the non-zero matrix entries of P are \mathscr{D}^T -related.

By Lemma 1.15, the element $p_{i,i}p_{j,j}$ is contained in a group \mathscr{H}^T -class for all $i, j \in I$. Moreover, by choosing i = j in (3.64), we find that $p_{i,i} \mathscr{H}^T p_{i,i}^2$ for all $i \in I$, and so the \mathscr{H}^T -class of each non-zero matrix entry is a group. Let $i, j \in I$ be arbitrary, let e be the idempotent such that $e \mathscr{H}^T p_{i,i}$ and let f be the idempotent such that $f \mathscr{H}^T p_{i,i}p_{j,j}$. Since e and f are \mathscr{R}^T -related idempotents, it follows that each is a relative left identity for the other, i.e. ef = f and fe = e. Let u be the group-theoretic inverse of $p_{i,i}$ in its \mathscr{H}^T -class. Note that e is an identity for both $p_{i,i}$ and u, and that $p_{i,i}u = e = up_{i,i}$. Since $e \mathscr{R}^T f$ and \mathscr{R} is a left congruence, it follows that $u = ue \mathscr{R}^T uf$. By Proposition 3.32, $(i, u, i) \mathscr{R}^S (i, uf, i)$. But these elements are idempotents, and since S is an inverse semigroup, which contains a unique idempotent in each \mathscr{R} -class, it follows that u = uf. Multiplying on the left by $p_{i,i}$, we deduce that e = f, and so in fact $p_{i,i} \mathscr{H}^T p_{i,i}p_{j,j}$. By an analogous argument, $p_{j,j} \mathscr{H}^T p_{i,i}p_{j,j}$, and so $p_{i,i} \mathscr{H}^T p_{j,j}$. Since i and j were chosen arbitrary, it follows that non-zero entries of P are \mathscr{H}^T -related, and that their \mathscr{H}^T -class is a group, whose idempotent we name e.

To show that e is the identity of T, let $t \in T$, $i \in I$ be arbitrary, and define $x = (i, t, i) \in S$. Then $x^{-1} = (i, u, i) \in S$ for some $u \in T$. Certainly x is an inverse of x^{-1} , but so too is y = (i, te, i), since

$$x^{-1}yx^{-1} = (i, up_{i,i}tep_{i,i}u, i)$$

= $(i, up_{i,i}tp_{i,i}u, i)$
= $x^{-1}xx^{-1} = x^{-1},$

and similarly $yx^{-1}y = y$. In addition, z = (i, et, i) is an inverse of x^{-1} . Since x^{-1} has a unique inverse, it follows that y = x = z, and so te = t = et. Thus T is a monoid with identity e, and the non-zero entries of P are units in T.

Finally, to show that T is inverse, it suffices to show that the idempotents of T commute. Let $e, f \in E(T)$ and $i \in I$ be arbitrary, and let $x = (i, ep_{i,i}^{-1}, i), y = (i, fp_{i,i}^{-1}, i) \in S$. Then x and y are idempotents of S, which is an inverse semigroup, and so they commute. Therefore xy = yx, which implies that $efp_{i,i}^{-1} = fep_{i,i}^{-1}$, and so ef = fe.

 (\Leftarrow) It follows by Theorem 3.52 that S is regular, and so it remains to show that the idempotents of S commute. The multiplicative zero of S is an idempotent that commutes with every element of the semigroup. Since the matrix P contains precisely one non-zero entry in each row and column, we may assume without loss of generality that $I = \Lambda$, and that P is diagonal. Therefore, a non-zero element $(i, t, k) \in I \times T \times I$ is idempotent if and only if i = k and $t = tp_{i,i}t$. It follows by the discussion preceding the statement of the theorem that $t = tp_{i,i}t$ if and only if $t = ep_{i,i}^{-1}$ for some idempotent $e \in E(T)$. Thus, any non-zero idempotent of S can be written as $(i, ep_{i,i}^{-1}, i)$, for some $i \in I$ and $e \in E(T)$. On the other hand, any element of this form is idempotent. It follows that

$$E(S) = \left\{ \left(i, ep_{i,i}^{-1}, i\right) : i \in I, e \in E(T) \right\} \cup \{0\}.$$

Let $x = (i, ep_{i,i}^{-1}, i)$ and $y = (j, fp_{j,j}^{-1}, j)$ be arbitrary non-zero idempotents of S. If i = j, then

$$xy = (i, efp_{i,i}^{-1}, i) = (i, fep_{i,i}^{-1}, i) = yx$$

since $p_{i,i}$ is a unit and the idempotents of T commute. Otherwise, xy = yx = 0.

Remark 3.65. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be an inverse Rees 0-matrix semigroup, and let $x = (i, t, \lambda) \in S$ be an arbitrary non-zero element. Define $j \in I$ to be the unique index such that $p_{\lambda,j} \neq 0$, and define $\mu \in \Lambda$ to be the unique index such that $p_{\mu,i} \neq 0$. Then

$$x^{-1} = (j, p_{\lambda,j}^{-1}t^{-1}p_{\mu,i}^{-1}, \mu).$$

As we did with Proposition 3.55 and Rees 0-matrix monoids, we may use Theorem 3.63 in conjunction with Proposition 3.5 to describe normalized inverse Rees 0-matrix semigroups.

Corollary 3.66. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be an inverse Rees 0-matrix semigroup, and let Q be a $I \times I$ matrix with 1_T on its leading diagonal, and 0 elsewhere. Then $S \cong \mathscr{M}^0[T; I, I; Q]$.

Proof. In Proposition 3.5, choose θ to be the identity isomorphism on T, choose appropriate bijections $\psi: I \longrightarrow I$ and $\chi: \Lambda \longrightarrow I$ to diagonalise P, and choose $u_i = p_{\lambda,i}$ and $v_{\lambda} = 1_T$ for all $i \in I$ and $\lambda \in \Lambda$. The result follows.

In Algorithm 3.67, we use Theorem 3.63 to describe a procedure for computing whether an arbitrary finite Rees 0-matrix semigroup is inverse. This assumes the ability to test whether the underlying semigroup is an inverse monoid, to compute its group of units, and to test membership in the group of units. This algorithm is used in the SEMIGROUPS package [101] for GAP [58], when the function IsInverseSemigroup is given a Rees 0-matrix semigroup.

Algorithm 3.67 Compute whether a finite Rees 0-matrix semigroup is inverse.							
Input: A finite Rees 0-matrix semigroup $\mathscr{M}^0[T; I, \Lambda; P]$.							
Output: TRUE if $\mathscr{M}^0[T; I, \Lambda; P]$ is inverse; else FALSE.							
1: if T is not an inverse monoid or $ I \neq \Lambda $ then							
: return FALSE							
: $G \leftarrow$ the group of units of T							
4: for $\lambda \in \Lambda$ and $i \in I$ do							
5: SEEN_I[i] \leftarrow FALSE \triangleright SEEN_I[i] \Leftrightarrow a non-zero matrix entry $p_{\mu,i}$ has been found.							
6: SEEN_ $\Lambda[\lambda] \leftarrow$ FALSE \triangleright SEEN_ $\Lambda[\lambda] \Leftrightarrow$ a non-zero matrix entry $p_{\lambda,j}$ has been found.							
7: for $\lambda \in \Lambda$ do							
8: for $i \in I$ do							
if $p_{\lambda,i} \neq 0$ then							
$\mathbf{if} \ p_{\lambda,i} \notin G \ \mathbf{or} \ \mathrm{SEEN}_\mathrm{I}[i] \ \mathbf{or} \ \mathrm{SEEN}_\mathrm{\Lambda}[\lambda] \ \mathbf{then}$							
return FALSE							
$\text{SEEN}_{I}[i] \leftarrow \text{TRUE}$							
$\operatorname{SEEN}_{\Lambda}[\lambda] \leftarrow \operatorname{TRUE}$							
14: if not SEEN_ $\Lambda[\lambda]$ then							
15: return FALSE							
16: return TRUE							

Proof of Algorithm 3.67. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be a finite Rees 0-matrix semigroup, where P is a $\Lambda \times I$ matrix over T^0 . By Theorem 3.63, S is inverse if and only if T is an inverse monoid, and each row and each column of P contains exactly one non-zero entry, which is a unit in T. Thus the matrix of an inverse Rees 0-matrix semigroup is square. Clearly Algorithm 3.67 returns TRUE when the input S is inverse, and so it remains to show that Algorithm 3.67 returns FALSE when the input S is inverse, that is, when one of the following holds:

1. T is not an inverse monoid, or

- 2. P contains any non-zero entry that is not a unit in T, or
- 3. P contains two or more non-zero entries in the same row or column, or
- 4. P contains only 0 in some row or column.

If T is not an inverse monoid, then the algorithm returns FALSE on line 1. If P has any non-zero non-unit entries, or if P contains two or more non-zero entries in some row or column, then this is detected on line 10. If P contains any row that consists entirely of zeroes, then this is detected on line 14. Finally, suppose that P has some column consisting entirely of zeroes. We may assume that no row or column of P contains two or more non-zero entries since this will be checked independently, as already discussed. Therefore, the number of non-zero entries of P is strictly fewer than |I|. If $|I| \neq |\Lambda|$, then the algorithm returns FALSE on line 2; if $|I| = |\Lambda|$, then it follows that some row of P also consists entirely of zeroes, and the algorithm returns FALSE on line 15, provided that it has not already done so.

The proof of the converse implication of Theorem 3.63 involved a description of the idempotents of an inverse Rees 0-matrix semigroup. We state this description as the following lemma for the sake of completeness.

Lemma 3.68. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be an inverse Rees 0-matrix semigroup, and let $\sigma : I \longrightarrow \Lambda$ be the unique bijection such that $p_{i\sigma,i} \neq 0$ for each $i \in I$. Then the set of idempotents of S is given by

$$F(S) = E(S) = \left\{ \left(i, \, ep_{i\sigma,i}^{-1}, \, i\sigma \right) \, : \, i \in I, \, e \in E(T) \right\} \cup \{0\}.$$

In particular, S contains $1 + |I| \cdot |E(T)|$ idempotents.

Proof. Given the proof of Theorem 3.63, it remains to count the idempotents of S. If two non-zero idempotents $x = (i, ep_{i\sigma,i}^{-1}, i\sigma)$ and $y = (j, fp_{j\sigma,j}^{-1}, j\sigma)$ of S are equal, then certainly i = j, and $ep_{i\sigma,i}^{-1} = fp_{i\sigma,i}^{-1}$. But the non-zero matrix entries of P are units, and so

$$e = ep_{i\sigma,i}^{-1}p_{i\sigma,i} = fp_{i\sigma,i}^{-1}p_{i\sigma,i} = f.$$

Therefore, there are $|I| \cdot |E(T)|$ non-zero idempotents of S, along with the zero.

In the SEMIGROUPS [101] package, the functions Idempotents and NrIdempotents use the characterisation given in Lemma 3.68 to compute the idempotents, and the number of idempotents, respectively, of an inverse Rees 0-matrix semigroup.

On any inverse semigroup S may be defined a *natural partial order*. If $a, b \in S$, then $a \leq b$ in the natural partial order on S if and only if there exists $e \in E(S)$ such that a = eb. See [76, Section 5.2] for more information about the natural partial order on an inverse semigroup. In the following lemma, we use the description of the idempotents of an inverse Rees 0-matrix semigroup given by Lemma 3.68 to describe the natural partial order on such a semigroup.

Lemma 3.69. Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be an inverse Rees 0-matrix semigroup, and let $x = (i, t, \lambda)$ and $y = (j, u, \mu)$ be arbitrary non-zero elements of S. Then $x \leq y$ in the natural partial order on S if and only if i = j, $\lambda = \mu$, and $t \leq u$ in the natural partial order on T.

Proof. As in Lemma 3.68, let $\sigma: I \longrightarrow \Lambda$ be the bijection such that $p_{i\sigma,i} \neq 0$ for all $i \in I$.

 (\Rightarrow) Since $x \leq y$, it follows by Lemma 3.68 that there exists $k \in I$ and $e \in E(T)$ such that

$$(i, t, \lambda) = (k, ep_{k\sigma,k}^{-1}, k\sigma)(j, u, \mu).$$

In particular, $p_{k\sigma,j} \neq 0$, which implies that k = j. Therefore $(i, t, \lambda) = (j, ep_{j\sigma,j}^{-1}p_{j\sigma,j}u, \mu)$, which implies that $i = j, \lambda = \mu$, and t = eu, i.e. $t \leq u$ in the natural partial order on T.

(\Leftarrow) By assumption, there exists $e \in E(T)$ such that t = eu. By Lemma 3.68, $z = (i, ep_{i\sigma,i}^{-1}, i\sigma) \in E(S)$, and it is easy to verify that x = zy. Therefore $x \leq y$.

Chapter 4

Computing maximal subsemigroups of a finite semigroup

4.1 Introduction

A maximal subsemigroup of a semigroup S is a proper subsemigroup of S that is not contained in any other proper subsemigroup of S. The main purpose of this chapter is to present practical algorithms for computing the maximal subsemigroups of an arbitrary finite semigroup. These algorithms are based on the paper from 1968 of Graham, Graham, and Rhodes [61], and are implemented in the SEMIGROUPS package [101] for GAP [58]. Part of the research described in this chapter was conducted in collaboration with Casey R. Donoven and James D. Mitchell, and is published in [35]. Additionally, some of the results in this chapter, particularly most of those in Section 4.5, have been published in [45], in collaboration with Jitender Kumar, James East, and James D. Mitchell.

A closely related notion is that of a maximal subgroup of a group. We first address a potential source of confusion, since the term maximal subgroup has different and conflicting meanings in semigroup theory and group theory. In semigroup theory, maximal subgroup is a term meaning group \mathscr{H} -class. However, in this thesis, we exclusively use the meaning from group theory, where a maximal subgroup of a group G is a proper subgroup of G that is contained in no other proper subgroup of G.

Maximal subgroups of both finite and infinite groups have been extensively investigated in the literature. In part, this is because of the importance of maximal subgroups to other aspects of group theory. For example, the Frattini subgroup of a group is the intersection of its maximal subgroups, and gives information about the generating sets of the group. Maximal subgroups are also closely related to primitive permutation representations. If G is a group acting primitively on some set, then the stabilizer of any point is a maximal subgroup; on the other hand, if H is a maximal subgroup of a group G, then G acts primitively on the set of cosets of H in G. The maximal subgroups of the finite symmetric groups are described, in some sense, by the O'Nan-Scott Theorem [109,116], given the Classification of Finite Simple Groups. See [6–8, 11, 14, 23, 92, 93, 100, 111] and the references therein for research concerning maximal subgroups of infinite groups.

If G is a finite group, then the subgroups of G are the non-empty subsemigroups of G. Therefore, except for the trivial group, the maximal subsemigroups of a finite group are its maximal subgroups, and so the notions of maximal subgroups and maximal subsemigroups are not really distinct in this case. The same does not hold in general for infinite groups; see Example 4.1. It follows that computing the maximal subgroups of a finite group is a special case of the problem we consider here. Indeed, in the algorithms that we present, certain cases reduce to the computation of maximal subgroups of particular group \mathscr{H} -classes of the given semigroup, even for semigroups that are not groups. There are well-developed techniques for finding the maximal subgroups of a finite group, such as those of Cannon and Holt [17] which are implemented in MAGMA [13], and those of Eick and Hulpke [47], which are implemented in GAP [58]; see also [3]. Therefore, in this thesis, we will suppose that it is possible to compute the maximal subgroups of any given finite group.

Example 4.1. Let \mathbb{Z} denote the group of integers under addition. It is clear that \mathbb{N}_0 is a subsemigroup of \mathbb{Z} that is not a group, as is \mathbb{N} . Indeed, \mathbb{N}_0 is a maximal subsemigroup of \mathbb{Z} , since if x an arbitrary negative integer, then $\langle \mathbb{N}_0, x \rangle = \mathbb{Z}$. Furthermore, since $\mathbb{N} \leq \mathbb{N}_0$ and $\mathbb{N}_0 \setminus \mathbb{N} = \{0\}$, it follows that \mathbb{N} is a maximal subsemigroup of \mathbb{N}_0 ; see Lemma 4.3.

There are many papers in the literature relating to maximal subsemigroups of semigroups that are not groups. Maximal subsemigroups of infinite semigroups, as with maximal subgroups of infinite groups, are very different from their finite counterparts. Every non-empty finite semigroup has at least one maximal subsemigroup, and every proper subsemigroup of a finite semigroup is contained in a maximal subsemigroup of that semigroup. However, the same is not true in general for infinite semigroups: there exist infinite semigroups with proper subsemigroups that are contained in no maximal subsemigroups, and infinite semigroups with no maximal subsemigroups at all; see Example 4.2. Analogous statements hold for groups. See [38,74,86] and the references therein for information about maximal subsemigroups of infinite semigroups. There are also numerous papers in the literature about finding maximal subsemigroups of particular classes of finite semigroups; see for example [25–30,68,69,82,124,129–131]. In Chapter 5, we classify the maximal subsemigroups of various families of finite transformation and diagram monoids, using the techniques presented in this chapter.

Example 4.2. Let $S = (1, \infty)$ be the semigroup consisting of the real numbers greater than 1 whose operation is the usual multiplication of real numbers. Certainly (x, ∞) is a proper subsemigroup of S for any $x \in S$. Let T be any proper subsemigroup of S. There exist elements $a, b \in S \setminus T$ with a < b. Let U be the union of T with the ideal of S consisting of all real numbers greater than or equal to b. Then U is a proper subsemigroup of S (since $a \notin U$) that properly contains T (since $T \subseteq U$ and $b \in U \setminus T$). It follows that every proper subsemigroups of S.

The most important paper on the topic of maximal subsemigroups of finite semigroups is arguably that of Graham, Graham, and Rhodes [61]. In this paper, the authors prove that every maximal subsemigroup of a finite semigroup has certain features, and that every maximal subsemigroup has one of a small number of types. This paper appears to have been overlooked for many years, and special cases of the results it contains have been repeatedly reproved. While the paper describes the possible forms of a maximal subsemigroup, it does not classify which forms arise in a given finite semigroup. Determining this is difficult, and until the publication of our paper [35], no practical mechanism for doing so had appeared in the literature.

The naive algorithm for computing the maximal subsemigroups of an arbitrary finite semigroup is to construct all proper subsemigroups of the semigroup, to find their partial order induced by containment, and then to find the maximal elements in that partial order. Computing every subsemigroup of a finite semigroup S could be accomplished by finding the subsemigroup generated by each subset of S. Alternatively, the subsemigroups of S can be found by first constructing each 1-generated subsemigroup $\langle x \rangle$, for $x \in S$, and then iteratively and systematically finding the subsemigroup generated by each distinct pair of subsemigroups, until all have been found. In general, given a finite semigroup S, the time complexity of such algorithms is at least $O(2^{|S|})$, since there exist finite semigroups in which every subset is a subsemigroup. Finding all the subsemigroups of a semigroup is a difficult problem, which has been considered in the literature in some natural cases. To illustrate, there are 132 069 776 subsemigroups of the full transformation monoid of degree 4 up to conjugacy, and 3 161 965 550 in total [41, Table 2] (note that \mathcal{T}_4 contains only 256 elements). The exact values are not known for \mathcal{T}_n when $n \geq 5$, although a large lower bound has been proven [16, Theorem 9.1]. This suggests that finding all subsemigroups is not a feasible general-purpose approach to finding maximal subsemigroups.

In this chapter, we present algorithms for computing the maximal subsemigroups of an arbitrary finite semigroup using the results of Graham, Graham, and Rhodes [61]. In order to solve this problem in general, we first develop algorithms for computing the maximal subsemigroups of an arbitrary finite regular Rees 0-matrix semigroup over a group. Certain aspects of the general solution reduce to other well-known computational problems, such as finding all maximal cliques in a graph, finding strongly connected components in a digraph, and, as mentioned above, computing the maximal subgroups of a group.

The algorithms that we present for computing the maximal subsemigroups of an arbitrary finite semigroup S require a description of the Green's structure of S. Throughout this chapter, we assume that we are able to compute the Green's structure of any finite semigroup; see Section 1.4 for more information about this problem. Therefore, if computing the Green's structure of S is not practical, perhaps because it requires too much time, or memory, then the algorithms presented here cannot be used to find the maximal subsemigroups of S. In many examples, we find that if it is practical to compute the Green's structure of S using the algorithms we present. In such examples, the time taken to determine the Green's structure of S is roughly comparable to that taken to find its maximal subsemigroups, given the Green's structure. Further details can be found in Section 4.6.2.

This chapter is organised as follows. In Section 4.2, we provide some preliminary results concerning the form of a maximal subsemigroup of a finite semigroup, and we state the main results of Graham, Graham, and Rhodes [61] in Proposition 4.10. In Section 4.3, we describe algorithms for finding maximal subsemigroups in any finite regular Rees 0-matrix semigroup over a group. In Section 4.4, we build on these results by developing algorithms for finding the maximal subsemigroups of an arbitrary finite semigroup defined by a generating set. In Section 4.5, we present special cases of some of these techniques for finite monoids. These results are particularly useful for Chapter 5. In Section 4.6, we state the overall algorithm for computing maximal subsemigroups of finite semigroups, and analyse its performance experimentally. Note that we do not formally analyse the time complexities of the algorithms that we present.

4.2 The form of a maximal subsemigroup

In this section, we present some results concerning the form of a maximal subsemigroup of a finite semigroup. In particular, we restate the main proposition of Graham, Graham, and Rhodes from their 1968 paper [61]. We use these results when developing algorithms for arbitrary semigroups in the later parts of this chapter, and when describing the maximal subsemigroups of specific finite monoids in the next.

Lemma 4.3. Let S be a semigroup and let M be a subsemigroup of S such that $|S \setminus M| = 1$. Then M is a maximal subsemigroup of S.

Proof. Any proper subsemigroup of S lacks at least one element of S. Therefore, M is the only proper subsemigroups of S that contains M.

Example 4.4. Let S be a non-empty left-zero semigroup. It is clear that any subset of S is a subsemigroup. In particular, the subset $S \setminus \{x\}$ is a subsemigroup of S for each $x \in S$, and is therefore a maximal subsemigroup by Lemma 4.3. Any other proper subsemigroup is properly contained in such a semigroup, and is therefore not maximal.

We find that the indecomposable elements of a semigroup give rise to maximal subsemigroups of the kind described in Lemma 4.3. Recall that an element x of a semigroup S is *decomposable* if $x \in S^2$, and *indecomposable* if not; see Section 2.2 for more details.

Lemma 4.5. Let S be a semigroup, and let $x \in S \setminus S^2$ be an indecomposable element of S. Then $S \setminus \{x\}$ is a maximal subsemigroup of S. Moreover, if $S = \langle S \setminus S^2 \rangle$, then these are the only maximal subsemigroups of S.

Proof. Suppose that $x \in S \setminus S^2$. Certainly $x \notin \langle S \setminus \{x\} \rangle$, since x cannot be written as a non-trivial product of elements in S. Therefore $S \setminus \{x\}$ is a subsemigroup of S; it is maximal by Lemma 4.3.

Suppose that S is generated by its indecomposable elements, and let M be a maximal subsemigroup of S. Since M is a proper subsemigroup, there exists an indecomposable element $x \in S \setminus M$. In particular, $M \subseteq S \setminus \{x\}$. But $S \setminus \{x\}$ is a maximal subsemigroup containing M, which implies that $M = S \setminus \{x\}$.

Example 4.6. The number 1 is the only indecomposable element of the natural numbers \mathbb{N} under addition. Moreover, $\mathbb{N} = \langle 1 \rangle$. By Lemma 4.5, the unique maximal subsemigroup of \mathbb{N} is $\mathbb{N} \setminus \{1\} = \{2, 3, 4, \ldots\}$.

As shown in Lemma 2.9, if an element x in a semigroup S is indecomposable, then $\{x\}$ is a maximal \mathscr{J} -class of S. Therefore, a maximal subsemigroup M of S of the kind described in Lemma 4.5 is formed by removing a \mathscr{J} -class of S. In particular, M contains all other \mathscr{J} classes of S. In the following lemma, we prove that a maximal subsemigroup of any semigroup S contains every \mathscr{J} -class of S, except for one. This result was stated and proved for finite semigroups in [61, PROPOSITION (1)]. It follows that if S is \mathscr{J} -trivial, then every maximal subsemigroup of S is found by removing a \mathscr{J} -class, and has the form described in Lemma 4.3.

Lemma 4.7 (cf. [61, PROPOSITION (1)]). Let S be an arbitrary semigroup and let M be a maximal subsemigroup of S. There exists a \mathcal{J} -class of S such that $S \setminus M \subseteq J$.

Proof. Let $x, y \in S \setminus M$ be arbitrary. Since M is a maximal subsemigroup of S, it follows that $S = \langle M, x \rangle$. Therefore $y \in \langle M, x \rangle \setminus M$, which implies that y can be expressed as a product in $M \cup \{x\}$ that involves x. In particular, y = axb for some $a, b \in S^1$. By a symmetric argument, there exist elements $s, t \in S^1$ such that x = syt. Hence $x \not = y$.

In this thesis, we call a maximal subsemigroup whose complement is contained in a \mathscr{J} -class J a maximal subsemigroup arising from J. Note that if U is a subset of a semigroup S, and $J \in S/\mathscr{J}$, then $S \setminus U \subseteq J$ if and only if $S \setminus J \subseteq U$. By Lemma 4.7, the problem of computing every maximal subsemigroup in a semigroup can be approached by determining the maximal subsemigroups that arise from each of its \mathscr{J} -classes in turn, or indeed in parallel.

In the following lemma, we classify the maximal subsemigroups arising from a \mathscr{J} -class of a semigroup S in terms of the generating sets of S. This result is useful in this thesis when describing the maximal subsemigroups of a semigroup whose generating sets are well-understood. In particular, Lemma 4.8 is used in the proofs of Theorems 5.7, 5.49, and 5.60.

Lemma 4.8. Let S be a semigroup, and let J be a \mathcal{J} -class of S. Suppose there exists an indexing set I and subsets $X_i \subseteq J$ for $i \in I$. Further suppose that no set X_i is contained in a

different set X_j , and that $S = \langle S \setminus J, A \rangle$ if and only if $A \cap X_i \neq \emptyset$ for all $i \in I$. Then the maximal subsemigroups of S arising from J are precisely the sets $S \setminus X_i$ for each $i \in I$.

Proof. Let $i \in I$ be arbitrary. We show that $S \setminus X_i$ is a subsemigroup of S; its maximality is then obvious. Let $x, y \in S \setminus X_i$. Since $S \setminus X_i$ does not generate S, but it contains $S \setminus J$ and an element $x_j \in X_j$ for each $j \in I \setminus \{i\}$, it follows that $xy \notin X_i$. Conversely, let M be a maximal subsemigroup of S arising from J. If $M \cap X_i \neq \emptyset$ for each i then, by assumption, $S = \langle M \rangle = M$, a contradiction. Thus $M \cap X_i = \emptyset$ for some i. In other words, $M \subseteq S \setminus X_i$. By the maximality of M in S, it follows that $M = S \setminus X_i$.

In the remainder of this section, we restrict our attention to finite semigroups. By [61], a maximal subsemigroup of a finite semigroup S is either a union of \mathscr{H} -classes S, or it intersects every \mathscr{H} -class of S non-trivially; see Lemma 4.9. It is not known whether this statement holds more generally for infinite semigroups. The techniques that we develop for finding maximal subsemigroups of a finite semigroup differ significantly according to whether the desired maximal subsemigroup is a union of \mathscr{H} -class of the semigroup, or intersects each \mathscr{H} -class of the semigroup non-trivially.

Lemma 4.9 ([61, PROPOSITION (2)]). A maximal subsemigroup of a finite semigroup S either intersects every \mathcal{H} -class of S non-trivially, or is a union of \mathcal{H} -classes of S.

Proof. Let M be a maximal subsemigroup of S. We will show that

$$T = \bigcup_{x \in M} H_x^S$$

is a subsemigroup of S. Since M is maximal, it then follows that either T = M, and M is a union of \mathcal{H} -classes of S, or that T = S, and M intersects every \mathcal{H} -class of S non-trivially.

Let $s, t \in T$ be arbitrary. Certainly $S \setminus J \subseteq M$ by Lemma 4.7, and $M \subseteq T$ by definition. Therefore, to show that T is a subsemigroup, it suffices to show that $st \in T$ whenever $st \in J$. So assume that $st \in J$, and that $s \in J$ or $t \in J$. In the first case, suppose that $s \in J$ and $t \in M$. Since $s \in J \cap T$, and a \mathscr{J} -class is a union of \mathscr{H} -classes, there exists by definition some $x \in M \cap J$ such that $s\mathscr{H}^S x$. By Lemma 1.10, $s \mathscr{R} st$, and so by Green's Lemma (Lemma 1.11), right multiplication by t defines an \mathscr{H} -class preserving bijection from L_s^S to L_{st}^S . In particular, $s\mathscr{H}^S xt \in M$, and so $st \in T$. A dual argument shows that $st \in T$ when $s \in M$ and $t \in J$. Finally, suppose that $s, t \in J$. Define x as above, and choose any $y \in M \cap J$ such that $t\mathscr{H}^S y$. By Lemma 1.10, $s \mathscr{R} st \mathscr{L} t$. By Lemma 1.15, $H_s^S H_t^S \subseteq H_{st}^S$. In particular, $xy\mathscr{H}^S st$, and since $xy \in M$, it follows that $st \in T$.

The main result of Graham, Graham, and Rhodes [61] describes, in the finite case, the possible intersections of a maximal subsemigroup with the \mathscr{J} -class from which it arises. In Proposition 4.10, we include a slightly reformulated version of this main result. The principal difference is that the isomorphism ϕ in Proposition 4.10 is arbitrary, whereas the result in Graham, Graham, and Rhodes specifies only that there exists some isomorphism with the required properties. Proposition 4.10 follows from the proof in [61]. Proposition 4.10(a) is part (3) of the main proposition [61], and Proposition 4.10(b) is Case 1 of part (4). Proposition 4.10(c) is adapted from Case 2 of part (4) of the main proposition of [61], in order to render the statements (i)–(iv) mutually-exclusive.

Proposition 4.10 (PROPOSITIONS (3) and (4) in [61]). Let S be a finite semigroup, let M be a maximal subsemigroup of S, and let J denote the \mathscr{J} -class of S that contains $S \setminus M$. If J is regular, define $\phi : J^* \longrightarrow \mathscr{M}^0[G; I, \Lambda; P]$ to be any isomorphism from the principal factor of J to a regular Rees 0-matrix semigroup over a group G. Then exactly one of the following holds:

- (a) J is non-regular and $M \cap J = \emptyset$;
- (b) J is regular and

 $(M \cap J)\phi \cup \{0\} \cong \mathscr{M}^0[H; I, \Lambda; Q],$

where H is a maximal subgroup of G and Q is a $\Lambda \times I$ matrix over H^0 . In this case, $(M \cap J)\phi \cup \{0\}$ is a maximal subsemigroup of $\mathscr{M}^0[G; I, \Lambda; P]$;

- (c) J is regular and $(M \cap J)\phi$ is equal to one of the following:
 - (i) (I × G × Λ) \ (I' × G × Λ') for some proper non-empty subsets I' ⊊ I and Λ' ⊊ Λ. In this case, (M ∩ J)φ ∪ {0} is a maximal subsemigroup of M⁰[G; I, Λ; P];
 - (ii) $I \times G \times \Lambda'$ for some proper non-empty subset Λ' of Λ ;
 - (iii) $I' \times G \times \Lambda$ for some proper non-empty subset I' of I;
 - (iv) Ø.

In the remainder of this chapter, we develop effective methods for determining the maximal subsemigroups of each possible form that arise in a finite semigroup, given its Green's structure. In order to be able to easily refer to these possible forms, we introduce the labels (M1)–(M5) as follows. Let S be any finite semigroup, and let M be a maximal subsemigroup of S. By Lemma 4.7, there exists a \mathscr{J} -class J of S such that $S \setminus M \subseteq J$. By Proposition 4.10, if $M \cap J$ is non-empty, then J is regular and precisely one of the following holds:

- (M1) $M \cap J$ has non-empty intersection with every \mathscr{H} -class in J (Proposition 4.10(b));
- (M2) $M \cap J = \bigcup_{A \in \mathfrak{L} \cup \mathfrak{R}} A$, where \mathfrak{L} and \mathfrak{R} are non-empty sets of \mathscr{L} and \mathscr{R} -classes of J, respectively (Proposition 4.10(c)(i));
- (M3) $M \cap J$ is a non-empty union of \mathscr{L} -classes in J (Proposition 4.10(c)(ii));
- (M4) $M \cap J$ is a non-empty union of \mathscr{R} -classes in J (Proposition 4.10(c)(iii));
- (M5) $M = S \setminus J$ (Proposition 4.10(a) and (c)(iv)).

A maximal subsemigroup of type (M1) intersects every \mathscr{H} -class of S non-trivially, whereas maximal subsemigroups of types (M2)–(M5) are unions of \mathscr{H} -classes of S; see Lemma 4.9. Note that a maximal subsemigroup of type (M5) can arise from either a regular or a non-regular \mathscr{J} -class; maximal subsemigroups of types (M1)–(M4) only arise from a regular \mathscr{J} -class.

In general, the collection of maximal subsemigroups arising from a particular regular \mathscr{J} class J can have any combination of types (M1)–(M4). However, if $S \setminus J$ is a maximal subsemigroup of S, then it is the only maximal subsemigroup to arise from J, since a maximal subsemigroup of type (M1)–(M4) properly contains $S \setminus J$. In other words, there is at most one maximal subsemigroup of type (M5) arising from J, and its existence precludes the occurrence of maximal subsemigroups of types (M1)–(M4).

4.3 Finite regular Rees 0-matrix semigroups over groups

In this section, we describe techniques for computing the maximal subsemigroups of an arbitrary finite regular Rees 0-matrix semigroup over a group. This topic has inherent interest. As discussed in Section 3.1, a finite semigroup is 0-simple if and only if it is isomorphic to a regular Rees 0-matrix semigroup over a group, and in some sense, finite 0-simple semigroups are the building blocks of finite semigroups. Furthermore, we use the algorithms for finite regular Rees

0-matrix semigroups over groups when computing the maximal subsemigroups of an arbitrary finite semigroup, which is the topic of Section 4.4.

The following result, which classifies the maximal subsemigroups of a finite regular Rees 0-matrix semigroup over a group, is central to the algorithms presented in this section. This proposition is a slight reformulation of [61, Remark 1] and [62, Theorem 4].

Proposition 4.11 (cf. [62, Theorem 4] and [61, Remark 1]). Let $S = \mathscr{M}^0[G; I, \Lambda; P]$ be a finite regular Rees 0-matrix semigroup over a group, and let M be any subset of S. Then M is a maximal subsemigroup of S if and only if one of the following conditions holds:

- (R1) $M = \{0\}$ and $|G| = |I| = |\Lambda| = 1;$
- (R2) $M = I \times G \times \Lambda$ and every entry of P is non-zero;
- (R3) $|\Lambda| > 1$, $M = S \setminus (I \times G \times \{\lambda\})$ for some $\lambda \in \Lambda$, and for each $i \in I$ there exists $\mu \in \Lambda \setminus \{\lambda\}$ such that $p_{\mu,i} \neq 0$;
- (R4) |I| > 1, $M = S \setminus (\{i\} \times G \times \Lambda)$ for some $i \in I$, and for each $\lambda \in \Lambda$ there exists $j \in I \setminus \{i\}$ such that $p_{\lambda,j} \neq 0$;
- (R5) $M = S \setminus ((I \setminus X) \times G \times (\Lambda \setminus Y))$, where X and Y are proper non-empty subsets of I and Λ , respectively, and $X \cup Y$ is a maximal independent subset of the Graham-Houghton graph of S;
- (R6) *M* is a subsemigroup of *S* isomorphic to a regular Rees 0-matrix semigroup of the form $\mathscr{M}^{0}[H; I, \Lambda; Q]$, where *H* is a maximal subgroup of *G* and *Q* is a $\Lambda \times I$ matrix over H^{0} .

In [62, Theorem 4] and [61, Remark 1], the characterisation of the maximal subsemigroups of type (R1) is incorrectly stated. In more detail, if $S = \mathscr{M}^0[G; I, \Lambda; P]$ is a finite regular Rees 0-matrix semigroup where G is a group, then these results claim that $\{0\}$ is a maximal subsemigroup of S if and only if $|I| = |\Lambda| = 1$ and G is a simple cyclic group. However, if $I = \{i\}$ and $\Lambda = \{\lambda\}$ and G is non-trivial, then the set $\{0, (i, p_{\lambda,i}^{-1}, \lambda)\}$ is a proper subsemigroup of S that properly contains $\{0\}$, and so $\{0\}$ is not a maximal subsemigroup of S.

By Lemma 4.9, a maximal subsemigroup of a finite semigroup is either a union of \mathcal{H} -classes of the semigroup, or it intersects every \mathcal{H} -class of the semigroup non-trivially. Maximal subsemigroups of types (R1)–(R5) are subsemigroups that are unions of \mathcal{H} -classes, while maximal subsemigroups of type (R6) intersect every \mathcal{H} -class. The maximal subsemigroups of type (R6) are the most complicated to describe.

In Sections 4.3.1–4.3.4, we describe how to compute maximal subsemigroups of an arbitrary finite regular Rees 0-matrix semigroup over a group that have types (R1)-(R6). In each case, we also describe how to specify generating sets for these maximal subsemigroups. Often, this requires the construction of generating sets for related regular Rees 0-matrix semigroups over groups; as discussed in Section 3.4, constructing such generating sets is well-understood and straightforward. Although finding minimal generating sets for these kinds of semigroups is well-studied [63, 114], we do not claim that the generating sets in this section are minimal.

In Section 4.3.5, we demonstrate many of the results of Sections 4.3.1–4.3.4 by finding the maximal subsemigroups of a specific Rees 0-matrix semigroup.

Finally, in Section 4.3.6 we describe how to compute the maximal subsemigroups of any finite regular Rees 0-matrix semigroup that have type (R6) and that contain a particular subset of elements; see Algorithm 4.44. This is required by Algorithm 4.86 from Section 4.4, which describes a procedure for computing the maximal subsemigroups of an arbitrary finite semigroup.

4.3.1 Maximal subsemigroups of types (R1) and (R2)

Let $S = \mathscr{M}^0[G; I, \Lambda; P]$ be an arbitrary finite regular Rees 0-matrix semigroup over a group G. It is trivial to determine whether there exists a maximal subsemigroup of S of type (R1) or type (R2). For the former, we test whether G, I, and Λ are trivial, and if they are, then $\{0\}$ is a maximal subsemigroup of S. For the latter, we simply check whether 0 appears as an entry in the matrix P, and if it does not, then $S \setminus \{0\} = I \times G \times \Lambda$ is a maximal subsemigroup of S.

Example 4.12. Let S_2 denote the symmetric group of degree 2, let I and Λ be index sets with two elements, and define

$$P = \begin{pmatrix} \operatorname{id}_2 & 0\\ 0 & (1\ 2) \end{pmatrix} \quad \text{and} \quad Q = \begin{pmatrix} \operatorname{id}_2 & (1\ 2)\\ (1\ 2) & \operatorname{id}_2 \end{pmatrix}.$$

If $S = \mathscr{M}^0[\mathcal{S}_2; I, \Lambda; P]$, then S has no maximal subsemigroup of type (R2), since the matrix P contains 0. However, every entry of Q is non-zero, so if $T = \mathscr{M}^0[\mathcal{S}_2; I, \Lambda; Q]$, then $T \setminus \{0\}$ is the unique maximal subsemigroup of T of type (R2). Neither S nor T has a maximal subsemigroup of type (R1), since their underlying groups and index sets are non-trivial.

It is not complicated to describe generating sets for maximal subsemigroups of types (R1) and (R2). The subset $\{0\}$ is a trivial subsemigroup of S, and so its unique generating is $\{0\}$ itself. If $S \setminus \{0\}$ is a maximal subsemigroup of S, then by Lemma 1.5(i), a subset X of S generates S if and only if $X \setminus \{0\}$ generates $S \setminus \{0\}$. Therefore, a generating set for $S \setminus \{0\}$ can be obtained from any generating set for S by removing the zero element. See Section 3.4 for more information about finding generating sets for finite regular Rees 0-matrix semigroups over groups.

4.3.2 Maximal subsemigroups of types (R3) and (R4)

Let $S = \mathscr{M}^0[G; I, \Lambda; P]$ be an arbitrary finite regular Rees 0-matrix semigroup over a group G. In Proposition 4.11, the existence of maximal subsemigroups of types (R3) and (R4) is characterised in terms of the existence of certain non-zero entries in the matrix P. It is possible to test these conditions directly. However, for computational purposes, it is simpler to reformulate these conditions in terms of the Graham-Houghton graph of S.

Lemma 4.13. Let $S = \mathscr{M}^0[G; I, \Lambda; P]$ be a finite regular Rees 0-matrix semigroup over a group G. The following hold:

- (i) If |Λ| > 1 and λ ∈ Λ, then S \ (I × G × {λ}) is a maximal subsemigroup of S if and only if λ is not adjacent to a vertex of degree one in the Graham-Houghton graph of S.
- (ii) If |I| > 1 and i ∈ I, then S \ ({i} × G × Λ) is a maximal subsemigroup of S if and only if i is not adjacent to a vertex of degree one in the Graham-Houghton graph of S.

Proof. We prove only (i), since (ii) is dual. Suppose that $|\Lambda| > 1$ and let $\lambda \in \Lambda$.

If $S \setminus (I \times G \times \{\lambda\})$ is a maximal subsemigroup of R, then it certainly has type (R3). Therefore, for each $i \in I$, there exists $\mu \in \Lambda \setminus \{\lambda\}$ such that $p_{\mu,i} \neq 0$. In other words, every vertex $i \in I$ in the Graham-Houghton graph of S that is adjacent to λ is also adjacent to some other vertex, and therefore has degree at least 2.

Conversely, suppose that every vertex $i \in I$ that is adjacent to λ in the Graham-Houghton graph of S has degree at least 2. Therefore, each such vertex is adjacent to some vertex $\mu \neq \lambda$. In other words, for each $i \in I$ there exists $\mu \in \Lambda \setminus \{\lambda\}$ such that $p_{\mu,i} \neq 0$. Therefore, by Proposition 4.11, the set $S \setminus (I \times G \times \{\lambda\})$ is a maximal subsemigroup of S of type (R3). \Box **Example 4.14.** Define $S = \mathscr{M}^0[\mathcal{S}_3; I, \Lambda; P]$, where \mathcal{S}_3 denotes the symmetric group of degree 3, $I = \{1', 2', 3'\}$ and $\Lambda = \{1, 2, 3\}$, and

$$P = \begin{pmatrix} \mathrm{id}_3 & (1\ 3\ 2) & (1\ 2) \\ (1\ 3) & (2\ 3) & 0 \\ 0 & \mathrm{id}_3 & 0 \end{pmatrix}.$$

See Figure 4.15 for a picture of the Graham-Houghton graph of S. Note that |I| > 1 and $|\Lambda| > 1$. As discussed in the caption of Figure 4.15, 1 and 2' are the only vertices in the Graham-Houghton graph of S that are adjacent to vertices of degree one. Therefore, by Lemma 4.13, the sets

 $S \setminus (I \times G \times \{2\})$ and $S \setminus (I \times G \times \{3\})$,

are the maximal subsemigroups of type (R3), and the maximal subsemigroups of type (R4) are

$$S \setminus (\{1'\} \times G \times \Lambda)$$
 and $S \setminus (\{3'\} \times G \times \Lambda)$.



Figure 4.15: The Graham-Houghton graph of the Rees 0-matrix semigroup $S = \mathcal{M}^0[S_3; I, \Lambda; P]$ from Example 4.14, where $I = \{1', 2', 3'\}$ and $\Lambda = \{1, 2, 3\}$. The only vertices of degree 1 in this graph are 3 and 3', which are adjacent to the vertices 2' and 1, respectively. There are four maximal independent subsets of the Graham-Houghton graph of $S: \{1, 2, 3\}, \{1', 2', 3'\}, \{1', 3', 3\}, \text{ and } \{3', 2, 3\}$. Therefore, by Lemma 4.13, there is one maximal subsemigroup of type (R3) and one maximal subsemigroup of type (R4), and by Proposition 4.11, there are two maximal subsemigroups of S of type (R5).

A method that uses Lemma 4.13 to compute the maximal subsemigroups of S that have type (R3) is given in Algorithm 4.16. One of the steps involved is the construction of the Graham-Houghton graph of S, which requires only the identification of the non-zero entries of the matrix P. The only other substantive calculation involved in this algorithm occurs in line 7, which tests whether the vertex $i \in I$ has a unique neighbour $\lambda \in \Lambda$. Finding the neighbours of a vertex in a graph is a task that is fundamental to almost every algorithm in graph theory. Therefore, given any useful representation of the Graham-Houghton graph, finding the neighbours of i is trivially easy. Indeed, a graph is often represented on a computer by storing the set of neighbours of each vertex.

There is an obvious analogue of Algorithm 4.16 that describes a method for computing the maximal subsemigroups of a finite regular Rees 0-matrix semigroup over a group that have type (R4). Given the preceding discussion, therefore, it is easy to find the maximal subsemigroups of S of types (R3) and (R4).

If $T = S \setminus (I \times G \times \{\lambda\})$ is a maximal subsemigroup of S of type (R3), for some $\lambda \in \Lambda$, then we may regard T as the regular Rees 0-matrix semigroup $\mathscr{M}^0[G; I, \Lambda \setminus \{\lambda\}; Q]$, where $Q = (p_{\mu,i})_{\mu \in \Lambda \setminus \{\lambda\}, i \in I}$ is the submatrix of P formed by removing row λ . Similarly, we can view a maximal subsemigroup of S of type (R4) as the regular Rees 0-matrix semigroup $\mathscr{M}^0[G; I \setminus \{i\}, \Lambda; Q']$, for some $i \in I$ and for some submatrix Q' of P. There are well-developed techniques for finding generating sets of Rees 0-matrix semigroups over groups; see Section 3.4 for a discussion of this topic. In this way, by finding a generating set for a corresponding Rees 0-matrix semigroup, it is straightforward to construct a generating set for a maximal subsemigroup of type (R3) or (R4). Algorithm 4.16 Maximal subsemigroups of a finite regular Rees 0-matrix semigroup over a group that have type (R3).

Input: A finite regular Rees 0-matrix semigroup $S = \mathscr{M}^0[G; I, \Lambda; P]$, where G is a group. **Output:** The set \mathfrak{M} of maximal subsemigroups of S of type (R3).

1: $\mathfrak{M} \leftarrow \emptyset$ 2: if $|\Lambda| = 1$ then return M. 3: 4: $\Gamma \leftarrow$ the Graham-Houghton graph of S 5: FORBIDDEN $\leftarrow \emptyset$ 6: for each $i \in I$ do if some $\lambda \in \Lambda$ is the unique neighbour of *i* then 7:FORBIDDEN \leftarrow FORBIDDEN $\cup \{\lambda\}$ 8: for each $\lambda \in \Lambda \setminus$ forbidden do 9: $\mathfrak{M} \leftarrow \mathfrak{M} \cup \{S \setminus (I \times G \times \{\lambda\})\}$ \triangleright Lemma 4.13 10: 11: return M.

4.3.3 Maximal subsemigroups of type (R5)

In this section, we discuss how to compute the maximal subsemigroups of an arbitrary finite regular Rees 0-matrix semigroup $S = \mathscr{M}^0[G; I, \Lambda; P]$, where G is a group, that have type (R5).

Certainly, the subsets I and Λ of the Graham-Houghton graph of S are maximal independent subsets. By Proposition 4.11, the remaining maximal independent subsets are in one-to-one correspondence with the maximal subsemigroups of type (R5). Therefore, computing these maximal subsemigroups is equivalent to computing the maximal cliques in the complement of the Graham-Houghton graph of S. This is a well-understood but computationally hard problem; see [15, 103, 125] for more information.

The methods in the SEMIGROUPS package [101] for GAP [58] for computing maximal subsemigroups of type (R5) use the clique-finding methods from the DIGRAPHS package [10], which were written by the author. These clique-finding methods are based on the Bron-Kerbosch Algorithm [15], and use the automorphism group of the graph to break the symmetry of the search space. In many cases, it seems that the majority of the time taken to compute the maximal subsemigroups of a finite regular Rees 0-matrix semigroup over a group is spent calculating the maximal independent subsets of the Graham-Houghton graph. See Section 4.6.2, especially Figure 4.93, for more information about the performance of the methods that are implemented in SEMIGROUPS.

Example 4.17. Let $S = \mathscr{M}^0[S_3; I, \Lambda; P]$ be defined as in Example 4.14; see Figure 4.15 for a picture of the Graham-Houghton graph of S. Two of the maximal independent subsets of the Graham-Houghton graph are I and Λ ; the remaining maximal independent subsets are $\{1', 3', 3\} = (I \setminus \{2'\}) \cup (\Lambda \setminus \{1, 2\})$ and $\{3', 2, 3\} = (I \setminus \{1', 2'\}) \cup (\Lambda \setminus \{1\})$. Therefore, by Proposition 4.11, the maximal subsemigroup of S of type (R5) are the sets

 $S \setminus (\{2'\} \times G \times \{1, 2\})$ and $S \setminus (\{1', 2'\} \times G \times \{1\}).$

An efficient way to specify a maximal subsemigroups of S that has type (R5) is by a generating set. Let X and Y be proper non-empty subsets of I and Λ , respectively, such that $X \cup Y$ is a maximal independent subset of the Graham-Houghton graph of S. The maximal

subsemigroup T of S corresponding to this maximal independent subset is given by

$$T = S \setminus ((I \setminus X) \times G \times (\Lambda \setminus Y))$$

= $(X \times G \times Y) \cup (I \times G \times (\Lambda \setminus Y)) \cup ((I \setminus X) \times G \times \Lambda) \cup \{0\}.$

Let $Q = (p_{\lambda,i})_{\lambda \in \Lambda \setminus Y, i \in I}$ and $Q' = (p_{\lambda,i})_{\lambda \in \Lambda, i \in I \setminus X}$. We may regard $(I \times G \times (\Lambda \setminus Y)) \cup \{0\}$ as a regular Rees 0-matrix semigroup $\mathscr{M}^0[G; I, \Lambda \setminus Y; Q]$, and we may regard $((I \setminus X) \times G \times \Lambda) \cup \{0\}$ as the regular Rees 0-matrix semigroup $\mathscr{M}^0[G; I \setminus X, \Lambda; Q']$. As discussed in Section 3.4, it is straightforward to construct a small generating set for a regular Rees 0-matrix semigroup over a group.

Let A be any subset of T that contains generating sets for $(I \times G \times (\Lambda \setminus Y)) \cup \{0\}$ and $((I \setminus X) \times G \times \Lambda) \cup \{0\}$, and that contains any element of $X \times G \times Y$. Let $(i, g, \lambda) \in X \times G \times Y$ be arbitrary, and let (k, h, γ) be the unique element of $A \cap (X \times G \times Y)$. Since $k \in X$, there exists $\mu \in \Lambda \setminus Y$ such that $p_{\mu,k} \neq 0$, and since $\gamma \in Y$, there exists $j \in I \setminus X$ such that $p_{\gamma,j} \neq 0$. Noting that

 $\left(i, \, gp_{\mu,i}^{-1}, \, \mu\right) \in \left(I \times G \times (\Lambda \setminus Y)\right) \subseteq \langle A \rangle, \quad \text{and} \quad \left(j, \, p_{\gamma,j}^{-1}h^{-1}, \, \lambda\right) \in \left((I \setminus X) \times G \times \Lambda\right) \subseteq \langle A \rangle,$

it follows that

$$(i, g, \lambda) = (i, gp_{\mu,i}^{-1}, \mu)(k, h, \gamma)(j, p_{\gamma,j}^{-1}h^{-1}, \lambda) \in \langle A \rangle$$

Therefore, A also contains a generating set for $(X \times G \times Y) \cup \{0\}$. In particular, $T = \langle A \rangle$.

In conclusion, to specify a generating set for T, it suffices to find a generating set for the two corresponding Rees 0-matrix semigroups described above, along with any element of $X \times G \times Y$.

4.3.4 Maximal subsemigroups of type (R6)

In this section, we develop an algorithm for computing the maximal subsemigroups of an arbitrary finite regular Rees 0-matrix semigroup over a group that have type (R6); these are the maximal subsemigroups that intersect every \mathcal{H} -class of the semigroup non-trivially. In this section, we assume that we are able to compute the maximal subgroups of any finite group.

As shown in Proposition 3.2, every finite regular Rees 0-matrix semigroup over a group can be normalized. See Section 3.2.1 for a discussion about computing normalizations of finite regular Rees 0-matrix semigroups over groups. Thus, without loss of generality, we assume throughout this section that $S = \mathscr{M}^0[G; I, \Lambda; P]$ is a normalized finite regular Rees 0-matrix semigroup over a group G. We fix the following notation, as used in Proposition 3.2. Let $I_1 \cup \Lambda_1, \ldots, I_n \cup \Lambda_n$ be the distinct connected components of the Graham-Houghton graph of S, where $I_k \subseteq I$ and $\Lambda_k \subseteq \Lambda$ for each k, and for each $k \in \{1, \ldots, n\}$, fix $i_k \in I_k$ and $\lambda_k \in \Lambda_k$ such that $p_{\lambda_k, i_k} = 1_G$, and define $G_k = \{g \in G : (i_k, g, \lambda_k) \in F(S)\}$.

By Proposition 4.11, a maximal subsemigroup of a finite regular Rees 0-matrix semigroup $\mathscr{M}^0[G; I, \Lambda; P]$ over a group G that has type (R6) is isomorphic to some regular Rees 0-matrix semigroup $\mathscr{M}^0[H; I, \Lambda; Q]$, where H is a maximal subgroup of G. More specifically, in [62, Theorems 3–4], Graham proves that if V is a maximal subgroup of G and there exist elements $g_1, \ldots, g_n \in G$ such that $G_k \leq g_k^{-1} V g_k$ for all k, then there exists a maximal subsemigroup of S that is isomorphic to $\mathscr{M}^0[V; I, \Lambda; Q]$, where Q is a matrix over V^0 that can be constructed from P. In this section, we develop these ideas in a way that allows us to describe and count, in an explicit and systematic way, the specific maximal subsemigroups of S that can be obtained from each maximal subgroup of G. We extend these techniques in Section 4.3.6, where we present an algorithm for computing the maximal subsemigroups of an arbitrary finite regular Rees 0-matrix semigroup over a group that have type (R6) and contain a given set.

In order to describe the Green's relations on a maximal subsemigroup of S of type (R6), we require the following lemma, which shows that any such subsemigroup is regular.

Lemma 4.18. Let T be a subsemigroup of S that intersects every \mathcal{H} -class of S non-trivially. Then T is regular.

Proof. Certainly 0 is a regular element, so let $x = (i, g, \lambda) \in T \setminus \{0\}$ be an arbitrary non-zero element of T. Since P contains a non-zero entry in every row and column, there exist indices $j \in I$ and $\mu \in \Lambda$ such that $p_{\lambda,j} \neq 0 \neq p_{\mu,i}$. Let $e = (j, p_{\lambda,j}^{-1}, \lambda) \in E(S)$. Then $e \in T$ by Lemma 1.14. Certainly xe = x. Furthermore, if $y \in T$ is an arbitrary element of the \mathscr{H}^S -class $\{j\} \times G \times \{\mu\}$, then $yx \in \{j\} \times G \times \{\lambda\}$. Therefore, $(yx)^k = e$ for some $k \in \mathbb{N}$, by the finiteness of S. In particular, $x\mathscr{L}^T e$. It follows that T is regular.

Let T be a subsemigroup of S. If T is a regular subsemigroup of S, then Green's \mathcal{L} -, \mathcal{R} -, and \mathcal{H} -relations on T are inherited from S [76, Proposition 2.4.2]. In particular, if T intersects every \mathcal{H} -class of S non-trivially, then by Lemmas 3.15 and 4.18, the following hold for all non-zero elements $x = (i, g, \lambda)$ and $y = (j, h, \mu)$ in T:

- (i) $x \mathscr{L}^T y$ if and only if $\lambda = \mu$;
- (ii) $x \mathscr{R}^T y$ if and only if i = j; and
- (iii) $x \mathscr{H}^T y$ if and only if i = j and $\lambda = \mu$.

In particular, this characterises the Green's relations on any maximal subsemigroup of type (R6).

Arbitrary subsemigroups that intersect every \mathscr{H} -class non-trivially

In order to be able to easily specify the subsemigroups of S that intersect every \mathscr{H} -class of S non-trivially, we introduce the following notation. For any subgroup $V \leq G$ and any elements $g_1, g_2, \ldots, g_n \in G$, we define the following subset of S:

$$SUB(V, g_1, g_2, \dots, g_n) = \bigcup_{k,l \in \{1, \dots, n\}} (I_k \times g_k^{-1} V g_l \times \Lambda_l) \cup \{0\}.$$
(4.19)

Note that $\text{SUB}(V, g_1, g_2, \ldots, g_n)$ contains |V| elements from each non-zero \mathscr{H} -class of S, since $|g^{-1}Vh| = |V|$ for all $g, h \in G$; see Corollary 4.24. Furthermore, we may assume without loss of generality that $g_1 = 1_G$; this observation is stated in the following lemma.

Lemma 4.20. $\operatorname{SUB}(V, g_1, g_2, \ldots, g_n) = \operatorname{SUB}(g_1^{-1}Vg_1, 1_G, g_1^{-1}g_2, \ldots, g_1^{-1}g_n).$

In the following proposition, we show that any subsemigroup of S that intersects each \mathscr{H} -class of S non-trivially has the form $SUB(V, 1_G, g_2, \ldots, g_n)$, for some subgroup V of G and elements $g_2, \ldots, g_n \in G$. It follows, therefore, that any maximal subsemigroup of S of type (R6) has such a form. In subsequent results, we classify the containment and equality of such semigroups, and we provide generating sets for them.

Proposition 4.21. Let T be a subset of S that intersects each \mathscr{H} -class of S non-trivially. Then T is a subsemigroup of S if and only if there exists a subgroup V of G containing G_1 , and for each $k \in \{2, \ldots, n\}$ there exists $g_k \in G$ such that $G_k \leq g_k^{-1}Vg_k$, and $T = \text{SUB}(V, 1_G, g_2, \ldots, g_n)$.

Proof. (\Rightarrow) For each $j \in I$ and $\mu \in \Lambda$, define $H_{j,\mu}$ to be the set $T \cap (\{j\} \times G \times \{\mu\})$, which is non-empty because $\{j\} \times G \times \{\mu\}$ is an \mathscr{H}^S -class of S. Since T is regular by Lemma 4.18, it follows that $H_{j,\mu}$ is an \mathscr{H}^T -class of T. Since $p_{\lambda_k,i_k} = 1_G$ for all $k \in \{1, \ldots, n\}$, it follows by Green's Lemma (Lemma 1.11) that

$$H_{j,\mu} = H_{j,\lambda_k} (i_k, h, \mu) = (j, h', \lambda_k) H_{i_k,\mu}, \quad \text{and so} \quad H_{j,\lambda_k} H_{i_k,\mu} = H_{j,\mu}$$
(4.22)

for all $j \in I$, $\mu \in \Lambda$, and for all $h, h' \in G$ such that $(i_k, h, \mu), (j, h', \lambda_k) \in T$. These equations are used repeatedly throughout this proof.

We first define the subgroup V of G and the elements g_2, \ldots, g_n required by the proposition. Since $p_{\lambda_1,i_1} = 1_G$, the \mathscr{H}^T -class H_{λ_1,i_1} is a group, and the function $\phi : H_{\lambda_1,i_1} \longrightarrow G$ given by $(i_1, g, \lambda_1) \mapsto g$ is an embedding. In particular, $H_{\lambda_1,i_1} = \{i_1\} \times V \times \{\lambda_1\}$, where $V = (H_{\lambda_1,i_1})\phi$ is a subgroup of G. Additionally, since T intersects each \mathscr{H}^S -class non-trivially, for each $k \in \{2, \ldots, n\}$ there exists some $g_k \in G$ such that $(i_1, g_k, \lambda_k) \in H_{i_1,\lambda_k}$. For convenience, we define $g_1 = 1_G$.

We next show that $G_k \leq g_k^{-1} V g_k$ for all k. Let $k \in \{1, \ldots, n\}$ be arbitrary. By (4.22),

$$H_{i_1,\lambda_k} = H_{i_1,\lambda_1}\left(i_1, g_k, \lambda_k\right) = \{i_1\} \times Vg_k \times \{\lambda_k\},$$

and similarly, $H_{i_k,\lambda_1} = \{i_k\} \times U_k \times \{\lambda_1\}$ for some non-empty subset $U_k \subseteq G$. Again using (4.22), it follows that

$$H_{i_1,\lambda_1} = (i_1, g_k, \lambda_k) H_{i_k,\lambda_1} = \{i_1\} \times g_k U_k \times \{\lambda_1\},\$$

but we have already shown that $H_{i_1,\lambda_1} = \{i_1\} \times V \times \{\lambda_1\}$. Therefore $U_k = g_k^{-1}V$, and by (4.22),

$$H_{i_k,\lambda_k} = H_{i_k,\lambda_1} H_{i_1,\lambda_k} = \{i_k\} \times g_k^{-1} V g_k \times \{\lambda_k\}.$$
(4.23)

By Lemma 1.14, T contains E(S), and so $F(S) = \langle E(S) \rangle \leq T$. It follows that

$$G_{k} = \left\{ g \in G : (i_{k}, g, \lambda_{k}) \in F(S) \right\}$$
 by definition; see Proposition 3.2(ii)
$$\leq \left\{ g \in G : (i_{k}, g, \lambda_{k}) \in T \right\}$$

$$= g_{k}^{-1} V g_{k},$$
 by (4.23)

as required.

Finally, since $0 \in T$, to show that $T = \text{SUB}(V, 1_G, g_2, \ldots, g_n)$, it suffices to show that if $j \in I_k$ and $\mu \in \Lambda_l$ for some $k, l \in \{1, \ldots, n\}$, then $H_{j,\mu} = \{j\} \times g_k^{-1} V g_l \times \{\mu\}$.

Let $k \in \{1, \ldots, n\}$ and $j \in I_k$ be arbitrary. Since j and λ_k are contained in the same connected component of the Graham-Houghton graph of S, there exists a path in this graph from j to λ_k . Specifically, there exists an alternating sequence $(j = a_1, b_1, \ldots, a_m, b_m = \lambda_k)$ of indices from I_k and Λ_k , respectively, such that $p_{b_l,a_l} \neq 0$ and $p_{b_l,a_{l+1}} \neq 0$ for all possible l. Therefore $x = \prod_{l=1}^m (a_l, p_{b_l,a_l}^{-1}, b_l) \in F(S) \leq T$. Indeed by Proposition 3.2(ii), $p_{b_l,a_l} \in G_k$ and $p_{b_l,a_{l+1}} \in G_k$ for all possible l, and so $x = (j, h, \lambda_k)$ for some $h \in G_k \leq g_k^{-1}Vg_k$. By (4.22),

$$H_{j,\lambda_k} = x \cdot H_{i_k,\lambda_k} = \{j\} \times h \cdot (g_k^{-1} V g_k) \times \{\lambda_k\}$$
$$= \{j\} \times g_k^{-1} V g_k \times \{\lambda_k\}.$$

It follows similarly that $H_{i_k,\mu} = \{i_k\} \times g_k^{-1} V g_k \times \{\mu\}$ for all $\mu \in \Lambda_k$.

Let $j \in I$ and $\mu \in \Lambda$ be arbitrary, and let $k, l \in \{1, \ldots, n\}$ be the unique indices such that $j \in I_k$ and $\mu \in \Lambda_l$. Then by applying the previous results with (4.22), we deduce that

$$\begin{split} H_{j,\mu} &= H_{j,\lambda_k} H_{i_k,\lambda_1} H_{i_1,\lambda_l} H_{i_l,\mu} \\ &= (\{j\} \times g_k^{-1} V g_k \times \{\lambda_k\}) (\{i_k\} \times g_k^{-1} V \times \{\lambda_l\}) (\{i_1\} \times V g_l \times \{\lambda_l\}) (\{i_l\} \times g_l^{-1} V g_l \times \{\mu\}) \\ &= \{j\} \times (g_k^{-1} V g_k) \cdot (g_k^{-1} V) \cdot (V g_l) \cdot (g_l^{-1} V g_l) \times \{\mu\} \\ &= \{j\} \times g_k^{-1} V g_l \times \{\mu\}. \end{split}$$

 (\Leftarrow) Let $x = (a, f, \mu)$ and $y = (b, h, \gamma)$ be arbitrary non-zero elements of T. If $p_{\mu,b} = 0$, then certainly $xy = 0 \in T$. Otherwise, b and μ are contained in the same connected component of the Graham-Houghton graph of S, and so $b \in I_k$ and $\mu \in \Lambda_k$ for some $k \in \{1, \ldots, n\}$. Furthermore,

since $T = \text{SUB}(V, g_1 = 1_G, g_2, \ldots, g_n)$ and $x, y \in T$, there exist $l, m \in \{1, \ldots, n\}$ such that $a \in I_l, f \in g_l^{-1}Vg_k, h \in g_k^{-1}Vg_m$, and $\gamma \in \Lambda_m$. By Proposition 3.2(ii) $p_{\mu,b} \in G_k$, and by assumption, $G_k \leq g_k^{-1}Vg_k$. Therefore,

$$f \cdot p_{\mu,b} \cdot h \in (g_l^{-1}Vg_k) \cdot G_k \cdot (g_k^{-1}Vg_m)$$
$$\subseteq (g_l^{-1}Vg_k) \cdot (g_k^{-1}Vg_k) \cdot (g_k^{-1}Vg_m)$$
$$= g_l^{-1}V^3g_m$$
$$= g_l^{-1}Vg_m.$$

It follows that

$$\begin{aligned} xy &= (a, fp_{\mu,b}h, \gamma) \\ &\in \{a\} \times g_l^{-1} V g_m \times \{\gamma\} \\ &\subseteq I_l \times g_l^{-1} V g_m \times \Lambda_m \\ &\subseteq \text{SUB}(V, 1_G, g_2, \dots, g_n) = T \end{aligned}$$

as required, and so T is a subsemigroup of S.

By Proposition 4.21, if T is a subsemigroup of S that intersects every \mathscr{H} -class of S nontrivially, then the set $V = \{g \in G : (i_1, g, \lambda_k) \in T\}$ is a subgroup of V, and T has the form $SUB(V, 1_G, g_2, \ldots, g_n)$ for some elements $g_2, \ldots, g_n \in G$. We call a semigroup of the form $SUB(V, 1_G, g_2, \ldots, g_n)$ a subsemigroup of S arising from V. The order of a subsemigroup of S arising from V is given in the following corollary to to Proposition 4.21, which is required in the proof of Theorem 4.28. The corollary holds since $|g^{-1}Vh| = |V|$ for all $g, h \in G$.

Corollary 4.24. Let T be a subsemigroup of S arising from V. Then every non-zero \mathcal{H} -class of T contains |V| elements. In particular, $|T| = |I| \cdot |V| \cdot |\Lambda| + 1$, and if T contains a non-zero \mathcal{H} -class of S, then T = S.

For the purposes of computation, it is useful to specify a semigroup by a generating set. Let V be a subgroup of G. By Proposition 4.21, any subsemigroup of S arising from V is equal to $SUB(V, 1_G, g_2, \ldots, g_n)$, for some elements $g_2, \ldots, g_n \in G$. The subset $SUB(V, 1_G, g_2, \ldots, g_n)$ is defined in (4.19) in terms of its elements, but it takes only the group V and the elements g_2, \ldots, g_n to uniquely determine $SUB(V, 1_G, g_2, \ldots, g_n)$. In the following lemma, we define a generating set for $SUB(V, 1_G, g_2, \ldots, g_n)$ in terms of V, g_2, \ldots, g_n , and the idempotents of S.

Lemma 4.25. Let V be a subgroup of G that contains G_1 , and suppose there exist elements $g_2, \ldots, g_n \in G$ such that $G_k \leq g_k^{-1}Vg_k$ for all $k \in \{2, \ldots, n\}$. For each $k \in \{2, \ldots, n\}$, define $x_k = (i_1, g_k, \lambda_k)$ and $y_k = (i_k, g_k^{-1}, \lambda_1)$. Let E be a generating set for the idempotent generated subsemigroup of S, and let Y be a generating set for V. Then the set

 $\{E, \{i_1\} \times Y \times \{\lambda_1\}, x_2, \dots, x_n, y_2, \dots, y_n\}$

generates SUB $(V, 1_G, g_2, \ldots, g_n)$.

Proof. Let X denote the set defined in the lemma, and let $T = \text{SUB}(V, 1_G, g_2, \ldots, g_n)$.

We first show that $X \subseteq T$. By the definition of T in (4.19), it is clear that T contains $\{i_1\} \times Y \times \{\lambda_1\}$ and the elements x_k and y_k for each k. By Proposition 4.21, T is a subsemigroup of S that intersects every \mathscr{H} -class of S non-trivially, and so T contains every idempotent of S by Lemma 1.14. Therefore T contains the idempotent generated subsemigroup of S, which contains E.

Next, we show that $T \subseteq \langle X \rangle$. Certainly $0 \in X$. Define $x_1 = y_1 = (i_1, 1_G, \lambda_1) \in X$ and $g_1 = 1_G$. By Proposition 4.21, T is a subsemigroup of S that intersects every \mathscr{H} -class of S

non-trivially, and the non-zero \mathscr{H} -classes of T are the sets $H_{j,\mu} = \{j\} \times g_k^{-1} V g_l \times \{\mu\}$ for each $k, l \in \{1, \ldots, n\}$ and $j \in I_k$ and $\mu \in \Lambda_l$. Note that $H_{i_1,\lambda_1} = \langle \{i_1\} \times Y \times \{\lambda_1\} \rangle \subseteq \langle X \rangle$.

Let $k, m \in \{1, \ldots, n\}, j \in I_k$, and $\mu \in \Lambda_m$ be arbitrary. It remains to prove that $H_{j,\mu} \in \langle X \rangle$. By (4.22),

$$H_{i_k,\lambda_k} = y_k \cdot H_{i_1,\lambda_1} \cdot x_k \subseteq \langle X \rangle,$$

and similarly $H_{i_m,\lambda_m} \in \langle X \rangle$. Furthermore, the elements $(j, 1_G, \lambda_k)$ and $(i_m, 1_G, \mu)$ are idempotent generated by (3.4), and $F(S) = \langle E \rangle \leq \langle X \rangle$. Therefore, by repeatedly applying (4.22),

$$\begin{aligned} H_{j,\mu} &= H_{j,\lambda_k} H_{i_k,\lambda_1} H_{i_1,\lambda_m} H_{i_m,\mu} \\ &= \big((j, 1_G, \lambda_k) \cdot H_{i_k,\lambda_k} \big) \big(y_k \cdot H_{i_1,\lambda_1} \big) \big(H_{i_1,\lambda_1} \cdot x_m \big) \big(H_{i_m,\lambda_m} \cdot (i_m, 1_G, \mu) \big) \\ &\subseteq \langle X \rangle. \end{aligned}$$

See Section 3.2.2, especially Algorithm 3.12, for information about computing a small generating set for the idempotent generated subsemigroup of a finite Rees 0-matrix semigroup over a group. Such a generating set can be created easily from a spanning forest of the Graham-Houghton graph.

In the following results, we classify the containment, and therefore the equality, of two subsets $SUB(U, 1_G, h_2, \ldots, h_n)$ and $SUB(V, 1_G, g_2, \ldots, g_n)$ of S, in terms of the subgroups U and V, and the elements h_2, \ldots, h_n , and g_2, \ldots, g_n .

Lemma 4.26. Let U and V be subgroups of G, and let $h_2, \ldots, h_n, g_2, \ldots, g_n \in G$ be arbitrary. Then $SUB(U, 1_G, h_2, \ldots, h_n) \subseteq SUB(V, 1_G, g_2, \ldots, g_n)$ if and only if $U \leq V$ and $Uh_k \subseteq Vg_k$ for all $k \in \{2, \ldots, n\}$.

Proof. For convenience, we define $g_1 = h_1 = 1_G$. By Proposition 4.21,

$$\begin{aligned} \operatorname{SUB}(U, 1_G, h_2, \dots, h_n) &\subseteq \operatorname{SUB}(V, 1_G, g_2, \dots, g_n) \\ \Leftrightarrow & (I_k \times h_k^{-1} U h_l \times \Lambda_l) \subseteq (I_k \times g_k^{-1} V g_l \times \Lambda_l) \text{ for all } k, l \in \{1, \dots, n\} \\ \Leftrightarrow & h_k^{-1} U h_l \subseteq g_k^{-1} V g_l \text{ for all } k, l \in \{1, \dots, n\} \\ \Leftrightarrow & U h_l \subseteq V g_l \text{ for all } k \in \{1, \dots, n\} \\ \Leftrightarrow & U \leq V \text{ and } U h_l \subseteq V g_l \text{ for all } k \in \{2, \dots, n\}. \end{aligned}$$

Corollary 4.27. Let U and V be subgroups of G, and let $h_2, \ldots, h_n, g_2, \ldots, g_n \in G$ be arbitrary. Then $SUB(U, 1_G, h_2, \ldots, h_n) = SUB(V, 1_G, g_2, \ldots, g_n)$ if and only if U = V and $Vh_k = Vg_k$ for all $k \in \{2, \ldots, n\}$.

Maximal subsemigroups that intersect every \mathcal{H} -class non-trivially

To find the maximal subsemigroups of $S = \mathcal{M}^0[G; I, \Lambda; P]$ that have type (R6), it suffices to find the maximal subsemigroups of S arising from each subgroup of G. As shown in the following theorem, which is the main result of this section, the maximal subsemigroups of S of type (R6) are precisely those subsemigroups of S that arise from maximal subgroups of G.

Theorem 4.28 (Maximal subsemigroups of type (R6)). Let $S = \mathscr{M}^0[G; I, \Lambda; P]$ be a normalized finite regular Rees 0-matrix semigroup where G is a group, let $n \in \mathbb{N}$ denote the number of connected components of the Graham-Houghton graph of S, and let I_1, \ldots, I_n and $i_1 \in I_1, \ldots, i_n \in I_n, \Lambda_1, \ldots, \Lambda_n$ and $\lambda_1 \in \Lambda_1, \ldots, \lambda_n \in \Lambda_n$, and G_1, \ldots, G_n be defined as in Proposition 3.2. Let T be any subset of S that intersects every \mathscr{H} -class of S non-trivially. Then T is a maximal subsemigroup of S if and only if there exists a maximal subgroup V of G and elements $g_1 = 1_G, g_2, \ldots, g_n \in G$ such that $G_k \leq g_k^{-1} V g_k$ for all k, and

$$T = \bigcup_{k,l \in \{1, \dots, n\}} (I_k \times g_k^{-1} V g_l \times \Lambda_l) \cup \{0\}.$$

Proof. (\Rightarrow) By Proposition 4.21, $T = \text{SUB}(V, 1_G, g_2, \ldots, g_n)$ has the required form, and so it remains to prove that the subgroup V is maximal in G. Let K be any subgroup of G with $V \leq K \leq G$. Then $U = \text{SUB}(K, 1_G, g_2, \ldots, g_n)$ is a subsemigroup of S that contains T by Proposition 4.21 and Lemma 4.26. Since T is maximal, either U = T or U = S. In the first case, Corollary 4.27 implies that K = V. Noting that $S = \text{SUB}(G, 1_G, g_2, \ldots, g_n)$, in the second case Corollary 4.27 implies that K = G. Therefore V is a maximal subgroup of G.

(\Leftarrow) By Proposition 4.21, it remains to prove that the semigroup $T = \text{SUB}(V, 1_G, g_2, \ldots, g_n)$ is maximal in S. Let U be any subsemigroup of S that properly contains T. Then U intersects every \mathscr{H} -class of S non-trivially, and so by Proposition 4.21, $U = \text{SUB}(K, 1_G, h_2, \ldots, h_n)$ for some subgroup K of G, and for some elements $h_2, \ldots, h_n \in G$. By Lemma 4.26, $V \leq K$, but since |T| < |U|, it follows by Corollary 4.24 that |V| < |K|. Since V is a maximal subgroup of G, we deduce that K = G. Corollary 4.24 implies that U = S, and so T is maximal in S. \Box

Given these results, we discuss how to compute the subsemigroups of S, and therefore the maximal subsemigroups of S, that arise from a given subgroup of G. Suppose that V is a subgroup of G that contains G_1 . By Proposition 4.21 and Corollary 4.27, in order to find the subsemigroups of S that arise from V, it suffices to find an arbitrary transversal \mathfrak{T} of the right cosets of V in G, and to find the sets

$$\left\{g \in \mathfrak{T} : G_k \le g^{-1} V g\right\}$$

for all $k \geq 2$. More explicitly, given any transversal \mathfrak{T} of the right cosets of V in G, the subsemigroups of S arising from V are in one-to-one correspondence with the Cartesian product

$$\prod_{k=2}^{n} \{ g \in \mathfrak{T} : G_k \le g^{-1} V g \}.$$
(4.29)

Note that for a normal subgroup V of G, $g^{-1}Vg = V$ for all $g \in G$. Therefore, if V is normal in G, \mathfrak{T} is a transversal of the right cosets of V in G, and $k \geq 2$, then

$$\left\{g \in \mathfrak{T} : G_k \le g^{-1} V g\right\} = \begin{cases} \mathfrak{T} & \text{if } G_k \le V, \\ \varnothing & \text{otherwise.} \end{cases}$$
(4.30)

Thus the sets $\{g \in \mathfrak{T} : G_k \leq g^{-1}Vg\}$ are particularly easy to compute when V is normal in G.

Moreover, if V is an arbitrary subgroup of G (not necessarily containing G_1), and there exists $t \in G$ such that $t^{-1}Vt$ is a conjugate of V containing G_1 , then we may use the set in (4.29) to find the subsemigroups of S that arise from $t^{-1}Vt$. In more detail, as argued above, if \mathfrak{U} is an arbitrary transversal of the right cosets of $t^{-1}Vt$ in G, then the subsemigroups of S that arise from $t^{-1}Vt$ are in one-to-one correspondence with the Cartesian product

$$\prod_{k=2}^{n} \left\{ g \in \mathfrak{U} : G_k \le g^{-1} \left(t^{-1} V t \right) g \right\}$$

However, if \mathfrak{T} is a transversal of the right cosets of V in G, then the set $\{t^{-1}g : g \in \mathfrak{T}\}$ is a transversal of the right cosets of $t^{-1}Vt$ in G. In particular, given any transversal \mathfrak{T} of the right cosets of V in G, the collection of all subsemigroups that arise from $t^{-1}Vt$ is in one-to-one correspondence with the Cartesian product

$$\prod_{k=2}^{n} \left\{ t^{-1}g : g \in \mathfrak{T} \text{ and } G_k \le \left(t^{-1}g\right)^{-1} \left(t^{-1}Vt\right) \left(t^{-1}g\right) \right\} = \prod_{k=2}^{n} \left\{t^{-1}g : g \in \mathfrak{T} \text{ and } G_k \le g^{-1}Vg \right\}.$$

Let V be a representative of any conjugacy class of subgroups of G. By the previous arguments, to find all subsemigroups in S that arise from conjugates of V, it suffices to compute

the set in (4.29), and to find the conjugates of V that contain G_1 . Of course, if V is normal in G, then the only conjugate of V is V itself. In full generality, two conjugates $g^{-1}Vg$ and $t^{-1}Vt$ of V are equal if and only if g and t are representatives of the same right coset of $N_G(V)$ in G, where $N_G(V)$ is the normalizer of G in V. Therefore, to find the conjugates of V that contain G_1 , we require any transversal \mathfrak{U} of the right cosets of $N_G(V)$ in G; given \mathfrak{U} , the set

$$\left\{t \in \mathfrak{U} : G_1 \le t^{-1} V t\right\} \tag{4.31}$$

specifies the required conjugates. In particular, if \mathfrak{G} is a transversal of the conjugacy classes of maximal subgroups of G, then by by finding the transversals \mathfrak{T} and \mathfrak{U} and the sets (4.29) and (4.31) for each representative $V \in \mathfrak{G}$, we find all maximal subsemigroups of S of type (R6). A method for finding the maximal subsemigroups of a finite regular Rees 0-matrix semigroup over a group that have type (R6), using the results presented in this section, is given in Algorithm 4.32.

Algorithm 4.32 Maximal subsemigroups of a finite regular Rees 0-matrix semigroup over a group that have type (R6).

Input: $S = \mathscr{M}^0[G; I, \Lambda; P']$, a finite regular Rees 0-matrix semigroup over a group G. **Output:** Generating sets for the maximal subsemigroups of S that have type (\mathbb{R}^6). 1: find the connected components $I_1 \cup \Lambda_1, \ldots, I_n \cup \Lambda_n$ of the Graham-Houghton graph of S \triangleright Section 3.2.1 and [62, Section 4] 2: compute a normalization $\Psi: S \longrightarrow \mathscr{M}^0[G; I, \Lambda; P]$ 3: for each k, fix $i_k \in I_k$ and $\lambda_k \in \Lambda_k$ such that $p_{\lambda_k, i_k} = 1_G$ \triangleright Proposition 3.2(i) 4: $G_k \leftarrow \langle \{ p_{\lambda,i} : i \in I_k, \lambda \in \Lambda_k, p_{\lambda,i} \neq 0 \} \rangle$ for all k \triangleright Proposition 3.2(ii) 5: $E \leftarrow$ generators for the idempotent generated subsemigroup of S \triangleright Algorithm 3.12 6: $\mathcal{C} \leftarrow$ a transversal of the conjugacy classes of maximal subgroups of G \triangleright [17, 47] 7: $\mathfrak{M} \leftarrow \varnothing$ 8: for $V \in \mathcal{C}$ do $Y \leftarrow$ a generating set for V 9: $\mathfrak{T} \leftarrow$ a transversal of the right cosets of V in G 10: $\mathfrak{U} \leftarrow$ a transversal of the right cosets of $N_G(V)$ in G 11: $T_1 \leftarrow \{t \in \mathfrak{U} : G_1 \le t^{-1} V t\}$ 12: for $k \in \{2, ..., n\}$ do 13: $T_k \leftarrow \{t \in \mathfrak{T} : G_k \le t^{-1}Vt\}$ 14: for $t \in T_1, g_2 \in T_2, ..., g_n \in T_n$ do 15: $X_1 \leftarrow \{i_1\} \times t^{-1}Yt \times \{\lambda_1\}$ 16: $X_{1} \leftarrow \{i_{1}\} \times t \quad \forall i \times \{\lambda_{1}\} \\ X_{2} \leftarrow \{(i_{1}, t^{-1}g_{2}, \lambda_{2}), \ldots, (i_{1}, t^{-1}g_{n}, \lambda_{n}), (i_{2}, g_{2}^{-1}t, \lambda_{1}), \ldots, (i_{n}, g_{n}^{-1}t, \lambda_{1})\} \\ \mathfrak{m} \leftarrow \mathfrak{m} + \{ \mathcal{L} + \mathbf{V}, \mathfrak{U}^{-1} + \mathcal{V}_{n}\mathfrak{U}^{-1} \} \qquad \rhd \text{ Lemma } 4.25$ 17:18: 19: return M

By the foregoing discussion and (4.30), we may count the subsemigroups of S that arise from conjugates of a given subgroup of V, without having to construct the subsemigroups of S themselves; we may also obtain upper bounds on their number.

Lemma 4.33. Let V be an arbitrary subgroup of G, let \mathfrak{T} be an arbitrary transversal of the right cosets of G in V, and let \mathfrak{U} be an arbitrary transversal of the right cosets of the normalizer $N_G(V)$ of V in G. Then the number of subsemigroups of S that arise from conjugates of V is

$$m = \left| \left\{ t \in \mathfrak{U} : G_1 \le t^{-1} V t \right\} \right| \cdot \prod_{k=2}^n \left| \left\{ g \in \mathfrak{T} : G_k \le g^{-1} V g \right\} \right|.$$

In particular, $m \leq [G: N_G(V)] \cdot [G:V]^{n-1}$. If V is normal in G, then $m \in \{0, [G:V]^{n-1}\}$.

We may simplify the bounds from Lemma 4.33 in the case that V is a maximal subgroup of G. Since $N_G(V)$ is subgroup of G containing V, either $N_G(V) = V$ and $[G : N_G(V)] = [G : V]$, or $N_G(V) = G$ and $[G : N_G(V)] = 1$. We use this observation in combination with (4.30) to state Corollary 4.34. The bounds given in this corollary are tight; see Examples 4.35 and 4.36.

Corollary 4.34. Let V be a maximal subgroup of G. If V is normal in G, then there are either 0 or $[G:V]^{n-1}$ maximal subsemigroups of S arising from V. Otherwise, there are at most $[G:V]^n$ maximal subsemigroups of S arising from conjugates of V.

Example 4.35. Let $n \in \mathbb{N}$ be arbitrary, and let $S = B(S_3, n)$ denote the $n \times n$ Brandt semigroup over S_3 ; see Section 3.5.3 for the definition. Let $\mathcal{A}_3 = \langle (1 \ 2 \ 3) \rangle$ denote the alternating group of degree 3, which consists of all even permutations in S_3 , and let $\mathcal{C}_2 = \langle (1 \ 2) \rangle$. Then \mathcal{A}_3 is a maximal subgroup of S_3 that is normal, and \mathcal{C}_2 is a maximal subgroup of S_3 with three conjugates. Since the subgroup G_k of S_3 is trivial for each $k \in \{1, \ldots, n\}$, it follows by Lemma 4.33 that the number of maximal subsemigroups of S arising from conjugates of \mathcal{A}_3 is $[S_3 : \mathcal{A}_3]^{n-1} = 2^{n-1}$, and the number of maximal subsemigroups of S arising from conjugates of \mathcal{C}_2 is $[S_3 : \mathcal{C}_2]^n = 3^n$.

Example 4.36. Let $S = \mathscr{M}^0[\mathcal{S}_2; I, \Lambda; P]$, where \mathcal{S}_2 is the symmetric group of degree 2, I and Λ are any sets with two elements, and P is the $\Lambda \times I$ matrix given by

$$\begin{pmatrix} \mathrm{id}_2 & \mathrm{id}_2 \\ \mathrm{id}_2 & (1\ 2) \end{pmatrix}.$$

Then S is a normalized finite regular Rees 0-matrix semigroup over a group whose Graham-Houghton graph has a single connected component. Therefore the entries of P generate the subgroup G_1 of S_2 ; see Proposition 3.2. But $G_1 = S_2$, and no maximal subgroup of S_2 contains S_2 . Therefore, by Lemma 4.33, no maximal subsemigroups of S have type (R6).

4.3.5 An example

In this section, we construct a specific Rees 0-matrix semigroup, and then demonstrate how the techniques of Sections 4.3.1–4.3.4 can be applied to calculate its maximal subsemigroups.

Let $S = \mathcal{M}^0[\mathcal{S}_5; I, \Lambda; P]$, where \mathcal{S}_5 is the symmetric group of degree 5, $I = \{1', \ldots, 6'\}$, $\Lambda = \{1, \ldots, 5\}$, and P is the $\Lambda \times I$ matrix

1	$(1\ 2)(4\ 5)$	0	0	0	0	0)	١
	0	$(1\ 5\ 4\ 2\ 3)$	$(3 \ 5)$	0	0	0	
	0	$(1\ 4\ 5)(2\ 3)$	$(2\ 5\ 3)$	0	0	0	
	0	0	0	$(1\ 5\ 2\ 3\ 4)$	$(1\ 2\ 5\ 3\ 4)$	$(1\ 2)(4\ 5)$	
	0	0	0	$(1\ 2)(3\ 5\ 4)$	0	$(1\ 3\ 2)(4\ 5)$	

Certainly S_5 is a group, and S is regular since P contains a non-zero entry in each of its rows and columns. A diagram of the Graham-Houghton graph of S is shown in Figure 4.37.

We use the ideas discussed in Section 4.3.1 to find the maximal subsemigroups of S that have types (R1) or (R2). There is no maximal subsemigroup of type (R1), since the group S_5 and the sets I and Λ are not all trivial – indeed, they are all non-trivial. In other words, $\{0\}$ is not a maximal subsemigroup of S. Since the matrix P contains 0, the subset $S \setminus \{0\}$ is not a subsemigroup of S, and so there is no maximal subsemigroup of S of type (R2).

We use Lemma 4.13 from Section 4.3.2 to find the maximal subsemigroups of S that have types (R3) or (R4). As discussed in Figure 4.37, the vertices in the Graham-Houghton graph of S that are adjacent to vertices of degree one are $1, 4 \in \Lambda$ and $1' \in I$. By Lemma 4.13(i),


Figure 4.37: A diagram of the Graham-Houghton graph of S, whose three connected components are $\{1', 1\}$, $\{2', 3', 2, 3\}$, and $\{4', 5', 6', 4, 5\}$. The vertices of degree one in the graph are 1, 1', and 5', which are adjacent to the vertices 1', 1, and 4, respectively. There are twelve maximal independent subsets, including I, Λ , $\{5', 1, 2, 3, 5\}$ and $\{1', 4', 5', 6', 2, 3\}$.

the maximal subsemigroups of S of type (R3) are the sets $S \setminus (I \times S_3 \times \{\lambda\})$ for each $\lambda \in \Lambda \setminus \{1, 4\}$, and by Lemma 4.13(ii), the maximal subsemigroups of S of type (R4) are the sets $S \setminus (\{i\} \times S_3 \times \Lambda)$ for each $i \in I \setminus \{i'\}$. In particular, there are three maximal subsemigroups of S that have type (R3), and five that have type (R4).

The maximal independent subsets of the Graham-Houghton graph of S are the index sets I and Λ , along with the sets:

$$\{1', 4', 5', 6', 2, 3\}, \ \{4', 5', 6', 1, 2, 3\}, \ \{2', 3', 5', 1, 5\}, \ \{5', 1, 2, 3, 5\}, \ \{1', 5', 2, 3, 5\}, \\ \{2', 3', 4', 5', 6', 1\}, \ \ \{1', 2', 3', 5', 5\}, \ \{1', 2', 3', 4, 5\}, \ \{1', 2, 3, 4, 5\}, \ \{2', 3', 1, 4, 5\}.$$

By the discussion in Section 4.3.3, these ten sets are in one-to-one correspondence with the maximal subsemigroups of 6 that have type (R5). If U is one of these sets, then the corresponding maximal subsemigroup of S is $S \setminus ((I \setminus U) \times G \times (\Lambda \setminus U))$. For example, if $U = \{1', 5', 2, 3, 5\}$, then the maximal subsemigroup of S of type (R5) that corresponds to U is $S \setminus (\{2', 3', 4', 6'\} \times G \times \{1, 4\})$. In particular, there are ten maximal subsemigroups of S that have type (R5).

It remains to calculate the maximal subsemigroups of S that have type (R6). For this, we use the techniques described in Section 4.3.4. First, it is necessary to find a normalization of S, which is an isomorphism from S to a normalized Rees 0-matrix semigroup; see Section 3.2.1 for more detail about this step. One normalized Rees 0-matrix semigroup that is isomorphic to S is $T = \mathscr{M}^0[\mathcal{S}_5; I, \Lambda; Q]$, where

$$Q = \begin{pmatrix} \mathrm{id}_5 & 0 & 0 & 0 & 0 & 0 \\ 0 & \mathrm{id}_5 & \mathrm{id}_5 & 0 & 0 & 0 \\ 0 & \mathrm{id}_5 & (1 \ 4 \ 2 \ 3 \ 5) & 0 & 0 & 0 \\ 0 & 0 & 0 & \mathrm{id}_5 & \mathrm{id}_5 & \mathrm{id}_5 \\ 0 & 0 & 0 & \mathrm{id}_5 & 0 & (2 \ 4)(3 \ 5) \end{pmatrix}.$$

The Graham-Houghton graph of T is equal to the Graham-Houghton graph of S, which is shown in Figure 4.37. It is clear by inspection that the three connected components of this graph are the sets $I_1 \cup \Lambda_1 = \{1', 1\}$, $I_2 \cup \Lambda_2 = \{2', 3', 2, 3\}$, and $I_3 \cup \Lambda_3 = \{4', 5', 6', 4, 5\}$. To find the maximal subsemigroups of T that have type (R6), for each $k \in \{1, 2, 3\}$ we require the group G_k corresponding to the k^{th} connected component $I_k \cup \Lambda_k$ of the Graham-Houghton graph of S. Each such group is generated by the non-zero matrix entries of Q corresponding to the relevant connected component. Therefore,

$$G_1 = {id_5}, \quad G_2 = \langle (1 \ 4 \ 2 \ 3 \ 5) \rangle, \text{ and } G_3 = \langle (2 \ 4)(3 \ 5) \rangle.$$

For each maximal subgroup V of S_5 , we must find the maximal subsemigroups of T that arise from V. Up to conjugacy, the maximal subgroups of S_5 are: the alternating group A_5 consisting of all even permutations in S_5 ; the symmetric group S_4 consisting of all permutations in S_5 that fix the point 5; a dihedral group $\langle (2 \ 3), (1 \ 2 \ 3)(4 \ 5) \rangle$ of order 12; and the group $\langle (1 \ 2 \ 3 \ 5 \ 4), (2 \ 3 \ 4 \ 5) \rangle$. For each such maximal subgroup V, we require a transversal \mathfrak{U} of the right cosets of $N_G(V)$ in G, a transversal \mathfrak{T} of the right cosets of V in G, and the sets

 $\{g \in \mathfrak{U} : G_1 \leq g^{-1}Vg\}$ and $\{g \in \mathfrak{T} : G_k \leq g^{-1}Vg\}$, for all $k \in \{2, 3\}$.

Case 1: \mathcal{A}_5 . Note that \mathcal{A}_5 is a normal subgroup of \mathcal{S}_5 that contains the subgroups G_1 , G_2 , and G_3 . In other words, every conjugate of \mathcal{A}_5 contains G_1 , G_2 , and G_3 , and so by Corollary 4.34, there are four maximal subsemigroups of T (and hence S) that arise from \mathcal{A}_5 .

Case 2: S_4 . Since G_2 contains a five-cycle, but there are no five-cycles in S_4 , it follows that $G_2 \not\leq g^{-1}S_4g$ for any $g \in G$. Therefore there are no maximal subsemigroups of T or S arising from conjugates of S_4 .

Case 3: $\langle (2 3), (1 2 3)(4 5) \rangle$. Since the group $\langle (2 3), (1 2 3)(4 5) \rangle$ contains no five-cycles, it follows that there are no maximal subsemigroups arising from its conjugates.

Case 4: $V = \langle (1 \ 2 \ 3 \ 5 \ 4), (2 \ 3 \ 4 \ 5) \rangle$. Since V is not normal in S_5 , the normalizer $N_{S_5}(V)$ is V itself. We choose the following transversal of the right cosets of V in S_5 :

$$\mathfrak{T} = \mathfrak{U} = \{ \mathrm{id}_5, (4\ 5), (3\ 4), (3\ 5), (3\ 4\ 5), (3\ 5\ 4) \}.$$

Certainly, all six conjugates of V contain G_1 , which is the trivial subgroup of S_5 . The only conjugate of V that contains G_2 is $(3 \ 5 \ 4)^{-1}V(3 \ 5 \ 4)$, and the two conjugates of V that contain G_3 are V itself and $(3 \ 5)^{-1}V(3 \ 5)$. Therefore, there are twelve $(= 6 \cdot 1 \cdot 2)$ maximal subsemigroups of T arising from conjugates of V, and twelve corresponding maximal subsemigroups of S.

Overall, there are 34 maximal subsemigroups of S: 3 of type (R3), 5 of type (R4), 10 of type (R5), and 16 of type (R6). Finding and constructing these maximal subsemigroups using the SEMIGROUPS package [101] for GAP [58], on a 2.66 GHz Intel Core i7 processor with 8GB of RAM, takes roughly 10 milliseconds.

4.3.6 Maximal subsemigroups of type (R6) that contain a given set

The culmination of this chapter is Algorithm 4.86, which describes a method for computing the maximal subsemigroups of an arbitrary finite semigroup. On line 20 of this algorithm, we require the maximal subsemigroups of the principal factor of a regular \mathscr{J} -class that contain a given set and intersect every \mathscr{H} -class of the principal factor non-trivially. Therefore, in order to use Algorithm 4.86, we require the ability to compute these kinds of maximal subsemigroups in the principal factor. It is straightforward to construct an isomorphism from the principal factor of a finite regular \mathscr{J} -class to some regular Rees 0-matrix semigroup over a group. Hence it suffices to describe an algorithm for computing the maximal subsemigroups of a finite regular Rees 0-matrix semigroup over a group that have type (R6), and that contain a given set, which is the topic of this section. We present an algorithm to solve this problem in Algorithm 4.44.

Throughout, we continue to use the notation and terminology from Section 4.3.4. More specifically, let $S = \mathscr{M}^0[G; I, \Lambda; P]$ denote a normalized finite regular Rees 0-matrix semigroup over a group G, let $n \in \mathbb{N}$ denote the number of connected components of the Graham-Houghton graph of S, and for each $k \in \{1, \ldots, n\}$, let $I_k, \Lambda_k, i_k, \lambda_k$, and G_k be defined as in Proposition 3.2. For a subgroup V of G and elements $g_1, \ldots, g_n \in G$, we again define the set $SUB(V, g_1, g_2, \ldots, g_n)$ as in (4.19).

Let A be any subset of S. Note that any subsemigroup of S that intersects every \mathscr{H} class of S non-trivially necessary contains 0, which is the unique element of the \mathscr{H}^S -class $\{0\}$. Therefore, we only concern ourselves with the non-zero elements of A.

One way to find the maximal subsemigroups of S of type (R6) that contain A is to use Algorithm 4.32 to construct *every* maximal subsemigroups of S that has type (R6), and then to use standard membership checking algorithms from computational semigroup theory to discard those maximal subsemigroups that do not contain A. However, such an approach is to be avoided when possible. By Corollary 4.34, the upper bound on the number of maximal subsemigroups arising from a given maximal subgroup of G is exponential in n. Therefore, the maximal subsemigroups of type (R6) can easily be far too numerous to construct, even when very few of them contain the set A. A more suitable solution would involve computing the desired maximal subsemigroups directly.

Let $T = \text{SUB}(V, g_1 = 1_G, g_2, \ldots, g_n)$ be any subsemigroup of S that intersects every \mathscr{H} class of S non-trivially; see Proposition 4.21. Observe that the elements of T are given in terms of the subgroup V of G and the group elements $g_2, \ldots, g_n \in G$; see (4.19). This suggests that we may also characterise whether T contains A in terms of these parameters. To that end, let $j, k, l \in \{1, \ldots, n\}$ with k < l, and define the sets

$$U_j = \{g : (i, g, \lambda) \in A \text{ for some } i \in I_j \text{ and } \lambda \in \Lambda_j\}, \text{ and}$$

$$(4.38)$$

$$U_{\{k,l\}} = \left\{g : (i, g, \lambda) \in A \text{ for some } i \in I_k \text{ and } \lambda \in \Lambda_l\right\}$$

$$(4.39)$$

$$\cup \{g^{-1} : (i, g, \lambda) \in A \text{ for some } i \in I_l \text{ and } \lambda \in \Lambda_k \}.$$

These sets are required in the following lemma.

Lemma 4.40. Let T be a subset of S that intersects each \mathscr{H} -class of S non-trivially. Then T is a (maximal) subsemigroup of S containing A if and only if there exists a (maximal) subgroup V of G and elements $g_1 = 1_G, g_2, \ldots, g_n \in G$ such that $T = \text{SUB}(V, 1_G, g_2, \ldots, g_n)$, and:

- (i) for all $k \in \{1, \ldots, n\}$, $(G_k \cup U_k) \subseteq g_k^{-1} V g_k$; and
- (ii) for all $k, l \in \{1, ..., n\}$ with $k < l, U_{\{k, l\}} \subseteq g_k^{-1} V g_l$.

Proof. By Proposition 4.21 and Theorem 4.28, T is a (maximal) subsemigroup of S if and only if there exists a (maximal) subgroup V of G and elements $g_1 = 1_G, g_2, \ldots, g_n \in G$ such that $G_k \leq g_k^{-1}Vg_k$ for all $k \in \{1, \ldots, n\}$ and such that $T = \text{SUB}(V, g_1, g_2, \ldots, g_n)$. It remains to prove that the remaining conditions hold in each case.

 (\Rightarrow) Let $k \in \{1, \ldots, n\}$ and let $g \in U_k$. Then $(i, g, \lambda) \in A \cap (I_k \times G \times \Lambda_k)$ for some $i \in I_k$ and $\lambda \in \Lambda_k$. But T contains A, and so $(i, g, \lambda) \in T \cap (I_k \times G \times \Lambda_k) = I_k \times g_k^{-1} V g_k \times \Lambda_k$. In particular, $g \in g_k^{-1} V g_k$. Since $g \in G$ was arbitrary, it follows that $U_k \subseteq g_k^{-1} V g_k$.

Let $k, l \in \{1, \ldots, n\}$ with k < l, and let $g \in U_{\{k,l\}}$. It suffices to show that $g \in g_k^{-1}Vg_l$. Either $(i, g, \lambda) \in A \cap (I_k \times G \times \Lambda_l)$ or $(i, g^{-1}, \lambda) \in A \cap (I_l \times G \times \Lambda_k)$, for some $i \in I$ and $\lambda \in \Lambda$. In the first case, since $A \subseteq T$, it follows that $g \in g_k^{-1}Vg_l$, and we are done. In the second case, it follows that $g^{-1} \in g_l^{-1}Vg_k$ and, by inversion, that $g \in g_k^{-1}Vg_l$.

 $(\Leftarrow) \text{ Let } x = (i, g, \lambda) \in A, \text{ and fix } k, l \in \{1, \ldots, n\} \text{ such that } i \in I_k \text{ and } \lambda \in \Lambda_l. \text{ It suffices to show that } g \in g_k^{-1} V g_l. \text{ If } k = l, \text{ then } g \in U_k, \text{ and by assumption, } U_k \subseteq g_k^{-1} V g_k. \text{ If } k < l, \text{ then } g \in U_{\{k,l\}}, \text{ and by assumption, } U_{\{k,l\}} \subseteq g_k^{-1} V g_l. \text{ Finally, if } k > l, \text{ then } g^{-1} \in U_{\{k,l\}}, \text{ and by assumption, } U_{\{k,l\}} \subseteq g_l^{-1} V g_k. \text{ Therefore, } g \in (g_l^{-1} V g_k)^{-1} = g_k^{-1} V g_l. \qquad \Box$

Throughout the rest of this section, let V be a subgroup of G, let \mathfrak{U} be a transversal of the right cosets of $N_G(V)$ in G, and let \mathfrak{T} be a transversal of the right cosets of V in G. Suppose there exist elements $g_1, \ldots, g_n \in G$ such that conditions (i) and (ii) of Lemma 4.40 hold. Then by Lemma 4.40, $\operatorname{SUB}(g_1^{-1}Vg_1, 1_G, g_1^{-1}g_2, \ldots, g_1^{-1}g_n)$ is a subsemigroup of S that contains A and arises from $g_1^{-1}Vg_1$. Conversely, suppose that $\operatorname{SUB}(t^{-1}Vt, 1_G, h_2, \ldots, h_n)$ is a subsemigroup of S that contains A and arises from some conjugate of V, where $t, h_2, \ldots, h_n \in G$. Define $g_1 = t, g_2 = t \cdot h_2, \ldots, g_n = t \cdot h_n$. Then conditions (i) and (ii) of Lemma 4.40 hold.

By this observation and by Corollary 4.27, to compute all subsemigroups of S that contain A and arise from conjugates of V, it suffices to test each selection of elements $g_1 \in \mathfrak{U}$ and $g_2, \ldots, g_n \in \mathfrak{T}$ against the conditions in Lemma 4.40. Therefore, an algorithm for computing the maximal subsemigroups of S that contain A and have type (R6) can be created from Algorithm 4.32 by adding a test for these conditions after line 15. However, this approach leaves room for improvement, since it ignores the links between the elements g_1, \ldots, g_n that are implied by Lemma 4.40(ii).

In order to take advantage of this information, we define the graph

$$\Gamma = (\{1, \dots, n\}, \{\{k, l\} : U_{k,l} \neq \emptyset\}).$$
(4.41)

Let $\{D_1, \ldots, D_m\}$ be the set of connected components of Γ , numbered so that $1 \in D_1$. We reformulate Lemma 4.40 in the following corollary, which shows that the search for subsemigroups of S that contain A and arise from conjugates of V can be broken down into smaller independent searches that correspond to the connected components of Γ .

Corollary 4.42. Let $g_1 \in \mathfrak{U}$ and $g_2, \ldots, g_n \in \mathfrak{T}$. Then $\operatorname{SUB}(g_1^{-1}Vg_1, 1_G, g_1^{-1}g_2, \ldots, g_1^{-1}g_n)$ is a subsemigroup of S containing A if and only if for each $i \in \{1, \ldots, m\}$, the following hold:

- (i) for all $k \in D_i$, $(G_k \cup U_k) \subseteq g_k^{-1} V g_k$; and
- (ii) for all $k, l \in D_i$ with $k < l, U_{\{k, l\}} \subseteq g_k^{-1} V g_l$.

Indeed, the connected components of Γ have further utility for our purposes. Let $g_1 \in \mathfrak{U}$ and $g_2, \ldots, g_n \in \mathfrak{T}$, and suppose that $\operatorname{SUB}(g_1^{-1}Vg_1, 1_G, g_1^{-1}g_2, \ldots, g_1^{-1}g_n)$ is a subsemigroup of S that contains A. In the following lemma, we show that the choice of $g_1 \in \mathfrak{U}$ determines the value of g_k , for each $k \in D_1 \setminus \{1\}$, and that for $j \in \{2, \ldots, m\}$, the choice of $g_{\min(D_j)} \in \mathfrak{T}$ determines the value of g_k , for each $k \in D_j \setminus \{\min(D_j)\}$.

Lemma 4.43. Let $g_1 \in \mathfrak{U}$ and $g_2, \ldots, g_n \in \mathfrak{T}$ be chosen arbitrarily, and suppose that the subset $\operatorname{SUB}(g_1^{-1}Vg_1, 1_G, g_1^{-1}g_2, \ldots, g_1^{-1}g_n)$ is a subsemigroup of S containing A. The following hold.

- (i) If $u \in U_{\{k,l\}}$, then $g_l \in \mathfrak{T}$ is the coset representative determined by $g_k u$ (when k < l) or $g_k u^{-1}$ (when k > l).
- (ii) For all $i \in \{1, ..., m\}$ and $k \in D_i$, the value of g_k is determined by $g_{\min(D_i)}$.
- *Proof.* (i). Suppose that k < l. By Lemma 4.40(ii), $U_{\{k,l\}} \subseteq g_k^{-1} V g_l$, and so $g_k u \in V g_l$. Suppose that k > l. Then $U_{\{k,l\}} \subseteq g_l^{-1} V g_k$ by Lemma 4.40(ii), and so $g_k u^{-1} \in V g_l$.
- (ii). If $k = \min(D_i)$ there is nothing to prove, so suppose that $k > \min(D_i)$. Since $\min(D_i)$ and k are connected in Γ , we may choose some path $(\min(D_i) = a_1, a_2, \ldots, a_q = k)$ in Γ from $\min(D_i)$ to k. By (i), $g_{a_2} \in \mathfrak{T}$ is the coset representative determined by $g_{\min(D_i)}u_2$, where $u_2 \in U_{\{\min(D_i), a_2\}}$ is arbitrary. By (i), $g_{a_3} \in \mathfrak{T}$ is the coset representative determined by $g_{a_2}u_3$, where $u_3 \in U_{\{a_2, a_3\}}$ (if $a_2 < a_3$) or $u_3^{-1} \in U_{\{a_2, a_3\}}$ (if $a_2 > a_3$) is arbitrary. But g_{a_2} is determined by $g_{\min(D_i)}u_2$, and so g_{a_3} is determined by $g_{\min(D_i)}u_2u_3$. Continuing in this way, we may choose a sequence of elements u_2, \ldots, u_q , where either u_j or u_j^{-1} is contained in $U_{\{a_{j-1}, a_j\}}$ for each j, such that $g_k \in \mathfrak{T}$ is the unique coset representative of V in G determined by $g_{\min(D_i)} \cdot u_2 \cdots u_q$.

A method for finding the maximal subsemigroups of a normalized finite regular Rees 0matrix semigroup over a group that have type (R6) and that contain a certain set is given in Algorithm 4.32. This algorithm relies on the preceding results, especially Corollary 4.42 and

 \triangleright The graph from (4.41)

Algorithm 4.44 Maximal subsemigroups of type (R_6) that contain a given set.

Input: $S = \mathscr{M}^0[G; I, \Lambda; P]$, a normalized finite regular Rees 0-matrix semigroup over a group, and a subset A of S.

Output: \mathfrak{M} , the maximal subsemigroups of S that have type (R6) and contain A.

1: $n \leftarrow$ the number of connected components of the Graham-Houghton graph of S

2: for each $k \in \{1, \ldots, n\}$, compute G_k as in Algorithm 4.32.

3: for each $j, k, l \in \{1, ..., n\}$ with k < l, construct U_j and $U_{\{k, l\}}$ \triangleright (4.38) and (4.39)

4: $\mathcal{C} \leftarrow$ a transversal of the conjugacy classes of maximal subgroups of $G \qquad \qquad \triangleright [17, 47]$

5:
$$\Gamma \leftarrow (\{1, \ldots, n\}, \{\{k, l\} : U_{\{k, l\}} \neq \emptyset\})$$

6: find the connected components D_1, \ldots, D_m of Γ , with $1 \in D_1$

```
7: \mathfrak{M} \leftarrow \emptyset
8: for V \in \mathcal{C} do
```

8: for $V \in \mathcal{C}$ do $T_i \leftarrow \emptyset$ for all $i \in \{1, \ldots, m\}$ 9: 10: $\mathfrak{T} \leftarrow$ a transversal of the right cosets of V in G $\mathfrak{U} \leftarrow a \text{ transversal of the right cosets of } N_G(V) \text{ in } G$ 11:for $(i, g) \in (\{1\} \times \mathfrak{U}) \cup (\{2, \ldots, m\} \times \mathfrak{T})$ do 12:13: $g_{\min(D_i)} \leftarrow g$ for each $k \in D_i \setminus {\min(D_i)}$, fix $g_k \in \mathfrak{T}$ as determined by $g_{\min(D_i)} \triangleright$ Lemma 4.43 14: $\begin{array}{l} \mathbf{if} \ (G_j \cup U_j) \subseteq g_j^{-1} V g_j \ \mathbf{and} \ U_{\{k,l\}} \subseteq g_k^{-1} V g_l \ \text{for all} \ j,k,l \in D_i \ \text{with} \ k < l \ \mathbf{then} \\ T_i \leftarrow T_i \cup \left\{ \{(k,g_k) \ : \ k \in D_i\} \} \\ \end{array} \right. \mathrel{\triangleright} T_i \ \text{contains functions} \ D_i - t_i \\ \end{array}$ 15: $\triangleright T_i$ contains functions $D_i \longrightarrow G$ 16:for $\sigma_1 \in T_1, \ldots, \sigma_m \in T_m$ do 17: $\triangleright \sigma$ defines a function $\{1, \ldots, n\} \longrightarrow G$ $\sigma \leftarrow \sigma_1 \cup \cdots \cup \sigma_m$ 18: $\mathfrak{M} \leftarrow \mathfrak{M} \cup \{ \operatorname{SUB}((1\sigma)^{-1}V(1\sigma), 1_G, (1\sigma)^{-1}(2\sigma), \dots, (1\sigma)^{-1}(n\sigma)) \} \triangleright \operatorname{Corollary} 4.42$ 19:20: return M.

Lemma 4.43. In Example 4.45, we demonstrate the application of Algorithm 4.44 to a small Rees 0-matrix semigroup.

Although we do not include the following optimisations in Algorithm 4.44, it is possible to use Lemma 4.43 to further prune the search space of subsemigroups of S that contain A and arise from conjugates of V. These techniques are implemented in the SEMI-GROUPS [101] package. Let $\{k, l\}$ be any edge in Γ with 1 < k < l. By Lemma 4.43(i), if $\operatorname{sub}(g_1^{-1}Vg_1, 1_G, g_1^{-1}g_2, \ldots, g_1^{-1}g_n)$ is a subsemigroup of S that contains A, then for any $u \in U_{\{k,l\}}$, the right cosets $Vg_k u$ and Vg_l coincide. In particular, for all $u, v \in U_{\{k,l\}}$, it follows that $Vg_k u = Vg_k v$, and so $(g_k^{-1}Vg_k)u = (g_k^{-1}Vg_k)v$. In other words, g_k is required to be chosen so that every element of $U_{\{k,l\}}$ defines the same right coset of $g_k^{-1}Vg_k$ in G. Moreover, if V is normal in G, then every element of $U_{\{k,l\}}$ defines the same right coset of V in G. It follows that, if V is a normal subgroup of G and the elements of some subset $U_{\{k,l\}}$ (with k < l) do not define the same right coset of V in G, then there are no subsemigroups of S that contain A and arise from V.

Example 4.45. Let $S = \mathscr{M}^0[\mathcal{S}_4; I, \Lambda; P]$, where \mathcal{S}_4 is the symmetric group of degree 4, $I = \{1', \ldots, 6'\}, \Lambda = \{1, \ldots, 6\}$, and P is the $\Lambda \times I$ matrix

$/\mathrm{id}_4$	id_4	0	0	0	0)
id_4	$(1\ 2)(3\ 4)$	0	0	0	0
0	0	id_4	0	0	0
0	0	0	id_4	id_4	0
0	0	0	id_4	$(1\ 2\ 3\ 4)$	0
$\left(0 \right)$	0	0	0	0	id_4

Let $A = \{0, (3', (1 \ 3 \ 2 \ 4), 3), (1, (1 \ 4)(2 \ 3), 3'), (6', (2 \ 4 \ 3), 4), (6', id_4, 6)\}$. In this example, we demonstrate how to compute the maximal subsemigroups of S that contain A and that have type (R6). The method that we use is essentially that described in Algorithm 4.44.

Notice that S is already normalized. The four connected components of the Graham-Houghton graph of S are $I_1 \cup \Lambda_1 = \{1', 2', 1, 2\}, I_2 \cup \Lambda_2 = \{3', 3'\}, I_3 \cup \Lambda_3 = \{4', 5', 4, 5\}$, and $I_4 \cup \Lambda_4 = \{6', 6\}$. For each $k \in \{1, 2, 3, 4\}$, we require the group G_k generated by the non-zero entries of P corresponding to the k^{th} connected component. These groups are

$$G_1 = \langle (1 \ 2)(3 \ 4) \rangle, \quad G_2 = G_4 = \{ \mathrm{id}_4 \}, \text{ and } G_3 = \langle (1 \ 2 \ 3 \ 4) \rangle$$

We also require the sets of the form U_j and $U_{\{k,l\}}$, defined in (4.38) and (4.39). The only non-empty sets of these kinds are

$$U_2 = \{(1 \ 3 \ 2 \ 4)\}, \quad U_4 = \{id_4\}, \quad U_{\{1, 2\}} = \{(1 \ 4)(2 \ 3)\}, \text{ and } U_{\{3, 4\}} = \{(2 \ 3 \ 4)\}.$$

Therefore, the connected components of the graph Γ from (4.41) are $\{1, 2\}$ and $\{3, 4\}$.

Finding the maximal subsemigroups of S that contain A and have type (R6) is equivalent to finding the maximal subsemigroups of S that contain A and that arise from conjugates of maximal subgroups of S_4 . Up to conjugacy, there are three maximal subgroups of S_4 : the alternating group \mathcal{A}_4 consisting of all even permutations in \mathcal{S}_4 ; the symmetric group \mathcal{S}_3 consisting of all permutations in \mathcal{S}_4 that fix the point 4; and $\mathcal{D}_4 = \langle (1 \ 2 \ 3 \ 4), (1 \ 2)(3 \ 4) \rangle$, a dihedral group of order 8. By using the techniques in Section 4.3.4, it is possible to show that there are no maximal subsemigroups of S that arise from conjugates of \mathcal{A}_4 or \mathcal{S}_3 , and that there are 27 maximal subsemigroups of S that arise from conjugates of \mathcal{D}_4 . Therefore, it remains to find the maximal subsemigroups of S that contain A and arise from \mathcal{D}_4 .

Since \mathcal{D}_4 is not a normal subgroup of \mathcal{S}_4 , we choose

$$\mathfrak{U} = \mathfrak{T} = \{ \mathrm{id}_4, \ (3\ 4), \ (2\ 3) \}$$

to be a transversal of the right cosets of \mathcal{D}_4 in \mathcal{S}_4 . This transversal was computed with GAP [58].

We begin by considering the possible choices for g_1 and g_2 , which correspond to the first connected component of Γ . By Lemma 4.43(ii), the choice of g_2 is determined by g_1 . More specifically, by Lemma 4.43(i), g_2 is the representative in \mathfrak{T} determined by $g_1 \cdot (1 \ 4)(2 \ 3)$. In each case, $g_1 = g_2$. The only conjugate of \mathcal{D}_4 that contains $G_2 \cup U_2 = \langle (1 \ 3 \ 2 \ 4) \rangle$ is $(2 \ 3)^{-1} \mathcal{D}_4(2 \ 3)$, and so the remaining possible choice is $g_1 = g_2 = (2 \ 3)$. Since $(2 \ 3)^{-1} V(2 \ 3)$ contains both $G_1 \cup U_1$ and $U_{\{1,2\}}$, it follows by Corollary 4.42 that $g_1 = g_2 = (2 \ 3)$ is the only valid choice corresponding to the first connected component of Γ .

Finally, we consider the possible choices for g_3 and g_4 . The only conjugate of \mathcal{D}_4 that contains $G_3 \cup U_3 = \langle (1 \ 2 \ 3 \ 4) \rangle$ is \mathcal{D}_4 itself. Therefore id₄ is the only possible choice for g_3 , and by Lemma 4.43(i), this determines that (3 4) is the only possibility for g_4 . Since $(3 \ 4)^{-1}\mathcal{D}_4(3 \ 4)$ contains $G_4 \cup U_4 = \{ id_4 \}$ and $U_{\{3,4\}} \subseteq id_4^{-1}\mathcal{D}_4(3 \ 4)$, it follows that the selection of $g_3 = id_4$ and $g_4 = (3 \ 4)$ is the unique possibility corresponding to the second connected component of Γ .

In conclusion, by Corollary 4.42, $\text{SUB}((2\ 3)^{-1}\mathcal{D}_4(2\ 3), 1_G, 1_G, (2\ 3), (2\ 4\ 3))$ is the unique maximal subsemigroup of S that contains A and has type (R6). A generating set for this maximal subsemigroup can be constructed using Lemma 4.25.

Finding this solution with the SEMIGROUPS package [101] for GAP [58], which uses a version of Algorithm 4.44, takes roughly 10 milliseconds on a 2.66 GHz Intel Core i7 processor with 8GB of RAM. On the other hand, using SEMIGROUPS to find every maximal subsemigroup of S of type (R6), and discarding those that do not contain A, takes roughly 50 milliseconds.

4.4 Arbitrary finite semigroups

In this section, we develop a framework for computing the maximal subsemigroups of an arbitrary finite semigroup, building on the results of Section 4.3. We consider each of the possible kinds of maximal subsemigroups (M1)-(M5) separately in Sections 4.4.1–4.4.5, with the exception of maximal subsemigroups of types (M3) and (M4). Maximal subsemigroups of types (M3) and (M4) are dual, and are considered jointly in Section 4.4.4.

In the framework that we put forth, the calculation of maximal subsemigroups of types (M3) and (M4) requires the prior calculation of the maximal subsemigroups of type (M2), and for a regular \mathscr{J} -class, the calculation of maximal subsemigroups of types (M5) requires the prior calculation of the maximal subsemigroups of types (M1)–(M4).

Throughout Section 4.4, S denotes an arbitrary finite semigroup with a generating set X, J denotes an arbitrary \mathscr{J} -class of S whose principal factor is $J^* = J \cup \{0\}$, and X' denotes the set of generators $x \in X$ such that $J_x > J$ in the partial order of \mathscr{J} -classes on S.

We use the following lemma repeatedly to prove the forthcoming results. This lemma gives necessary and sufficient conditions for a subset T of S containing $S \setminus J$ to be a subsemigroup of S. Condition (i) requires that $T \cap J$ contain the elements of J that are generated by $S \setminus J$; condition (ii) requires that $(T \cap J) \cup \{0\}$ define a subsemigroup of J^* ; and condition (iii) requires that $T \cap J$ be stabilized under left and right multiplication by $S \setminus J$, in some sense.

Lemma 4.46. Let T be a subset of S such that $S \setminus T \subseteq J$. Then T is a subsemigroup of S if and only if

- (i) $\langle X' \rangle \subseteq T;$
- (ii) if $x, y \in T \cap J$, then $xy \in J$ implies that $xy \in T$; and
- (iii) if $x \in T \cap J$ and $y \in \langle X' \rangle$, then $xy \in J$ implies that $xy \in T$, and $yx \in J$ implies that $yx \in T$.

Proof. (\Rightarrow) The conditions hold since T is a subsemigroup of S that contains X'.

(\Leftarrow) Let $x, y \in T$ be arbitrary. Since T contains $S \setminus J$, it suffices to show that $xy \in T$ whenever $xy \in J$, so suppose that $xy \in J$. We first prove that $x, y \in J \cup \langle X' \rangle$. Since Xgenerates $S, x = x_1 \cdots x_n$, for some $n \in \mathbb{N}$ and $x_i \in X$. Note that $x_i \in J \cup X'$ for each i by Lemma 1.12. If $x_i \in X'$ for each i, then $x \in \langle X' \rangle$. Otherwise, $x_i \in J$ for some i; by Lemma 1.12, $J = J_{xy} \leq J_x \leq J_{x_i} = J$, and $x \in J$. Similarly, $y \in J \cup \langle X' \rangle$. If $x, y \in \langle X' \rangle$, then $xy \in \langle X' \rangle \subseteq T$ by (i). If $x, y \in J$, then $xy \in T$ by (ii). For the remaining cases, $xy \in T$ by (iii).

4.4.1 Maximal subsemigroups that intersect every \mathcal{H} -class: (M1)

In this section, we consider those maximal subsemigroups of the finite semigroup S that arise from the exclusion of elements in a regular \mathscr{J} -class, and that intersect every \mathscr{H} -class of S nontrivially. In other words, we are considering maximal subsemigroups of type (M1). Throughout this section, we suppose that J is a regular \mathscr{J} -class of S.

By Proposition 4.10(b), if M is a maximal subsemigroup of S of type (M1) that arises from J, then $(M \cap J) \cup \{0\}$ is a maximal subsemigroup of the principal factor J^* . Recall that the principal factor J^* of a regular \mathscr{J} -class is a 0-simple semigroup; let $\mathscr{M}^0[G; I, \Lambda; P]$ be a regular Rees 0matrix semigroup over a group that is isomorphic to J^* . It follows that a maximal subsemigroup of S that arises from J and has type (M1) gives a maximal subsemigroup of $\mathscr{M}^0[G; I, \Lambda; P]$ that has type (R6). The converse does not necessarily hold in general; in this section, we develop necessary and sufficient criteria to characterise the circumstances in which it does hold. In particular, in Proposition 4.48 we characterise the maximal subsemigroups of type (M1) in terms of the maximal subsemigroups of J^* that have type (R6) and that contain a particular set. Computing such maximal subsemigroups was the topic of Section 4.3.6 and Algorithm 4.44. We use this algorithm in Algorithm 4.86 when computing maximal subsemigroups of type (M1)

We first describe the subsemigroups of S that contain $S \setminus J$ and intersect every \mathcal{H} -class of J non-trivially, before considering those that are maximal.

Lemma 4.47. Let T be a subset of S containing $S \setminus J$, and suppose that T intersects every \mathscr{H} -class of S non-trivially. Let E be a set consisting of one idempotent from each \mathscr{L} -class of J. Then T is a subsemigroup of S if and only if

- (i) $EX' \cap J \subseteq T$;
- (ii) if $x, y \in T \cap J$, then $xy \in J$ implies $xy \in T$ (i.e. Lemma 4.46(ii) holds).

Proof. (\Rightarrow) Since T is a subsemigroup, Lemma 4.46(ii) holds. Since T is finite and intersects every \mathscr{H} -class of S non-trivially, T contains every idempotent of S by Lemma 1.14. By definition, T contains $S \setminus J$, which contains X', and so $EX' \subseteq T$. In particular, $EX' \cap J \subseteq T$.

(\Leftarrow) It suffices to show that the remaining conditions of Lemma 4.46 hold. In order to do this, we first show that the intersection $E\langle X'\rangle \cap J$ of the set product $E\langle X'\rangle$ with J is contained in T. Let $x \in E\langle X'\rangle \cap J$. By definition, there exist an idempotent $e_1 \in E$ and a sequence of generators $x_1, \ldots, x_n \in X'$ such that $x = e_1x_1\cdots x_n$. Since e_1 and the product $e_1x_1\cdots x_n$ are both members of J, it follows by Lemma 1.12 that the intermediate product $e_1x_1\cdots x_k$ is a member of J for every $k \in \{1, \ldots, n\}$. Hence, by definition of E, for each k < n there exists an idempotent $e_{k+1} \in E$ such that $e_{k+1} \mathscr{L} e_1x_1\cdots x_k$. In particular, $(e_1x_1\cdots x_k)e_{k+1} = e_1x_1\cdots x_k$ for each k < n, since an idempotent is a right identity for its \mathscr{L} -class. Therefore $x = \prod_{k=1}^n e_k x_k$. Furthermore, for each $k \in \{1, \ldots, n\}$ the element $e_k x_k$ is contained in J since

$$J = J_{e_k} \ge J_{e_k x_k} \ge J_x = J.$$

Therefore $e_k x_k \in EX' \cap J \subseteq T$ By repeated application of condition (ii), it follows that $x = \prod_{k=1}^n e_k x_k \in T \cap J$. Since $x \in E\langle X' \rangle \cap J$ was arbitrary, it follows that $E\langle X' \rangle \subseteq T$.

Note that condition (ii) is equivalent to the statement that $(T \cap J) \cup \{0\}$ is a subsemigroup of the principal factor J^* . Since $(T \cap J) \cup \{0\}$ intersects every \mathscr{H} -class of J^* and J^* is finite, it follows by Lemma 1.14 that T contains every idempotent of J.

To prove that condition (i) of Lemma 4.46 holds, let $x \in \langle X' \rangle \cap J$. There exists an idempotent $f \in T \cap J$ in the \mathscr{R}^S -class of x, and so fx = x. By definition of E, there exists an idempotent $e \in E$ such that $e \mathscr{L} f$, and so x = fx = (fe)x = f(ex). Certainly $ex \in J$ by Lemma 1.12, and since $E\langle X' \rangle \cap J \subseteq T$, it follows that $ex \in T \cap J$. By assumption, $(T \cap J) \cup \{0\}$ is a subsemigroup of J^* , and so $x = f(ex) \in T$.

To prove that condition (iii) of Lemma 4.46 holds, let $x \in T \cap J$ and $y \in \langle X' \rangle$. First suppose that $xy \in J$. By assumption, there exists an idempotent $e \in E$ such that x = xe. Since $xy = x(ey) \in J$ it follows that

$$J = J_e \ge J_{ey} \ge J_{x(ey)} = J_{xy} = J$$

by Lemma 1.12, and so $ey \in J$. Furthermore, $ey \in E\langle X' \rangle \cap J \subseteq T$. Since $x, ey \in T \cap J$, $xy \in J$, and $(T \cap J) \cup \{0\}$ is a subsemigroup of J^* , it follows that $xy = x(ey) \in T \cap J$. Finally suppose that $yx \in J$. Since J is a regular \mathscr{J} -class, there exists an idempotent $f \in T \cap J$ such that f(yx) = yx. By definition of E, there exists $e \in E$ such that fe = f, and so yx = f(yx) = (fe)yx = f(ey)x. Note that $ey \in J$ since

$$J = J_e \ge J_{ey} \ge J_{feyx} = J_{yx} = J_{yx}$$

and $ey \in T$ since $E\langle X' \rangle \cap J \subseteq T$. Finally, $f, ey, x \in T \cap J$ and $(T \cap J) \cup \{0\}$ is a subsemigroup of J^* , and so $yx = f(ey)x \in T$. \Box

Given a description of the subsemigroups of S that contain $S \setminus J$ and intersect every \mathcal{H} -class of J non-trivially, it is straightforward to classify those subsemigroups that are maximal. We do this in the following proposition, which is the main result of this section.

Proposition 4.48 (Maximal subsemigroups of type (M1)). Let S be a finite semigroup with generating set X, let J be a regular \mathcal{J} -class of S, and let $X' = \{x \in X : J < J_x\}$. Let T be any subset of S that intersects every \mathcal{H} -class of S non-trivially and contains $S \setminus J$, and let E be a set consisting of one idempotent from each \mathcal{L} -class of J. Then T is a maximal subsemigroup of S if and only if $(T \cap J) \cup \{0\}$ is a maximal subsemigroup of J^* containing $EX' \cap J$.

Proof. (\Rightarrow) Let U be a subset of J such that $U \cup \{0\}$ is a maximal subsemigroup of J^* containing $(T \cap J) \cup \{0\}$. Then by Lemma 4.47, the set $M = (S \setminus J) \cup U$ is a proper subsemigroup of S containing T. Since T is maximal, it follows that T = M, and $U = T \cap J$.

 (\Leftarrow) Let M be a maximal subsemigroup of S containing T. By Lemma 4.47, $(M \cap J) \cup \{0\}$ is a proper subsemigroup of J^* containing $(T \cap J) \cup \{0\}$. Since the latter is a maximal subsemigroup of J^* , it follows that $T \cap J = M \cap J$, and hence T = M.

Note that results dual to Lemma 4.47 and Proposition 4.48 hold if we replace EX' by X'F, where F is a set consisting of one idempotent from each \mathscr{R} -class of J.

We may use Proposition 4.48 to describe an algorithm to calculate the maximal subsemigroups of type (M1) arising from J. We first compute the set $EX' \cap J$ and a normalization $\Psi : J^* \longrightarrow \mathscr{M}^0[G; I, \Lambda; P]$ of the principal factor of J to some regular Rees 0-matrix semigroup over a group; see Section 3.2.1. We then search for the maximal subsemigroups of $\mathscr{M}^0[G; I, \Lambda; P]$ that contain $(EX' \cap J)\Psi$ and have type (R6). Finding such maximal subsemigroups was the topic of Section 4.3.6 and Algorithm 4.44. Therefore, Algorithm 4.44 can be used to compute the maximal subsemigroups of S that arise from J and have type (M1).

A generating set for any such maximal subsemigroup is given by a generating set for $\langle S \setminus J \rangle$, along with the preimage under Ψ of a generating set for the corresponding maximal subsemigroup of $\mathscr{M}^0[G; I, \Lambda; P]$ that has type (R6) (minus the element 0, if present). A generating set for $\langle S \setminus J \rangle$ is given by the union of $X \setminus J$ with a generating set for the ideal $\{x \in S : J_x < J\}$. Generating sets for maximal subsemigroups of type (R6) can be constructed by using Lemma 4.25.

In Chapter 5, we describe the maximal subsemigroups of some particular families of finite monoids. In several of these cases, the most difficult step is the calculation of the maximal subsemigroups that have type (M1). We present results tailored to these monoids in Section 4.5.2. However, in many instances in Chapter 5, the following lemma can be used to show that no maximal subsemigroups of type (M1) arise. Given the Green's structure of a semigroup, it is straightforward to check whether a given \mathcal{J} -class is \mathcal{H} -trivial. Therefore, part (a) of the lemma can be also be used more widely when searching for maximal subsemigroups of type (M1), in order to avoid unnecessary computation.

Lemma 4.49. Let S be a finite semigroup, let J be a \mathcal{J} -class of S. If

- (a) each \mathscr{H} -class of J is trivial, or
- (b) $J \subseteq \langle (S \setminus J) \cup E(J) \rangle$,

then there are no maximal subsemigroups of S of type (M1) that arise from J.

Proof. Let T be an arbitrary subsemigroup of S that contains $S \setminus J$ and intersects each \mathscr{H} class of J non-trivially. If T is a maximal subsemigroup, then in particular, T is a proper subsemigroup. Thus, it suffices in each case to prove that T contains J. By definition, Tcontains an element from each \mathscr{H} -class of J. If each such \mathscr{H} -class is trivial, then T contains J, proving part (a). By Lemma 1.14, T contains every idempotent of S, and so T contains E(J). Therefore, if $\langle (S \setminus J) \cup E(J) \rangle$ contains J, then T contains J, proving part (b).

4.4.2 Graphs and digraphs for regular \mathcal{J} -classes

In this section, we introduce two graphs and two digraphs that can be constructed from any regular \mathscr{J} -class in a finite semigroup. Throughout this section, we suppose that J is a regular \mathscr{J} -class of the finite semigroup $S = \langle X \rangle$. Recall that $X' \subseteq X$ consists of those generators $x \in X$ such that $J_x > J$. We use the graphs associated with J to characterise the subsemigroups of S that arise from the exclusion of elements in J, and are unions of \mathscr{H} -classes of S.

In Sections 4.4.3–4.4.5, we build on this characterisation to give necessary and sufficient conditions, in terms of the properties of the associated graphs, for a subset of S containing $S \setminus J$ to be a maximal subsemigroup of type (M2), (M3), (M4), or (M5). This formulation in terms of graphs makes the problem of computing maximal subsemigroups of these types more tractable. In particular, we can take advantage of several well-known and mature algorithms from graph theory, such as those for computing strongly connected components in a digraph (see [55, 121] or [117, Section 4.2]) and finding all maximal cliques in a graph [15, 103, 125].

The following digraphs are central to the results in the forthcoming sections. We define $\Gamma_{\mathscr{L}}(S, J)$ to be digraph formed by removing the loops from the quotient of the digraph

$$(J/\mathscr{L}, \{(L_a, L_b) \in J/\mathscr{L} \times J/\mathscr{L} : L_a x = L_b \text{ for some } x \in X'\})$$

$$(4.50)$$

by its strongly connected components. Note in particular that $\Gamma_{\mathscr{L}}(S, J)$ is an acyclic digraph. We define a colouring π of $\Gamma_{\mathscr{L}}(S, J)$ so that any vertex V containing an \mathscr{L} -class that has non-empty intersection with $\langle X' \rangle$ has $\pi(V) = 1$, and every other vertex U in $\Gamma_{\mathscr{L}}(S, J)$ has $\pi(U) = 0$. The digraph $\Gamma_{\mathscr{R}}(S, J)$ is defined dually.

Open Problem 4.51. If A_L and A_R are arbitrary finite acyclic digraphs, does there exist a finite semigroup S with a regular \mathscr{J} -class J such that $\Gamma_{\mathscr{L}}(S, J) \cong A_L$ and $\Gamma_{\mathscr{R}}(S, J) \cong A_R$?

We require two additional graphs. The first graph, $\Delta(S, J)$, is isomorphic to a quotient of the Graham-Houghton graph of the principal factor of J. We define $\Delta(S, J)$ to have vertex set equal to the disjoint union of the vertices of $\Gamma_{\mathscr{L}}(S, J)$ and $\Gamma_{\mathscr{R}}(S, J)$. In the special case that Jconsists of a single \mathscr{H} -class, we follow the convention that the strongly connected component of \mathscr{L} -classes $\{J\}$ and the strongly connected component of \mathscr{R} -classes $\{J\}$ are distinct, so that $\Delta(S, J)$ contains two vertices. The edges of $\Delta(S, J)$ are defined as follows. If U is a vertex in $\Gamma_{\mathscr{L}}(S, J)$ and V is a vertex in $\Gamma_{\mathscr{R}}(S, J)$, then $\{U, V\}$ is an edge of $\Delta(S, J)$ if and only if the intersection of some \mathscr{L} -class in U with an \mathscr{R} -class in V is a group \mathscr{H} -class.

The second graph, $\Theta(S, J)$, has the same vertex set as $\Delta(S, J)$, and it contains the edge $\{U, V\}$ if and only if there is an element of $\langle X' \rangle$ in the intersection of some \mathscr{L} -class in U with an \mathscr{R} -class in V, or vice versa. Note that the colourings of $\Gamma_{\mathscr{L}}(S, J)$ and $\Gamma_{\mathscr{R}}(S, J)$ can be deduced from the edges of $\Theta(S, J)$. The graphs $\Delta(S, J)$ and $\Theta(S, J)$ are bipartite, since the vertices in $\Gamma_{\mathscr{R}}(S, J)$ and the vertices in $\Gamma_{\mathscr{R}}(S, J)$ form maximal independent subsets whose union is the whole vertex set.

Example 4.52. Define S to be the transformation semigroup of degree 7 generated by the set X, which consists of the following transformations:

$$\begin{aligned} x_1 &= \begin{pmatrix} 1 & 2 & 3 & 4 & 5 & 6 & 7 \\ 1 & 3 & 4 & 1 & 5 & 5 & 5 \end{pmatrix}, \quad x_2 = \begin{pmatrix} 1 & 2 & 3 & 4 & 5 & 6 & 7 \\ 1 & 4 & 1 & 3 & 5 & 5 & 5 \end{pmatrix}, \\ x_3 &= \begin{pmatrix} 1 & 2 & 3 & 4 & 5 & 6 & 7 \\ 3 & 3 & 1 & 2 & 5 & 5 & 5 \end{pmatrix}, \quad x_4 = \begin{pmatrix} 1 & 2 & 3 & 4 & 5 & 6 & 7 \\ 4 & 4 & 2 & 3 & 5 & 5 & 5 \end{pmatrix}, \\ x_5 &= \begin{pmatrix} 1 & 2 & 3 & 4 & 5 & 6 & 7 \\ 1 & 1 & 3 & 4 & 5 & 5 & 6 \end{pmatrix}, \quad x_6 = \begin{pmatrix} 1 & 2 & 3 & 4 & 5 & 6 & 7 \\ 1 & 2 & 2 & 4 & 5 & 6 & 7 \\ 1 & 2 & 2 & 4 & 5 & 6 & 7 \end{pmatrix}, \\ x_7 &= \begin{pmatrix} 1 & 2 & 3 & 4 & 5 & 6 & 7 \\ 1 & 4 & 3 & 4 & 5 & 6 & 7 \end{pmatrix}, \quad x_8 = \begin{pmatrix} 1 & 2 & 3 & 4 & 5 & 6 & 7 \\ 1 & 2 & 4 & 4 & 5 & 6 & 7 \end{pmatrix}. \end{aligned}$$

Let J denote the \mathscr{J} -class J_{x_1} . The following calculations were performed with the GAP [58] package SEMIGROUPS [101]. The \mathscr{J} -class J is regular, and contains the generators x_1, x_2, x_3 , and x_4 . The remaining generators are contained in \mathscr{J} -classes that are greater than J in the \mathscr{J} -class partial order, and so $X' = \{x_5, x_6, x_7, x_8\}$. The sets of \mathscr{L} - and \mathscr{R} -classes of J are

$$J/\mathscr{L} = \{L_{x_1}, L_{x_3}, L_{x_4}, L_{x_1x_6}\} \text{ and } J/\mathscr{R} = \{R_{x_1}, R_{x_2}, R_{x_3}, R_{x_8x_2}, R_{x_6x_2}, R_{x_7x_3}\}.$$

There are four strongly connected components of \mathscr{L} -classes in J, each of which consists of a single \mathscr{L} -class. Hence the digraph $\Gamma_{\mathscr{L}}(S, J)$ has four vertices, one for each strongly connected component. There are also four strongly connected components of \mathscr{R} -classes, two of which are singletons, and two of which are not; these strongly connected components form the vertices of $\Gamma_{\mathscr{R}}(S, J)$. See Figure 4.53 for a description of the edges of $\Gamma_{\mathscr{L}}(S, J)$ and $\Gamma_{\mathscr{R}}(S, J)$.



Figure 4.53: The digraphs $\Gamma_{\mathscr{L}}(S, J)$, left, and $\Gamma_{\mathscr{R}}(S, J)$, right, from Example 4.52.

Since $\Gamma_{\mathscr{L}}(S, J)$ and $\Gamma_{\mathscr{R}}(S, J)$ each have four vertices, it follows that the bipartite graphs $\Delta(S, J)$ and $\Theta(S, J)$ each have eight vertices. These graphs are shown in Figures 4.54 and 4.55. The set of edges of $\Delta(S, J)$ was determined by computation of the idempotents in J. There are four elements in $\langle X' \rangle \cap J$: $x_5^2 \in L_{x_1} \cap R_{x_3}, x_4 x_1 x_6 \in L_{x_1 x_6} \cap R_{x_3}, x_7 x_4 x_1 x_6 \in L_{x_1 x_6} \cap R_{x_7 x_3}$, and $x_7 x_4 x_1 \in L_{x_1} \cap R_{x_7 x_3}$. The \mathscr{L} - and \mathscr{R} -classes of these elements determine the edges in the graph $\Theta(S, J)$, along with the colours of the vertices in $\Gamma_{\mathscr{L}}(S, J)$ and $\Gamma_{\mathscr{R}}(S, J)$. The vertices of $\Gamma_{\mathscr{L}}(S, J)$ with colour 1 are $\{L_{x_1}\}$ and $\{L_{x_1 x_6}\}$, whilst the only vertex of $\Gamma_{\mathscr{R}}(S, J)$ with colour 1 is $\{R_{x_3}, R_{x_7 x_3}\}$.

When the semigroup S and the \mathscr{J} -class J is obvious from the surrounding context, we abbreviate the names of the digraphs $\Gamma_{\mathscr{L}}(S, J)$ and $\Gamma_{\mathscr{R}}(S, J)$, and the graphs $\Delta(S, J)$ and $\Theta(S, J)$, to $\Gamma_{\mathscr{L}}$, $\Gamma_{\mathscr{R}}$, Δ , and Θ , respectively.



Figure 4.55: The graph $\Theta(S, J)$ from Example 4.52.

The digraphs $\Gamma_{\mathscr{L}}$ and $\Gamma_{\mathscr{R}}$ and the graphs Δ and Θ can be constructed by applying graph algorithms to the left and right Cayley digraphs of S, with respect to its generating set X. The time complexity of finding $\Gamma_{\mathscr{L}}$, $\Gamma_{\mathscr{R}}$, Δ , and Θ by using the Cayley digraphs of S is O(|S||X|). As discussed in Section 1.4, this is the same as the time complexity of determining the left and right Cayley digraphs of S themselves using the Froidure-Pin Algorithm [54]. However, in practice, finding $\Gamma_{\mathscr{L}}$, $\Gamma_{\mathscr{R}}$, Δ , and Θ using the Cayley digraphs of S is much quicker than determining the Cayley digraphs themselves. For certain types of semigroups, such as a semigroup generated by a set of transformations, the \mathscr{J} -class itself and the vertices and edges of $\Gamma_{\mathscr{L}}$, $\Gamma_{\mathscr{R}}$, and Δ can be determined without finding the Cayley digraphs of S and without necessarily exhaustively enumerating the semigroup; see [37] and see Section 1.4.3 for further details. In what follows, we assume that $\Gamma_{\mathscr{L}}$, $\Gamma_{\mathscr{R}}$, Δ , and Θ , and the colourings of $\Gamma_{\mathscr{L}}$ and $\Gamma_{\mathscr{R}}$, are known a priori.

Note that the definitions of $\Gamma_{\mathscr{L}}$ and $\Gamma_{\mathscr{R}}$ are given in terms of the subset X' of the generating set X. This suggests that the descriptions of these digraphs vary according to the chosen generating set for S. However, for our purposes, these variations are irrelevant.

By Green's Lemma [Lemma 1.11] and Lemma 1.10, if L_a and L_b are distinct \mathscr{L} -classes in J, then L_b is reachable from L_a in the digraph in (4.50) if and only if $sx \in L_b$ for some $s \in L_a$ and $x \in \langle S \setminus J \rangle$; see Lemma 4.56 for the proof of a closely related statement. In particular, it follows that the strongly connected components of the digraph in (4.50) do not depend on the chosen generating set for S, and so neither does the description of the vertices of $\Gamma_{\mathscr{L}}$. Similarly, the description of the vertices of $\Gamma_{\mathscr{R}}$ does not depend on X, and consequently the same is true of Δ and Θ . The definition of the edges of Δ is certainly independent of the generating set. Since $\langle S \setminus J \rangle \cap J = \langle X' \rangle \cap J$, it follows that the description of the edges of Θ , and the colourings of $\Gamma_{\mathscr{L}}$ and $\Gamma_{\mathscr{R}}$, are independent of X, too.

On the other hand, in general, the descriptions of the edges of $\Gamma_{\mathscr{L}}$ and $\Gamma_{\mathscr{R}}$ depend on X. However, in the forthcoming results, we are only concerned with whether a certain vertex is reachable from another in $\Gamma_{\mathscr{L}}$ or $\Gamma_{\mathscr{R}}$, and in particular, whether certain vertices are sources or sinks. The following lemma shows that this is independent of the chosen generating set. Lemma 4.56. The following hold.

- (i) The vertex containing $L_b \in J/\mathscr{L}$ in $\Gamma_{\mathscr{L}}(S, J)$ is reachable from the vertex containing L_a if and only if $L_a = L_b$, or there exists $s \in L_a$ and $x \in \langle S \setminus J \rangle$ such that $sx \in L_b$.
- (ii) The vertex containing $R_b \in J/\mathscr{R}$ in $\Gamma_{\mathscr{R}}(S, J)$ is reachable from the vertex containing R_a if and only if $R_a = R_b$, or there exists $s \in R_a$ and $x \in \langle S \setminus J \rangle$ such that $xs \in R_b$.

Proof. We prove only (i), since (ii) is dual.

 (\Rightarrow) Suppose that $L_a \neq L_b$. It follows that L_b is reachable from L_a in the digraph (4.50). Therefore, there exists a path $(L_a = L_1, \ldots, L_n = L_b)$ in this digraph from L_a to L_b . By definition of (4.50), there exists a sequence of generators $x_1, \ldots, x_{n-1} \in X'$ such that $L_a x_1 \cdots x_{n-1} = L_b$. In particular, if $s \in L_a$ is arbitrary and $x = x_1 \cdots x_{n-1} \in \langle S \setminus J \rangle$, then $sx \in L_b$.

 (\Leftarrow) If $L_a = L_b$, then there is nothing to prove. Suppose there exists $s \in L_a$ and $x \in \langle S \setminus J \rangle$ such that $sx \in L_b$. Let $x = x_1 \cdots x_n$ be a factorization of x over $S \setminus J$. Note that

$$J = J_{sx} \leq J_x \leq J_{x_i}$$
 for each $i \in \{1, \ldots, n\}$

by Lemma 1.12, and $x_i \notin J$. In particular, $x_i \in X'$ for each *i*. By Lemma 1.10, $s \mathscr{R} sx_1 \cdots x_i$ for each $i \in \{1, \ldots, n\}$, and by Green's Lemma,

$$(L_a, L_a x_1, \cdots, L_a x_1 \cdots x_n = L_b)$$

is a path in (4.50) from L_a to L_b . The result follows.

We may use the digraphs $\Gamma_{\mathscr{L}}$ and $\Gamma_{\mathscr{R}}$ to understand the subsemigroups of S that contain $S \setminus J$ and that are unions of \mathscr{H} -classes of S. By Green's Lemma and Lemma 1.10, if T is a subsemigroup of S that contains $S \setminus J$, then $T \cap J$ contains an \mathscr{L} -class L if and only if $T \cap J$ contains every \mathscr{L} -class in every vertex of $\Gamma_{\mathscr{L}}$ that is reachable from the vertex containing L. The analogous statement holds for \mathscr{R} -classes. Furthermore, $T \cap J$ contains an \mathscr{H} -class H if and only if $T \cap J$ contains every \mathscr{H} -class that is the intersection of an \mathscr{L} -class and an \mathscr{R} -class that are contained in vertices that are reachable from the vertices containing the \mathscr{L} -classes and \mathscr{R} -classes containing H in $\Gamma_{\mathscr{L}}$ and $\Gamma_{\mathscr{R}}$, respectively. This observation is fundamental to the forthcoming results, and is demonstrated in the following example.

Example 4.57. Let S and J be the semigroup and the \mathscr{J} -class, respectively, from Example 4.52. Suppose that T is a subsemigroup of S such that $S \setminus T \subseteq J$. By analysing the digraph $\Gamma_{\mathscr{L}}$, depicted in Figure 4.53, we see that if T contains the \mathscr{L} -class L_{x_4} , then T also contains the \mathscr{L} -classes L_{x_1} and $L_{x_1x_6}$, since the vertices containing these \mathscr{L} -classes are reachable in the digraph from the vertex $\{L_{x_4}\}$. Likewise, if T contains the \mathscr{R} -class $R_{x_8x_2}$, then by considering the digraph $\Gamma_{\mathscr{R}}$, depicted in Figure 4.53, we see that T also contains the \mathscr{R} -classes $R_{x_6x_2}$, R_{x_3} , and $R_{x_7x_3}$. If we consider these digraphs together, then we see that T contains the \mathscr{H} -classes $L_{x_1} \cap R_{x_3}$ if and only if T also contains the \mathscr{H} -classes $L_{x_1} \cap R_{x_7x_3}$, $L_{x_1x_6} \cap R_{x_3}$, and $L_{x_1x_6} \cap R_{x_7x_3}$.

Suppose that T is a subset of S that contains $S \setminus J$, and suppose that there exist proper subsets $A \subsetneq J/\mathscr{L}$ and $B \subsetneq J/\mathscr{R}$ such that $T \cap J$ is the union of the Green's classes in A and in B. In other words, $T \cap J$ is a possibly empty union of \mathscr{L} -classes and/or \mathscr{R} -classes of J. In the following proposition, we give necessary and sufficient conditions for T to be a subsemigroup of S, in terms of the graphs and digraphs from this section.

Proposition 4.58. Let S be a finite semigroup, and let T be a proper subset of S such that $S \setminus T$ is contained in a regular \mathscr{J} -class J of S. Suppose there exist proper subsets $A \subsetneq J/\mathscr{L}$ and $B \subsetneq J/\mathscr{R}$ such that $T \cap J$ is the union of the Green's classes in A and B. Then T is a subsemigroup of S if and only if the following hold:

- (i) A and B are unions of vertices of $\Delta(S, J)$;
- (ii) if $U \subseteq A$ is a vertex of $\Gamma_{\mathscr{L}}(S, J)$ and V is an out-neighbour of U, then $V \subseteq A$;
- (iii) if $U \subseteq B$ is a vertex of $\Gamma_{\mathscr{R}}(S, J)$ and V is an out-neighbour of U, then $V \subseteq B$;
- (iv) if $\{U, V\}$ is an edge in $\Theta(S, J)$, then $U \subseteq A \cup B$ or $V \subseteq A \cup B$;
- (v) the vertices contained in $A \cup B$ form an independent subset of $\Delta(S, J)$.

Proof. (\Rightarrow) As mentioned after Lemma 4.56, if $T \cap J$ contains an \mathscr{L} -class L, then $T \cap J$ contains every \mathscr{L} -class in every vertex of $\Gamma_{\mathscr{L}}$ that is reachable from the vertex containing L. An analogous statement holds for \mathscr{R} -classes and $\Gamma_{\mathscr{R}}$. Parts (i), (ii), and (iii) follow immediately from these observations.

If $\{U, V\}$ is an edge in Θ , then by definition, there exists an element $x \in \langle X' \rangle$ in the intersection of some \mathscr{L} -class L_x in U and some \mathscr{R} -class R_x in V. By Lemma 4.46, since T is a subsemigroup, $x \in T$ and so either $L_x \in A$ and U is contained in A; or $R_x \in B$ and V is contained in B. Therefore part (iv) holds.

If $A = \emptyset$ or $B = \emptyset$, then part (v) holds by the definition of Δ , so suppose otherwise. In order to reach a contradiction, suppose that $A \cup B$ does not form an independent subset of Δ . Therefore, there exists an \mathscr{L} -class $L \in A$ and an \mathscr{R} -class $R \in B$ such that $L \cap R$ is a group. Since A and B are proper subsets of Green's classes of J, we may choose $x \in L$ such that $R_x \notin B$, and we may choose $y \in R$ such that $L_y \notin A$. Note that $x, y \in T$. By Lemma 1.15, $xy \in R_x \cap L_y$. But $L_{xy} = L_y \notin A$ and $R_{xy} = R_x \notin B$, and so $xy \notin T$, contradicting the assumption that T is a subsemigroup. Therefore (v) holds.

 (\Leftarrow) It suffices to show that T satisfies the conditions (i), (ii), and (iii) of Lemma 4.46.

To verify that condition (i) of Lemma 4.46 holds, let $x \in \langle X' \rangle \cap J$. Then there exists an edge in Θ between the vertex U containing L_x and the vertex V containing R_x . By assumption (iv) of the proposition, either $U \subseteq A$ or $V \subseteq B$ (or both). In either case, it follows that $x \in T$.

For condition (ii), suppose that $x, y \in T \cap J$ and $xy \in J$. By Lemma 1.10, $xy \in R_x \cap L_y$, and so the \mathscr{H} -class $L_x \cap R_y$ is a group by Lemma 1.15. By assumption (v), the vertices contained in $A \cup B$ form an independent subset of Δ . Hence either $L_x \notin A$ or $R_y \notin B$. If $R_y \notin B$, then, since $T \cap J$ is a union of \mathscr{L} - and \mathscr{R} -classes of J and $y \in T \cap J$, we conclude that $L_y \in A$. By Lemma 1.10, $L_{xy} = L_y \in A$, and $xy \in T$. If $L_x \notin A$, then the proof is analogous.

To show that the final condition of Lemma 4.46 holds, let $x \in T \cap J$ and $y \in \langle X' \rangle$. Since $x \in T \cap J$, either $L_x \in A$ or $R_x \in B$. Suppose that $xy \in J$. In this case, note that Green's Lemma and Lemma 1.10 imply that $L_{xy} = L_y x$, and Lemma 1.10 implies that $R_{xy} = R_x$. If $L_x \in A$, then the vertex containing L_{xy} is reachable in $\Gamma_{\mathscr{L}}$ from the vertex containing L_x , and so $L_{xy} \in A$ and $xy \in T$. Otherwise, $R_x \in B$, and so $R_{xy} = R_x \in B$ and $xy \in T$. The proof that $yx \in J$ implies $yx \in T$ is similar.

Note that if the subset T from the statement of Proposition 4.58 is a maximal subsemigroup of S, then T has type (M2) when $A \neq \emptyset$ and $B \neq \emptyset$; T has type (M3) when $A \neq \emptyset$ and $B = \emptyset$; T has type (M4) when $A = \emptyset$ and $B \neq \emptyset$; and T has type (M5) when $A = \emptyset$ and $B = \emptyset$. We use Proposition 4.58 in the subsequent sections to obtain descriptions of the maximal subsemigroups of S that have types (M2)–(M5).

4.4.3 Maximal subsemigroups that are unions of \mathscr{L} - and \mathscr{R} -classes: (M2)

In this section, we discuss how to compute maximal subsemigroups that have type (M2), building on the work of Section 4.4.2. Throughout this section, we assume that J is a regular \mathscr{J} -class of the finite semigroup S. The following proposition, which is a slight adaptation of Proposition 4.58, gives necessary and sufficient conditions for the existence of a maximal subsemigroup of type (M2).

Proposition 4.59 (Maximal subsemigroups of type (M2)). Let S be a finite semigroup, and let T be a proper subset of S such that $S \setminus T$ is contained in a regular \mathscr{J} -class J of S. Suppose that there exist proper non-empty subsets $A \subsetneq J/\mathscr{L}$ and $B \subsetneq J/\mathscr{R}$ such that $T \cap J$ is the union of the Green's classes in A and B. Then T is a maximal subsemigroup of S if and only if conditions (i)–(v) of Proposition 4.58 hold, and the independent subset of $\Delta(S, J)$ formed by the vertices contained in $A \cup B$ is maximal in $\Delta(S, J)$.

Proof. (\Rightarrow) Let $\phi : J^* \longrightarrow \mathscr{M}^0[G; I, \Lambda; P]$ be an isomorphism from the principal factor of J to a Rees 0-matrix semigroup over a group. We may assume without loss of generality that $I = J/\mathscr{R}$ and $\Lambda = J/\mathscr{L}$, and that $L\phi = I \times G \times \{L\}$ and $R\phi = \{R\} \times G \times \Lambda$ for any \mathscr{L} -class L in J and for any \mathscr{R} -class R in J. Therefore

$$(M \cap J)\phi = (B \times G \times \Lambda) \cup (I \times G \times A) = (I \times G \times \Lambda) \setminus ((I \setminus B) \times G \times (\Lambda \setminus A))$$

By Proposition 4.10, T is a maximal subsemigroup of type (M2), and $(M \cap J)\phi \cup \{0\}$ is a maximal subsemigroup of $\mathscr{M}^0[G; I, \Lambda; P]$ that has type (R5). By Proposition 4.11, $A \cup B$ is a maximal independent subset of the Graham-Houghton graph of $\mathscr{M}^0[G; I, \Lambda; P]$. Recall that Δ is a quotient of this graph. Therefore, the vertices of Δ contained in $A \cup B$ form a maximal independent subset of Δ , as required. The remaining conditions hold by Proposition 4.58.

 (\Leftarrow) By Proposition 4.58, it remains to prove that the subsemigroup T is maximal in S. Let M be a maximal subsemigroup of S that contains T. By the assumption that $T \cap J$ is a union of non-empty sets of \mathscr{L} - and \mathscr{R} -classes, M has type (M2). Let A' and B' be the sets of \mathscr{L} - and \mathscr{R} -classes of J, respectively, that are contained in M. Note that $A \subseteq A'$ and $B \subseteq B'$ since M contains T, and so $A \cup B \subseteq A' \cup B'$. By the forward implication, the vertices of Δ in $A' \cup B'$ form a maximal independent subset of Δ . But the vertices of Δ in $A \cup B$ form a maximal independent subset of Δ . But the vertices of Δ in $A \cup B$ form a maximal independent subset of Δ by assumption. Therefore A = A' and B = B', and so T = M, and T is maximal in S.

We use Proposition 4.59 to describe an algorithm for computing the maximal subsemigroups of S that have type (M2) and that arise from J.

The first step is to determine the maximal independent subsets of the bipartite graph Δ . As discussed in Section 4.3.3, this is equivalent to finding the maximal cliques in the complement of Δ . The Bron-Kerbosch Algorithm [15], which is implemented by the author in the DIGRAPHS package [10] for GAP [58], is a recursive algorithm for finding the maximal cliques in any finite graph. Roughly speaking, the algorithm begins with a singleton clique, and recursively attempts to extend the given clique to a larger clique by adding another vertex.

We may use Proposition 4.58 to help to guide this recursion. By Proposition 4.58(ii) and (iii), we are only interested in those cliques K with following property: if $U \in K$ and V is an outneighbour of U in $\Gamma_{\mathscr{L}}$ or $\Gamma_{\mathscr{R}}$, then $V \in K$. As such, the search tree can be pruned to exclude any branch starting at a clique containing a vertex U whose out-neighbours in $\Gamma_{\mathscr{L}}$ and $\Gamma_{\mathscr{R}}$ do not extend the given clique, or where we have already discovered every maximal clique containing one of these out-neighbours. By executing the Bron-Kerbosch Algorithm in this modified way, we produce the maximal independent subsets of Δ , each of which corresponds to sets of Green's classes $A \subseteq J/\mathscr{L}$ and $B \subseteq J/\mathscr{R}$ that satisfy Proposition 4.58(i)–(iii) and (v).

The second step is then to check which of these sets A and B satisfy part (iv) of Proposition 4.58, which is routine. Given proper non-empty sets A and B satisfying all the conditions in Proposition 4.58, the final step is to specify a generating set for the corresponding maximal subsemigroup; see Proposition 4.61 for more details about this step.

Example 4.60. Let S be the transformation semigroup of degree 7, and let J be the \mathcal{J} -class of S, defined in Example 4.52. Consider the graph Δ , which is shown in Figure 4.54.

There are seven maximal independent subsets of Δ in total. Two of these correspond to all of the vertices of $\Gamma_{\mathscr{L}}$ and all of the vertices of $\Gamma_{\mathscr{R}}$, respectively. Three further maximal independent subsets correspond to sets A and B of \mathscr{L} -classes and \mathscr{R} -classes that either do not satisfy Proposition 4.58(ii) or do not satisfy Proposition 4.58(iii). For example, the maximal independent subset $Q = \{\{L_{x_1}\}, \{R_{x_1}\}, \{R_{x_2}\}\}$ does not satisfy Proposition 4.58(ii), since $\{L_{x_1x_6}\}$ is an out-neighbour of $\{L_{x_1}\}$ in $\Gamma_{\mathscr{L}}$, but $\{L_{x_1x_6}\} \notin Q$; see Figure 4.53. Furthermore, Q does not satisfy Proposition 4.58(iii), since $\{R_{x_8x_2}, R_{x_6x_2}\}$ is an out-neighbour of $\{R_{x_2}\}$ in $\Gamma_{\mathscr{R}}$ that is not contained in Q.

The remaining two maximal independent subsets of Δ correspond to non-empty sets A and B that satisfy Proposition 4.58(i)–(iii). The first of these is $\{\{L_{x_1}\}, \{L_{x_1x_6}\}, \{R_{x_1}\}\}$, which corresponds to the sets $A_1 = \{L_{x_1}, L_{x_1x_6}\} \subseteq J/\mathscr{L}$ and $B_1 = \{R_{x_1}\} \subseteq J/\mathscr{R}$; the second corresponds to the sets $A_2 = \{L_{x_1x_6}\}$ and $B_2 = \{R_{x_1}, R_{x_3}, R_{x_7x_3}\}$.

It remains to test whether these subsets satisfy Proposition 4.58(iv). There are two edges in the graph Θ , as shown in Figure 4.55:

$$\{\{L_{x_1}\}, \{R_{x_3}, R_{x_7x_3}\}\}, \text{ and } \{\{L_{x_1x_6}\}, \{R_{x_3}, R_{x_7x_3}\}\}.$$

For the first edge, $\{L_{x_1}\} \subseteq A_1$ and $\{R_{x_3}, R_{x_7x_3}\} \subseteq B_2$; for the second edge, $\{L_{x_1x_6}\} \subseteq A_1$ and $\{R_{x_3}, R_{x_7x_3}\} \subseteq B_2$. In other words, the sets A_1 and B_1 , and the sets A_2 and B_2 , satisfy Proposition 4.58(iv). Therefore, there are two maximal subsemigroups of S of type (M2) arising from J: the set consisting of $S \setminus J$ along with the union of the \mathscr{L} -classes in A_1 and the union of the \mathscr{R} -classes in B_1 , and the set consisting of $S \setminus J$ along with the union of the \mathscr{L} -classes in A_2 and the union of the \mathscr{R} -classes in B_2 .

Proposition 4.61. Let T be a proper subsemigroup of a finite semigroup $S = \langle X \rangle$ that arises from a regular \mathscr{J} -class J of S. Suppose that there exist non-empty proper subsets $A \subsetneq J/\mathscr{L}$ and $B \subsetneq J/\mathscr{R}$ such that $T \cap J$ is the union of the Green's classes in A and in B. Then T is generated by any set consisting of:

- (i) $X \setminus J$;
- (ii) a semigroup generating set for the ideal $\{x \in S : J_x < J\}$ of S;
- (iii) for each source U in $\Gamma_{\mathscr{L}}(S, J)$ contained in A, an element t such that $R_t \in B$ and $L_t \in U$;
- (iv) a generating set for a group \mathscr{H} -class H_x , where $L_x \in A$;
- (v) for each source V in the induced subdigraph of $\Gamma_{\mathscr{R}}(S, J)$ on the complement of B, an element $z \in L_x$ such that $R_z \in V$; and
- (vi) for each source U in the induced subdigraph of $\Gamma_{\mathscr{L}}(S, J)$ on A, an element $y \in R_x$ such that $L_y \in U$;
- (vii) a generating set for a group \mathscr{H} -class $H_{x'}$, where $R_{x'} \in B$;
- (viii) for each source U' in the induced subdigraph of $\Gamma_{\mathscr{R}}(S, J)$ on B, an element $y' \in L_{x'}$ such that $R_{y'} \in U'$;
- (ix) for each source V' in the induced subdigraph of $\Gamma_{\mathscr{L}}(S, J)$ on the complement of A, an element $z' \in R_{x'}$ such that $L_{z'} \in V'$.

Proof. Let Y be a set of the kind described in the proposition. Clearly every element of Y is contained in T, and so $\langle Y \rangle$ is a subsemigroup of T. Furthermore, the inclusion of the generators in (i) and (ii) implies that $\langle Y \rangle$ contains $S \setminus J$. To show that $T \leq \langle Y \rangle$, let $a \in T \cap J$. By the definition of T, either $L_a \in A$ or $R_a \in B$.

First suppose that $L_a \in A$ and $R_a \notin B$. The vertex of $\Gamma_{\mathscr{L}}$ containing L_a is reachable from some vertex U that is a source of the induced subdigraph of $\Gamma_{\mathscr{L}}$ on A. Hence there exists an element $y \in Y$ of type (vi) such that $L_y \in U$ and $y \mathscr{R} x$, where x is an element in the group \mathscr{H} -class from part (iv). By Lemma 4.56, either $L_a = L_y$, or there exists $r' \in \langle S \setminus J \rangle \subseteq \langle Y \rangle$ such that $yr' \in L_a$ and, by Lemma 1.10, $yr' \in R_y = R_x$. In either case, there exists an element $r \in \langle Y \rangle$ such that $r \mathscr{R} x$ and $r \mathscr{L} a$. Likewise, by using a generator of type (v), there exists an element $s \in \langle Y \rangle$ such that $s \mathscr{L} x$ and $s \mathscr{R} a$. Since H_x is a group, it follows by Green's Lemma that $a \in H_a = sH_xr \subseteq \langle Y \rangle$. If $L_a \notin A$ and $R_a \in B$, then the proof that $a \in \langle Y \rangle$ is similar.

For the final case, suppose that $L_a \in A$ and $R_a \in B$. The vertex of $\Gamma_{\mathscr{L}}$ containing L_a is reachable from some vertex U that is a source of $\Gamma_{\mathscr{L}}$. If $U \subseteq A$, then by (iii) there exists an element $t \in Y$ such that $R_t \in B$ and $L_t \in U$. If $U \not\subseteq A$, then by the previous paragraph, every element t' such that $R_{t'} \in B$ and $L_{t'} \notin A$ is contained in $\langle Y \rangle$. In either case, there exists an element $t \in \langle Y \rangle$ such that $R_t \in B$ and $L_t \in U$.

If $a \mathscr{L} t$, then $t \in L_a \cap R_t$. Otherwise, by Lemmas 4.56 and 1.10, there exists an element $r' \in \langle S \setminus J \rangle \subseteq \langle Y \rangle$ such that $tr' \in L_a \cap R_t$. In either case, there exists $r \in \langle Y \rangle$ such that $r \in L_a \cap R_t$. By the regularity of J, there exists an idempotent $e \in R_t$, and since $R_e = R_t \in B$ and $A \cup B$ corresponds to an independent subset of Δ , it follows that $L_e \notin A$. By the arguments of the second paragraph, the \mathscr{H} -classes H_e and $H_s = L_e \cap R_a$ are both contained in $\langle Y \rangle$. Since H_e is a group, it follows by Green's Lemma and Lemma 1.15 that $a \in H_a = sH_er \subseteq \langle Y \rangle$.

Note that if $L_a \in A$ and $R_a \in B$, and there exists an idempotent $e \in J$ such that $L_e \notin A$ and $R_e \notin B$, then $a \in H_a = (L_a \cap R_e)(L_e \cap R_a)$. Therefore the generators in (iii) are redundant in this case, which occurs if and only if the complement of $A \cup B$ in $(J/\mathscr{L}) \cup (J/\mathscr{R})$ corresponds to a non-independent subset of Δ .

4.4.4 Maximal subsemigroups that are unions of \mathscr{L} - or \mathscr{R} -classes: (M3)–(M4)

In this section, we build on the results of Sections 4.4.2 and 4.4.3 in order to describe the subsemigroups, and therefore the maximal subsemigroups, that can be obtained by the removal of either \mathscr{L} -classes or \mathscr{R} -classes from a regular \mathscr{J} -class of a finite semigroup. In other words, we describe how to compute maximal subsemigroups of types (M3) and (M4).

Consider Proposition 4.58. If the set of \mathscr{R} -classes B in the statement of this proposition is empty, then the subset T is a union of \mathscr{L} -classes of the semigroup, and the criteria for T to be a subsemigroup can be simplified. In more detail, conditions (iii) and (v) of Proposition 4.58 are immediately satisfied, as is the second part of condition (i). Furthermore, the reference to the set B in condition (iv) can be removed. By performing these simplifications, we obtain the following corollary to Proposition 4.58.

Corollary 4.62. Let S be a finite semigroup with a regular \mathscr{J} -class J, and let T be a subset of S that contains $S \setminus J$. Suppose there exists a proper subset $A \subseteq J/\mathscr{L}$ such that $T \cap J$ is the union of the \mathscr{L} -classes in A. Then T is a subsemigroup of S if and only if A is a union of vertices of $\Gamma_{\mathscr{L}}(S, J)$, including every vertex with colour 1, and Proposition 4.58(ii) holds.

We use this corollary in the proof of the following proposition.

Proposition 4.63 (Maximal subsemigroups of type (M3)). Let S be a finite semigroup with a regular \mathcal{J} -class J, and let T be a subset of S that contains $S \setminus J$. Suppose there exists a proper

non-empty subset $A \subsetneq J/\mathscr{L}$ such that $T \cap J$ is the union of the \mathscr{L} -classes in A. Then T is a maximal subsemigroup of S if and only if the complement of A is a source of $\Gamma_{\mathscr{L}}(S, J)$ with colour 0, and there is no maximal subsemigroup of S of type (M2) whose corresponding subset of J/\mathscr{L} is A.

Proof. (\Rightarrow) By Corollary 4.62, A is a union of vertices of $\Gamma_{\mathscr{L}}$. Since $\Gamma_{\mathscr{L}}$ is finite and acyclic, it follows that any induced subdigraph of $\Gamma_{\mathscr{L}}$ is finite and acyclic. Therefore, there exists a sink in the induced subdigraph of $\Gamma_{\mathscr{L}}$ on the vertices not contained in A. Note that this vertex is not necessarily a sink in $\Gamma_{\mathscr{L}}$ itself. Let A' be the subset of J/\mathscr{L} formed from the union of A with the set of \mathscr{L} -classes contained in this sink, and define T' to be the subset of S such that $S \setminus T' \subseteq J$ and $T' \cap J$ is the union of the \mathscr{L} -classes in A'. Then either $A' = J/\mathscr{L}$, or, by Corollary 4.62, T' is a proper subsemigroup of S that properly contains T. Since T is maximal, it follows that $A' = J/\mathscr{L}$, and so the complement of A forms a single vertex of $\Gamma_{\mathscr{L}}$. This vertex is a source of $\Gamma_{\mathscr{L}}$ by condition (ii) of Proposition 4.58, and has colour 0 by Corollary 4.62.

Since T is a maximal subsemigroup of S of type (M3), it is not contained in a maximal subsemigroup of S of type (M2).

 (\Leftarrow) Let M be a maximal subsemigroup of S that contains T. Since T contains a union of \mathscr{L} -classes of J, M has type (M2) or (M3). In either case, it follows that M contains the \mathscr{L} -classes in A. However, A lacks only one vertex of $\Gamma_{\mathscr{L}}$, and since a maximal subsemigroup is a proper subsemigroup, it follows that M contains no additional \mathscr{L} -classes. Such a maximal subsemigroup of type (M2) does not exist by assumption, and so M = T has type (M3). \Box

Analogues of Corollary 4.62 and Proposition 4.63 hold which concern subsemigroups that are unions of \mathscr{R} -classes of a finite semigroup. For completeness, we state the analogue of Proposition 4.63, which describes the maximal subsemigroups of type (M4).

Proposition 4.64 (Maximal subsemigroups of type (M4)). Let S be a finite semigroup with a regular \mathscr{J} -class J, and let T be a subset of S that contains $S \setminus J$. Suppose there exists a proper non-empty subset $B \subsetneq J/\mathscr{R}$ such that $T \cap J$ is the union of the \mathscr{R} -classes in B. Then T is a maximal subsemigroup of S if and only if the complement of B is a source of $\Gamma_{\mathscr{R}}(S, J)$ with colour 0, and there is no maximal subsemigroup of S of type (M2) whose corresponding subset of J/\mathscr{R} is B.

We describe an algorithm that uses Proposition 4.63 to compute the maximal subsemigroups of type (M3) arising from a regular \mathscr{J} -class J of a finite semigroup S. An analogous algorithm, using Proposition 4.64, permits the computation of maximal subsemigroups of type (M4). The first step is to compute the maximal subsemigroups of S of type (M2) that arise from J, as described in Section 4.4.3. In doing so, we record the sets of \mathscr{L} -classes and \mathscr{R} -classes of J that correspond to each maximal subsemigroup of type (M2). The second step is to search for the sources of the digraph $\Gamma_{\mathscr{L}}(S, J)$ with colour 0. For each source, we test whether its complement occurs as the set of \mathscr{L} -classes of some maximal subsemigroup of type (M2), and discard if so. By Proposition 4.63, the complements of the remaining sources define the maximal subsemigroups of S that arise from J and have type (M3). The final step is to specify a generating set for each such maximal subsemigroup; these generating sets can be constructed by using Proposition 4.66.

Example 4.65. Let S and J be the transformation semigroup and the \mathscr{J} -class, respectively, from Example 4.52. The digraph $\Gamma_{\mathscr{L}}$, depicted in Figure 4.53, contains two sources with colour 0, $\{L_{x_3}\}$ and $\{L_{x_4}\}$. The complements of these vertices in J/\mathscr{L} are $A_1 = \{L_{x_1}, L_{x_4}, L_{x_1x_6}\}$ and $A_2 = \{L_{x_1}, L_{x_3}, L_{x_1x_6}\}$, respectively. In Example 4.60, we found that there is no maximal subsemigroup of S of type (M2) whose set of \mathscr{L} -classes is equal to A_1 or A_2 . By Proposition 4.63, there are two maximal subsemigroups of S of type (M3) arising from J: the set consisting of

 $S \setminus J$ and the union of the \mathscr{L} -classes in A_1 , and the set consisting of $S \setminus J$ and the union of the \mathscr{L} -classes in A_2 .

In $\Gamma_{\mathscr{R}}$, the sources are $\{R_{x_1}\}$ and $\{R_{x_2}\}$, and they have colour 0. In Example 4.60, we found that there is no maximal subsemigroup of S of type (M2) whose set of \mathscr{R} -classes is equal to the complement of either of these sources. By Proposition 4.64, $S \setminus R_{x_1}$ and $S \setminus R_{x_2}$ are the maximal subsemigroups of S of type (M4) that arise from J.

The following proposition can be used to describe generating sets for maximal subsemigroups of type (M3). Generating sets for maximal subsemigroups of type (M4) are obtained analogously. The proof of Proposition 4.66 is similar to, but simpler than, the proof of Proposition 4.61, and is omitted.

Proposition 4.66. Let T be a subsemigroup of a finite semigroup $S = \langle X \rangle$ such that $S \setminus T$ is contained in a regular \mathscr{J} -class J of S. Suppose that there exists a non-empty proper subset $A \subsetneq J/\mathscr{L}$ such that $T \cap J$ is the union of the \mathscr{L} -classes in A. Then T is generated by any set consisting of:

- (i) $X \setminus J$;
- (ii) a semigroup generating set for the ideal $\{x \in S : J_x < J\}$ of S;
- (iii) a generating set for a group \mathscr{H} -class H_x , where $L_x \in A$;
- (iv) for each source U in the induced subdigraph of $\Gamma_{\mathscr{L}}(S, J)$ on A, an element $y \in R_x$ such that $L_y \in U$; and
- (v) for every source V in $\Gamma_{\mathscr{R}}(S, J)$, an element $z \in L_x$ such that $R_z \in V$.

4.4.5 Maximal subsemigroups by removing a \mathcal{J} -class: (M5)

In this section, we discuss how to compute maximal subsemigroups of type (M5), which are formed by removing a \mathscr{J} -class from a finite semigroup. Recall that J is an arbitrary \mathscr{J} -class of the finite semigroup $S = \langle X \rangle$, and that $X' = \{x \in X : J < J_x\}$.

If the set of \mathscr{L} -classes A and the set of \mathscr{R} -classes B in the statement of Proposition 4.58 are both empty, then conditions (i)–(iii) and (v) of Proposition 4.58 are vacuously satisfied, which leaves only condition (iv). Thereby, we obtain the following corollary.

Corollary 4.67. Suppose that J is regular. Then the subset $S \setminus J$ is a subsemigroup of S if and only if the graph $\Theta(S, J)$ has no edges.

As mentioned previously, the existence of a maximal subsemigroup of type (M5) arising from J precludes the existence of maximal subsemigroups of types (M1)–(M4) arising from J, since any of these kinds of maximal subsemigroups properly contains the set $S \setminus J$. Therefore, $S \setminus J$ is a maximal subsemigroup of S if and only if $S \setminus J$ is a subsemigroup, and no maximal subsemigroups of types (M1)–(M4) arise from J. Alternatively, $S \setminus J$ is a maximal subsemigroup if and only if at least one maximal subsemigroup arises from J, but no maximal subsemigroups of types (M1)–(M4) arise from J. We use the following lemma to determine whether at least one maximal subsemigroup arises from J.

Lemma 4.68. The following are equivalent:

- (i) Some proper subsemigroup of S contains $S \setminus J$.
- (ii) Some maximal subsemigroup of S arises from J.

- (iii) Every generating set for S has non-empty intersection with J.
- (iv) $\langle S \setminus J \rangle \neq S$.
- (v) $J \cap X \not\subseteq \langle X' \rangle$.
- *Proof.* (i) \Rightarrow (ii) Let T be a proper subsemigroup of S that contains $S \setminus J$. Since J is finite, T is contained in a maximal subsemigroup of S, which arises from J.
- (ii) \Rightarrow (iii) Let M be a maximal subsemigroup of S arising from J, and let A be any subset of S that is disjoint from J. Then $\langle A \rangle \leq \langle S \setminus J \rangle \leq M \neq S$, and so A does not generate S.
- (iii) \Rightarrow (iv) Since $S \setminus J$ is disjoint from J, it follows that $\langle S \setminus J \rangle \neq S$.
- (iv) \Rightarrow (v) (Contrapositive). Certainly $S \setminus J$ contains $X \setminus J$, and X' in particular. Therefore, if $J \cap X \subseteq \langle X' \rangle$, then $\langle S \setminus J \rangle$ contains X, which is a generating set for S, and so $\langle S \setminus J \rangle = S$.
- (v) \Rightarrow (i) Since $\langle S \setminus J \rangle \cap J = \langle X' \rangle \cap J$, it follows that $\langle S \setminus J \rangle$ does not contain $J \cap X$. In particular, $\langle S \setminus J \rangle$ is a proper subsemigroup of S that contains $S \setminus J$.

Note that when the generating set X is minimal with respect to containment, $J \cap X \subseteq \langle X' \rangle$ if and only if $J \cap X = \emptyset$. Therefore condition (v) can be tested more easily in this case.

We use the following proposition, which is the main result of this section, when computing maximal subsemigroups of type (M5).

Proposition 4.69 (Maximal subsemigroups of type (M5)). Let $S = \langle X \rangle$ be a finite semigroup, let J be a \mathscr{J} -class of S with representative x, let $X' = \{s \in X : J < J_s\}$, and suppose that no maximal subsemigroups of types (M1)–(M4) arise from J. The following are equivalent:

- (i) $S \setminus J$ is a maximal subsemigroup of S;
- (ii) any of the conditions in Lemma 4.68 holds; and
- (iii) $x \notin \langle X' \rangle$.

Furthermore, if J is regular, then the following is also equivalent to the above conditions:

(iv) the graph $\Theta(S, J)$ has no edges.

Proof. The equivalence of (i), (ii), and (iv) has been established by the above discussion and Corollary 4.67. It remains to prove the equivalence of (iii).

- (i) \Rightarrow (iii) Since $X' \subseteq S \setminus J$, it follows that $\langle X' \rangle \leq \langle S \setminus J \rangle = S \setminus J$. In particular, $x \notin \langle X' \rangle$.
- (iii) \Rightarrow (ii) Since $\langle X' \rangle \cap J = \langle S \setminus J \rangle \cap J$, it follows that $x \notin \langle S \setminus J \rangle$. Therefore $\langle S \setminus J \rangle \neq S$, i.e. Lemma 4.68(iv) holds.

By Proposition 4.10, a maximal subsemigroup arising from a non-regular \mathscr{J} -class has type (M5). Therefore, if J is non-regular, then the assumption concerning maximal subsemigroups of types (M1)–(M4) in the statement of Proposition 4.69 can be ignored.

In Algorithm 4.86, we use Proposition 4.69(ii) in combination with Lemma 4.68(v) to compute the maximal subsemigroups of type (M5) that arise from a regular non-maximal \mathscr{J} -class; see lines 17 and 27. For a non-regular non-maximal \mathscr{J} -class, we simply test whether Proposition 4.69(iii) holds. If $S \setminus J$ is a subsemigroup of S, then a generating set for $S \setminus J$ is given by the union of $X \setminus J$ with a semigroup generating set for the ideal { $x \in S : J_x < J$ } of S.

Maximal subsemigroups and minimal generating sets

Before continuing, we briefly elucidate a few connections between the minimal generating sets of a semigroup and the maximal subsemigroups that are formed by removing a \mathcal{J} -class.

Lemma 4.70. Let $S = \langle X \rangle$ be a semigroup, suppose that $S \setminus J$ is a maximal subsemigroup of S for some $J \in S/\mathscr{J}$, and let $x \in J$ be arbitrary. Then $(X \setminus J) \cup \{x\}$ generates S.

Proof. Let J' be an arbitrary \mathscr{J} -class of S. If $J' \leq J$, then, by the contrapositive of Lemma 1.12(iii), the elements of J' cannot be expressed involving elements of $X \cap J$. In other words, $J' \subseteq \langle X \setminus J \rangle$. Let $I = \{s \in S : J_s < J\}$. Since I is an ideal of S, the union $\langle (X \setminus J) \cup \{x\} \rangle \cup I$ is a subsemigroup of S that contains every \mathscr{J} -class of S, except for possibly J. Hence

$$T = \langle (X \setminus J) \cup \{x\} \rangle \cup I = \langle (X \setminus J) \cup \{x\} \rangle \cup (S \setminus J).$$

Since T properly contains the maximal subsemigroup $S \setminus J$, it follows that T = S. In particular, $T \cap J = J$. But $T \cap J = \langle (X \setminus J) \cup \{x\} \rangle \cap J$. Therefore $\langle (X \setminus J) \cup \{x\} \rangle$ contains J, and hence contains the generating set X.

Corollary 4.71. Let S be a semigroup that has a minimal generating set X with respect to containment, let J be a \mathscr{J} -class of S, and suppose that $S \setminus J$ is a maximal subsemigroup of S. Then $|J \cap X| = 1$.

Since a finite semigroup has minimal generating sets, we may use these results to describe the minimal generating sets of a finite semigroup whose maximal subsemigroups have type (M5). The result implies that a finite \mathscr{J} -trivial semigroup has a unique minimal generating set [36].

Corollary 4.72 (cf. [36]). Let S be a finite semigroup whose maximal subsemigroups have type (M5), and let $\mathfrak{J} = \{J \in S | \mathscr{J} : S \setminus J \text{ is a maximal subsemigroup of } S\}$. Then the minimal generating sets of S (with respect to both cardinality and containment) are the transversals of \mathfrak{J} . In particular, rank $(S) = |\mathfrak{J}|$.

Proof. In a finite semigroup, a generating set that is minimal with respect to cardinality is necessarily minimal with respect to containment. The generating sets in the statement have common cardinality $|\mathfrak{J}|$. Therefore, if they comprise all generating sets that are minimal with respect to containment, then the two notions of minimality are equivalent in this case. By assumption and by Lemma 4.68, a minimal generating set for S (with respect to containment) contains only elements of \mathscr{J} -classes in \mathfrak{J} . By Lemma 4.70, the result follows.

4.5 Arbitrary finite monoids

In Chapter 5, we describe and count the maximal subsemigroups of several families of particular finite monoids. For almost all of these monoids, the maximal subsemigroups arise either from the group of units, or from a \mathscr{J} -class that is covered (in the partial order of \mathscr{J} -classes) by the group of units. While the techniques of the previous section certainly apply to finite monoids, many of the cases that they treat do not arise for these kinds of \mathscr{J} -classes. Thus, the purpose of this section is to simplify some of the results of Section 4.4, under the assumption that the \mathscr{J} -class in question is either equal to, or is covered by, the group of units of a monoid.

4.5.1 Maximal subsemigroups from the group of units

The maximal subsemigroups of a finite monoid that are the easiest to describe are those that arise from its group of units. Such maximal subsemigroups exist by Lemma 4.68, since the

non-units in a finite monoid form an ideal, and so every generating set of the monoid contains a unit. As shown in the following lemma, maximal subsemigroups of this kind can be calculated from the group of units in isolation, without reference to the remainder of the monoid.

Lemma 4.73. Let S be a finite monoid with group of units G. Then the maximal subsemigroups of S arising from G are the sets $(S \setminus G) \cup U$, for each maximal subsemigroup U of G. In other words, if G is trivial, then the unique maximal subsemigroup of S arising from G is $S \setminus G$, which has type (M5); if G is non-trivial, then the maximal subsemigroups of S arising from G are the sets $(S \setminus G) \cup U$, for each maximal subgroup U of G, which have type (M1).

Proof. Since $S \setminus G$ is an ideal of S, it follows that, for a subset U of G, $(S \setminus G) \cup U$ is a subsemigroup of S if and only if U is a subsemigroup of G. Since this correspondence between subsemigroups of S containing $S \setminus G$ and subsemigroups of G preserves inclusion, the result follows. Note that a *subsemigroup* of a finite group is a *subgroup*, unless it is empty; the only group to possess the empty semigroup as a maximal subsemigroup is the trivial group. \Box

4.5.2 Maximal subsemigroups from a \mathcal{J} -class covered by the group of units

Before presenting the main results of this section, we first state the following corollary to Lemma 4.8. This corollary is used several times in Chapter 5 to describe the maximal subsemigroups that arise from a \mathcal{J} -class covered by the group of units of a monoid.

Corollary 4.74. Let S be a finite monoid with group of units G, and suppose there exists a non-empty subset A of $S \setminus G$ with the property that $S = \langle G, x \rangle$ if and only if $x \in A$. Then the maximal subsemigroups of S are those that arise from the group of units and $S \setminus A$.

Proof. It remains to describe the maximal subsemigroups that do not arise from the group of units. Let $x \in A$. Since $S = \langle G, x \rangle$ and G is closed under multiplication, the principal ideal of S generated by x is $S \setminus G$. Since x was arbitrary, every element of A generates the same principal ideal, and so A is contained in some \mathscr{J} -class J of S. By Lemma 4.68, the remaining maximal subsemigroups of S arise from J. By applying Lemma 4.8 with k = 1 and $X_1 = A$, we find that the unique maximal subsemigroup in this case is $S \setminus A$.

In the remainder of this section, we let S denote a finite monoid with generating set X, let G denote the group of units of S, and let J be a \mathscr{J} -class of S that is covered by the group of units; that is, if $J' \in S/\mathscr{J}$ and J < J', then J' = G. Define

$$X' = \{ x \in X : J < J_x \} = G \cap X.$$

The problem of finding the maximal subsemigroups of S that arise from J is simpler than the general case. This is because elements of \mathcal{J} -classes that are strictly greater than J are units, which generate no elements within J, and because, roughly speaking, the action of units on J by multiplication is easier to understand than the action of arbitrary elements.

Maximal subsemigroups of type (M5)

We first show that at least one maximal subsemigroup of S arises from a \mathscr{J} -class covered by the group of units. Since $X' = \{x \in X : J < J_x\} \subseteq G$, it follows that $\langle X' \rangle \cap J \subseteq G \cap J = \emptyset$, and so we may obtain the following corollary to Proposition 4.69.

Corollary 4.75 (Maximal subsemigroups of type (M5)). Let S be a finite monoid with group of units G, and let J be a \mathcal{J} -class of S that is covered by G. Then $S \setminus J$ is a maximal subsemigroup of S if and only if no maximal subsemigroups of types (M1)–(M4) arise from J.

Maximal subsemigroups of types (M2)–(M4)

Suppose that J is regular, and that the generating set X contains the group of units G. Therefore X' = G. The digraphs $\Gamma_{\mathscr{L}}(S, J)$ and $\Gamma_{\mathscr{R}}(S, J)$ and the graphs $\Delta(S, J)$ and $\Theta(S, J)$ were introduced in Section 4.4.2 in order to classify the maximal subsemigroups that arise from J and have types (M2)–(M5). In the present case, when J is a regular \mathscr{J} -class covered by the group of units, these graphs and digraphs have particularly straightforward descriptions.

By (4.50), $\Gamma_{\mathscr{L}}(S, J)$ is the digraph formed by removing the loops from the quotient of

$$\left(J/\mathscr{L}, \left\{(L_x, L_y) \in J/\mathscr{L} \times J/\mathscr{L} : L_x g = L_y \text{ for some } g \in G\right\}\right)$$

$$(4.76)$$

by its strongly connected components. Let $L \in J/\mathscr{L}$, $x \in L$, and $g \in G$ be arbitrary. Since $xg \mathscr{R} x$, it follows by Green's Lemma [Lemma 1.11] that $L_xg = L_{xg}$. Furthermore, \mathscr{L} is a right congruence, and so $L_x = L_y$ implies that $L_{xg} = L_{yg}$. Therefore, the function $(J/\mathscr{L} \times G) \longrightarrow J/\mathscr{L}$ defined by $L \cdot g = Lg$ for all $L \in J/\mathscr{L}$ and $g \in G$, where $x \in L$ is arbitrary, gives a well-defined right action of G on J/\mathscr{L} . In other words, G acts on the \mathscr{L} -classes of J by right multiplication.

The strongly connected components of the digraph in (4.76) are simply the orbits of G under its right action on J/\mathscr{L} . Therefore, the digraph $\Gamma_{\mathscr{L}}(S, J)$ has no edges, and its vertices are the orbits of the right action of G on J/\mathscr{L} by right multiplication. Analogously, the vertices of the digraph $\Gamma_{\mathscr{R}}(S, J)$ are the orbits of the left action of G on J/\mathscr{R} by left multiplication, and the digraph has no edges. As before, the vertices of $\Delta(S, J)$ and $\Theta(S, J)$ are the disjoint union of the vertices of $\Gamma_{\mathscr{L}}(S, J)$ and $\Gamma_{\mathscr{R}}(S, J)$. Since $\langle X' \rangle \cap J = G \cap J = \emptyset$, it follows that $\Theta(S, J)$ has no edges, and that every vertex in $\Gamma_{\mathscr{L}}(S, J)$ and $\Gamma_{\mathscr{R}}(S, J)$ has colour 0.

The results in Section 4.4 concerning the existence of maximal subsemigroups of types (M2)– (M4) are formulated in terms of the vertices and edges of the graphs and digraphs mentioned above. In the present case, when J is a regular \mathscr{J} -class of a finite monoid covered by the units, the conditions that concern the edges or the colour of vertices in $\Gamma_{\mathscr{L}}(S, J)$, $\Gamma_{\mathscr{R}}(S, J)$, or $\Theta(S, J)$ are vacuously satisfied; the remaining conditions on these graphs and digraphs can be reformulated in terms of $\Delta(S, J)$.

We begin by reformulating Proposition 4.59, which concerns maximal subsemigroups of type (M2). This reformulation follows from the statement of Proposition 4.59, since the conditions in Proposition 4.58(ii)–(iv) concern only the edges of $\Gamma_{\mathscr{L}}(S, J)$, $\Gamma_{\mathscr{R}}(S, J)$, and $\Theta(S, J)$, and can therefore be ignored.

Proposition 4.77 (Maximal subsemigroups of type (M2)). Let S be a finite monoid, let J be a regular \mathscr{J} -class of S covered by the group of units, and let T be a subset of S that contains $S \setminus J$. Suppose there exist proper non-empty subsets $A \subsetneq J/\mathscr{L}$ and $B \subsetneq J/\mathscr{R}$ such that $T \cap J$ is the union of the Green's classes in A and B. Then T is a maximal subsemigroup of S if and only if A and B are unions of vertices that form a maximal independent subset of $\Delta(S, J)$.

By Proposition 4.77, the maximal subsemigroups of S of type (M2) arising from J are in one-to-one correspondence with the maximal independent subsets of $\Delta(S, J)$, excluding the set of all orbits of \mathscr{L} -classes, and the set of all orbits of \mathscr{R} -classes. We deduce the following.

Corollary 4.78. The number of maximal subsemigroups of S of type (M2) arising from J is two less than the number of maximal independent subsets of $\Delta(S, J)$.

We reformulate Proposition 4.63 as follows.

Proposition 4.79 (Maximal subsemigroups of type (M3)). Let S be a finite monoid, let J be a regular \mathcal{J} -class of S covered by the group of units of S, and let T be a subset of S such that

contains $S \setminus J$. Suppose there exists a proper non-empty subset $A \subsetneq J/\mathscr{L}$ such that $T \cap J$ is the union of the \mathscr{L} -classes in A. Then T is a maximal subsemigroup of S if and only if $(J/\mathscr{L}) \setminus A$ is a vertex in $\Delta(S, J)$ that is not adjacent to a vertex of degree one.

Proof. (\Rightarrow) By Proposition 4.63, $(J/\mathscr{L}) \setminus A$ is a vertex in $\Delta(S, J)$. Suppose that $(J/\mathscr{L}) \setminus A$ is adjacent to a vertex $B' \subseteq J/\mathscr{R}$ of degree one. The vertices in $A \cup B'$ form an independent subset of $\Delta(S, J)$. This independent subset is contained in a maximal independent subset corresponding to $A \cup B$, for some non-empty proper subset $B \subsetneq J/\mathscr{R}$ that contains B'. By Proposition 4.77, a maximal subsemigroup of S corresponding to the maximal independent subset subset $A \cup B$ properly contains T. Therefore no such vertex B' exists.

 (\Leftarrow) Every vertex in $\Gamma_{\mathscr{L}}(S, J)$ is a source with colour 0, and so by Proposition 4.63 it suffices to show that no maximal subsemigroup of type (M2) contains T. Equivalently, by Proposition 4.77, if suffices to show that there exists no non-empty proper subset $B \subsetneq J/\mathscr{R}$ such that $A \cup B$ forms a maximal independent subset of $\Delta(S, J)$. Let $B' \subseteq J/\mathscr{R}$ be a vertex of $\Delta(S, J)$. Since J is regular, B' is adjacent to some orbit of \mathscr{L} -classes. If B' is adjacent to $(J/\mathscr{L}) \setminus A$, then by assumption, B' is also adjacent to some vertex in A. In any case, B' is adjacent to some vertex in A, and so $A \cup B'$ does not form an independent subset of $\Delta(S, J)$. In particular, there exists no subset $B \subsetneq J/\mathscr{R}$ such that $A \cup B$ forms a maximal independent subset of $\Delta(S, J)$.

There is a natural dual to this proposition, which characterises the maximal subsemigroups of S that arise from J and that have type (M4) in terms of the orbits of \mathscr{R} -classes in $\Delta(S, J)$ that are not adjacent to vertices of degree one.

By the regularity of J, every vertex in $\Delta(S, J)$ has degree at least one. Therefore, by Proposition 4.79, the number of maximal subsemigroups of S of type (M3) is the number of orbits of \mathscr{L} -classes that are adjacent in $\Delta(S, J)$ only to orbits of \mathscr{R} -classes with degree at least two. In the case that every orbit of \mathscr{R} -classes has degree two or more in $\Delta(S, J)$, then the number of maximal subsemigroups of type (M3) is simply the number of orbits of \mathscr{L} -classes. The analogous statements hold for maximal subsemigroups of type (M4). By the same token, the existence of maximal subsemigroups is restricted when there is a single orbit of \mathscr{L} -classes or a single orbit of \mathscr{R} -classes (i.e. when the group acts transitively).

Lemma 4.80. If G acts transitively on the \mathscr{L} -classes of J, then no maximal subsemigroups of types (M2) or (M3) arise from J. Similarly, if G acts transitively on the \mathscr{R} -classes of J, then no maximal subsemigroups of types (M2) or (M4) arise from J.

Proof. Suppose that G acts transitively on the \mathscr{L} -classes of J. Since J is regular, for every \mathscr{L} -class L of J there exists an \mathscr{R} -class R of J such that $L \cap R$ is a group, and vice versa. Therefore there are no isolated vertices in $\Delta(S, J)$, and so each vertex of \mathscr{R} -classes is adjacent only to the unique vertex of \mathscr{L} -classes. It follows that the set of all \mathscr{L} -classes and the set of all \mathscr{R} -classes are the only maximal independent subsets of $\Delta(S, J)$. By Corollary 4.78, there are no maximal subsemigroups of type (M2) arising from J, and by Proposition 4.79, there are no maximal subsemigroups of type (M3). The proof of the second statement is dual.

When S is a regular *-monoid, the \mathscr{L} -classes and \mathscr{R} -classes of a \mathscr{J} -class are in bijective correspondence via the * anti-isomorphism, and so the graph $\Delta(S, J)$ is particularly easy to describe in this case.

Lemma 4.81. Let S be a regular *-monoid and let J be a \mathscr{J} -class covered by the group of units G of S. A collection of \mathscr{L} -classes $\{L_{x_1}, \ldots, L_{x_n}\}$ is a vertex of $\Delta(S, J)$ if and only if the collection of \mathscr{R} -classes $\{R_{x_1^*}, \ldots, R_{x_n^*}\}$ is a vertex of $\Delta(S, J)$, and any pair of such vertices is adjacent in $\Delta(S, J)$.

Proof. If L_x and L_y are \mathscr{L} -classes of J in the same vertex of $\Delta(S, J)$, then there exists $g \in G$ such that $L_x g = L_y$. Therefore

$$g^*R_{x^*} = g^*L_x^* = (L_xg)^* = L_y^* = R_{y^*},$$

and so R_{x^*} and R_{y^*} belong to the same vertex of $\Delta(S, J)$. By an analogous argument, it follows that the first statement holds. The second statement holds since, for any element $x \in J$, the \mathscr{H} -class $L_x \cap R_{x^*}$ contains the projection x^*x , and is therefore a group. In particular, the vertex of $\Delta(S, J)$ containing L_x is adjacent to the vertex of $\Delta(S, J)$ that contains R_{x^*} . \Box

The situation is further simplified when every idempotent of J is a projection, which occurs, for instance, when S is inverse.

Corollary 4.82. Let S be a finite regular *-monoid with group of units G, and let J be a \mathscr{J} -class of S that is covered by G and whose only idempotents are projections. Suppose that $\{O_1, \ldots, O_n\}$ are the orbits of the right action of G on the \mathscr{L} -classes of J by right multiplication Then the maximal subsemigroups of S arising from J have types (M1), (M2), or (M5). A maximal subsemigroup of type (M2) is the union of $S \setminus J$ and the union of the Green's classes

$$\{L : L \in O_i, \ i \in A\} \cup \{L^* : L \in O_i, \ i \notin A\},\$$

where A is any proper non-empty subset of $\{1, \ldots, n\}$. In particular, there are $2^n - 2$ maximal subsemigroups of type (M2), and no maximal subsemigroups of types (M3) or (M4).

Proof. By definition of $\Delta(S, J)$, the vertices of \mathscr{L} -classes of $\Delta(S, J)$ are $\{O_1, \ldots, O_n\}$, and so by Lemma 4.81, the vertices of \mathscr{R} -classes are $\{\{L^* : L \in O_i\} : i \in \{1, \ldots, n\}\}$. Since every idempotent of J is a projection, for each \mathscr{L} -class L_x of J, the only group \mathscr{H} -class contained in L_x is $L_x \cap R_{x^*}$, and so the vertex containing L_x is only adjacent to the vertex containing R_{x^*} . Therefore the edges of $\Delta(S, J)$ are $\{O_i, \{L^* : L \in O_i\}\}$ for each $i \in \{1, \ldots, n\}$. In particular, each vertex of $\Delta(S, J)$ has degree one, and it follows from Proposition 4.79 and its dual that no maximal subsemigroups of types (M3) or (M4) arise from J. Furthermore, given the description of $\Delta(S, J)$, it is clear that a maximal independent subset of $\Delta(S, J)$ is formed by choosing any one vertex from each of the n edges, and so there are 2^n maximal independent subsets. The description and number of maximal subsemigroups of type (M2) follows by Proposition 4.77 and Corollary 4.78.

Maximal subsemigroups of type (M1)

A few of the monoids that we consider in Chapter 5 exhibit maximal subsemigroups of type (M1) that arise from a regular \mathscr{J} -class covered by the group of units. The results in Section 4.4.1 neatly characterise the existence of maximal subsemigroups of type (M1) in terms of certain maximal subsemigroups of the relevant principal factor. However, while this is very useful in general, especially from the point of view of computation, we desire more specialised tools to easily describe the maximal subsemigroups that arise in the particular examples included in Chapter 5. In Proposition 4.85, we present a result that will be useful for these cases. To prove this proposition, we require the following definition.

Definition 4.83. Let S be a finite regular *-monoid with group of units G, and let $A \subseteq S$. Then the setwise stabilizer of A in G, $\operatorname{Stab}_G(A)$, is the subgroup $\{g \in G : Ag = A\}$ of G.

Let S and G be defined as in Definition 4.83, and let $A \subseteq S$. Note that $\operatorname{Stab}_G(A)$ is defined to be the set of elements of G that stabilize A on the *right*. However, with $A^* = \{a^* : a \in A\}$, the set of elements of G that stabilize A on the *left* is equal to $\operatorname{Stab}_G(A^*)$, since

$$\{g \in G : gA = A\} = \{g \in G : A^*g^* = A^*\} = \operatorname{Stab}_G(A^*)^* = \operatorname{Stab}_G(A^*)^{-1} = \operatorname{Stab}_G(A^*).$$

Thus, for a subset A of S that satisfies $A^* = A$, such as for the \mathcal{H} -class of a projection,

$$\operatorname{Stab}_G(A) = \{ g \in G : Ag = A = gA \}.$$

In Proposition 4.85, we require the set $eStab_G(H_e) = \{es : s \in Stab_G(H_e)\}$, where e is a projection of the regular *-monoid S, and the \mathscr{J} -class J_e is covered by G. Any submonoid of S that contains both e and G also contains $eStab_G(H_e)$. In particular, every maximal subsemigroup of type (M1) arising from J_e contains G and all idempotents in J_e by Lemma 1.14, and hence contains $eStab_G(H_e)$. A stronger result, necessary for the proof of Proposition 4.85, is given by the following lemma.

Lemma 4.84. Let S be a finite monoid with group of units G, let $e \in E(S)$, and let T be a submonoid of S that contains both e and G. Then the set $eStab_G(H_e^S)$ is a subgroup of H_e^T .

Proof. Since e is an idempotent, $H_e^T = T \cap H_e^S$. Clearly $e\operatorname{Stab}_G(H_e^S) \subseteq eG \subseteq T$. Let $g \in \operatorname{Stab}_G(H_e^S)$. Then $eg \in H_e^S$ by definition, and so $e\operatorname{Stab}_G(H_e^S) \subseteq H_e^S$. Thus $e\operatorname{Stab}_G(H_e^S) \subseteq T \cap H_e^S = H_e^T$, and the subset is non-empty since $e = e1_S \in \operatorname{eStab}_G(H_e^S)$. Since S is finite, it remains to show that $e\operatorname{Stab}_G(H_e^S)$ is closed under multiplication. Let $g, g' \in \operatorname{Stab}_G(H_e^S)$. Since $eg \in H_e^S$ and e is the identity of H_e^S , it follows that (eg)e = eg. Thus

$$(eg)(eg') = (ege)g' = (eg)g' = e(gg') \in e\operatorname{Stab}_G(H_e^S).$$

We use these notions and results to state and prove the following proposition.

Proposition 4.85. Let S be a finite regular *-monoid with group of units G, let J be a \mathcal{J} -class of S that is covered by G, and let H_e^S be the \mathcal{H} -class of a projection $e \in J$. Suppose that G acts transitively on the \mathcal{R} -classes or the \mathcal{L} -classes of J, and that J contains one idempotent per \mathcal{L} -class and one idempotent per \mathcal{R} -class (i.e. every idempotent of J is a projection). Then the maximal subsemigroups of S arising from J are either:

- (a) $(S \setminus J) \cup GUG = \langle S \setminus J, U \rangle$, for each maximal subgroup U of H_e^S that contains $eStab_G(H_e^S)$ (type (M1)), or
- (b) $S \setminus J$, if no maximal subsemigroups of type (M1) exist (type (M5)).

Proof. Since S is a regular *-monoid, G acts transitively on the \mathscr{L} -classes of J if and only if G acts transitively on the \mathscr{R} -classes of J. Hence there are no maximal subsemigroups of types (M2), (M3), or (M4) arising from J, by Lemma 4.80. By Corollary 4.75, it remains to describe the maximal subsemigroups of type (M1).

Let U be a maximal subgroup of H_e^S that contains $e\operatorname{Stab}_G(H_e^S)$, and define $M_U = (S \setminus J) \cup GUG$. To prove that M_U is a maximal subsemigroup of S, we first show that M_U is a proper subset of S, then that it is a subsemigroup, and finally that it is maximal in S. Since G acts transitively on the \mathscr{L} - and \mathscr{R} -classes of J and M_U contains $S \setminus J$, it follows that the set M_U intersects every \mathscr{H} -class of S non-trivially. Once we have shown that M_U is a subsemigroup, it will obviously follows that M_U is generated by $(S \setminus J) \cup U$, since $G \subseteq S \setminus J$.

To prove that M_U is a proper subset of S, it suffices to show that $GUG \cap H_e^S \subseteq U$. Let $x \in GUG \cap H_e^S$. Since $x \in GUG$, we may write $x = \alpha u\beta$ for some $\alpha, \beta \in G$ and $u \in U$. Since $u, \alpha u\beta \in H_e^S$, it is straightforward to show that $\alpha u, u\beta \in H_e^S$. Thus

$$\alpha H_e^S = \alpha (u H_e^S) = (\alpha u) H_e^S = H_e^S, \quad \text{ and } \quad H_e^S \beta = (H_e^S u) \beta = H_e^S (u \beta) = H_e^S.$$

In other words, α and β stabilize H_e^S on the left and right, respectively. Thus $\alpha, \beta \in \operatorname{Stab}_G(H_e^S)$, and

$$x = ex = e\alpha u\beta = e\alpha ue\beta \in \left(e\mathrm{Stab}_G(H_e^S)\right) \cup \left(e\mathrm{Stab}_G(H_e^S)\right) \subseteq U^3 = U.$$

In order to show that M_U is a subsemigroup, it suffices to show that $xy \in M_U$ whenever $x, y \in G \cup GUG$, because $S \setminus (G \cup J)$ is an ideal of S. If $x \in G$ and $y \in G$, then certainly $xy \in G$. If $x \in G$ and $y \in GUG$, then $xy \in G^2UG = GUG$ and $yx \in GUG^2 = GUG$. For the final case, assume that $x, y \in GUG$ and that $xy \in J$. By definition, $x = \alpha u\beta$ and $y = \sigma v\tau$ for some $\alpha, \beta, \sigma, \tau \in G$ and $u, v \in U$. It suffices to show that $\beta\sigma \in \operatorname{Stab}_G(H_e^S)$, because then

$$xy = (\alpha u\beta)(\sigma v\tau) = \alpha(ue)\beta\sigma v\tau = \alpha u(e\beta\sigma)v\tau \in GU\left(e\mathrm{Stab}_G(H_e^S)\right)UG$$
$$\subseteq GU^3G = GUG.$$

Since H_e^S is a group containing u and v, it follows that $u^*u = vv^* = e$. Thus

$$e\beta\sigma e = u^*u\beta\sigma vv^* = u^*\alpha^{-1}(\alpha u\beta\sigma v\tau)\tau^{-1}v^* = u^*\alpha^{-1}(xy)\tau^{-1}v^*.$$

Together with $xy = \alpha u(e\beta\sigma e)v\tau$, it follows that $e\beta\sigma e \in J$. By Lemma 1.10, $e\beta\sigma e \in R_e^S$. Since the elements $e\beta\sigma$ and e, and their product $e\beta\sigma e$, are all contained in R_e^S , Corollary 1.16 implies that $H_{e\beta\sigma}^S$ is a group. By assumption, R_e^S contains only one group \mathscr{H} -class, which is H_e^S . Thus $e\beta\sigma \in H_e^S$, and so $H_e^S\beta\sigma = (H_e^S e)\beta\sigma = H_e^S(e\beta\sigma) = H_e^S$, i.e. $\beta\sigma \in \operatorname{Stab}_G(H_e^S)$, as required.

Let M be a maximal subsemigroup of S that contains M_U . By Proposition 4.10, $M \cap H_e^S$ is a maximal subgroup of H_e^S , and the intersection of M with any \mathscr{H} -class of J contains exactly $|M \cap H_e^S|$ elements. Since $M \cap H_e^S$ contains U, the maximality of U in H_e^S implies that $U = M \cap H_e^S$. Since the group G acts transitively on the \mathscr{L} - and \mathscr{R} -classes of J, the intersection of GUG with any \mathscr{H} -class of J contains |U| elements. Thus $|M| \leq |M_U|$, and $M = M_U$.

Conversely, suppose that M is a maximal subsemigroup of S of type (M1) arising from J. By Proposition 4.10, $U = M \cap H_e^S = H_e^M$ is a maximal subgroup of H_e^S , and U contains $e\operatorname{Stab}_G(H_e^S)$ by Lemma 4.84. Since M contains G, U, and $S \setminus J$, it contains the maximal subsemigroup $M_U = (S \setminus J) \cup GUG$. But M is a proper subsemigroup, and so $M = M_U$. \Box

4.6 The algorithm

In this section, we describe the overall algorithm for computing the maximal subsemigroups of any given finite semigroup S represented by a generating set X. This algorithm is composed of the procedures described in Sections 4.3 and 4.4 and is fully implemented in the SEMI-GROUPS package [101] for GAP [58]; the underlying algorithms for graphs and digraphs are implemented in the DIGRAPHS [10] package for GAP. Pseudocode is given in Algorithm 4.86. We describe the corresponding functionality of the SEMIGROUPS package in Section 4.6.1, and we analyse the performance of these functions on a range of examples in Section 4.6.2.

We first compute the \mathscr{J} -classes of S that contain an element of X. If a particular \mathscr{J} -class J has empty intersection with X, then by Lemma 4.68(iv), there is no maximal subsemigroup of S arising from J. A \mathscr{J} -class of S that intersects X non-trivially does not necessarily give rise to maximal subsemigroups. However, as discussed after the proof of Lemma 4.68, if the generating set X is minimal, with respect to containment, then a maximal subsemigroup arises from a \mathscr{J} -class if and only if that \mathscr{J} -class has non-empty intersection with X.

Maximal and non-maximal \mathcal{J} -classes are treated separately.

Suppose that J is a maximal \mathscr{J} -class. Note that every maximal \mathscr{J} -class J intersects X nontrivially, since the subset $S \setminus J$ is an ideal of S. If |J| = 1, then by Lemma 4.3, the only maximal subsemigroup to arise from J is $S \setminus J$, which has type (M5). Suppose that |J| > 1. If follows that J is necessarily regular, and so the principal factor $J^* = J \cup \{0\}$ is a 0-simple semigroup. We may therefore compute a normalization Ψ from J^* to a regular Rees 0-matrix semigroup over a group $\mathscr{M}^0[G; I, \Lambda; P]$; see Section 3.2.1. Since $S \setminus J$ is an ideal, it is straightforward to see that a maximal subsemigroup of S arising from J has the form $(S \setminus J) \cup U$, where $U \subseteq J$ Algorithm 4.86 Maximal subsemigroups of a finite semigroup defined by generating set. **Input:** $S = \langle X \rangle$, a finite semigroup with generating set X. **Output:** \mathfrak{M} , the maximal subsemigroups of S. 1: $\mathfrak{M} \leftarrow \varnothing$ 2: if $S = \emptyset$ then return M. 3: 4: for $J_x \in \{J \in S | \mathcal{J} : J \cap X \neq \emptyset\}$ do $X' \leftarrow \left\{ y \in X \, : \, J_y > J_x \right\}$ 5: if $X' = \emptyset$ then 6: $\triangleright J_x$ is a maximal \mathscr{J} -class of Sif $|J_x| = 1$ then 7: $\mathfrak{M} \leftarrow \mathfrak{M} \cup \{S \setminus \{x\}\}$ \triangleright Lemma 4.3 8: else $\triangleright J_x$ is necessarily regular 9: compute a normalization $\Psi: J_x^* \longrightarrow \mathscr{M}^0[G; I, \Lambda; P]$ 10: \triangleright Section 3.2.1 $\mathfrak{X} \leftarrow$ maximal subsemigroups of $(J_x^*)\Psi$ of types (R3)–(R6) \triangleright Sections 4.3.3–4.3.4 11:for $M \in \mathfrak{X}$ do 12: $\mathfrak{M} \leftarrow \mathfrak{M} \cup \left\{ (S \setminus J_x) \cup (M \setminus \{0\}) \Psi^{-1} \right\}$ 13: $\triangleright J_x$ is a non-maximal \mathscr{J} -class else 14:if J_x is non-regular and $x \notin \langle X' \rangle$ then 15: $\mathfrak{M} \leftarrow \mathfrak{M} \cup \{S \setminus J_x\}$ ▷ Type (M5), Proposition 4.69(iii) 16:else if J_x is regular and $J_x \cap X \not\subseteq \langle X' \rangle$ then \triangleright Lemma 4.68(v) 17:compute a normalization $\Psi: J_x^* \longrightarrow \mathscr{M}^0[G; I, \Lambda; P]$ \triangleright Section 3.2.1 18: $E \leftarrow \{\text{one idempotent from each } \mathscr{L}\text{-class of } J_x\}$ 19: $\mathfrak{X} \leftarrow$ maximal subsemigroups of $(J_x^*)\Psi$ of type (R6) containing $(EX')\Psi \triangleright$ Alg. 4.44 20: for $M \in \mathfrak{X}$ do 21: $\mathfrak{M} \leftarrow \mathfrak{M} \cup \{ (S \setminus J_x) \cup (M \setminus \{0\}) \Psi^{-1} \}$ \triangleright Type (M1), Proposition 4.48 22: compute $\Gamma_{\mathscr{L}}(S, J_x), \Gamma_{\mathscr{R}}(S, J_x), \Delta(S, J_x)$, and $\Theta(S, J_x)$ \triangleright Section 4.4.2 23: $\mathfrak{M} \leftarrow \mathfrak{M} \cup$ maximal subsemigroups of type (M2) \triangleright Section 4.4.3 24:25: $\mathfrak{M} \leftarrow \mathfrak{M} \cup$ maximal subsemigroups of types (M3) and (M4) \triangleright Section 4.4.4 if no maximal subsemigroups of types (M1)–(M4) arise from J_x then 26: $\mathfrak{M} \leftarrow \mathfrak{M} \cup \{S \setminus J_x\}$ \triangleright Type (M5), Proposition 4.69(ii) 27: 28: return M.

and $U \cup \{0\}$ is a maximal subsemigroup of J^* . In other words, M is a maximal subsemigroup of S arising from J if and only if $(M \cap J)\Psi \cup \{0\}$ is a maximal subsemigroup of $\mathscr{M}^0[G; I, \Lambda; P]$ of type (R3), (R4), (R5), or (R6). Therefore, it suffices to compute the maximal subsemigroups of $\mathscr{M}^0[G; I, \Lambda; P]$ of these kinds, using the techniques of Section 4.3.

Suppose that J is a non-maximal \mathscr{J} -class of S that intersects X non-trivially. We first compute the set $X' = \{x \in X : J < J_x\}$. If J is non-regular, then by Proposition 4.69(iii), $S \setminus J$ is a maximal subsemigroup of type (M5) if and only if $x \notin \langle X' \rangle$, where x is an arbitrary element of J. No other kinds of maximal subsemigroup arise from a non-regular \mathscr{J} -class. If J is regular, then by Lemma 4.68, maximal subsemigroups of S arise from J if and only if $J \cap X \not\subseteq \langle X' \rangle$. If this condition holds, then we compute the maximal subsemigroups of types (M1)–(M4), as discussed in Sections 4.4.1, 4.4.3, and 4.4.4. If no such maximal subsemigroups exist, then by Proposition 4.69(ii), $S \setminus J$ is a maximal subsemigroup of S of type (M5).

Open Problem 4.87. Develop tools for computing subsemigroups that are maximal with respect to some property, such as maximal commutative or maximal regular subsemigroups of a finite semigroup; or maximal inverse subsemigroups of a finite inverse semigroup.

4.6.1 Relevant functionality in the Semigroups package for GAP

The techniques described in this chapter for computing maximal subsemigroups of a finite semigroup have been implemented by the author in the SEMIGROUPS package [101] for GAP [58]. The functions in the SEMIGROUPS package can be applied to any non-empty finite semigroup that can be constructed in GAP, subject to the time and space constraints of the computer. There are technical difficulties in GAP that arise when computing with an empty semigroup. For this reason, the SEMIGROUPS package finds only *non-empty* maximal subsemigroups. Note that only a semigroup of order 1 has the empty semigroup as a maximal subsemigroup.

Let S to be a non-empty finite semigroup. The command MaximalSubsemigroups(S); returns a list of the non-empty maximal subsemigroups of S. The behaviour of the function MaximalSubsemigroups may be customised by providing, as an additional argument, a record opts that describes these customisations via its components. The following independent customisations are available, and may be used together in any combination:

• If opts.number is true, then MaximalSubsemigroups(S, opts); returns the number of maximal subsemigroups of S that it finds, rather than constructing the maximal subsemigroups. It can be computationally expensive to find a generating set for a maximal subsemigroup (see Propositions 4.61 and 4.66). Therefore, in many cases, it can be much quicker to count the maximal subsemigroups of S directly than it is to construct them.

Note that the command NrMaximalSubsemigroups(S); is equivalent to the command MaximalSubsemigroups(S, rec(number := true));

- If opts.contain is a subset of S, then the command MaximalSubsemigroups(S, opts); returns the maximal subsemigroups of S that contain this set of elements. Note that this is often much more efficient that computing all maximal subsemigroups of S and then retaining only those that contain the desired set. In particular, a version of Algorithm 4.44 is used in this instance when S is a regular Rees 0-matrix semigroup over a group.
- If opts.D is a \mathscr{D} -class of S (or, equivalently, a \mathscr{J} -class of S, since S is finite), then MaximalSubsemigroups(S, opts); finds the maximal subsemigroups of S that arise from D. Since Algorithm 4.86 independently searches for the maximal subsemigroups arising from each \mathscr{D} -class, the maximal subsemigroups that arise from D can be computed without finding maximal subsemigroups that arise from any other \mathscr{D} -class.
- If S is a regular Rees matrix or Rees 0-matrix semigroup over a group, and opts.types is a subset of [1 .. 6], then MaximalSubsemigroups(S, opts); finds only those maximal subsemigroups of S of the types amongst (R1)-(R6) that correspond to opts.types. For instance, if opts.contain is [1, 3, 4, 6], then MaximalSubsemigroups(S, opts); finds the maximal subsemigroups of S that have types types (R1), (R3), (R4), or (R6).

The following GAP code demonstrates how to perform a custom maximal subsemigroups computation with the SEMIGROUPS package. This code produces the answer 24, which is the number of maximal subsemigroups of S that arise from D_x and contain the set $\{x, y\}$.

```
S := SingularTransformationMonoid(5);
x := Transformation([2, 2, 3, 4, 5]);
y := Transformation([5, 4, 1, 2, 1]);
opts := rec(number := true, contain := [x, y], D := DClass(S, x));
MaximalSubsemigroups(S, opts);
```

Further details and examples are included in the manual of the SEMIGROUPS [101] package.

4.6.2 Performance measurements

In this section, we experimentally investigate the performance of the algorithms presented in this chapter. The computations described in this section were run using the SEMIGROUPS package [101] for GAP [58], on a 2.66 GHz Intel Core i7 processor with 8GB of RAM.

Given a semigroup S represented by a set of generators, we compare the time taken to compute the Green's structure of S with that taken to find the maximal subsemigroups of S, given the Green's structure. As argued previously, if it is not possible to compute the Green's structure of a given semigroup within the limitations of the hardware, then there is little of significance that can be computed about that semigroup. Therefore, this comparison is appropriate. Additionally, we include the length of time that was spent finding maximal cliques or maximal subgroups of group \mathcal{H} -classes during the computation of the maximal subsemigroups, when these times are not negligible. We analyse the performance of computing the maximal subsemigroups of the following semigroups.

- For each $n \in \{2, ..., 11\}$, we compute the maximal subsemigroups of the full transformation monoid \mathcal{T}_n , as defined in Section 1.3.5. There are n^n elements and $n \mathscr{J}$ -classes in \mathcal{T}_n , and rank $(\mathcal{T}_n) = 3$ when n > 2. See Section 5.2.3 for a description of the maximal subsemigroups of \mathcal{T}_n in general, and see Table 5.8 for their number.
- For each $n \in \{11, \ldots, 20\}$, we calculate the maximal subsemigroups of the inverse monoid \mathcal{PORI}_n , which consists of all partial permutations of degree n that preserve or reverse the cyclic orientation of $\{1, \ldots, n\}$; see Section 5.2.1 for a precise definition. There are

$$1 + n \binom{2n}{n} - \frac{n^2(n^2 - 2n + 3)}{2}$$

elements and n + 1 \mathscr{J} -classes in \mathcal{PORI}_n , and $\operatorname{rank}(\mathcal{PORI}_n) = 3$ when n > 2. See Section 5.2.8 and Table 5.37 for the maximal subsemigroups of \mathcal{PORI}_n .

- We compute the maximal subsemigroups of the Jones monoid \mathcal{J}_n , as defined in Section 5.3.1, for each $n \in \{6, \ldots, 20\}$. If n > 1, then $\operatorname{rank}(\mathcal{J}_n) = n 1$, and the order of \mathcal{J}_n is the n^{th} Catalan number [43, Section 9], $\frac{1}{n+1}\binom{2n}{n}$. The number of \mathcal{J} -classes in \mathcal{J}_n is $\lfloor n/2 \rfloor$. See Section 5.3.7 and Table 5.59 for the maximal subsemigroups of \mathcal{J}_n , in general.
- We compute the maximal subsemigroups of 100 transformation semigroups of degree 9, each generated by 9 transformations chosen uniformly at random from \mathcal{T}_9 . See Table 4.88 for some data about these semigroups.

We chose to analyse the performance of the algorithms with \mathcal{T}_n , \mathcal{PORI}_n , and \mathcal{J}_n because these monoids are well-studied in the literature, because their internal representations in the SEMIGROUPS package are different, and because the methods implemented in the SEMIGROUPS package exhibit different behaviour for these monoids. In Figures 4.89, 4.90, and 4.91, we compare the time taken to compute the partial order of the \mathscr{J} -classes of these monoids with the time taken to find the maximal subsemigroups, given the partial order of \mathscr{J} -classes. The partial order of \mathscr{J} -classes of a semigroup is a fundamental part of its Green's structure, and was computed with the method described in [37, Algorithm 14], which is implemented in the SEMIGROUPS package. In each case, the largest value of n considered was the largest for which the Green's structure could be computed within the limitations of the given hardware.¹

For $S \in \{\mathcal{T}_n, \mathcal{PORI}_n\}$, the time taken to compute the maximal subsemigroups of S is dominated neither by the time taken to compute maximal cliques, nor by the time taken

¹Note that $|\mathcal{T}_{11}| = 285\ 311\ 670\ 611,\ |\mathcal{PORI}_{20}| = 2\ 756\ 930\ 503\ 801,\ \text{and}\ |\mathcal{J}_{20}| = 6\ 564\ 120\ 420.$

to compute maximal subgroups; rather, the majority of the time is spent constructing and processing the digraphs $\Gamma_{\mathscr{L}}(S, J)$ and $\Gamma_{\mathscr{R}}(S, J)$ and the graphs $\Delta(S, J)$ and $\Theta(S, J)$ from Section 4.4, where J is the \mathscr{J} -class of S consisting of elements of rank n-1. For these monoids, the graphs $\Delta(S, J)$ have 2 vertices, and so it can be immediately deduced that there are no maximal cliques from which maximal subsemigroups may arise. On the other hand, for the Jones monoids, the majority of the time in the computation of maximal subsemigroups is spent searching for maximal cliques, as is shown in Figure 4.91.

The monoids \mathcal{T}_n , \mathcal{PORI}_n , and \mathcal{J}_n are atypical examples of semigroups: for instance, they are all regular monoids, and each of their ideals is principal. Because of this, we opted to include some data relating to 'random' semigroups in order to demonstrate what is perhaps more typical behaviour of the algorithms for computing maximal subsemigroups. It is not clear what constitutes a reasonable notion of randomness, however, we believe that the notions used here are somewhat meaningful for the discussion here. The random semigroups were chosen to be generated by nine transformations of degree nine, because semigroups of this kind approach the limit of what is possible to compute with the given hardware.

					Standard
	Min	Max	Mean	Median	deviation
Size	125333	85014449	5657333	2484319	9865603
Number of \mathcal{J} -classes	8	89858	5336	2550	10187
Number of maximal subsemigroups	9	640	25	13	65
Time for maximal subsemigroups (ms)	8	272132	15998	5003	37944

Table 4.88: Information about the 100 transformation semigroups of degree 9 that we consider, each of which was generated by by 9 transformations that were chosen uniformly at random.

Figure 4.92 compares the time taken to compute the maximal subsemigroups of the 100 random transformation semigroups described above with the time taken to compute their partial orders of \mathscr{J} -classes. Each point on the horizontal axis of this graph corresponds to one of these semigroups. The points are sorted along the horizontal axis in increasing order, according to the ratio of the time taken to compute the maximal subsemigroups to the time taken to compute the partial order of its \mathscr{J} -classes. While there are some instances where computing the maximal subsemigroups is several orders of magnitude slower than computing the Green's structure, for the majority, the times taken are roughly comparable.

For each $n \in \{1, \ldots, 20\}$, we also analysed the performance of computing the maximal subsemigroups of 100 regular Rees 0-matrix semigroups $\mathscr{M}^0[G; I, \Lambda; P]$, where in each instance, $|I| = |\Lambda| = n, G$ was a permutation group chosen uniformly at random from the representatives of conjugacy classes of subgroups of \mathcal{S}_{10} , the number of connected components of the Graham-Houghton graph was chosen uniformly from $\{1, \ldots, n\}$, and the entries of the matrix P were chosen randomly, subject to this constraint. The entries of P were not chosen uniformly, nor do we claim that these Rees 0-matrix semigroups represent a uniform sample of such semigroups. The algorithms presented in Section 4.3, as implemented in the SEMIGROUPS [101] package, were used to compute these maximal subsemigroups.

The Green's structure of a regular Rees 0-matrix semigroup over a group can be determined immediately from its definition, and so the time taken to determine this was not used for comparison. In Figure 4.93, for each dimension considered, the mean time taken to compute the maximal subsemigroups of the 100 semigroups is shown alongside the mean time spent computing maximal cliques as part of the calculation. For the Rees 0-matrix semigroups we considered, it appears that the time taken to compute maximal cliques approaches the time taken to compute maximal subsemigroups, as the dimension increases; maximal cliques correspond to maximal subsemigroups of type (R5). The mean time spent computing maximal subgroups, which is a step in the computation of maximal subsemigroups of type (M1) was negligible in these examples.



Figure 4.89: For $n \in \{2, ..., 11\}$, this graph compares the time taken to compute the partial order of \mathscr{J} -classes of \mathcal{T}_n with both the time taken to compute its maximal subsemigroups, and the time spent during this finding the maximal subgroups of group \mathscr{H} -classes in \mathcal{T}_n .



Figure 4.90: For $n \in \{10, ..., 20\}$, this graph compares the time taken to compute the partial order of \mathscr{J} -classes of \mathcal{PORI}_n with both the total time taken to compute its maximal subsemigroups, and the amount of time that was spent as part of this finding maximal subgroups of group \mathscr{H} -classes in \mathcal{PORI}_n . As *n* increases, it appears that the time taken to compute the partial order of \mathscr{J} -classes of \mathcal{PORI}_n increases much more rapidly than the time taken to compute the maximal subsemigroups of \mathcal{PORI}_n , given the partial order of \mathscr{J} -classes.



Figure 4.91: For $n \in \{10, ..., 20\}$, this graph compares the time taken to compute the partial order of \mathscr{J} -classes of \mathcal{J}_n with the total time taken to compute its maximal subsemigroups, and the time spent as part of this in computing maximal cliques. It appears that, as *n* increases, the length of time spent computing maximal cliques approaches the total time taken.



Figure 4.92: A graph showing, for 100 random 9-generated subsemigroups of \mathcal{T}_9 , the ratio of the time taken to compute the maximal subsemigroups with the time taken to compute the partial order of \mathscr{J} -classes. Each cross represents a semigroup. If a cross sits above the horizontal line, then more time was spent computing the maximal subsemigroups of the corresponding semigroup than was spent computing its partial order of \mathscr{J} -classes. For most of these semigroups, the times taken for both steps was roughly comparable.



Figure 4.93: For each matrix dimension $|I| = |\Lambda| \in \{1, ..., 20\}$, this graph compares the mean time taken to compute the maximal subsemigroups of 100 random regular Rees 0-matrix semigroups $\mathscr{M}^0[G; I, \Lambda; P]$ with the mean time taken to compute the maximal cliques in the duals of their Graham-Houghton graphs. As the matrix dimension increases, it seems that the time spent computing maximal cliques approaches the total time for maximal subsemigroups.
Chapter 5

Maximal subsemigroups of finite transformation and diagram monoids

5.1 Introduction

In this chapter, we describe and count the maximal subsemigroups of several families of wellknown finite transformation monoids and diagram monoids, using the tools provided in Chapter 4. The research underpinning this chapter was conducted in collaboration with James East, Jitender Kumar, and James D. Mitchell. A paper containing the results of this chapter is available on arXiv [44]; a shorter version of that paper, containing the most significant subset of these results, is published in the *Journal of Algebra* [45]. The research presented in this chapter is included with the permission of the co-authors.

Having obtained various theoretical results in Chapter 4 that concern the maximal subsemigroups of an arbitrary finite semigroup or monoid, a natural continuation of this work is to apply these results and techniques to study the maximal subsemigroups of particular semigroups.

The problem of describing or counting the maximal subsemigroups of certain transformation semigroups and monoids has been studied extensively in the literature. Much of this research has been led by I. Dimitrova, V. H. Fernandes, and co-authors; see [25-31,56,69], and the references therein. The techniques used to prove these results were typically rather disparate, and did not lend themselves well to generalisation. However, we can use the framework described in Chapter 4 in order to recover these results in a unified way, and moreover, to prove new results. Our strategy also allows us to clearly see the common features in the maximal subsemigroups of similar monoids. We especially make use of the tools of Section 4.5, which concern the maximal subsemigroups of an arbitrary finite monoid.

The same techniques can also be applied to describe the maximal subsemigroups of many other finite semigroups and monoids. To further demonstrate the utility of this framework, we describe the maximal subsemigroups of various families of diagram monoids. Diagram monoids are submonoids of the partition monoid, and they are currently well-studied in the literature; see [34, 39, 40, 42, 43, 46, 51, 70, 94, 98]. However, with the exception of those of the dual symmetric inverse monoid [94], the maximal subsemigroups of these diagram monoids had not been described in the literature.

Some of the results in this chapter were obtained with the help of the SEMIGROUPS package [101] for GAP [58], using the tools for computing maximal subsemigroups described in Section 4.6.1. More specifically, by applying the function MaximalSubsemigroups to several finite monoids from a particular family, it was often straightforward to analyse the results and make a conjecture about their general description. This chapter is structured as follows. In Section 5.1.1, we present some preliminary results concerning the maximal subgroups of the groups of units of the monoids that appear in the later sections. In Section 5.2, we study partial transformation monoids. We define various families of monoids that consist of order- and orientation-preserving and -reversing partial transformations in Section 5.2.1, and we provide some prerequisite information about them in Section 5.2.2. We describe the maximal subsemigroups of these monoids in Sections 5.2.3-5.2.8. We study monoids of partitions in Section 5.3. This section is structured similarly: in Sections 5.3.1 and 5.3.2, we define and present prerequisite results about the diagram monoids whose maximal subsemigroups we subsequently classify in Sections 5.3.3-5.3.7. In particular, we classify the maximal subsemigroups of the partition, Jones, Motzkin, and Brauer monoids, and several related monoids. In Section 5.4, we present Table 5.62. This table gives a summary of the information in this chapter. More specifically, for each of the families of monoids considered in this chapter, the table lists the corresponding sequence of the numbers of maximal subsemigroups, with references to the theorems where the maximal subsemigroups are described.

5.1.1 Maximal subgroups of cyclic, dihedral, and symmetric groups

It follows by Lemma 4.73 that, in order to classify the maximal subsemigroups of the monoids in this chapter, we require a description of the maximal subgroups of their groups of units. The groups of units that appear in this chapter are cyclic, dihedral, and symmetric groups. The conjugacy classes of maximal subgroups of the finite symmetric groups are described in [87] and counted in [88]; see [120, A066115] for the sequence of their number. In general, however, there is no known simple formula for the number of *all* maximal subgroups of a finite symmetric group. Throughout this chapter, we use the notation s_k to denote the number of maximal subsemigroups of the symmetric group of degree k [120, A290138]. For the maximal subgroups of the cyclic and dihedral groups, we present the following well-known results. The proofs of these results require descriptions of the subgroups of the cyclic and dihedral groups. These descriptions are straightforward to prove.

Lemma 5.1. Let $n \in \mathbb{N}$, $n \geq 2$, and let $G = \langle \alpha \rangle$ be a cyclic group of order n. The maximal subgroups of G are the subgroups $\langle \alpha^p \rangle$, for each prime divisor p of n. In particular, the total number of maximal subgroups G is the number of prime divisors of n.

Proof. The subgroups of G are $\langle \alpha^d \rangle$, for each positive divisor d of n, and $\langle \alpha^d \rangle \leq \langle \alpha^{d'} \rangle$ if and only if d' divides d. Therefore, the maximal subgroups of G are the subgroups $\langle \alpha^d \rangle$, where d > 1 is a divisor of n whose positive divisors are 1 and d: these are the prime divisors of n. \Box

Some of the families of transformation monoids considered in this chapter have trivial groups of units, or groups of units of order 2. In the first case, by Lemma 4.73, no further work is required. In the second case, since a group of order 2 is cyclic, it follows by Lemma 5.1 that its unique maximal subgroup is trivial, consisting of the identity element of the group.

The following lemma was stated incorrectly in [45, Lemma 2.5], but appears correctly below.

Lemma 5.2. Let $n \in \mathbb{N}$, $n \geq 3$, and let $G = \langle \sigma, \rho \rangle$ be a dihedral group of order 2n, where σ has order 2, ρ has order n, and $\sigma \rho = \rho^{-1}\sigma$. The maximal subgroups of G are the subgroups $\langle \rho \rangle$ and $\langle \rho^p, \sigma \rho^i \rangle$, for each prime divisor p of n and each integer $i \in \{0, \ldots, p-1\}$. In particular, the number of maximal subgroups of G is one more than the sum of the prime divisors of n.

Proof. The subgroups of G are $\langle \sigma^d \rangle$ and $\langle \sigma^d, \sigma \rho^i \rangle$, for each positive divisor d of n and each integer $i \in \{0, \ldots, d-1\}$. Any subgroup of the form $\langle \rho^d \rangle$ is contained in $\langle \rho \rangle$, which is maximal since its index in G is 2. The subgroup $\langle \rho^d, \sigma \rho^i \rangle$ is not contained in $\langle \rho \rangle$, and is contained in

 $\langle \rho^{d'}, \sigma \rho^{i'} \rangle$ if and only if d' divides d and i = i'. Therefore, the maximal subgroups of G in this second form are the subgroups $\langle \rho^d, \sigma \rho^i \rangle$, where d is a prime divisor of n and $0 \le i \le d-1$. \Box

5.2 Partial transformation monoids

In this section, we describe and count the maximal subsemigroups of several well-known families of partial transformation monoids. We define these monoids in Section 5.2.1, and present some prerequisite results and background information about them in Section 5.2.2; the main results are given in Sections 5.2.3–5.2.8. For certain families, the descriptions of their maximal subsemigroups, and the corresponding proofs, are alike. In such cases, we collect the results concerning these families, in order to highlight their similarities.

5.2.1 Definitions

As defined in Section 1.3.5, \mathcal{PT}_n is the monoid of all partial transformations of degree n under composition of partial functions; its submonoid \mathcal{T}_n consists of all transformations of degree n, and its inverse submonoid \mathcal{I}_n consists of all partial permutations of degree n. In Sections 5.2.3– 5.2.8, we describe and count the maximal subsemigroups of these monoids, as well as those of a collection of other submonoids of \mathcal{PT}_n , which we define here. These other monoids consist of partial transformations that preserve or reverse the usual order, or orientation, of $\{1, \ldots, n\}$.

Let $n \in \mathbb{N}$, and let α be a partial transformation of degree n. Clearly, there is a unique way to write dom(α) = $\{i_1, \ldots, i_k\}$, where $i_j \in \{1, \ldots, n\}$ for each j, and $i_1 < \cdots < i_k$. We say that α is order-preserving if $i_1\alpha \le \cdots \le i_k\alpha$, and order-reversing if $i_1\alpha \ge \cdots \ge i_k\alpha$. Similarly, we say that α is orientation-preserving if there exists at most one $l \in \{1, \ldots, k-1\}$ such that $i_l\alpha > i_{l+1}\alpha$, and α is orientation-reversing if there exists at most one $l \in \{1, \ldots, k-1\}$ such that $i_l\alpha < i_{l+1}\alpha$. In a certain informal sense, an orientation-preserving or -reversing partial transformations preserves or reverses the 'cyclic ordering' of its domain. Note that an order-preserving partial transformation is orientation-preserving, and similarly, an orderreversing partial transformation is orientation-reversing. Rules for the composition of such partial transformations are shown in Table 5.3.

0	Order-preserving	Orde	er-reversing	
Order-preserving Order-reversing	Order-preserving Order-reversing	Order-reversing Order-preserving ving Orientation-rever		
0	Orientation-preser	ving	Orientation-reversing	
Orientation-preserving Orientation-reversing	Orientation-preservin Orientation-reversing	g	Orientation-reversing Orientation-preserving	

Table 5.3: Rules describing the composition of order-preserving/-reversing partial transformations, and the composition of orientation-preserving/-reversing partial transformations. It follows that we may create subsemigroups of \mathcal{PT}_n that consist of such partial transformations.

Given these notions, and the rules described in Table 5.3, it is natural to define the largest submonoids of \mathcal{PT}_n that consist of such elements. These monoids are:

- $\mathcal{PO}_n = \{ \alpha \in \mathcal{PT}_n : \alpha \text{ is order-preserving} \},$
- $\mathcal{POD}_n = \{ \alpha \in \mathcal{PT}_n : \alpha \text{ is order-preserving or -reversing} \},$

- $\mathcal{POP}_n = \{ \alpha \in \mathcal{PT}_n : \alpha \text{ is orientation-preserving} \}, \text{ and }$
- $\mathcal{POR}_n = \{ \alpha \in \mathcal{PT}_n : \alpha \text{ is orientation-preserving or -reversing} \}.$

We also define their largest submonoids consisting of transformations, and their largest inverse submonoids consisting of partial permutations. That is, we define their intersections with \mathcal{T}_n :

- $\mathcal{O}_n = \mathcal{PO}_n \cap \mathcal{T}_n$,
- $\mathcal{OD}_n = \mathcal{POD}_n \cap \mathcal{T}_n$,
- $\mathcal{OP}_n = \mathcal{POP}_n \cap \mathcal{T}_n$, and
- $\mathcal{OR}_n = \mathcal{POR}_n \cap \mathcal{T}_n;$

and we define their intersections with \mathcal{I}_n :

- $\mathcal{POI}_n = \mathcal{PO}_n \cap \mathcal{I}_n$,
- $\mathcal{PODI}_n = \mathcal{POD}_n \cap \mathcal{I}_n$,
- $\mathcal{POPI}_n = \mathcal{POP}_n \cap \mathcal{I}_n$, and
- $\mathcal{PORI}_n = \mathcal{POR}_n \cap \mathcal{I}_n$.

These monoids have been studied extensively in the literature; see for example [28, 31], and the references therein, where the notation that we use for these monoids was introduced. In Sections 5.2.4–5.2.8, we classify the maximal subsemigroups of these monoids.

We may easily describe the permutations, and therefore the largest subgroups of S_n , that satisfy each of these properties. Recall that the symmetric group S_n is the group consisting of all permutations of degree n; S_n is the group of units of \mathcal{PT}_n , \mathcal{T}_n , and \mathcal{I}_n . The identity permutation of degree n, id_n, is the unique order-preserving permutation of degree n; it is the identity element of each of the monoids that we define above. There is also a unique order-reversing permutation of degree n, which we write in disjoint cycle notation as $(1 \ n)(2 \ n-1) \cdots (\lfloor (n+1)/2 \rfloor \lceil (n+1)/2 \rceil)$. A permutation is orientation-preserving if and only if it is a power of the n-cycle $(1 \ 2 \ \dots \ n)$, and an orientation-reversing permutation is the product of an orientation-preserving permutation with the unique order-reversing permutation. We define

$$\mathcal{C}_n = \langle (1 \ 2 \ \dots \ n) \rangle \tag{5.4}$$

to be the subgroup of S_n consisting of all orientation-preserving permutations; it is cyclic of order n, and so its maximal subgroups are given by Lemma 5.1. We also define

$$\mathcal{D}_n = \langle (1 \ 2 \ \dots \ n), \ (1 \ n)(2 \ n-1) \cdots (\lfloor (n+1)/2 \rfloor \ \lceil (n+1)/2 \rceil) \rangle.$$
(5.5)

to be the group of orientation-preserving or -reversing permutations of degree n. When $n \geq 3$, \mathcal{D}_n is dihedral of order 2n, and so its maximal subgroups are given by Lemma 5.2. Note that $\mathcal{C}_2 = \mathcal{D}_2 = \langle (1 \ 2) \rangle$.

Thus we may describe the groups of units of the monoids that we have defined. The groups of units of \mathcal{PO}_n , \mathcal{O}_n , and \mathcal{POI}_n are the trivial group $\{id_n\}$; the groups of units of \mathcal{POD}_n , \mathcal{OD}_n , and \mathcal{PODI}_n are $\langle (1 \ n)(2 \ n - 1) \cdots (\lfloor (n + 1)/2 \rfloor \lceil (n + 1)/2 \rceil) \rangle$; the groups of units of \mathcal{POP}_n , \mathcal{OP}_n , and \mathcal{POPI}_n are \mathcal{C}_n ; and the groups of units of \mathcal{POR}_n , \mathcal{OR}_n , and \mathcal{PORI}_n are \mathcal{D}_n .

The maximal subsemigroups of several of these monoids have previously been described in the literature, including those of \mathcal{OP}_n [28, Theorem 1.6], \mathcal{OR}_n [28, Theorem 2.6], \mathcal{POI}_n [56, Theorem 2], and \mathcal{PODI}_n [29, Theorem 4]. The maximal subsemigroups of \mathcal{PT}_n , \mathcal{T}_n , and \mathcal{I}_n are

well-known folklore. Furthermore, the maximal subsemigroups of the singular ideal $\mathcal{O}_n \setminus \{\mathrm{id}_n\}$ of \mathcal{O}_n were described in [25, Theorem 2], and those of the singular ideal $\mathcal{PO}_n \setminus \{\mathrm{id}_n\}$ of \mathcal{PO}_n were described in in [31, Theorem 1]. Additionally, the maximal subsemigroups of the singular ideal of \mathcal{OD}_n were found in [69, Theorem 2], but since the group of units of \mathcal{OD}_n is non-trivial, this is a fundamentally different problem than finding the maximal subsemigroups of \mathcal{OD}_n itself. Until the publication of the research detailed in this section (in [45]), the maximal subsemigroups of \mathcal{OD}_n , \mathcal{POP}_n , \mathcal{POPI}_n , \mathcal{POR}_n , and \mathcal{PORI}_n remained open problems.

Nevertheless, we describe the maximal subsemigroups of all the monoids from this section. The purpose of reproving the known results is to demonstrate that they may all be obtained in a largely unified manner, using the tools described in Chapter 4. Furthermore, the descriptions of the maximal subsemigroups of some of these monoids, such as those of \mathcal{PO}_n and \mathcal{POD}_n , and those of \mathcal{O}_n and \mathcal{OD}_n , are closely linked, and so it is instructive to present them together.

5.2.2 Preliminaries

Let $n \in \mathbb{N}$, $n \geq 2$. In order to describe the maximal subsemigroups of the submonoids of \mathcal{PT}_n from Section 5.2.1, we require some additional facts and notation. We first require a description of the Green's relations on \mathcal{PT}_n . These are given by:

- $\alpha \mathscr{L} \beta$ if and only if $\operatorname{im}(\alpha) = \operatorname{im}(\beta)$,
- $\alpha \mathscr{R} \beta$ if and only if $\ker(\alpha) = \ker(\beta)$, and
- $\alpha \not \subseteq \beta$ if and only if rank $(\alpha) = \operatorname{rank}(\beta)$,

for $\alpha, \beta \in \mathcal{PT}_n$; see [57, Theorem 4.5.1] for a proof. Note that $\ker(\alpha) = \ker(\beta)$ implies that $\operatorname{dom}(\alpha) = \operatorname{dom}(\beta)$, since the kernel of a partial transformation is defined to be an equivalence on its domain. Each of the submonoids of \mathcal{PT}_n defined in Section 5.2.1 is regular, and so it follows easily that the Green's relations on these monoids are characterised in the same way.

By Lemma 4.68, we also require generating sets for these monoids. Any order-reversing partial transformation α is the product of some order-preserving partial transformation and the permutation $\sigma = (1 \ n)(2 \ n - 1) \cdots (\lfloor (n + 1)/2 \rfloor \lceil (n + 1)/2 \rceil)$, since $\alpha = (\alpha \sigma^{-1})\sigma$, and $\alpha \sigma^{-1}$ is order-preserving. Similarly, any orientation-preserving partial transformation is the product of some order-preserving partial transformation with some permutation in C_n . The analogous statement holds for orientation-reversing partial transformations. Therefore, if S is any of the monoids of order- or orientation-preserving or -reversing partial transformations defined in Section 5.2.1, then S is generated by its units and its order-preserving elements. However, \mathcal{O}_n is generated by id_n and its idempotents of rank n - 1 [1], and \mathcal{POI}_n is generated by id_n and its elements of rank n - 1 [48, Lemma 2.7]. It follows that S is generated by its units and its order-preserving elements of rank n - 1. The monoids \mathcal{PT}_n , \mathcal{T}_n , and \mathcal{I}_n are generated by \mathcal{S}_n and their elements of rank n - 1 [57, Theorems 3.1.3–3.1.5].

Therefore, by Lemma 4.68, in order to describe the maximal subsemigroups of any of the monoids from Section 5.2.1, we must find those maximal subsemigroups that arise from the group of units, and those that arise from the \mathscr{J} -class containing elements of rank n-1. The results of Section 4.5.1 apply in the former case, those of Section 4.5.2 apply in the latter case.

To describe the maximal subsemigroups arising from the \mathcal{J} -class consisting of elements of rank n-1, we require the following notation for the Green's classes of this \mathcal{J} -class. We define

$$J_{n-1} = \left\{ \alpha \in \mathcal{PT}_n : \operatorname{rank}(\alpha) = n - 1 \right\}$$

to be the \mathscr{J} -class of \mathcal{PT}_n that consists of all partial transformations of rank n-1. A partial transformation of rank n-1 lacks exactly one element from its image, and is either a partial

permutation that also lacks exactly one element from its domain, or is a transformation with a unique non-trivial kernel class, which contains two points. Thus for distinct points $i, j \in \{1, \ldots, n\}$, we define the Green's \mathscr{L} - and \mathscr{R} -classes of J_{n-1} :

- $L_i = \{ \alpha \in J_{n-1} : i \notin im(\alpha) \}$, which is an \mathscr{L} -class of J_{n-1} ;
- $R_i = \{ \alpha \in J_{n-1} : i \notin \operatorname{dom}(\alpha) \}$, which is an \mathscr{R} -class of J_{n-1} of partial permutations; and
- $R_{\{i, j\}} = \{ \alpha \in J_{n-1} : (i, j) \in \ker(\alpha) \}$, which is an \mathscr{R} -class of J_{n-1} of transformations.

An \mathscr{H} -class of the form $L_i \cap R_j$ is a group if and only if i = j, and an \mathscr{H} -class of the form $L_i \cap R_{\{j,k\}}$ is a group if and only if $i \in \{j, k\}$ [57, Theorem 2.7.2].

We observe that the non-trivial kernel class of an order-preserving or -reversing transformation of rank n-1 has the form $\{i, i+1\}$ for some $i \in \{1, \ldots, n-1\}$, and that the non-trivial kernel class of an orientation-preserving or -reversing transformation of rank n-1 has the same form, or is equal to $\{1, n\}$. Any non-empty subset of $\{1, \ldots, n\}$ appears as the image of some partial transformation in each of the monoids defined in Section 5.2.1.

Let S be one of the submonoids of \mathcal{PT}_n defined in Section 5.2.1. By the preceding discussion, it follows that the set $J_{n-1} \cap S$ is a regular \mathscr{J} -class of S, that the \mathscr{L} -classes of $J_{n-1} \cap S$ are the sets of the form $L_i \cap S$, and that the \mathscr{R} -classes of $J_{n-1} \cap S$ are those non-empty sets of the forms $R_i \cap S$ and $R_{\{i,j\}} \cap S$, for distinct $i, j \in \{1, \ldots, n\}$.

Often, the principal obstacle to describing the maximal subsemigroups of S that arise from $J_{n-1} \cap S$ is to determine the maximal independent subsets of the graph $\Delta(S, J_{n-1} \cap S)$, which was introduced in Section 4.4.2; see also Section 4.5.2 for more information. In every case in this section, the relevant \mathscr{J} -class is $J_{n-1} \cap S$, and so we henceforth refer to this graph as simply $\Delta(S)$. Whenever we present a picture of $\Delta(S)$, such as in Figure 5.16, we label an \mathscr{L} -class as L_i , rather than as $L_i \cap S$, and so on, in order to avoid cluttering the image. This approach also has the advantage of highlighting the similarities between the graphs of related monoids, and allows us to more easily see that some graphs may be obtained as induced subgraphs of others.

To describe $\Delta(S)$, we must calculate the left and right actions of the group of units G of S on the \mathscr{R} -classes and on the \mathscr{L} -classes of $J_{n-1} \cap S$, respectively. The following lemma shows that these actions correspond to natural right actions of G on points and pairs in $\{1, \ldots, n\}$.

Lemma 5.6. Let S be one of the submonoids of \mathcal{PT}_n defined in Section 5.2.1, and let G be its group of units.

- (a) Let $\Omega \subseteq \{1, \ldots, n\}$. Then Ω is an orbit of G on $\{1, \ldots, n\}$ if and only if $\{L_i \cap S : i \in \Omega\}$ is an orbit of the right action of G on $(J_{n-1} \cap S)/\mathscr{L}$.
- (b) Let $\Omega \subseteq X = \{i : R_i \cap S \neq \emptyset\}$. Then Ω is an orbit of G on X if and only if $\{R_i \cap S : i \in \Omega\}$ is an orbit of the left action of G on the \mathscr{R} -classes of $J_{n-1} \cap S$ that contain partial permutations.
- (c) Let $\Omega \subseteq Y = \{\{i, j\} : i \neq j, R_{\{i, j\}} \cap S \neq \emptyset\}$. Then Ω is an orbit of G on Y if and only if $\{R_{\{i, j\}} : \{i, j\} \in \Omega\}$ is an orbit of the left action of G on the \mathscr{R} -classes of $J_{n-1} \cap S$ that contain transformations.

Proof. To prove part (a), let $i, j \in \{1, ..., n\}$ and let $\alpha \in L_i \cap S$. Then, with respect to the right action of G on $\{1, ..., n\}$,

i and *j* lie in the same orbit \Leftrightarrow $i\sigma = j$ for some $\sigma \in G$

$$\Leftrightarrow \quad j \notin \operatorname{im}(\alpha \sigma) \text{ for some } \sigma \in G$$

- $\Leftrightarrow \quad \alpha \sigma \in L_j \cap S \text{ for some } \sigma \in G$
- $\Leftrightarrow (L_i \cap S)\sigma = L_j \cap S \text{ for some } \sigma \in G.$

The proof of part (b) is similar to the proof of part (a). To prove part (c), let $\{i, j\}, \{k, l\} \in Y$ be arbitrary, and let $\alpha \in R_{\{i, j\}} \cap S$. Then, with respect to the right action of G on Y,

$$\{i, j\} \text{ and } \{k, l\} \text{ lie in the same orbit} \iff \{k\sigma, l\sigma\} = \{i, j\} \text{ for some } \sigma \in G \\ \Leftrightarrow k\sigma\alpha = l\sigma\alpha \text{ for some } \sigma \in G \\ \Leftrightarrow \sigma\alpha \in R_{\{k, l\}} \cap S \text{ for some } \sigma \in G \\ \Leftrightarrow \sigma(R_{\{i, j\}} \cap S) = R_{\{k, l\}} \cap S \text{ for some } \sigma \in G. \square$$

The right actions of the groups $\{\mathrm{id}_n\}$, $\langle (1 \ n)(2 \ n-1)\cdots(\lfloor (n+1)/2 \rfloor \lceil (n+1)/2 \rceil) \rangle$, \mathcal{C}_n , \mathcal{D}_n , and \mathcal{S}_n (as defined in Section 5.2.1) on the points, and pairs of points in $\{1, \ldots, n\}$, are easy to understand, since a permutation of degree n is defined in terms of its action on $\{1, \ldots, n\}$. Therefore, if S is one of the monoids mentioned in Section 5.2.1, then the actions of its group of units on the \mathscr{R} - and \mathscr{L} -classes of $J_{n-1} \cap S$ may be readily understood via Lemma 5.6.

5.2.3 $\mathcal{PT}_n, \mathcal{T}_n, \text{ and } \mathcal{I}_n$

To start, we describe and count the maximal subsemigroups of \mathcal{PT}_n , \mathcal{T}_n , and \mathcal{I}_n . Of course, the maximal subsemigroups of the these monoids are well-known folklore, and so the following result is by no means novel. However, we include this theorem and the subsequent discussion for completeness, and to begin to show the utility of the results presented in Chapter 4.

Theorem 5.7. Let $n \in \mathbb{N}$, $n \geq 2$, be arbitrary. The following hold.

- (a) The maximal subsemigroups of \mathcal{PT}_n are the sets:
 - (i) $(\mathcal{PT}_n \setminus S_n) \cup U$, where U is a maximal subgroup of S_n (type (M1));
 - (ii) $\mathcal{PT}_n \setminus \{ \alpha \in \mathcal{T}_n : \operatorname{rank}(\alpha) = n 1 \}$ (type (M4)); and
 - (iii) $\mathcal{PT}_n \setminus \{ \alpha \in \mathcal{I}_n : \operatorname{rank}(\alpha) = n 1 \}$ (type (M4)).
- (b) The maximal subsemigroups of \mathcal{T}_n are the sets:
 - (i) $(\mathcal{T}_n \setminus \mathcal{S}_n) \cup U$, where U is a maximal subgroup of \mathcal{S}_n (type (M1)); and
 - (ii) $\mathcal{T}_n \setminus \{ \alpha \in \mathcal{T}_n : \operatorname{rank}(\alpha) = n 1 \}$ (type (M5)).
- (c) The maximal subsemigroups of \mathcal{I}_n are the sets:
 - (i) $(\mathcal{I}_n \setminus \mathcal{S}_n) \cup U$, where U is a maximal subgroup of \mathcal{S}_n (type (M1)); and
 - (ii) $\mathcal{I}_n \setminus \{ \alpha \in \mathcal{I}_n : \operatorname{rank}(\alpha) = n 1 \}$ (type (M5)).

In particular, for $n \ge 2$, there are $s_n + 2$ maximal subsemigroups of \mathcal{PT}_n , and $s_n + 1$ maximal subsemigroups of both \mathcal{T}_n and \mathcal{I}_n .

	2	3	4	5	6	7	8	9	10	11	12	n
\mathcal{PT}_n	3	6	10	24	55	186	355	1378	3979	363906	396500	$s_n + 2$
$\mathcal{T}_n, \ \mathcal{I}_n$	2	5	9	23	54	185	354	1377	3978	363905	396499	$s_n + 1$

Table 5.8: The numbers of maximal subsemigroups of the monoids \mathcal{PT}_n , \mathcal{T}_n , and \mathcal{I}_n , for n = 2, ..., 12, along with the general formulae. Recall that s_n denotes the number of maximal subgroups of the symmetric group \mathcal{S}_n [120, A290138]. See Theorem 5.7 for a description of the maximal subsemigroups of these monoids.

In the proof of this theorem, we make use of the following well-known facts [57, Theorems 3.1.3–3.1.5]. A subset of \mathcal{T}_n or \mathcal{I}_n is a generating set if and only if the subset contains a generating set for \mathcal{S}_n , and any transformation or partial permutation, respectively, of rank n-1. In addition, a subset of \mathcal{PT}_n is a generating set for \mathcal{PT}_n if and only if it contains a generating set for \mathcal{T}_n and a generating set for \mathcal{I}_n .

Proof. Since S_n is the group of units of \mathcal{PT}_n , \mathcal{T}_n , and \mathcal{I}_n , it follows by Lemma 4.73 that, in each case, the maximal subsemigroups arising from the group of units are those described in the statement of the theorem, and that there are s_n of them. It remains to describe the maximal subsemigroups of each monoid that arise from the \mathscr{J} -class of rank n-1. By Corollary 4.74, the unique maximal subsemigroup of \mathcal{T}_n or \mathcal{I}_n to arise from its \mathscr{J} -class of rank n-1 has type (M5). By using Lemma 4.8 with k = 2, $X_1 = J_{n-1} \cap \mathcal{T}_n$, and $X_2 = J_{n-1} \cap \mathcal{I}_n$, we find that the two maximal subsemigroups of \mathcal{PT}_n arising from J_{n-1} are those described.

For each of these monoids, the descriptions of the maximal subsemigroups that arise from the \mathscr{J} -class of rank n-1 can also be obtained with the techniques and results from Section 4.5.2. Certainly, these monoids are each generated by their units and their idempotents of rank n-1, and so by Lemma 4.49(b), no maximal subsemigroups of type (M1) arise from this \mathscr{J} -class. The right action of \mathcal{S}_n on the \mathscr{L} -classes of J_{n-1} given by right multiplication is transitive. On the other hand, there are two orbits under the left action of \mathcal{S}_n on the \mathscr{R} -classes of J_{n-1} : one of these contains the \mathscr{R} -classes of transformations, and the other contains the \mathscr{R} -classes of partial permutations. Thus we may deduce the definitions of $\Delta(\mathcal{PT}_n)$, $\Delta(\mathcal{T}_n)$, and $\Delta(\mathcal{I}_n)$; pictures of these graphs are given in Figures 5.9, 5.10, and 5.11, respectively.



Figure 5.9: The graph $\Delta(\mathcal{PT}_n)$. The group of units \mathcal{S}_n acts transitively on the \mathscr{L} -classes of J_{n-1} , and so there is a single vertex corresponding to this orbit. However, \mathcal{S}_n has two orbits under its action on the \mathscr{R} -classes of J_{n-1} , and so there are two vertices that contain \mathscr{R} -classes.



Figure 5.10: The graph $\Delta(\mathcal{T}_n)$. Note that this graph may be obtained as the induced subgraph of $\Delta(\mathcal{PT}_n)$ on its orbits of Green's classes that contain transformations.



Figure 5.11: The graph $\Delta(\mathcal{I}_n)$. Note that this graph may be obtained as the induced subgraph of $\Delta(\mathcal{PT}_n)$ on its orbits of Green's classes that contain partial permutations.

By Lemma 4.80, there are no maximal subsemigroups of any of these monoids of types (M2) or (M3), and there are no maximal subsemigroups of \mathcal{T}_n or \mathcal{I}_n of type (M4). Therefore, by Corollary 4.75, the only maximal subsemigroup of \mathcal{T}_n or \mathcal{I}_n to arise from the \mathscr{J} -class of rank n-1 has type (M5). However, by the dual of Proposition 4.79, there are two maximal subsemigroups of \mathcal{PT}_n of type (M4) that arise from J_{n-1} , which are formed by removing either the partial permutations, or the transformations, of degree n-1.

5.2.4 \mathcal{PO}_n and \mathcal{POD}_n

The maximal subsemigroups of \mathcal{PO}_n were described in [31, Theorem 1]. The maximal subsemigroups of \mathcal{POD}_n had not been described in the literature prior to [45]. Using our approach, we find that the maximal subsemigroups of \mathcal{POD}_n are closely linked to those of \mathcal{PO}_n .

The main results of this section are the following theorems; to state them, we first reiterate the notation and background information given in Section 5.2.2. Let $n \in \mathbb{N}$, $n \geq 2$, be arbitrary, and let $S \in \{\mathcal{PO}_n, \mathcal{POD}_n\}$. Then $J_{n-1} \cap S$ is a regular \mathscr{J} -class of S, the \mathscr{L} -classes of $J_{n-1} \cap S$ are the sets $\{L_i \cap S : i \in \{1, \ldots, n\}\}$, and the \mathscr{R} -classes are the sets $\{R_i \cap S : i \in \{1, \ldots, n\}\}$ and $\{R_{\{i, i+1\}} \cap S : i \in \{1, \ldots, n-1\}\}$. We also use the fact that \mathcal{PO}_n is idempotent generated [60, Theorem 3.13], and that \mathcal{POD}_n is generated by \mathcal{PO}_n and the orderreversing permutation $(1 \ n)(2 \ n-1) \cdots (\lfloor (n+1)/2 \rfloor \lceil (n+1)/2 \rceil).$

Theorem 5.12. Let $n \in \mathbb{N}$, $n \geq 2$, be arbitrary. The maximal subsemigroups of \mathcal{PO}_n are:

- (a) $\mathcal{PO}_n \setminus \{ \mathrm{id}_n \}$ (type (M5));
- (b) the union of $\mathcal{PO}_n \setminus J_{n-1}$ and the union of the sets in

$$\{L_i \cap \mathcal{PO}_n : i \in A\} \cup \{R_i \cap \mathcal{PO}_n : i \notin A\} \cup \{R_{\{i, i+1\}} \cap \mathcal{PO}_n : i, i+1 \notin A\},\$$

where A is any non-empty proper subset of $\{1, \ldots, n\}$ (type (M2)) and;

(c) $\mathcal{PO}_n \setminus R$, where R is any \mathscr{R} -class in $J_{n-1} \cap \mathcal{PO}_n$ (type (M4)).

In particular, for $n \in \mathbb{N}$, there are $2^n + 2n - 2$ maximal subsemigroups of \mathcal{PO}_n .

Theorem 5.13. Let $n \in \mathbb{N}$, $n \geq 2$, be arbitrary. The maximal subsemigroups of \mathcal{POD}_n are:

- (a) $\mathcal{POD}_n \setminus \{(1 \ n)(2 \ n-1)\cdots(\lfloor (n+1)/2 \rfloor \lceil (n+1)/2 \rceil)\}$ (type (M1));
- (b) the union of $\mathcal{POD}_n \setminus J_{n-1}$ and the union of the sets in

$$\{ (L_i \cup L_{n-i+1}) \cap \mathcal{POD}_n : i \in A \} \cup \{ (R_i \cup R_{n-i+1}) \cap \mathcal{POD}_n : i \notin A \} \\ \cup \{ (R_{\{i, i+1\}} \cup R_{\{n-i, n-i+1\}}) \cap \mathcal{POD}_n : i, i+1 \notin A \},$$

where A is any non-empty proper subset of $\{1, \ldots, \lceil n/2 \rceil\}$ (type (M2));

- (c) $\mathcal{POD}_n \setminus (R_i \cup R_{n-i+1})$, for $i \in \{1, \ldots, \lceil n/2 \rceil\}$ (type (M4)); and
- (d) $\mathcal{POD}_n \setminus (R_{\{i, i+1\}} \cup R_{\{n-i, n-i+1\}}), \text{ for } i \in \{1, \dots, \lfloor n/2 \rfloor\} \text{ (type (M4))}.$

In particular, for $n \in \mathbb{N}$, there are $2^{\lceil n/2 \rceil} + n - 1$ maximal subsemigroups of \mathcal{POD}_n .

The most substantial step involved in the proofs of these theorems is the description of the maximal independent subsets of $\Delta(\mathcal{PO}_n)$ and $\Delta(\mathcal{POD}_n)$.

Since \mathcal{PO}_n has a trivial group of units, the orbits of \mathscr{L} -classes and \mathscr{R} -classes of $J_{n-1} \cap \mathcal{PO}_n$ are singletons, and $\Delta(\mathcal{PO}_n)$ is isomorphic to the Graham-Houghton graph of the principal factor of $J_{n-1} \cap \mathcal{PO}_n$. A picture of $\Delta(\mathcal{PO}_n)$ is shown in Figure 5.15.

	1	2	3	4	5	6	7	8	9	10	11	12	n
\mathcal{PO}_n	2	6	12	22	40	74	140	270	528	1042	2068	4118	$2^n + 2n - 2$
\mathcal{POD}_n	2	3	6	7	12	13	22	23	40	41	74	75	$2^{\lceil n/2\rceil} + n - 1$

Table 5.14: The numbers of maximal subsemigroups of the monoids \mathcal{PO}_n and \mathcal{POD}_n , for $n = 1, \ldots, 12$, along with the general formulae. Note that the number of maximal subsemigroups of \mathcal{PO}_n equals the number of maximal subsemigroups of \mathcal{POD}_{2n-1} . See Theorems 5.12 and 5.13 for descriptions of the maximal subsemigroups of \mathcal{POD}_n and \mathcal{POD}_n , respectively.



Figure 5.15: A visualisation of the graph $\Delta(\mathcal{PO}_n)$. The group of units of \mathcal{PO}_n is trivial, and so $\Delta(\mathcal{PO}_n)$ is isomorphic to the Graham-Houghton graph of the principal factor of $\mathcal{PO}_n \cap J_{n-1}$. The vertices of degree 1 in $\Delta(\mathcal{PO}_n)$ are those orbits whose \mathscr{R} -classes are composed of partial permutations.



Figure 5.16: The graph $\Delta(\mathcal{POD}_n)$, when *n* is odd. Note that the orbits $\{L_{(n+1)/2}\}$ and $\{R_{(n+1)/2}\}$ are singletons, and that $\Delta(\mathcal{POD}_n) \cong \Delta(\mathcal{PO}_{(n+1)/2})$; see Figure 5.15. The vertices of degree 1 in the graph are those orbits that contain \mathscr{R} -classes composed of partial permutations.



Figure 5.17: The graph $\Delta(\mathcal{POD}_n)$, when *n* is even. Note that the orbit $\{R_{\{n/2, n/2+1\}}\}$ is a singleton; this orbit and the orbits of \mathscr{R} -classes of partial permutations are the vertices of degree 1 in the graph.

The group of units of \mathcal{POD}_n is generated by $(1 \ n)(2 \ n-1)\cdots(\lfloor (n+1)/2 \rfloor \lceil (n+1)/2 \rceil)$, the unique order-reversing permutation. Since $\langle (1 \ n)(2 \ n-1)\cdots(\lfloor (n+1)/2 \rfloor \lceil (n+1)/2 \rceil) \rangle$ has $\lceil n/2 \rceil$ orbits on the set $\{1, \ldots, n\}$, it follows by Lemma 5.6 that there are $\lceil n/2 \rceil$ corresponding orbits of \mathscr{L} -classes and $\lceil n/2 \rceil$ orbits of \mathscr{R} -classes of partial permutations. Furthermore, there are $\lfloor n/2 \rfloor$

orbits of $\langle (1 n)(2 n-1)\cdots (\lfloor (n+1)/2 \rfloor \lceil (n+1)/2 \rceil) \rangle$ on the set $\{\{i, i+1\} : i \in \{1, \ldots, n-1\}\}$, and these orbits correspond to $\lfloor n/2 \rfloor$ orbits of \mathscr{R} -classes of transformations. Due to the division by 2 in each case, the properties of $\Delta(\mathcal{POD}_n)$ depend on the parity of n. A picture of $\Delta(\mathcal{POD}_n)$ is shown in Figure 5.16 for odd n, and in Figure 5.17 for even n; see these pictures for a description of the edges of this graph. When n is odd, the graph $\Delta(\mathcal{POD}_n)$ is isomorphic to $\Delta(\mathcal{PO}_{\lceil n/2 \rceil})$.

Given descriptions of $\Delta(\mathcal{PO}_n)$ and $\Delta(\mathcal{POD}_n)$, we establish the following lemmas.

Lemma 5.18. Let K be any subset of then vertices of the graph $\Delta(\mathcal{PO}_n)$. Then K is a maximal independent subset of $\Delta(\mathcal{PO}_n)$ if and only if

$$K = \left\{ \{L_i \cap \mathcal{PO}_n\} : i \in A \right\} \cup \left\{ \{R_i \cap \mathcal{PO}_n\} : i \notin A \right\} \cup \left\{ \{R_{\{i, i+1\}} \cap \mathcal{PO}_n\} : i, i+1 \notin A \right\}$$

for some subset A of $\{1, \ldots, n\}$.

Proof. (\Rightarrow) Suppose that K is a maximal independent subset of $\Delta(\mathcal{PO}_n)$. There exists a subset $A \subseteq \{1, \ldots, n\}$ of indices such that $\{\{L_i \cap \mathcal{PO}_n\} : i \in A\}$ is the collection of \mathscr{L} -class vertices in K. Since a vertex of the form $\{R_i \cap \mathcal{PO}_n\}$ is adjacent in $\Delta(\mathcal{PO}_n)$ only to the vertex $\{L_i \cap \mathcal{PO}_n\}$, it follows by the maximality of K that $\{R_i \cap \mathcal{PO}_n\} \in K$ if and only if $i \notin A$. Similarly, since an orbit of the form $\{R_{\{i, i+1\}} \cap \mathcal{PO}_n\}$ is adjacent in $\Delta(\mathcal{PO}_n)$ only to the orbits $\{L_i \cap \mathcal{PO}_n\}$ and $\{L_{i+1} \cap \mathcal{PO}_n\}$, it follows that $\{R_{\{i, i+1\}} \cap \mathcal{PO}_n\} \in K$ if and only if $i \notin A$ and $i+1 \notin A$. We have considered all vertices of $\Delta(\mathcal{PO}_n)$, and so K has the required form.

 (\Leftarrow) It is easy to verify that K is a maximal independent subset of $\Delta(\mathcal{PO}_n)$.

The following lemma is very similar to Lemma 5.18, and is therefore stated without proof.

Lemma 5.19. Let K be any collection of vertices of the graph $\Delta(\mathcal{POD}_n)$. Then K is a maximal independent subset of $\Delta(\mathcal{POD}_n)$ if and only if K is equal to

$$\{ \{L_i \cap \mathcal{POD}_n, L_{n-i+1} \cap \mathcal{POD}_n\} : i \in A \} \cup \{ \{R_i \cap \mathcal{POD}_n, R_{n-i+1} \cap \mathcal{POD}_n\} : i \notin A \} \\ \cup \{ \{R_{\{i, i+1\}} \cap \mathcal{POD}_n, R_{\{n-i, n-i+1\}} \cap \mathcal{POD}_n\} : i, i+1 \notin A \},$$

for some subset A of $\{1, \ldots, \lceil n/2 \rceil\}$.

We may now prove the main results of this section.

Proof of Theorems 5.12 and 5.13. The group of units of \mathcal{PO}_n is trivial, and the group of units of \mathcal{POD}_n has order 2. By Lemma 4.73, the maximal subsemigroup that arises from the group of units in each instance is the one described.

Let $S \in \{\mathcal{PO}_n, \mathcal{POD}_n\}$. Since \mathcal{POD}_n is generated by \mathcal{PO}_n and the order-reversing permutation, and since \mathcal{PO}_n is idempotent generated, it follows by Lemma 4.49(b) that there are no maximal subsemigroups of type (M1) arising from $J_{n-1} \cap S$. It follows directly from Proposition 4.77, and Lemmas 5.18 and 5.19, that the maximal subsemigroups of type (M2) are those described in the theorem. There exist vertices of degree 1 in $\Delta(S)$: these are orbits of \mathscr{R} -classes, and each orbit of \mathscr{L} -classes is adjacent to a vertex of degree 1. Thus by Proposition 4.79 and its dual, there are no maximal subsemigroups of type (M3) arising from S, but each orbit of \mathscr{R} -classes can be removed to provide a maximal subsemigroup of type (M4). There are 2n - 1 maximal subsemigroups of this type for \mathcal{PO}_n , and there are n of this type for \mathcal{POD}_n . By Corollary 4.75, there is no maximal subsemigroup of S of type (M5).

5.2.5 \mathcal{O}_n and \mathcal{OD}_n

The maximal subsemigroups of the singular ideal $\mathcal{O}_n \setminus \{ \mathrm{id}_n \}$ of \mathcal{O}_n were incorrectly described and counted in [130]: the formula $n^2 - 2n + 2$ presented for the number of maximal subsemigroups is correct for $2 \leq n \leq 5$, but gives only a lower bound when $n \geq 6$; see Table 5.28. A correct description, although no formula for the sequence of the numbers, was later given in [25, Theorem 2]. Since the group of units of \mathcal{O}_n is trivial, the maximal subsemigroups of its singular ideal correspond in an obvious way to the maximal subsemigroups of \mathcal{O}_n . Furthermore, the maximal subsemigroups of the singular ideal $\mathcal{OD}_n \setminus \langle (1 n)(2 n-1) \cdots (\lfloor (n+1)/2 \rfloor \lceil (n+1)/2 \rceil) \rangle$ of \mathcal{OD}_n were described in [69, Theorem 2]. However, the group of units of \mathcal{OD}_n acts on \mathcal{OD}_n non-trivially, and so the correspondence between the maximal subsemigroups of the singular ideal and the maximal subsemigroups of \mathcal{OD}_n itself is lost. Thus [69] solves an essentially different problem than the description of the maximal subsemigroups of \mathcal{OD}_n .

We summarise the information about \mathcal{O}_n and \mathcal{OD}_n from Section 5.2.2. Let $n \in \mathbb{N}, n \geq 2$, and let $S \in \{\mathcal{O}_n, \mathcal{OD}_n\}$. Then S is a regular monoid, the set $J_{n-1} \cap S$ is a regular \mathscr{J} -class of S, the set of \mathscr{L} -classes of $J_{n-1} \cap S$ is $\{L_i \cap S : i \in \{1, \ldots, n\}\}$, and the set of \mathscr{R} -classes of $J_{n-1} \cap S$ is $\{R_{\{i, i+1\}} \cap S : i \in \{1, \ldots, n-1\}\}$. Additionally, \mathcal{O}_n is generated by its idempotents of rank n-1 [1], and \mathcal{OD}_n is generated by \mathcal{O}_n and the order-reversing permutation [49].

Since $\mathcal{O}_n = \mathcal{PO}_n \cap \mathcal{T}_n$, we may identify a Green's class of $J_{n-1} \cap \mathcal{O}_n$ with the Green's class of $J_{n-1} \cap \mathcal{PO}_n$ that contains it, so that $L_i \cap \mathcal{O}_n$ corresponds with $L_i \cap \mathcal{PO}_n$, and $R_{\{i, i+1\}} \cap \mathcal{O}_n$ corresponds with $R_{\{i, i+1\}} \cap \mathcal{PO}_n$. In this way, we obtain $\Delta(\mathcal{O}_n)$ as the induced subgraph of $\Delta(\mathcal{PO}_n)$ on those orbits of Green's classes that contain transformations. In other words, the definition of $\Delta(\mathcal{O}_n)$ is contained in that of $\Delta(\mathcal{PO}_n)$. The graph $\Delta(\mathcal{O}_n)$ contains *n* singleton orbits of \mathscr{L} -classes, and n-1 singleton orbits of \mathscr{R} -classes; see Figure 5.20 for a picture.

In a similar way, $\Delta(\mathcal{OD}_n)$ may be obtained as the induced subgraph of $\Delta(\mathcal{POD}_n)$ on its orbits of Green's classes that contain transformations. This graph contains *n* vertices; a picture of $\Delta(\mathcal{OD}_n)$ is shown in Figure 5.21 for odd *n*, and in Figure 5.17 for even *n*.



Figure 5.20: The graph $\Delta(\mathcal{O}_n)$, which is a path graph of order 2n - 1. Since the group of units of \mathcal{O}_n is trivial, $\Delta(\mathcal{O}_n)$ is isomorphic to the Graham-Houghton graph of the principal factor of $J_{n-1} \cap \mathcal{O}_n$. Note that $\Delta(\mathcal{O}_n)$ may be obtained as the induced subgraph of $\Delta(\mathcal{PO}_n)$ on its orbits of Green's classes that contain transformations; see Figure 5.15. The vertices of degree 1 in $\Delta(\mathcal{O}_n)$ are $\{L_1\}$ and $\{L_n\}$.



Figure 5.21: The graph $\Delta(\mathcal{OD}_n)$, when *n* is odd. This is a path graph of order *n*. Note that $\Delta(\mathcal{OD}_n) \cong \Delta(\mathcal{O}_{n+1/2})$, and that $\Delta(\mathcal{OD}_n)$ may be obtained as the induced subgraph of $\Delta(\mathcal{POD}_n)$ on its orbits of Green's classes that contain transformations; see Figure 5.16. The vertices of degree 1 in $\Delta(\mathcal{OD}_n)$ are $\{L_1, L_n\}$ and $\{L_{(n+1)/2}\}$.



Figure 5.22: The graph $\Delta(\mathcal{OD}_n)$, when *n* is even. This is a path graph of order *n*. Note that $\Delta(\mathcal{OD}_n)$ may be obtained as the induced subgraph of $\Delta(\mathcal{POD}_n)$ on its orbits of Green's classes that contain transformations; see Figure 5.17. The vertices of degree 1 in $\Delta(\mathcal{OD}_n)$ are $\{L_1, L_n\}$ and $\{R_{\{n/2, n/2+1\}}\}$.

For $k \in \mathbb{N}$, define the *path graph* of order k to be the graph with vertex set $\{1, \ldots, k\}$, and edge set

$$\{\{i, i+1\} : i \in \{1, \ldots, k-1\}\}$$

The vertices of degree 1 in the path graph of order k are the end-points, 1 and k. It is easy to see that $\Delta(\mathcal{O}_n)$ is isomorphic to the path graph of order 2n - 1, via the isomorphism that maps the orbit $\{L_i \cap \mathcal{O}_n\}$ to the vertex 2i - 1, and maps the orbit $\{R_{\{i, i+1\}} \cap \mathcal{O}_n\}$ to the vertex 2i. Similarly, $\Delta(\mathcal{OD}_n)$ is isomorphic to the path graph of order n. We can describe and count the number of maximal independent subsets of a path graph, and therefore the maximal independent subsets of $\Delta(\mathcal{O}_n)$ and $\Delta(\mathcal{OD}_n)$, by using the following results.

Lemma 5.23. Let $n \in \mathbb{N}$, $n \geq 2$, be arbitrary, let Γ be the path graph of order n, and let U be a subset of the vertices of Γ . Then U is a maximal independent subset of Γ if and only if the following conditions hold:

- (a) the least vertex in U is either 1 or 2; and
- (b) for each $i \in U \cap \{1, ..., n-1\}, i+1 \notin U$; and
- (c) for each $i \in U \cap \{1, \ldots, n-2\}$, exactly one of i+2 and i+3 is contained in U.

Proof. Since vertices in Γ are adjacent if and only if they are consecutive, U is an independent subset of Γ if and only if (b) holds. It is easy to verify that an independent subset U of Γ is maximal when conditions (a) and (c) hold. Conversely, if U satisfies (b) but contains neither 1 nor 2, then $U \cup \{1\}$ is an independent subset properly containing U, and U is not maximal. Similarly, suppose that U satisfies (b) and contains some $i \in \{1, \ldots, n-2\}$, but contains neither i + 2 nor i + 3. Then since U also does not contain i + 1, it follows that $U \cup \{i + 2\}$ is an independent subset properly containing U, and U is not maximal. Thus, if U is a maximal independent subset of Γ , then conditions (a) and (c) hold.

There are two special maximal independent subsets of a path graph: the subset of all even vertices, and the subset of all odd vertices. These maximal independent subsets correspond to the subset of all orbits of \mathscr{L} -classes, and the subset of all orbits of \mathscr{R} -classes, of $\Delta(\mathcal{O}_n)$ and $\Delta(\mathcal{OD}_n)$. These are the unique maximal independent subsets that do not give rise to maximal subsemigroups of \mathcal{O}_n and \mathcal{OD}_n of type (M2): see Proposition 4.77 and Corollary 4.78.

Corollary 5.24. The number of maximal independent subsets of the path graph is counted by the sequence $(A_n)_{n \in \mathbb{N}}$, defined by

$$A_1 = 1, \ A_2 = A_3 = 2, \ and \ A_n = A_{n-2} + A_{n-3} \ for \ n \ge 4.$$
 (5.25)

Note that A_n is the $(n+6)^{\text{th}}$ term of the Padovan sequence [120, A000931], which satisfies the same recurrence, with different initial values.

Proof. For $n \in \mathbb{N}$, define Γ_n to be the path graph of order n, and let f(n) denote the number of maximal independent subsets of Γ_n . It is straightforward to verify that f(1) = 1, and f(2) = f(3) = 2, so suppose that $n \geq 4$. By Lemma 5.23, if U is a maximal independent subset of Γ_{n-3} , then $U \cup \{n-1\}$ is a maximal independent subset of Γ_n , and if U is a maximal independent subset of Γ_{n-2} , then $U \cup \{n\}$ is a maximal independent subset of Γ_n . Thus distinct maximal independent subsets of Γ_{n-3} and Γ_{n-2} give rise to distinct maximal independent subsets of Γ_n , and $f(n) \geq f(n-2) + f(n-3)$. Conversely, if U is a maximal independent subset of Γ_n , then by Lemma 5.23, U contains either n-1 or n. If $n-1 \in U$, then $n-2 \notin U$, which implies that $U \setminus \{n-1\}$ is a maximal independent subset of Γ_{n-3} . Otherwise, $U \setminus \{n\}$ is a maximal independent subset of Γ_{n-2} . Thus $f(n) \leq f(n-2) + f(n-3)$, and f(n) = A(n). \Box

We may now describe and count the maximal subsemigroups of \mathcal{O}_n and \mathcal{OD}_n .

Theorem 5.26. Let $n \in \mathbb{N}$, $n \geq 2$, be arbitrary. The maximal subsemigroups of \mathcal{O}_n are:

- (a) $\mathcal{O}_n \setminus \{ \mathrm{id}_n \}$ (type (M5));
- (b) the union of $\mathcal{O}_n \setminus J_{n-1}$ and the union of the Green's classes in

$$\{L_{(i+1)/2} \cap \mathcal{O}_n : i \in A, \ i \ is \ odd\} \cup \{R_{\{i/2, (i/2)+1\}} \cap \mathcal{O}_n : i \in A, \ i \ is \ even\},\$$

where A is a maximal independent subset of the path graph of order 2n - 1 that contains both odd and even numbers, as described in Lemma 5.23 (type (M2));

- (c) $\mathcal{O}_n \setminus L$, where L is any \mathscr{L} -class in $J_{n-1} \cap \mathcal{O}_n$ (type (M3)); and
- (d) $\mathcal{O}_n \setminus R_{\{i, i+1\}}$, where $i \in \{2, ..., n-2\}$ (type (M4)).

In particular, for $n \ge 3$ there are $A_{2n-1} + 2n - 4$ maximal subsemigroups of \mathcal{O}_n , where A_{2n-1} is as defined in (5.25).

Theorem 5.27. Let $n \in \mathbb{N}$, $n \geq 3$, be arbitrary. The maximal subsemigroups of \mathcal{OD}_n are:

- (a) $\mathcal{OD}_n \setminus \{(1 \ n)(2 \ n-1) \cdots (\lfloor (n+1)/2 \rfloor \lceil (n+1)/2 \rceil)\}$ (type (M1));
- (b) the union of $\mathcal{OD}_n \setminus J_{n-1}$ and the union of the sets in

$$\{ (L_{(i+1)/2} \cup L_{n+1-(i+1)/2}) \cap \mathcal{OD}_n : i \in A, \ i \ is \ odd \} \\ \cup \{ (R_{\{i/2, (i/2)+1\}} \cup R_{\{n-(i/2), n+1-(i/2)\}}) \cap \mathcal{OD}_n : i \in A, \ i \ is \ even \},$$

where A is a maximal independent subset of the path graph of order n that contains both odd and even numbers, as described in Lemma 5.23 (type (M2));

(c)
$$\mathcal{OD}_n \setminus (L_i \cup L_{n-i+1})$$
, where
$$\begin{cases} i \in \{1, \dots, (n+1)/2\} & \text{if } n \text{ is odd,} \\ i \in \{1, \dots, n/2 - 1\} & \text{if } n \text{ is even} \end{cases}$$
 (type (M3)); and

(d)
$$\mathcal{OD}_n \setminus (R_{\{i,i+1\}} \cup R_{\{n-i,n-i+1\}}), where \begin{cases} i \in \{2, \dots, (n-3)/2\} & \text{if } n \text{ is odd,} \\ i \in \{2, \dots, n/2\} & \text{if } n \text{ is even} \end{cases}$$
(type (M4)).

In particular, for $n \ge 4$, there are $A_n + n - 3$ maximal subsemigroups of \mathcal{OD}_n , where A_n is as defined in (5.25).

	1	2	3	4	5	6	7	8	9	10	11	12	n
\mathcal{O}_n	1	3	6	11	18	29	47	77	128	216	369	636	$A_{2n-1} + 2n - 4$
\mathcal{OD}_n	1	2	3	4	6	8	11	14	18	23	29	37	$A_n + n - 3$

Table 5.28: The numbers of maximal subsemigroups of the monoids \mathcal{O}_n and \mathcal{OD}_n , for $n = 1, \ldots, 12$, along with the general formulae. The sequence $(A_n)_{n \in \mathbb{N}}$ is defined in (5.25). Note that the number of maximal subsemigroups of \mathcal{O}_n equals the number of maximal subsemigroups of \mathcal{OD}_{2n-1} . See Theorems 5.26 and 5.27 for descriptions of the maximal subsemigroups of \mathcal{O}_n and \mathcal{OD}_n , respectively.

Proof of Theorems 5.26 and 5.27. The group of units of \mathcal{O}_n is trivial, and the group of units of \mathcal{OD}_n has order 2. By Lemma 4.73, the maximal subsemigroup that arises from the group of units in each instance is the one described.

Since \mathcal{O}_n is generated by id_n and its idempotents of rank n-1, and since \mathcal{OD}_n is generated by these idempotents and its non-identity permutation, it follows by Lemma 4.49(b) that there are no maximal subsemigroups of type (M1) arising from the \mathscr{J} -class consisting of elements of rank n-1. Since $\Delta(\mathcal{O}_n)$ and $\Delta(\mathcal{OD}_n)$ are path graphs of length 2n-1 and n, respectively, it follows by Proposition 4.77 and Lemma 5.23 that the maximal subsemigroups of type (M2) are those described in the theorems. By Corollary 4.78 and Corollary 5.24, the number of maximal subsemigroups of type (M2) is $A_{2n-1} - 2$ for \mathcal{O}_n , and $A_n - 2$ for \mathcal{OD}_n .

Let $S \in \{\mathcal{O}_n, \mathcal{OD}_n\}$. By Corollary 4.75, and since $n \geq 4$, there is no maximal subsemigroup of S of type (M5). To describe the maximal subsemigroups of types (M3) and (M4), it suffices to identify the two vertices of $\Delta(S)$ that are adjacent to the end-points of $\Delta(S)$. From this, the description of the maximal subsemigroups of types (M3) and (M4) follows from Proposition 4.79 and its dual. In particular, the total number of both types of maximal subsemigroups is two less than the number of vertices of $\Delta(S)$.

5.2.6 \mathcal{POI}_n and \mathcal{PODI}_n

The maximal subsemigroups of \mathcal{POI}_n are described and counted in [56, Theorem 2], and those of \mathcal{PODI}_n are described and counted in [29, Theorem 4]. Extensive additional information about \mathcal{POI}_n may be found in [48, 56]. We reprove these results in this section, for completeness.

We recall the following information from Section 5.2.2. Let $n \in \mathbb{N}$, $n \geq 2$, be arbitrary, and let $S \in \{\mathcal{POI}_n, \mathcal{PODI}_n\}$. Then S is an inverse monoid, and $J_{n-1} \cap S$ is a \mathscr{J} -class of S. Since \mathcal{POI}_n is generated by id_n and its elements of rank n-1 [48, Lemma 2.7], and since \mathcal{PODI}_n is generated by \mathcal{POI}_n and the permutation $(1 \ n)(2 \ n-1)\cdots(\lfloor (n+1)/2 \rfloor \lceil (n+1)/2 \rceil)$, it follows by Lemma 4.68 that the maximal subsemigroups of S arise from its group of units and from its \mathscr{J} -class $J_{n-1} \cap S$. By definition, $\mathcal{POI}_n = \mathcal{PO}_n \cap \mathcal{I}_n$, and $\mathcal{PODI}_n = \mathcal{POD}_n \cap \mathcal{I}_n$, and so given the description of the Green's classes of \mathcal{PO}_n and \mathcal{PODI}_n from Section 5.2.4, it follows that

$$(J_{n-1} \cap S)/\mathscr{L} = \{L_i \cap S : i \in \{1, \dots, n\}\}, \text{ and } (J_{n-1} \cap S)/\mathscr{R} = \{R_i \cap S : i \in \{1, \dots, n\}\}.$$

In Theorems 5.29 and 5.30, we describe the maximal subsemigroups of these inverse monoids.

Theorem 5.29. Let $n \in \mathbb{N}$, $n \geq 2$, be arbitrary. The maximal subsemigroups of \mathcal{POI}_n are:

- (a) $\mathcal{POI}_n \setminus \{ \mathrm{id}_n \}$ (type (M5)); and
- (b) the union of $\mathcal{POI}_n \setminus J_{n-1}$ and the union of

$$\{L_i \cap \mathcal{POI}_n : i \in A\} \cup \{R_i \cap \mathcal{POI}_n : i \notin A\},\$$

where A is any non-empty proper subset of $\{1, \ldots, n\}$ (type (M2)).

In particular, for $n \geq 2$, there are $2^n - 1$ maximal subsemigroups of \mathcal{POI}_n .

Proof. The inverse monoid \mathcal{POI}_n is \mathscr{H} -trivial, and so by Lemma 4.49(a) there are no maximal subsemigroups of type (M1), and by Lemma 4.73, the unique maximal subsemigroup arising from the group of units is formed by removing id_n. Since the group of units of \mathcal{POI}_n is trivial, its action on the \mathscr{L} -classes of $J_{n-1} \cap \mathcal{POI}_n$ is also trivial. Therefore, there are *n* singleton orbits. The \mathscr{R} -class $R_i \cap \mathcal{POI}_n$ is equal to $(L_i \cap \mathcal{POI}_n)^{-1}$, and so by Corollary 4.82, the maximal subsemigroups of \mathcal{POI}_n arising from $J_{n-1} \cap \mathcal{POI}_n$ are those of type (M2) described in the theorem.

Theorem 5.30. Let $n \in \mathbb{N}$, $n \geq 3$, be arbitrary. For $i, j \in \{1, ..., n\}$, define $\alpha_{i,j}$ to be the order-preserving partial permutation with domain $\{1, ..., n\} \setminus \{i\}$ and image $\{1, ..., n\} \setminus \{j\}$, and define $\beta_{i,j}$ to be the order-reversing partial permutation with this domain and image. The maximal subsemigroups of \mathcal{PODI}_n are:

- (a) $\mathcal{PODI}_n \setminus \{(1 \ n)(2 \ n-1) \cdots (\lfloor (n+1)/2 \rfloor \lceil (n+1)/2 \rceil)\}$ (type (M1));
- (b) $(\mathcal{PODI}_n \setminus J_{n-1}) \cup I_A$, where n is even,

$$\begin{split} I_{A} &= \big\{ \alpha_{i,j}, \ \beta_{i,n-j+1}, \ \beta_{n-i+1,j}, \ \alpha_{n-i+1,n-j+1} : i, j \in A \text{ or } i, j \notin A \big\} \\ & \cup \big\{ \beta_{i,j}, \ \alpha_{i,n-j+1}, \ \alpha_{n-i+1,j}, \ \beta_{n-i+1,n-j+1} : i \in A, \ j \notin A \text{ or } i \notin A, \ j \in A \big\}, \end{split}$$

and A is any subset of $\{2, \ldots, n/2\}$ (type (M1)); and

(c) the union of $\mathcal{PODI}_n \setminus J_{n-1}$ and the union of the sets in

$$\{(L_i \cup L_{n-i+1}) \cap \mathcal{PODI}_n : i \in A\} \cup \{(R_i \cup R_{n-i+1}) \cap \mathcal{PODI}_n : i \notin A\},\$$

where A is any non-empty proper subset of $\{1, \ldots, \lceil n/2 \rceil\}$ (type (M2)).

In particular, for $n \ge 3$ there are $3 \cdot 2^{(n/2)-1} - 1$ maximal subsemigroups of \mathcal{PODI}_n when n is even, and $2^{(n+1)/2} - 1$ when n is odd.

	1	2	3	4	5	6	7	8	9	10	n	
\mathcal{POI}_n	2	3	7	15	31	63	127	255	511	1023	$2^n - 1$	
\mathcal{PODI}_n	2	2	3	5	7	11	15	23	31	47	$\begin{cases} 3 \cdot 2^{(n/2)-1} - 1\\ 2^{(n+1)/2} - 1 \end{cases}$	even n odd n

Table 5.31: The numbers of maximal subsemigroups of the monoids \mathcal{POI}_n and \mathcal{PODI}_n , for n = 1, ..., 10, along with the general formulae. Note that the number of maximal subsemigroups of \mathcal{POI}_n equals the number of maximal subsemigroups of \mathcal{POI}_{2n-1} . See Theorems 5.29 and 5.30 for descriptions of the maximal subsemigroups of \mathcal{POI}_n and \mathcal{PODI}_n , respectively.

Proof. Since the group of units of \mathcal{PODI}_n is generated by the permutation $(1 \ n)(2 \ n - 1) \cdots (\lfloor (n+1)/2 \rfloor \lceil (n+1)/2 \rceil)$, which has order two, it follows by Lemma 4.73 that the unique maximal subsemigroup arising from the group of units of \mathcal{PODI}_n is formed by removing this non-identity permutation.

The graph $\Delta(\mathcal{PODI}_n)$ may be obtained as the induced subgraph of $\Delta(\mathcal{POD}_n)$ on those orbits of Green's classes that contain partial permutations. In particular, the orbits of the group of units $\langle (1 \ n)(2 \ n-1) \cdots (\lfloor (n+1)/2 \rfloor \lceil (n+1)/2 \rceil) \rangle$ on the \mathscr{L} -classes of $J_{n-1} \cap \mathcal{PODI}_n$ are the sets $\{L_i \cap \mathcal{PODI}_n, L_{n-i+1} \cap \mathcal{PODI}_n\}$, for each $i \in \{1, \ldots, \lceil n/2 \rceil\}$. By Corollary 4.82, the maximal subsemigroups of \mathcal{PODI}_n that arise from $J_{n-1} \cap \mathcal{PODI}_n$ are those $2^{\lceil n/2 \rceil} - 2$ maximal subsemigroups of type (M2) described in the statement of the theorem, as well as any maximal subsemigroups of type (M1).

It remains to describe the maximal subsemigroups of \mathcal{PODI}_n that arise from $J_{n-1} \cap \mathcal{PODI}_n$ and that have type (M1); suppose that M is such a maximal subsemigroup. By Proposition 4.10(b), the intersection of M with each \mathscr{H} -class of $J_{n-1} \cap \mathcal{PODI}_n$ is non-empty, and each of these intersections has some common size, $q \geq 1$. Since an \mathscr{H} -class in $J_{n-1} \cap \mathcal{PODI}_n$ contains two elements, and M is a proper subsemigroup of \mathcal{PODI}_n that lacks only elements from $J_{n-1} \cap \mathcal{PODI}_n$, it follows that q = 1. In other words, the intersection of M with each \mathscr{H} -class of $J_{n-1} \cap \mathcal{PODI}_n$ contains a single element. For $i, j \in \{1, \ldots, n\}$, let $\delta_{i,j}$ denote the unique element of M that is contained in the \mathscr{H} -class $\mathcal{PODI}_n \cap (L_i \cap R_j) = \{\alpha_{i,j}, \beta_{i,j}\}$ of \mathcal{PODI}_n . In other words, $M \cap (L_i \cap R_j) = \{\delta_{i,j}\}$.

Since M contains γ , it follows that $\delta_{i,j} \in M$ if and only if $\delta_{i,j} \cdot \gamma \in M$, $\gamma \cdot \delta_{i,j} \in M$, and $\gamma \cdot \delta_{i,j} \in M$. In particular, $\alpha_{i,j} \in M$ if and only if $\alpha_{i,j}$, $\beta_{i,n-j+1}$, $\beta_{n-i+1,j}$, $\alpha_{n-i+1,n-j+1} \in M$, and $\beta_{i,j} \in M$ if and only if $\beta_{i,j}$, $\alpha_{i,n-j+1}$, $\alpha_{n-i+1,j}$, $\beta_{n-i+1,n-j+1} \in M$. For odd n, this leads to the statement that $\alpha_{(n+1)/2,(n+1)/2} \in M$ if and only if $\beta_{(n+1)/2,(n+1)/2} \in M$, contradicting the fact that M contains exactly 1 element in each \mathscr{H} -class of $J_{n-1} \cap \mathcal{PODI}_n$. Hence n is even.

Given these observations, in order to describe M, it suffices to specify $\delta_{i,j}$ for each $i, j \in \{1, \ldots, n/2\}$. Indeed, our description can be even more concise. We observe that $\delta_{i,i} = \alpha_{i,i}$, since M contains every idempotent of \mathcal{PODI}_n . This implies that $\delta_{i,j} \cdot \delta_{j,i} = \delta_{i,i} = \alpha_{i,i}$, and so $\delta_{i,j} = \alpha_{i,j}$ if and only if $\delta_{j,i} = \alpha_{j,i}$. Furthermore, $\delta_{i,j} = \delta_{i,1}\delta_{1,j}$. Thus, to specify $\delta_{i,j}$ for each $i, j \in \{1, \ldots, n\}$, it suffices to specify $\delta_{1,i}$ for each $i \in \{2, \ldots, n/2\}$. Let $A = \{i \in \{2, \ldots, n/2\} : \delta_{1,i} = \beta_{1,j}\}$. A routine calculation shows that $M = (\mathcal{PODI}_n \setminus J_{n-1}) \cup I_A$, where I_A is the set defined in the statement of the theorem.

Conversely, for an even number $n \geq 4$ and a subset $A \subseteq \{2, \ldots, n/2\}$, it is tedious, but routine, to verify that $(\mathcal{PODI}_n \setminus J_{n-1}) \cup I_A$ is a subsemigroup of \mathcal{PODI}_n ; by construction, it intersects every \mathscr{H} -class of \mathcal{PODI}_n non-trivially. Any maximal subsemigroup of \mathcal{PODI}_n that contains $(\mathcal{PODI}_n \setminus J_{n-1}) \cup I_A$ has type (M1), and so by the preceding arguments, we see that it is equal to $(\mathcal{PODI}_n \setminus J_{n-1}) \cup I_A$. Thus $(\mathcal{PODI}_n \setminus J_{n-1}) \cup I_A$ is a maximal subsemigroup of \mathcal{PODI}_n of type (M1).

For two subsets $A, A' \subseteq \{2, \ldots, n/2\}$, it is clear from the definitions that $I_A = I_{A'}$ if and only if A = A'. Thus there are $2^{(n/2)-1}$ maximal subsemigroups of type (M1) when n is even, and none when n is odd.

5.2.7 $\mathcal{POP}_n, \mathcal{POR}_n, \mathcal{OP}_n, \text{ and } \mathcal{OR}_n$

The maximal subsemigroups of \mathcal{OP}_n were described in [28, Theorem 1.6], and the maximal subsemigroups of \mathcal{OR}_n were described in [28, Theorem 2.6]. Prior to the publication of [45], no description of the maximal subsemigroups of \mathcal{POP}_n and \mathcal{POR}_n had appeared in the literature.

To state the results of this section, we require the following notation and information from Section 5.2.2. Let $n \in \mathbb{N}$, $n \geq 2$, be arbitrary, and let $S \in \{\mathcal{POP}_n, \mathcal{POR}_n, \mathcal{OP}_n, \mathcal{OR}_n\}$. Then $J_{n-1} \cap S$ is a regular \mathscr{J} -class of S. Since \mathcal{O}_n and \mathcal{PO}_n are each generated by $\{id_n\}$ and their idempotents of rank n-1 [60], it follows that S is generated by its units and idempotents of rank n-1.

The \mathscr{L} -classes contained in $J_{n-1} \cap S$ are the intersections $L_i \cap S$, for each $i \in \{1, \ldots, n\}$. If

 $S \in \{\mathcal{POP}_n, \mathcal{POR}_n\}$, then the \mathscr{R} -classes of $J_{n-1} \cap S$ are the sets $R_i \cap S$, for each $i \in \{1, \ldots, n\}$, and $R_{\{i,i+1\}} \cap S$, for each $i \in \{1, \ldots, n-1\}$, along with the set $R_{\{1,n\}} \cap S$. If $S \in \{\mathcal{OP}_n, \mathcal{OR}_n\}$, then $S \leq \mathcal{T}_n$, and so the \mathscr{R} -classes of $J_{n-1} \cap S$ are the sets $R_{\{i,i+1\}} \cap S$ for each $i \in \{1, \ldots, n-1\}$ and $R_{\{1,n\}} \cap S$. The group of units of \mathcal{POP}_n and \mathcal{OP}_n is \mathcal{C}_n , defined in (5.4), and the group of units of \mathcal{POR}_n and \mathcal{OR}_n is \mathcal{D}_n , defined in (5.5).

The following theorems are the main results of this section.

Theorem 5.32. Let $n \in \mathbb{N}$, $n \geq 3$, be arbitrary, let $S \in \{\mathcal{POP}_n, \mathcal{POR}_n\}$, and let G be the group of units of S. The maximal subsemigroups of S are:

- (a) $(S \setminus G) \cup U$, where U is a maximal subgroup of G (type (M1)); and
- (b) $S \setminus \{ \alpha \in S \cap \mathcal{T}_n : \operatorname{rank}(\alpha) = n 1 \}$ (type (M4)); and
- (c) $S \setminus \{ \alpha \in S \cap \mathcal{I}_n : \operatorname{rank}(\alpha) = n 1 \}$ (type (M4)).

In particular, for $n \ge 3$, there are $|\mathbb{P}_n| + 2$ maximal subsemigroups of \mathcal{POP}_n , and $3 + \sum_{p \in \mathbb{P}_n} p$ maximal subsemigroups of \mathcal{POR}_n , where \mathbb{P}_n is the set of primes that divide n.

Proof. By Lemmas 4.73, 5.1, and 5.2, the maximal subsemigroups that arise from the group of units are those described in the statement of the theorem; there are $|\mathbb{P}_n|$ such maximal subsemigroups of \mathcal{POP}_n , and $1 + \sum_{p \in \mathbb{P}_n} p$ such maximal subsemigroups of \mathcal{POR}_n .

Let $S \in \{\mathcal{POP}_n, \mathcal{POR}_n\}$, and let G be the group of units of S. Since \mathcal{PO}_n is idempotent generated, and $S = \langle \mathcal{PO}_n, G \rangle$, it follows by Lemma 4.49(b) that there are no maximal subsemigroups of type (M1) arising from $J_{n-1} \cap S$.

The remainder of the proof is similar to the discussion in Section 5.2.3 after the proof of Theorem 5.7. The group of units G of S acts transitively on the \mathscr{L} -classes of $J_{n-1} \cap S$, and so there are no maximal subsemigroups of types (M2) and (M3) by Lemma 4.80. On the other hand, G has two orbits on the set of \mathscr{R} -classes of $J_{n-1} \cap S$: it transitively permutes the \mathscr{R} -classes of transformations, and it transitively permutes the \mathscr{R} -classes of partial permutations. By the dual of Proposition 4.79, the two maximal subsemigroups of S of type (M4) are found by removing either the partial permutations, or the transformations, of rank n-1. By Corollary 4.75, there is no maximal subsemigroup of type (M5).

Theorem 5.33. Let $n \in \mathbb{N}$, $n \geq 3$, be arbitrary, let $S \in \{\mathcal{OP}_n, \mathcal{OR}_n\}$, and let G be the group of units of S. The maximal subsemigroups of S are:

- (a) $(S \setminus G) \cup U$, where U is a maximal subgroup of G (type (M1)); and
- (b) $S \setminus \{ \alpha \in S : \operatorname{rank}(\alpha) = n 1 \}$ (type (M5)).

In particular, for $n \geq 3$, there are $|\mathbb{P}_n| + 1$ maximal subsemigroups of \mathcal{OP}_n , and $2 + \sum_{p \in \mathbb{P}_n} p$ maximal subsemigroups of \mathcal{OR}_n , where \mathbb{P}_n is the set of primes that divide n.

Proof. Let $S \in \{\mathcal{OP}_n, \mathcal{OR}_n\}$, and let G be the group of units of S. The description and number of the maximal subsemigroups arising from G follows by the same arguments that were used in the proof of Theorem 5.32. Let $\alpha \in J_{n-1} \cap S$ be arbitrary, and let ε be an arbitrary idempotent of \mathcal{O}_n of rank n-1. Since G acts transitively on the \mathscr{L} - and \mathscr{R} -classes of $J_{n-1} \cap S$, there exist permutations $\sigma, \tau \in G$ such that $\sigma \alpha \tau \in H^S_{\varepsilon}$. Therefore $\varepsilon = (\sigma \alpha \tau)^k$ for some $k \in \mathbb{N}$. Since ε was chosen arbitrarily, it follows that $\langle G, \alpha \rangle$ contains every idempotent of S of rank n-1. But Sis generated by its units and idempotents of rank n-1, and so $S = \langle G, \alpha \rangle$. The description of the remaining maximal subsemigroup follows by Corollary 4.74.

	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	n
\mathcal{POP}_n	2	3	3	3	3	4	3	3	3	4	3	4	3	4	4	$ \mathbb{P}_n + 2$
\mathcal{OP}_n	1	2	2	2	2	3	2	2	2	3	2	3	2	3	3	$ \mathbb{P}_n + 1$
\mathcal{POR}_n	2	3	6	5	8	8	10	5	6	10	14	8	16	12	11	$3 + \sum_{p \in \mathbb{P}_n} p$
\mathcal{OR}_n	1	2	5	4	7	7	9	4	5	9	13	7	15	11	10	$2 + \sum_{p \in \mathbb{P}_n}^{r} p$

Table 5.34: The numbers of maximal subsemigroups of the monoids \mathcal{POP}_n , \mathcal{OP}_n , \mathcal{POR}_n , and \mathcal{OR}_n , for n = 1, ..., 15, along with the general formulae. Recall that \mathbb{P}_n denotes the set of primes that divide n. See Theorems 5.32 and 5.33 for descriptions of the maximal subsemigroups of these monoids.

5.2.8 \mathcal{POPI}_n and \mathcal{PORI}_n

The maximal subsemigroups of the inverse monoids \mathcal{POPI}_n and \mathcal{PORI}_n exhibit maximal subsemigroups of type (M1) arising from a \mathscr{J} -class covered by the group of units, and to which we can apply Proposition 4.85.

Let $n \in \mathbb{N}$, $n \geq 2$, be arbitrary, and let $S \in \{\mathcal{POPI}_n, \mathcal{PORI}_n\}$. Then $J_{n-1} \cap S$ is a regular \mathscr{J} -class of S consisting of partial permutations, and S is generated by its units and its elements of this \mathscr{J} -class. By definition, $\mathcal{POPI}_n = \mathcal{POP}_n \cap \mathcal{I}_n$ and $\mathcal{PORI}_n = \mathcal{POR}_n \cap \mathcal{I}_n$. Therefore, the group of units of \mathcal{POPI}_n is \mathcal{C}_n and the group of units of \mathcal{PORI}_n is \mathcal{D}_n , and it follows from Section 5.2.7 that the \mathscr{L} -classes and \mathscr{R} -classes of $J_{n-1} \cap S$ are $\{L_i \cap S : i \in \{1, \ldots, n\}\}$ and $\{R_i \cap S : i \in \{1, \ldots, n\}\}$, respectively.

We describe the maximal subsemigroups of \mathcal{POPI}_n and \mathcal{PORI}_n in the following theorems.

Theorem 5.35. Let $n \in \mathbb{N}$, $n \geq 3$, be arbitrary, and define the partial permutation of degree n

$$\zeta_n = \begin{pmatrix} 1 & 2 & \cdots & n-2 & n-1 & n \\ 2 & 3 & \cdots & n-1 & 1 & - \end{pmatrix}.$$

For $k \in \mathbb{N}$, let \mathbb{P}_k denote the set of all primes that divide k. Then the maximal subsemigroups of \mathcal{POPI}_n are the sets:

- (a) $(\mathcal{POPI}_n \setminus \mathcal{C}_n) \cup U$, where U is a maximal subgroup of \mathcal{C}_n (type (M1)); and
- (b) $\langle \mathcal{POPI}_n \setminus J_{n-1}, \zeta_n^p \rangle$, for each $p \in \mathbb{P}_{n-1}$ (type (M1)).

In particular, for $n \geq 3$, there are $|\mathbb{P}_n| + |\mathbb{P}_{n-1}|$ maximal subsemigroups of \mathcal{POPI}_n .

Proof. By Lemmas 4.73 and 5.1, the $|\mathbb{P}_n|$ maximal subsemigroups arising from the group of units C_n are those stated in the theorem. Since \mathcal{POPI}_n is inverse, its idempotents are projections, and certainly C_n acts transitively on the \mathscr{L} -classes and \mathscr{R} -classes of $J_{n-1} \cap \mathcal{POPI}_n$. Define

$$H = H_{\zeta_n}^{\mathcal{POPI}_n} = \left\{ \alpha \in \mathcal{POPI}_n : \operatorname{dom}(\alpha) = \operatorname{im}(\alpha) = \{1, \dots, n-1\} \right\}.$$

Then H is a group \mathscr{H} -class in the \mathscr{J} -class $J_{n-1} \cap \mathcal{POPI}_n$. Note that H is a cyclic group of order n-1, generated by the partial permutation ζ_n , with identity $e = \zeta_n^{n-1}$. Since the conditions of Proposition 4.85 are satisfied, we may apply its conclusions. Therefore, the maximal subsemigroups that arise from $J_{n-1} \cap \mathcal{POPI}_n$ are the subsemigroups $\langle \mathcal{POPI}_n \setminus J_{n-1}, U \rangle$, for each maximal subgroup U of H that contains $e \cdot \operatorname{Stab}_{\mathcal{C}_n}(H)$ (see Definition 4.83), or $\mathcal{POPI}_n \setminus J_{n-1}$, if no such maximal subgroups exist. The setwise stabilizer $\operatorname{Stab}_{\mathcal{C}_n}(H)$ is equal to the pointwise stabilizer $\{\sigma \in \mathcal{C}_n : n\sigma = n\}$ of n in \mathcal{C}_n , which is trivial. Therefore, any maximal subgroup of H gives rise to a maximal subsemigroup of \mathcal{POPI}_n ; by Lemma 5.1, the maximal subgroups of H are $\langle \zeta_n^p \rangle$ for each $p \in \mathbb{P}_{n-1}$, and as $n \geq 3$, the result follows. \Box **Theorem 5.36.** Let $n \in \mathbb{N}$, $n \geq 4$, be arbitrary, and define the partial permutations

$$\zeta_n = \begin{pmatrix} 1 & 2 & \cdots & n-2 & n-1 & n \\ 2 & 3 & \cdots & n-1 & 1 & - \end{pmatrix}, \quad and \quad \tau_n = \begin{pmatrix} 1 & 2 & \cdots & n-1 & n \\ n-1 & n-2 & \cdots & 1 & - \end{pmatrix}$$

For $k \in \mathbb{N}$, let \mathbb{P}_k denote the set of all primes that divide k. Then the maximal subsemigroups of \mathcal{PORI}_n are the sets:

- (a) $(\mathcal{PORI}_n \setminus \mathcal{D}_n) \cup U$, where U is a maximal subgroup of \mathcal{D}_n (type ((M1))); and
- (b) $\langle \mathcal{PORI}_n \setminus J_{n-1}, \zeta_n^p, \tau_n \rangle$, for each $p \in \mathbb{P}_{n-1}$ (type (M1)).

In particular, for $n \ge 4$, there are $1 + |\mathbb{P}_{n-1}| + \sum_{p \in \mathbb{P}_n} p$ maximal subsemigroups of \mathcal{PORI}_n .

Proof. By Lemmas 4.73 and 5.2, and since $n \ge 4$, the description of the maximal subsemigroups arising from the group of units \mathcal{D}_n is as stated in the theorem, and there are $\sum_{p \in \mathbb{P}_n} p + 1$ such maximal subsemigroups.

Since \mathcal{PORI}_n is inverse, each of its idempotents is a projection, and \mathcal{D}_n acts transitively on the \mathscr{L} -classes and \mathscr{R} -classes of $J_{n-1} \cap \mathcal{PORI}_n$. Therefore we may use Proposition 4.85 to describe the maximal subsemigroups of \mathcal{PORI}_n that arise from $J_{n-1} \cap \mathcal{PORI}_n$. Define e to be the idempotent partial permutation with domain and image $\{1, \ldots, n-1\}$, and define

$$H = H_e^{\mathcal{PORI}_n} = \{ \alpha \in \mathcal{PORI}_n : \operatorname{dom}(\alpha) = \operatorname{im}(\alpha) = \{1, \dots, n-1\} \}.$$

Then *H* is a group \mathscr{H} -class of \mathcal{PORI}_n contained in $J_{n-1} \cap \mathcal{PORI}_n$. Note that since $n \geq 4$, *H* is a dihedral group of order 2(n-1), and it is generated by the partial permutations ζ_n and τ_n . An element of \mathcal{D}_n belongs to the setwise stabilizer $\operatorname{Stab}_{\mathcal{D}_n}(H)$ if and only if it fixes the point *n*. Thus $\operatorname{Stab}_{\mathcal{D}_n}(H)$ contains only the identity permutation id_n , which fixes every point in $\{1, \ldots, n\}$, and the permutation $(1 n - 1)(2 n - 2) \cdots (\lfloor n/2 \rfloor \lceil n/2 \rceil)$ that fixes *n* and reverses the order of $\{1, \ldots, n-1\}$. In particular,

$$e \cdot \operatorname{Stab}_{\mathcal{D}_n}(H) = \{e \cdot h : h \in \operatorname{Stab}_{\mathcal{D}_n}(H)\} = \{e, \tau_n\}$$

Since any subgroup of H contains e, it follows from Proposition 4.85 that the maximal subsemigroups arising from $\mathcal{PORI}_n \cap J_{n-1}$ are $\langle \mathcal{PORI}_n \setminus J_{n-1}, U \rangle$, for each maximal subgroup U of H that contains τ_n . By Lemma 5.2, the maximal subgroups of H are $\langle \zeta_n \rangle$ and the subgroups $\langle \zeta_n^p, \tau_n \zeta_n^i \rangle$, where $p \in \mathbb{P}_{n-1}$ and $0 \le i \le p-1$. Thus the maximal subgroups of H that contain τ_n are those in the latter form where i = 0. It follows that the maximal subsemigroups arising from $\mathcal{PORI}_n \cap J_{n-1}$ are those stated in the theorem, and that there are $|\mathbb{P}_{n-1}|$ such maximal subsemigroups.

	1	2	3	4	5	6	7	8	9	10	11	12	n
\mathcal{POPI}_n	2	2	2	2	2	3	3	2	2	3	3	3	$ \mathbb{P}_n + \mathbb{P}_{n-1} $
\mathcal{PORI}_n	2	2	5	4	7	$\overline{7}$	10	4	5	9	14	7	$1 + \mathbb{P}_{n-1} + \sum_{p \in \mathbb{P}_{-}} p$

Table 5.37: The numbers of maximal subsemigroups of the monoids \mathcal{POPI}_n and \mathcal{PORI}_n , for n = 1, ..., 12, along with the general formulae. Note that \mathbb{P}_n denotes the set of primes that divide n. See Theorems 5.35 and 5.36 for descriptions of the maximal subsemigroups of \mathcal{POPI}_n and \mathcal{PORI}_n , respectively.

5.3 Diagram monoids

In this section, we describe and count the maximal subsemigroups of the partition monoid, and several of its submonoids. We define these monoids in Section 5.3.1, and provide some additional necessary information about them in Section 5.3.2. We then classify the maximal subsemigroups of these monoids in Sections 5.3.3-5.3.7.

5.3.1 Definitions

Recall from Section 1.1 that an equivalence is a reflexive, symmetric, and transitive relation on a set, and recall that a partition of a set is a collection of non-empty disjoint subsets of that set whose union is the whole set. Since the equivalence classes of an equivalence form a partition, and vice versa, we may consider these notions to be interchangeable.

Let $n \in \mathbb{N}$ be arbitrary. A partition of degree n is an equivalence on the set

$$\{1, \ldots, n\} \cup \{1', \ldots, n'\}$$

The equivalence classes of a partition of degree n are called *blocks*, and a block is called *trans*verse if it contains points from both $\{1, \ldots, n\}$ and $\{1', \ldots, n'\}$. A *block bijection* is a partition of some degree whose blocks are all transverse, and a block bijection of degree n is *uniform* if each of its blocks contains an equal number of points of $\{1, \ldots, n\}$ and $\{1', \ldots, n'\}$.

Example 5.38. Define partitions γ and δ of degree 5 by

 $\gamma = \{\{1, 1'\}, \{2, 2', 4'\}, \{3\}, \{3', 5'\}, \{4, 5\}\}, \text{ and } \delta = \{\{1, 2, 2', 3', 5'\}, \{3, 1'\}, \{4, 5, 4'\}\}.$

There are five blocks in γ , only two of which are transverse, and δ has three blocks, which are all transverse. Therefore δ is a block bijection, but γ is not. However, δ is not a uniform block bijection, since, for example, its block $\{4, 5, 4'\}$ contains two points from $\{1, \ldots, 5\}$, but only one point from $\{1', \ldots, 5'\}$.

The topic of this thesis is semigroup theory, and so of course we require an associative operation on partitions of the same degree. Let $n \in \mathbb{N}$ be arbitrary, and let α and β be partitions of degree n. Calculating the product $\alpha\beta$ involves several steps, and requires three auxiliary partitions, each one being a partition of a different set. From α , we create a partition α^{\vee} of the set $\{1, \ldots, n\} \cup \{1'', \ldots, n''\}$, by replacing every occurrence of each i' in α by i''. Similarly, by replacing each i in β by i'', we obtain from β a partition β^{\wedge} of $\{1'', \ldots, n''\} \cup \{1', \ldots, n''\}$. We define $(\alpha\beta)^{\diamond}$ to be the smallest equivalence on $\{1, \ldots, n\} \cup \{1', \ldots, n'\} \cup \{1'', \ldots, n''\}$ that contains the relation $\alpha^{\vee} \cup \beta^{\wedge}$. In other words, since $\alpha^{\vee} \cup \beta^{\wedge}$ is reflexive and symmetric, $(\alpha\beta)^{\diamond}$ is the least transitive relation that contains $\alpha^{\vee} \cup \beta^{\wedge}$. Finally, the product $\alpha\beta$ is the intersection of $(\alpha\beta)^{\diamond}$ with $(\{1, \ldots, n\} \cup \{1', \ldots, n'\}) \times (\{1, \ldots, n\} \cup \{1', \ldots, n'\})$. See [70, Section 1] for a different approach to defining this multiplication.

This operation may be shown to be associative, and so the set \mathcal{P}_n of all partitions of degree n forms a semigroup under this operation. The partition $\{\{i, i'\} : i \in \{1, \ldots, n\}\}$ is the identity element of this semigroup. We call this element the *identity partition of degree n*. Therefore \mathcal{P}_n is a monoid, named the *partition monoid of degree n*. A *diagram monoid* is the name given to any submonoid of a partition monoid.

Example 5.39. Let γ and δ be the partitions of degree 5 defined in Example 5.38. We form the product $\delta\gamma$ as follows. First, we create the partitions

$$\delta^{\vee} = \{\{1, 2, 2'', 3'', 5''\}, \{3, 1''\}, \{4, 5, 4''\}\}, \text{ and} \\ \gamma^{\wedge} = \{\{1'', 1'\}, \{2'', 2', 4'\}, \{3''\}, \{3', 5'\}, \{4'', 5''\}\}.$$

The least equivalence on $\{1, \ldots, 5\} \cup \{1', \ldots, 5'\} \cup \{1'', \ldots, 5''\}$ that contains $\delta^{\vee} \cup \gamma^{\wedge}$ is

$$(\delta\gamma)^{\diamond} = \{\{1, 2, 4, 5, 2, 2', 4', 2'', 3'', 4'', 5''\}, \{3, 1', 1''\}, \{3', 5'\}, \{3''\}\};\$$

its intersection with $\{1, \ldots, 5\} \cup \{1', \ldots, 5'\}$ is the product

$$\delta\gamma = \{\{1, 2, 4, 5, 2', 4'\}, \{3, 1'\}, \{3', 5'\}\}.$$

Let $n \in \mathbb{N}$ and $\alpha \in \mathcal{P}_n$. We define α^* to be the partition of degree *n* created from α by replacing the point *i* by *i'* in the block in which it appears, and by replacing the point *i'* by *i*, for all $i \in \{1, \ldots, n\}$. For example, if $\alpha = \{\{1, 3'\}, \{3\}, \{2, 1', 2'\}\}$, a partition of degree 3, then $\alpha^* = \{\{1', 3\}, \{3'\}, \{2', 1, 2\}\}$. For arbitrary partitions $\alpha, \beta \in \mathcal{P}_n$, the following hold:

$$(\alpha^*)^* = \alpha, \ \alpha \alpha^* \alpha = \alpha, \ \text{and} \ (\alpha \beta)^* = \beta^* \alpha^*.$$

In particular, \mathcal{P}_n is a regular *-monoid, as defined in Section 1.3.3.

There is a canonical embedding $\psi : S_n \longrightarrow \mathcal{P}_n$, where a permutation $\alpha \in S_n$ is mapped to the partition $\{\{i, (i\alpha)'\} : i \in \{1, ..., n\}\}$. It is easy to verify that the units of \mathcal{P}_n are those partitions whose blocks have the form $\{i, j'\}$, for some $i, j \in \{1, ..., n\}$. It follows that the image of ψ is the group of units of \mathcal{P}_n . Therefore, throughout this section, we identify the permutation group S_n with the group of units of \mathcal{P}_n . In particular, we use disjoint cycle notation to refer to units in \mathcal{P}_n , and we use id_n to denote the identity element of \mathcal{P}_n .

In order to introduce the notions of planar and annular partitions, which are necessary to define some of the monoids whose maximal subsemigroups we classify, we define a canonical ordering

$$n' < (n-1)' < \dots < 1' < 1 < 2 < \dots < n \tag{5.40}$$

on the set $\{1, \ldots, n\} \cup \{1', \ldots, n'\}$. A partition $\alpha \in \mathcal{P}_n$ is *planar* if there do not exist distinct blocks A and X of α , and points $a, b \in A$ and $x, y \in X$, such that a < x < b < y. More generally, if $\rho_n = (1 \ 2 \ \ldots \ n)$, then a partition $\alpha \in \mathcal{P}_n$ is said to be *annular* if $\alpha = \rho_n^k \cdot \beta \cdot \rho_n^l$ for some planar partition $\beta \in \mathcal{P}_n$ and for some indices $k, l \in \mathbb{N}_0$. A partition may be represented graphically; planar and annular partitions have special representations in this context. For more information about this, see [4, 70].

Let $n \in \mathbb{N}$ be arbitrary. In Sections 5.3.3–5.3.7, we classify the maximal subsemigroups of \mathcal{P}_n , and the following submonoids:

- $\mathcal{PB}_n = \{ \alpha \in \mathcal{P}_n : \text{ each block of } \alpha \text{ contains 1 or 2 points} \}$, the *partial Brauer monoid of degree n*, introduced in [97];
- $\mathcal{B}_n = \{ \alpha \in \mathcal{P}_n : \text{ each block of } \alpha \text{ contains 2 points} \}$, the *Brauer monoid of degree n*, introduced in [97];
- $\mathcal{I}_n^* = \{ \alpha \in \mathcal{P}_n : \alpha \text{ is a block bijection} \}$, the dual symmetric inverse monoid of degree n, introduced in [52];
- $\mathfrak{F}_n = \{ \alpha \in \mathcal{P}_n : \alpha \text{ is a uniform block bijection} \}$, the uniform block bijection monoid of degree n, also known as the factorisable dual symmetric inverse monoid of degree n, see [50] for more details;
- $\mathcal{PP}_n = \{ \alpha \in \mathcal{P}_n : \alpha \text{ is planar} \}$, the planar partition monoid of degree n, introduced in [70];
- $\mathcal{M}_n = \{ \alpha \in \mathcal{PB}_n : \alpha \text{ is planar} \}$, the *Motzkin monoid of degree n*, see [9] for more details;

- $\mathcal{J}_n = \{ \alpha \in \mathcal{B}_n : \alpha \text{ is planar} \}$, the Jones monoid of degree n, also known as the Temperley-Lieb monoid, introduced in [84]; and
- $\mathcal{A}\mathcal{J}_n = \{ \alpha \in \mathcal{B}_n : \alpha \text{ is annular} \}$, the annular Jones monoid of degree n, introduced in [4].

Each of these diagram monoids is closed under the * operation, and is therefore a regular *semigroup. Furthermore, the monoids \mathcal{I}_n^* and \mathfrak{F}_n , which consist of block bijections, are inverse. The groups of units of \mathcal{PB}_n , \mathcal{B}_n , \mathcal{I}_n^* , and \mathfrak{F}_n are the symmetric group \mathcal{S}_n ; the group of units of \mathcal{AJ}_n is the cyclic group \mathcal{C}_n , as defined in (5.4); and the groups of units of \mathcal{PP}_n , \mathcal{M}_n , and \mathcal{J}_n are the trivial group {id_n}. See Figure 5.41 for a diagram showing the containment of these submonoids of \mathcal{P}_n , and their groups of units.

By [70], the planar partition monoid \mathcal{PP}_n is isomorphic to the Jones monoid \mathcal{J}_{2n} . Therefore, we will not separately determine the maximal subsemigroups of \mathcal{PP}_n , since their descriptions can be deduced from the results in Section 5.3.7. The maximal subsemigroups of \mathcal{I}_n^* were described in [94, Theorem 19], but we reprove this result for completeness. The maximal subsemigroups of the remaining monoids had not been classified until the publication of [45].



Figure 5.41: A Hasse diagram showing the containment of the diagram monoids defined in Section 5.3.1, along with their groups of units. We classify the maximal subsemigroups of these monoids in Sections 5.3.3–5.3.7.

5.3.2 Preliminaries

In order to prove the descriptions of the maximal subsemigroups of the diagram monoids defined in Section 5.3.1, we require some additional information about these monoids, which we introduce in this section.

Let $n \in \mathbb{N}$ be arbitrary. Define the function $\phi : \mathcal{PT}_n \longrightarrow \mathcal{P}_n$ such that if $\alpha \in \mathcal{PT}_n$, then the non-singleton blocks of $\alpha \phi$ are $\{i'\} \cup i\alpha^{-1}$, for each $i \in \operatorname{im}(\alpha)$. It is clear that ϕ is injective, since a partial transformation α is uniquely determined by its image and the pre-image $i\alpha^{-1}$ of each point i in $\operatorname{im}(\alpha)$. The restriction of ϕ to \mathcal{T}_n gives an embedding, as does the restriction of ϕ to \mathcal{I}_n . However, ϕ is not a homomorphism in general. Nevertheless, this injective function gives us a means of regarding partitions as generalisations of partial transformations. Furthermore, planar and annular partitions generalise order- and orientation-preserving partial transformations: for any $\alpha \in \mathcal{PT}_n$, α is order-preserving if and only if $\alpha \phi$ is planar, and α is orientation-preserving if and only if $\alpha \phi$ is annular. Let $\alpha \in \mathcal{P}_n$ be arbitrary. We require the following notions. The *rank* of α is the number of transverse blocks that it contains, and is denoted by $\operatorname{rank}(\alpha)$. The *kernel* of α , $\ker(\alpha)$, is the restriction of the equivalence α to $\{1, \ldots, n\}$. We also define $\operatorname{dom}(\alpha)$, the *domain* of α , to be the subset of $\{1, \ldots, n\}$ consisting of those points that are contained in a transverse block of α . Given these definitions, we define $\operatorname{coker}(\alpha) = \ker(\alpha^*)$ and $\operatorname{codom}(\alpha) = \operatorname{dom}(\alpha^*)$, the *cokernel* and *codomain* of α , respectively.

Example 5.42. Let ξ be the partition of degree 5 given by

$$\xi = \{\{1, 3\}, \{2, 4, 2'\}, \{5, 1', 3', 4'\}, \{5'\}\}.$$

Then the rank of ξ is two, since it contains two transverse blocks. The equivalence classes of the kernel of ξ are {{1, 3}, {2, 4}, {5}}, and the domain of ξ is {2, 4, 5}. Furthermore, $\operatorname{codom}(\xi) = \{1, 2, 3, 4\}$, and the equivalence classes of $\operatorname{coker}(\xi)$ are {{1, 3, 4}, {2}, {5}}.

For the majority of the monoids defined in Section 5.3.1, their Green's relations are completely determined by domain, kernel, and rank, as shown in the following lemma.

Lemma 5.43. Let $n \in \mathbb{N}$, let $S \in \{\mathcal{P}_n, \mathcal{PB}_n, \mathcal{B}_n, \mathcal{I}_n^*, \mathcal{M}_n, \mathcal{J}_n\}$, and let $\alpha, \beta \in S$. Then:

- (a) $\alpha \mathscr{R} \beta$ if and only if dom(α) = dom(β) and ker(α) = ker(β);
- (b) $\alpha \mathscr{L} \beta$ if and only if $\operatorname{codom}(\alpha) = \operatorname{codom}(\beta)$ and $\operatorname{coker}(\alpha) = \operatorname{coker}(\beta)$; and
- (c) $\alpha \not \subseteq \beta$ if and only if rank $(\alpha) = \operatorname{rank}(\beta)$.

See [97], [127, Theorem 17], and [33, Theorem 2.4] for the proof of this lemma. For $n \in \mathbb{N}$ and $k \in \{0, 1, \ldots, n\}$, we define

$$J_k = \{ \alpha \in \mathcal{P}_n : \operatorname{rank}(\alpha) = k \}$$

to be the \mathcal{J} -class of \mathcal{P}_n that comprises the partitions of rank k.

If S is a regular *-semigroup and $x, y \in S$, then $x \mathscr{L} y$ if and only if $x^* \mathscr{R} y^*$, and so parts (a) and (b) of Lemma 5.43 are equivalent. For the uniform block bijection monoid \mathfrak{F}_n , parts (a) and (b) certainly hold, since \mathfrak{F}_n is a regular submonoid of \mathcal{P}_n [76, Proposition 2.4.2] (indeed, it is inverse). However, while part (c) does not hold for \mathfrak{F}_n in general, it does hold for uniform block bijections of ranks n or n-1 [52, Section 3]. The description also holds for the annular Jones monoid \mathcal{AJ}_n .

It follows that, in general, if S is any of the diagram monoids defined in Section 5.3.1, then S has a unique \mathscr{J} -class J that is covered by the group of units of S. In the case that S is either the partition monoid or the Jones monoid, in order to determine the maximal subsemigroups of S that arise from this \mathscr{J} -class J, we require the graph $\Delta(S, J)$, as defined in Section 4.4.2 and further discussed in Section 4.5. Given a description of the \mathscr{L} -classes and \mathscr{R} -classes of J, to describe $\Delta(S, J)$, it remains to describe the left action of the group of units on the \mathscr{R} -classes of J by left multiplication. Since S is a regular *-monoid, a description of the right action of the group of units on the \mathscr{L} -classes of J by right multiplication is obtained as a consequence.

If $\alpha \in \mathcal{P}_n$ and $\sigma \in \mathcal{S}_n$, then the blocks are $\sigma \alpha$ are

$$(A \cap \{1, \ldots, n\})\sigma^{-1} \cup (A \cap \{1', \ldots, n'\}),$$

for each block A of α . Therefore,

$$\operatorname{dom}(\sigma\alpha) = \left\{ i\sigma^{-1} : i \in \operatorname{dom}(\alpha) \right\} \quad \text{and} \quad \operatorname{ker}(\sigma\alpha) = \left\{ \left(i\sigma^{-1}, j\sigma^{-1} \right) : (i, j) \in \operatorname{ker}(\alpha) \right\}.$$
(5.44)

Given this description and Lemma 5.43, the left action of a subgroup of S_n on the \mathscr{R} -classes of a particular \mathscr{J} -class is straightforward to determine.

5.3.3 The partition monoid \mathcal{P}_n

Let $n \in \mathbb{N}$, $n \geq 2$, be arbitrary. In this section, we classify the maximal subsemigroups of the partition monoid of degree n. To do so, we require the following information about the Green's classes of \mathcal{P}_n contained in its \mathscr{J} -class J_{n-1} . Let $\alpha \in J_{n-1}$. By definition, α contains n-1 transverse blocks. Since each transverse block contains at least two points, and since there are only 2n points in $\{1, \ldots, n\} \cup \{1', \ldots, n'\}$, there are few possible combinations for the kernel and domain for α . In particular, either ker(α) is trivial and dom(α) = $\{1, \ldots, n\} \setminus \{i\}$ for some $i \in \{1, \ldots, n\}$, or dom(α) = $\{1, \ldots, n\}$ and $\{i, j\}$ is the unique non-trivial kernel class of α , for some distinct point $i, j \in \{1, \ldots, n\}$. By Lemma 5.43, these properties describe the \mathscr{R} -classes of J_{n-1} . Since the \mathscr{L} -classes and \mathscr{R} -classes of a regular *-semigroup correspond via the * operation, this also gives a description of the \mathscr{L} -classes of J_{n-1} . Thus, for distinct $i, j \in \{1, \ldots, n\}$, we make the following definitions:

- $R_i = \{ \alpha \in J_{n-1} : \operatorname{dom}(\alpha) = \{1, \ldots, n\} \setminus \{i\} \}$, which is an \mathscr{R} -class of J_{n-1} ;
- $R_{\{i, j\}} = \{ \alpha \in J_{n-1} : (i, j) \in \ker(\alpha) \}$, which is an \mathscr{R} -class of J_{n-1} ;
- $L_i = R_i^* = \{ \alpha \in J_{n-1} : \operatorname{codom}(\alpha) = \{1, \ldots, n\} \setminus \{i\} \}$, which is an \mathscr{L} -class of J_{n-1} ;
- $L_{\{i,j\}} = R^*_{\{i,j\}} = \{\alpha \in J_{n-1} : (i,j) \in \operatorname{coker}(\alpha)\}$, which is an \mathscr{L} -class of J_{n-1} .

In general, it is slightly complicated to describe the group \mathscr{H} -classes of \mathcal{P}_n in terms of kernels, domains, cokernels, and codomains. However, in the \mathscr{J} -class J_{n-1} , it is straightforward. An \mathscr{H} -class of the form $L_i \cap R_j$ is a group if and only if i = j, an \mathscr{H} -class of the form $L_i \cap R_{\{j,k\}}$ or $R_i \cap L_{\{j,k\}}$ is a group if and only if $i \in \{j, k\}$, and an \mathscr{H} -class of the form $L_{\{i,j\}} \cap R_{\{k,l\}}$ is a group if and only if $i \in \{j, k\}$.

The main result of this section is the following theorem. In the proof, we use the fact that the ideal $\mathcal{P}_n \setminus \mathcal{S}_n$ is generated by its idempotents of rank n-1 [40, Section 6],

Theorem 5.45. Let $n \in \mathbb{N}$, $n \geq 2$, be arbitrary. The maximal subsemigroups of \mathcal{P}_n are:

- (a) $(\mathcal{P}_n \setminus \mathcal{S}_n) \cup U$, where U is a maximal subgroup of \mathcal{S}_n (type (M1));
- (b) $\mathcal{P}_n \setminus \{ \alpha \in \mathcal{P}_n : \operatorname{rank}(\alpha) = n 1 \text{ and } \ker(\alpha) \text{ is trivial} \}$ (type (M4));
- (c) $\mathcal{P}_n \setminus \{ \alpha \in \mathcal{P}_n : \operatorname{rank}(\alpha) = n 1 \text{ and } \operatorname{dom}(\alpha) = \{1, \ldots, n\} \}$ (type (M4));
- (d) $\mathcal{P}_n \setminus \{ \alpha \in \mathcal{P}_n : \operatorname{rank}(\alpha) = n 1 \text{ and } \operatorname{coker}(\alpha) \text{ is trivial} \}$ (type (M3)); and
- (e) $\mathcal{P}_n \setminus \{ \alpha \in \mathcal{P}_n : \operatorname{rank}(\alpha) = n 1 \text{ and } \operatorname{codom}(\alpha) = \{1, \ldots, n\} \}$ (type (M3)).

In particular, for $n \geq 2$, there are $s_n + 4$ maximal subsemigroups of \mathcal{P}_n .



Figure 5.46: The graph $\Delta(\mathcal{P}_n, J_{n-1})$, which is a complete bipartite graph with four vertices. The two maximal independent subsets are the set of orbits of \mathscr{L} -classes, and the set of orbits of \mathscr{R} -classes. Each vertex in the graph has degree two.

Proof. Since \mathscr{J} -equivalence in \mathcal{P}_n is determined by rank, and \mathcal{P}_n is generated by its units and idempotents of rank n-1, it follows by Lemma 4.68 that the maximal subsemigroups of \mathcal{P}_n arise from its group of units and its \mathscr{J} -class J_{n-1} . By Lemma 4.73, the s_n maximal subsemigroups that arise from the group of units are those stated in the theorem with type (M1).

By Lemma 4.49(b), there are no maximal subsemigroups of \mathcal{P}_n arising from J_{n-1} with type (M1). Given (5.44), it is clear that \mathcal{S}_n transitively permutes the \mathscr{R} -classes of J_{n-1} with trivial kernel, and it transitively permutes the \mathscr{R} -classes of J_{n-1} with domain $\{1, \ldots, n\}$. Hence there are two orbits of \mathscr{R} -classes under this right action, and so there are two orbits of \mathscr{L} -classes under the corresponding left action. We may therefore deduce the description of $\Delta(\mathcal{P}_n) =$ $\Delta(\mathcal{P}_n, J_{n-1})$; see Figure 5.46. Since $\Delta(\mathcal{P}_n)$ is a complete bipartite graph, with only two maximal independent subsets, Corollary 4.78 implies that there are no maximal subsemigroups of type (M2). Each vertex of $\Delta(\mathcal{P}_n)$ has degree 2, and so by Proposition 4.79, there are two maximal subsemigroups of type (M3), formed by removing each orbit of \mathscr{L} -classes in turn. Similarly, there are two maximal subsemigroups of type (M4).

5.3.4 The Brauer monoid \mathcal{B}_n and the uniform block bijection monoid \mathfrak{F}_n

Let $n \in \mathbb{N}$, $n \geq 2$. In this section, we describe and count the maximal subsemigroups of \mathcal{B}_n and \mathfrak{F}_n . The main results of this section are the following theorems. In the proofs of these theorems, we exploit the following results. By [4], \mathcal{B}_n is generated by its group of units \mathcal{S}_n and any projection of rank n - 2, and by [94, Section 5], \mathfrak{F}_n is generated by it groups of units \mathcal{S}_n and any projection of rank n - 1.

Theorem 5.47. Let $n \in \mathbb{N}$, $n \geq 2$, be arbitrary. The maximal subsemigroups of \mathcal{B}_n are:

- (a) $(\mathcal{B}_n \setminus \mathcal{S}_n) \cup U$, where U is a maximal subgroup of \mathcal{S}_n (type (M1)); and
- (b) $\mathcal{B}_n \setminus \{ \alpha \in \mathcal{B}_n : \operatorname{rank}(\alpha) = n 2 \}$ (type (M5)).

In particular, for $n \geq 2$, there are $s_n + 1$ maximal subsemigroups of \mathcal{B}_n .

Theorem 5.48. Let $n \in \mathbb{N}$, $n \geq 2$, be arbitrary. The maximal subsemigroups of \mathfrak{F}_n are:

- (a) $(\mathfrak{F}_n \setminus S_n) \cup U$, where U is a maximal subgroup of S_n (type (M1)); and
- (b) $\mathfrak{F}_n \setminus \{ \alpha \in \mathfrak{F}_n : \operatorname{rank}(\alpha) = n 1 \}$ (type (M5)).

In particular, for $n \geq 2$, there are $s_n + 1$ maximal subsemigroups of \mathfrak{F}_n .

Proof of Theorems 5.47 and 5.48. By Lemma 4.68, the maximal subsemigroups of \mathcal{B}_n arise from its group of units and its \mathscr{J} -class $J_{n-2} \cap \mathcal{B}_n$, and the maximal subsemigroups of \mathfrak{F}_n arise from its group of units and from its \mathscr{J} -class $J_{n-1} \cap \mathfrak{F}_n$. In both cases, by Lemma 4.73, there is one maximal subsemigroup corresponding to each maximal subgroup of \mathcal{S}_n , as stated in the theorem, and so there are s_n such maximal subsemigroups.

To describe the remaining maximal subsemigroups of \mathcal{B}_n , let $\alpha \in J_{n-2} \cap \mathcal{B}_n$. The nontransverse blocks of α are $\{i, j\}$ and $\{k', l'\}$ for some $i, j, k, l \in \{1, \ldots, n\}$ with $i \neq j$ and $k \neq l$. Let $\tau \in \mathcal{S}_n$ be a permutation that contains the blocks $\{k, i'\}$ and $\{l, j'\}$. Therefore the non-transverse blocks of $\alpha \tau$ are $\{i, j\}$ and $\{i', j'\}$, and so $(\alpha \tau)^m$ is a projection of rank n-2for some $m \in \mathbb{N}$. Thus $\langle \mathcal{S}_n, \alpha \rangle \supseteq \langle \mathcal{S}_n, (\alpha \tau)^m \rangle = \mathcal{B}_n$, since \mathcal{B}_n is generated by \mathcal{S}_n and any projection of rank n-2, and so $\mathcal{B}_n = \langle \mathcal{S}_n, \alpha \rangle$. By a very similar argument, $\mathfrak{F}_n = \langle \mathcal{S}_n, \beta \rangle$ for any uniform block bijection of rank n-1. By Corollary 4.74, in each case, the remaining maximal subsemigroups are those stated in the theorems.

5.3.5 The partial Brauer monoid \mathcal{PB}_n

Let $n \in \mathbb{N}$, $n \geq 2$. Recall that \mathcal{I}_n , the symmetric inverse monoid, is the monoid consisting of all partial permutations of degree n. The restriction to \mathcal{I}_n of the injective map $\phi : \mathcal{PT}_n \longrightarrow \mathcal{P}_n$, defined at the start of Section 5.3.2, is an embedding of \mathcal{I}_n into \mathcal{P}_n . In particular, $\mathcal{I}_n \phi$ is a submonoid of \mathcal{PB}_n isomorphic to \mathcal{I}_n . Therefore, in this section, we will identify \mathcal{I}_n with its image under ϕ , and regard \mathcal{I}_n as a submonoid of \mathcal{PB}_n .

By [33, Proposition 3.16], \mathcal{PB}_n is generated by its elements with rank at least n-2, and any generating set contains elements of ranks n, n-1, and n-2. Therefore, by Lemma 4.68, in order to describe the maximal subsemigroups of \mathcal{PB}_n , we require a description of the elements of \mathcal{PB}_n with these ranks. Partitions of degree n and rank n are units, and the group of units of \mathcal{PB}_n is \mathcal{S}_n .

Let $\alpha \in J_{n-1} \cap \mathcal{PB}_n$. By definition, α contains n-1 transverse blocks, which contain a point of $\{1, \ldots, n\}$ and a point of $\{1', \ldots, n'\}$. Therefore, the remaining blocks of α are the singletons $\{i\}$ and $\{j'\}$, for some $i, j \in \{1, \ldots, n\}$. In other words, α is the image of some partial permutation of rank n-1 under the embedding ϕ . Since α was arbitrary, and $\mathcal{I}_n \subseteq \mathcal{PB}_n$, it follows that $J_{n-1} \cap \mathcal{PB}_n = J_{n-1} \cap \mathcal{I}_n$.

Let $\alpha \in J_{n-2} \cap \mathcal{PB}_n$. Then α contains n-2 transverse blocks, and so a pair of points of $\{1, \ldots, n\}$ and a pair of points of $\{1', \ldots, n'\}$ are not contained in transverse blocks. Each one of these pairs forms either a block of size 2, or two singleton blocks. In particular, dom (α) lacks two points i and j, and either ker (α) is trivial, or $\{i, j\}$ is the unique non-trivial kernel class of α . A similar statement holds for the codomain and cokernel of α .

By [33, Proposition 3.15], $\mathcal{PB}_n = \langle S_n, J_{n-1} \cap \mathcal{I}_n, J_{n-2} \cap \mathcal{B}_n \rangle$. As stated in the proof of Theorem 5.7, \mathcal{I}_n is generated by \mathcal{S}_n and any partial permutation of rank n-1, and as stated in the proof of Theorem 5.47, \mathcal{B}_n is generated by \mathcal{S}_n and any of its elements of rank n-2. Therefore, $\mathcal{PB}_n = \langle S_n, \alpha, \beta \rangle$, for any $\alpha \in J_{n-1} \cap \mathcal{I}_n$ and $\beta \in J_{n-2} \cap \mathcal{B}_n$.

Theorem 5.49. Let $n \in \mathbb{N}$, $n \geq 2$, be arbitrary. The maximal subsemigroups of \mathcal{PB}_n are:

- (a) $(\mathcal{PB}_n \setminus \mathcal{S}_n) \cup U$, where U is a maximal subgroup of \mathcal{S}_n (type (M1));
- (b) $\mathcal{PB}_n \setminus \{ \alpha \in \mathcal{PB}_n : \operatorname{rank}(\alpha) = n 1 \}$ (type (M5));
- (c) $\mathcal{PB}_n \setminus \{\alpha \in \mathcal{PB}_n : \operatorname{rank}(\alpha) = n 2 \text{ and } \operatorname{coker}(\alpha) \text{ is non-trivial} \}$ (type (M3)); and
- (d) $\mathcal{PB}_n \setminus \{ \alpha \in \mathcal{PB}_n : \operatorname{rank}(\alpha) = n 2 \text{ and } \ker(\alpha) \text{ is non-trivial} \}$ (type (M4)).

In particular, for $n \in \mathbb{N}$, there are $s_n + 3$ maximal subsemigroups of \mathcal{PB}_n .

Proof. By Lemma 5.43, Green's \mathscr{J} -relation in \mathcal{PB}_n is determined by rank. Given the description of the generating sets of \mathcal{PB}_n , it follows by Lemma 4.68 that the \mathscr{J} -classes of \mathcal{PB}_n from which there arise maximal subsemigroups are the group of units, $J_{n-1} \cap \mathcal{PB}_n$, and $J_{n-2} \cap \mathcal{PB}_n$. The group of units of \mathcal{PB}_n is the symmetric group \mathcal{S}_n , and so by Lemma 4.73, the s_n maximal subsemigroups that arise in this case are those described.

Since the \mathscr{J} -class $J_{n-1} \cap \mathcal{PB}_n = J_{n-1} \cap \mathcal{I}_n$ is covered by the group of units, $\mathcal{PB}_n \setminus J_{n-1}$ is a subsemigroup of \mathcal{PB}_n . However, as stated above, $\mathcal{PB}_n = \langle S_n, \alpha, \beta \rangle$, for any $\alpha \in J_{n-1} \cap \mathcal{I}_n$ and $\beta \in J_{n-2} \cap \mathcal{B}_n$. In particular, $\langle \mathcal{PB}_n \setminus J_{n-1}, \alpha \rangle = \mathcal{PB}_n$ if and only if $\alpha \in J_{n-1} \cap \mathcal{PB}_n$. By Lemma 4.8, the only maximal subsemigroup of \mathcal{PB}_n to arise from this \mathscr{J} -class has type (M5).

In order to determine the maximal subsemigroups of \mathcal{PB}_n that arise from its \mathcal{J} -class of rank n-2, we define the subsets

$$X = \{ \alpha \in \mathcal{PB}_n : \operatorname{rank}(\alpha) = n - 2 \text{ and } \ker(\alpha) \text{ is non-trivial} \}, \text{ and}$$
$$X^* = \{ \alpha^* : \alpha \in X \} = \{ \alpha \in \mathcal{PB}_n : \operatorname{rank}(\alpha) = n - 2 \text{ and } \operatorname{coker}(\alpha) \text{ is non-trivial} \}.$$

Note that X is a union of \mathscr{R} -classes of $J_{n-2} \cap \mathcal{PB}_n$, and X^* is a union of \mathscr{L} -classes of $J_{n-2} \cap \mathcal{PB}_n$. Let A be any subset of $J_{n-2} \cap \mathcal{PB}_n$ such that $(\mathcal{PB}_n \setminus J_{n-2}) \cup A$ generates \mathcal{PB}_n , and let $\alpha \in X$ be arbitrary. Then α can the written as a product $\alpha = \beta_1 \cdots \beta_k$ of some of these generators. The generators β_1, \ldots, β_k all have rank at least n-2 by Lemma 1.12. Every element in \mathcal{PB}_n of rank n and n-1 has a trivial kernel, and the subset of partitions with trivial kernel in \mathcal{P}_n forms a subsemigroup. Thus, there exists some $r \in \{1, \ldots, k\}$ such that $\operatorname{rank}(\beta_r) = n-2$ and $\operatorname{ker}(\beta_r)$ is non-trivial – in other words, $\beta_r \in X$. A dual argument shows that $A \cap X^* \neq \emptyset$.

Conversely, let A be any subset of $J_{n-2} \cap \mathcal{PB}_n$ that intersects X and X^* non-trivially, and fix elements $\alpha \in A \cap X$ and $\beta \in A \cap X^*$. It is straightforward to see that there exists some permutation $\sigma \in S_n$ such that $\alpha\sigma\beta$ has non-trivial kernel and cokernel, and has rank n-2. In particular, $\alpha\sigma\beta \in J_{n-2} \cap \mathcal{B}_n$. Since $\mathcal{PB}_n = \langle S_n, \alpha, \beta \rangle$, for any $\alpha \in J_{n-1} \cap \mathcal{I}_n$ and $\beta \in J_{n-2} \cap \mathcal{B}_n$, it follows that $\langle \mathcal{PB}_n \setminus J_{n-2}, A \rangle = \mathcal{PB}_n$. By Lemma 4.8, the maximal subsemigroups of \mathcal{PB}_n arising from $J_{n-2} \cap \mathcal{PB}_n$ are $\mathcal{PB}_n \setminus X^*$ and $\mathcal{PB}_n \setminus X$; these maximal subsemigroups have types (M3) and (M4), respectively.

5.3.6 The dual symmetric inverse monoid \mathcal{I}_n^*

The maximal subsemigroups of the dual symmetric inverse monoid were first described by Maltcev [94, Theorem 19]. We reprove this result in the following theorem. See the same paper for much more information about \mathcal{I}_n^* , including the fact that for $n \geq 3$, $\mathcal{I}_n^* = \langle S_n, \alpha \rangle$ if and only if $\alpha \in J_{n-1} \cap (\mathcal{I}_n^* \setminus \mathfrak{F}_n)$ [94, Proposition 16].

Theorem 5.50. Let $n \in \mathbb{N}$, $n \geq 3$, be arbitrary. The maximal subsemigroups of \mathcal{I}_n^* are:

- (a) $(\mathcal{I}_n^* \setminus \mathcal{S}_n) \cup U$, where U is a maximal subgroup of \mathcal{S}_n (type (M1)); and
- (b) $\mathcal{I}_n^* \setminus \{ \alpha \in \mathcal{I}_n^* : \operatorname{rank}(\alpha) = n 1 \text{ and } \alpha \text{ is not uniform} \}$ (type (M1)).

In particular, for $n \geq 3$ there are $s_n + 1$ maximal subsemigroups of \mathcal{I}_n^* .

Proof. By Lemma 4.68, the maximal subsemigroups of \mathcal{I}_n^* arise from its group of units and it's \mathscr{J} -class $J_{n-1} \cap \mathcal{I}_n^*$. The group of units of \mathcal{I}_n^* is \mathcal{S}_n , and so by Lemma 4.73, the maximal subsemigroups arising from the group of units are those described. Using Corollary 4.74 with $X = J_{n-1} \cap (\mathcal{I}_n^* \setminus \mathfrak{F}_n)$, we find the description of the remaining maximal subsemigroup. Note that \mathcal{I}_n^* is the monoid of all block bijections, and \mathfrak{F}_n is the monoid of all uniform block bijections, and so the set X consists of the block bijections of rank n-1 that are not uniform.

The maximal subsemigroup of \mathcal{I}_n^* that arises from its \mathscr{J} -class $J_{n-1} \cap \mathcal{I}_n^*$ can also be found by using Proposition 4.85, since \mathcal{S}_n acts transitively on the \mathscr{R} -classes of $J_{n-1} \cap \mathcal{I}_n^*$, and each of the idempotents in this \mathscr{J} -class is a projection.

	1	2	3	4	5	6	7	8	9	10	11	12	n
\mathcal{P}_n	2	5	8	12	26	57	188	357	1380	3981	363908	396502	$s_n + 4$
\mathcal{PB}_n	2	4	7	11	25	56	187	356	1379	3980	363907	396501	$s_n + 3$
$\mathcal{B}_n, \mathfrak{F}_n, \mathcal{I}_n^*$	1	2	5	9	23	54	185	354	1377	3978	363905	396499	$s_n + 1$

Table 5.51: The numbers of maximal subsemigroups of the monoids \mathcal{P}_n , \mathcal{PB}_n , \mathcal{B}_n , \mathfrak{F}_n , and \mathcal{I}_n^* , for $n = 1, \ldots, 12$, along with the general formulae. Recall that s_n denotes the number of maximal subgroups of the symmetric group \mathcal{S}_n [120, A290138]. See Theorems 5.45–5.50 for the descriptions of the maximal subsemigroups of these monoids; see also Table 5.8.

5.3.7 The Jones monoid \mathcal{J}_n and the annular Jones monoid $\mathcal{A}\mathcal{J}_n$

Let $n \in \mathbb{N}$ be arbitrary. In this section, we classify the maximal subsemigroups of the Jones and annular Jones monoids \mathcal{J}_n and $\mathcal{A}\mathcal{J}_n$. Recall that \mathcal{J}_n consists of all planar partitions in the Brauer monoid \mathcal{B}_n , and $\mathcal{A}\mathcal{J}_n$ consists of all annular partitions in \mathcal{B}_n ; therefore \mathcal{J}_n is a submonoid of $\mathcal{A}\mathcal{J}_n$. Since the planar partition monoid \mathcal{PP}_n is isomorphic to \mathcal{J}_{2n} [70], by determining the maximal subsemigroups of the Jones monoids, we also find those of the planar partition monoids.

First, we classify the maximal subsemigroups of \mathcal{J}_n . Suppose that $n \geq 2$. The Green's relations on \mathcal{J}_n are given by Lemma 5.43, and it was proved in [12] that \mathcal{J}_n is generated by $\{\mathrm{id}_n\}$ and its projections of rank n-2. Therefore, by Lemma 4.68, the maximal subsemigroups of \mathcal{J}_n arise from its group of units, and from its \mathscr{J} -class $J_{n-2} \cap \mathcal{J}_n$. Since there are no elements of rank n-1 in \mathcal{J}_n , this \mathscr{J} -class is covered by the group of units, and so we may apply the results of Section 4.5.2 to describe the maximal subsemigroups that arise in this case.

To construct the graph $\Delta(\mathcal{J}_n) = \Delta(\mathcal{J}_n, J_{n-2} \cap \mathcal{J}_n)$, we require a description of the Green's classes of $J_{n-2} \cap \mathcal{J}_n$. Let $\alpha \in J_{n-2} \cap \mathcal{J}_n$ be arbitrary. Then α has n-2 transverse blocks, and these contain two points, and by planarity, the non-transverse blocks have the form $\{i, i+1\}$ and $\{j', (j+1)'\}$ for some $i, j \in \{1, \ldots, n-1\}$. By Lemma 5.43, the \mathscr{J} -class contains contains n-1 \mathscr{R} -classes, and n-1 \mathscr{L} -classes; the planarity of \mathcal{J}_n implies that \mathcal{J}_n is \mathscr{H} -trivial.

For $i \in \{1, \ldots, n-1\}$, we define

- $R_i = \{ \alpha \in \mathcal{J}_n : \operatorname{rank}(\alpha) = n 2 \text{ and } \{i, i + 1\} \text{ is a block of } \alpha \}, \text{ which is an } \mathscr{R}\text{-class; and } \{i, i + 1\} \text{ or } \alpha \in \mathcal{J}_n \}$
- $L_i = \{ \alpha \in \mathcal{J}_n : \operatorname{rank}(\alpha) = n 2 \text{ and } \{i', (i+1)'\} \text{ is a block of } \alpha \}, \text{ which is an } \mathscr{L}\text{-class.}$

The intersection of the \mathscr{L} -class L_i and the \mathscr{R} -class R_j is a group if and only if $|i-j| \leq 1$. Since the group of units of \mathcal{J}_n is trivial, its actions on the Green's classes of $J_{n-2} \cap \mathcal{J}_n$ are trivial. A picture of $\Delta(\mathcal{J}_n)$ is shown in Figure 5.52. The maximal independent subsets of $\Delta(\mathcal{J}_n)$ are described in Lemma 5.53, and counted in Corollary 5.55.



Figure 5.52: The graph $\Delta(\mathcal{J}_n, \mathcal{J}_n \cap \mathcal{J}_{n-2})$. Since the group of units of \mathcal{J}_n is trivial, its actions on its \mathscr{L} -classes and \mathscr{R} -classes are trivial, and so its orbits are singletons.

Lemma 5.53. Let $n \in \mathbb{N}$, $n \geq 2$, be arbitrary, and let U be a subset of the vertices of $\Delta(\mathcal{J}_n)$. Then U is a maximal independent subset of $\Delta(\mathcal{J}_n)$ if and only if all of the following hold:

- (a) U contains exactly one of $\{L_1\}$ and $\{R_1\}$; and
- (b) U contains exactly one of $\{L_{n-1}\}$ and $\{R_{n-1}\}$; and
- (c) if $\{L_i\} \in U$ for some $i \in \{1, ..., n-2\}$, then the vertex in $U \setminus \{\{L_i\}\}$ with the least index greater than or equal to i is either $\{L_{i+1}\}$ or $\{R_{i+2}\}$; and
- (d) if $\{R_i\} \in U$ for some $i \in \{1, ..., n-2\}$, then the vertex in $U \setminus \{\{R_i\}\}$ with the least index greater than or equal to i is either $\{R_{i+1}\}$ or $\{L_{i+2}\}$.

Proof. Suppose that U is a maximal independent subset of $\Delta(\mathcal{J}_n)$. We first show that condition (a) holds; the proof that condition (b) holds is analogous. Since $\{L_1\}$ and $\{R_1\}$ are adjacent in $\Delta(\mathcal{J}_n)$, they are not both contained in U; the same is true of $\{L_2\}$ and $\{R_2\}$. If $\{L_2\} \notin U$, then either $\{L_1\} \in U$, or $\{L_1\} \notin U$, and the maximality of U implies that $\{R_1\} \in U$. If instead $\{R_2\} \notin U$, then it follows similarly that $\{L_1\} \in U$ or $\{R_1\} \in U$.

To prove that condition (c) holds, let $i \in \{1, \ldots, n-2\}$ be arbitrary, and suppose that $\{L_i\} \in U$. Consider the vertex in $U \setminus \{\{L_i\}\}$ with least index greater than or equal to i; some such vertex exists, since U contains either $\{L_{n-1}\}$ or $\{R_{n-1}\}$, by (b). Certainly this vertex is neither $\{R_i\}$ nor $\{R_{i+1}\}$, since these are neighbours of $\{L_i\}$ in $\Delta(\mathcal{J}_n)$. If additionally $\{R_{i+2}\} \notin U$, then the maximality of U implies that $\{L_{i+1}\} \in U$. Noting that U does not contain both $\{L_{i+2}\}$ and $\{R_{i+2}\}$, it follows that the vertex in $U \setminus \{\{L_i\}\}$ with smallest index greater than or equal to i is either $\{L_{i+1}\}$ or $\{R_{i+2}\}$, as required. Condition (d) holds by an analogous argument.

Conversely, if U satisfies conditions (a)–(d) of the lemma, then it is straightforward to verify that U is an independent subset, and that it is maximal. \Box

In order to count the maximal subsemigroups of $\Delta(\mathcal{J}_n)$, we introduce the famous *Fibonacci* sequence [120, A000045], $(F_n)_{n \in \mathbb{N}}$, which is defined by

$$F_1 = F_2 = 1$$
, and $F_n = F_{n-1} + F_{n-2}$ for $n \ge 3$. (5.54)

In [43, Theorem 9.9], it was shown that the ideal $\mathcal{J}_n \setminus \{ \mathrm{id}_n \}$ of \mathcal{J}_n has exactly F_n minimal idempotent generating sets. In the following corollary to Lemma 5.53, we show that the Fibonacci numbers also count the maximal independent subsets of $\Delta(\mathcal{J}_n)$.

Corollary 5.55. Let $n \ge 2$. The number of maximal independent subsets of $\Delta(\mathcal{J}_n)$ is $2F_{n-1}$.

Proof. The result may be verified directly for $n \in \{2, 3\}$, so suppose that $n \ge 4$. By the symmetry of $\Delta(\mathcal{J}_n)$, the maximal independent subsets that contain $\{L_1\}$ are in one-to-one correspondence with the maximal independent subsets that contain $\{R_1\}$. Therefore, we count a(n), the number of maximal independent subsets of $\Delta(\mathcal{J}_n)$ that contain $\{L_1\}$; by Lemma 5.53(a), the total number of maximal independent subsets of $\Delta(\mathcal{J}_n)$ is 2a(n).

For $i \in \{1, 2\}$, define Λ_{n-i} be the induced subgraph of $\Delta(\mathcal{J}_n)$ on the vertices

$$\{\{L_{i+1}\},\ldots,\{L_{n-1}\},\{R_{i+1}\},\ldots,\{R_{n-1}\}\}.$$

In other words, Λ_{n-1} is the graph formed from $\Delta(\mathcal{J}_n)$ be removing the vertices $\{L_1\}$ and $\{R_1\}$, and Λ_{n-2} is formed from Λ_{n-1} by additionally removing the vertices $\{L_2\}$ and $\{R_2\}$. It is obvious that Λ_{n-i} is isomorphic to $\Delta(\mathcal{J}_{n-i})$, and so the number of maximal independent subsets of Λ_{n-1} that contain $\{L_2\}$ is a(n-1), and the number of maximal independent subsets of Λ_{n-2} that contain $\{R_3\}$ is a(n-2).

Let U be a maximal independent subset of $\Delta(\mathcal{J}_n)$ that contains $\{L_1\}$. By Lemma 5.53, U contains either $\{L_2\}$ or $\{R_3\}$, but not both. If U contains $\{L_2\}$, then $U \setminus \{\{L_1\}\}$ is a maximal independent subset of Λ_{n-1} that contains $\{L_2\}$, while if U contains $\{R_3\}$, then $U \setminus \{\{L_1\}\}$ is a maximal independent subset of Λ_{n-2} that contains $\{R_3\}$. Hence $a(n) \leq a(n-1) + a(n-2)$. Conversely, each maximal independent subset of Λ_{n-1} that contains $\{L_2\}$, and each maximal independent subset of Λ_{n-2} that contains $\{R_3\}$, gives rise to a unique maximal independent subset of $\Delta(\mathcal{J}_n)$ containing $\{L_1\}$, by the addition of $\{L_1\}$. Hence a(n) = a(n-1) + a(n-2). Since $a(2) = F_1$ and $a(3) = F_2$, we deduce that $a(n) = F_{n-1}$.

Note that the sets $\{\{L_1\}, \ldots, \{L_{n-1}\}\}$ and $\{\{R_1\}, \ldots, \{R_{n-1}\}\}$ are the only maximal independent subsets of $\Delta(\mathcal{J}_n)$ that do not contain both orbits of \mathscr{L} -classes and orbits of \mathscr{R} -

classes. By Proposition 4.77, these are the two maximal independent subsets of $\Delta(\mathcal{J}_n)$ that do not give rise to maximal subsemigroups of \mathcal{J}_n of type (M2).

Theorem 5.56. Let $n \in \mathbb{N}$, $n \geq 3$, be arbitrary. The maximal subsemigroups of \mathcal{J}_n are:

- (a) $\mathcal{J}_n \setminus {\mathrm{id}_n}$ (type (M5));
- (b) The union of J_n \ J_{n-2} and the union of the Green's classes in a maximal independent subset of Δ(J_n) that contains both ℒ-classes and ℛ-classes (type (M2));
- (c) $\mathcal{J}_n \setminus L$, where L is any \mathscr{L} -class in \mathcal{J}_n containing elements of rank n-2 (type (M3)); and
- (d) $\mathcal{J}_n \setminus R$, where R is any \mathscr{R} -class in \mathcal{J}_n containing elements of rank n-2 (type (M4)).

In particular, for $n \ge 3$, there are $2F_{n-1} + 2n - 3$ maximal subsemigroups of \mathcal{J}_n , where F_{n-1} is the $(n-1)^{th}$ term of the Fibonacci sequence, as defined in (5.54).

Proof. The group of units of \mathcal{J}_n is the trivial group $\{\mathrm{id}_n\}$, and so by Lemma 4.73, the unique maximal subsemigroup that arises from the group of units is formed by removing it.

It remains to describe the maximal subsemigroups that arise from the \mathscr{J} -class $J_{n-2} \cap \mathscr{J}_n$. The Jones monoid \mathscr{J}_n is \mathscr{H} -trivial, and so by Lemma 4.49(a), \mathscr{J}_n has no maximal subsemigroups of type (M1). The description of the maximal subsemigroups of type (M2) follows directly from Proposition 4.77 and Lemma 5.53, and their number $2F_{n-1} - 2$ follows by Corollaries 4.78 and 5.55. Since $n \geq 3$, each vertex of $\Delta(\mathscr{J}_n)$ has degree 2 or 3, and so by Proposition 4.79, any \mathscr{L} -class of rank n-2 can be removed to form a maximal subsemigroup of type (M3), and similarly, any \mathscr{R} -class of rank n-2 can be removed to form a maximal subsemigroup of type (M4). Hence there are n-1 maximal subsemigroups of each type. By Corollary 4.75, there is no maximal subsemigroup of type (M5).

In the second part of this section, we classify the maximal subsemigroups of the annular Jones monoid \mathcal{AJ}_n . In order to do so, we require an understanding of its generating sets. Let $n \in \mathbb{N}, n \geq 2$, and define ρ_n to be the *n*-cycle $(1 \ 2 \ \dots \ n)$. Recall from Section 5.3.1 that $\mathcal{C}_n = \langle \rho_n \rangle$ is the group of units of \mathcal{AJ}_n , and that a partition $\alpha \in \mathcal{P}_n$ is annular if $\alpha = \rho_n^k \beta \rho_n^l$, for some planar partition $\beta \in \mathcal{PP}_n$ and for some $k, l \in \mathbb{N}_0$. The elements of \mathcal{AJ}_n are the annular partitions of degree n whose blocks contain two points.

Let $\alpha = \rho_n^k \beta \rho_n^l \in \mathcal{AJ}_n$ be arbitrary, where β is planar and $k, l \in \mathbb{N}_0$. Since ρ_n^k and ρ_n^l are units, it follows that $\beta = \rho_n^{-k} \cdot \alpha \cdot \rho_n^{-l} \in \mathcal{AJ}_n$, and so the blocks of β have size two. In other words, $\beta \in \mathcal{J}_n$. Therefore

$$\mathcal{A}\mathcal{J}_n = \left\{ \rho_n^k \beta \rho_n^l : \beta \in \mathcal{J}_n \text{ and } k, l \in \{1, \dots, n\} \right\} = \langle \mathcal{J}_n, \rho_n \rangle.$$
(5.57)

For each $i \in \{1, \ldots, n-1\}$, define ξ_i to be the projection in $\mathcal{A}\mathcal{J}_n$ of rank n-2 whose nontransverse blocks are $\{i, i+1\}$ and $\{i', (i+1)'\}$, and define ξ_n to be the remaining projection of rank n-2 in $\mathcal{A}\mathcal{J}_n$; its non-transverse blocks are $\{1, n\}$ and $\{1', n'\}$. For all $i, j \in \{1, \ldots, n\}$, it is straightforward to verify that $\xi_i = \rho_n^{j-i}\xi_j\rho_n^{i-j}$. Since \mathcal{J}_n is generated by id_n and its projections of rank n-2 [12], it follows by this discussion and by (5.57) that $\mathcal{A}\mathcal{J}_n = \langle \rho_n, \xi_i \rangle$ for any $i \in \{1, \ldots, n\}$.

We proceed with the same technique that was used in the proof of Theorems 5.47 and 5.48. Let $\alpha \in J_{n-2} \cap \mathcal{AJ}_n$ be arbitrary. Working modulo n, there exist indices $i, j \in \{1, \ldots, n\}$ such that for all $k \in \mathbb{N}_0$, the non-transverse blocks of $\alpha \rho_n^k$ are $\{i, i+1\}$ and $\{(j\rho_n^k)', (j\rho_n^k+1)'\}$. Thus we may fix $k \in \mathbb{N}_0$ such that the non-transverse blocks of $\alpha \rho_n^k$ are $\{i, i+1\}$ and $\{i', (i+1)'\}$. Since \mathcal{AJ}_n is finite, some power of $\alpha \rho_n^k$ is equal to the projection ξ_i . Therefore

$$\mathcal{AJ}_n \ge \langle \rho_n, \alpha \rangle \ge \langle \rho_n, \alpha \rho_n^k \rangle \ge \langle \rho_n, \xi_i \rangle = \mathcal{AJ}_n,$$

and so $\langle \rho_n, \alpha \rangle = \langle C_n, \alpha \rangle = \mathcal{A}\mathcal{J}_n$. Since $\mathcal{A}\mathcal{J}_n$ contains no elements of rank n-1, any generating set for $\mathcal{A}\mathcal{J}_n$ contains elements of rank n-2. Therefore, it follows that $\langle C_n, \alpha \rangle$ if and only if $\alpha \in J_{n-2} \cap \mathcal{A}\mathcal{J}_n$. This result is required in the proof of the following theorem.

Theorem 5.58. Let $n \in \mathbb{N}$, $n \geq 2$, be arbitrary. The maximal subsemigroups of \mathcal{AJ}_n are:

- (a) $(\mathcal{A}\mathcal{J}_n \setminus \mathcal{C}_n) \cup U$, where U is a maximal subgroup of \mathcal{C}_n (type (M1)); and
- (b) $\mathcal{AJ}_n \setminus \{ \alpha \in \mathcal{AJ}_n : \operatorname{rank}(\alpha) = n 2 \}$ (type (M5)).

In particular, for $n \ge 2$, there are $|\mathbb{P}_n| + 1$ maximal subsemigroups of \mathcal{AJ}_n , where \mathbb{P}_n denotes the set of primes that divide n.

Proof. The group of units of \mathcal{AJ}_n is \mathcal{C}_n , and so the description and number of the maximal subsemigroups that arise from the group of units follows Lemmas 5.1 and 4.73. Since $\mathcal{AJ}_n = \langle \mathcal{C}_n, \alpha \rangle$ if and only if $\alpha \in J_{n-2} \cap \mathcal{AJ}_n$, it follows by Corollary 4.74 that the sole remaining maximal subsemigroup is $\mathcal{AJ}_n \setminus J_{n-2}$.

	1	2	3	4	5	6	7	8	9	10	11	12	13	n
\mathcal{J}_n	1	2	5	9	13	19	27	39	57	85	129	199	311	$2F_{n-1} + 2n - 3$
\mathcal{AJ}_n	1	2	2	2	2	3	2	2	2	3	2	3	2	$ \mathbb{P}_n + 1$

Table 5.59: The numbers of maximal subsemigroups of the Jones and annular Jones monoids \mathcal{J}_n and \mathcal{AJ}_n , for $n = 1, \ldots, 13$, along with the general formulae. The sequence $(F_n)_{n \in \mathbb{N}}$ is the Fibonacci sequence, defined in (5.54), and \mathbb{P}_n is the set of all primes dividing n. See Theorems 5.56 and 5.58 for descriptions of the maximal subsemigroups of \mathcal{J}_n and \mathcal{AJ}_n , respectively.

5.3.8 The Motzkin monoid \mathcal{M}_n

In the final section of this chapter, we classify the maximal subsemigroups of the Motzkin monoid, \mathcal{M}_n , which consists of all planar partitions in \mathcal{PB}_n . We first present some prerequisite information about this monoid.

By Lemma 4.68, in order to describe the maximal subsemigroups of \mathcal{M}_n , it is necessary to find the \mathscr{J} -classes that intersect every generating set of \mathcal{M}_n non-trivially. Let $n \in \mathbb{N}$, $n \geq 2$, be arbitrary. By [33, Proposition 4.2], the Motzkin monoid is generated by its elements of rank at least n-2, and any generating set for \mathcal{M}_n contains elements of ranks n, n-1, and n-2. By Lemma 5.43, Green's \mathscr{J} -relation on \mathcal{M}_n is determined by rank, and so the maximal subsemigroups of \mathcal{M}_n arise from the three \mathscr{J} -classes of \mathcal{M}_n that correspond to these ranks.

We therefore require a description of the elements of \mathcal{M}_n that have rank at least n-2. The unique element of rank n in \mathcal{M}_n is id_n . An arbitrary element of rank n-1 in \mathcal{M}_n has trivial kernel and cokernel, and is determined by the unique point that it lacks from its domain and the unique point that it lacks from its codomain. By Lemma 5.43, this describes the \mathscr{L} - and \mathscr{R} -classes of the \mathscr{J} -class $J_{n-1} \cap \mathcal{M}_n$. More specifically, two elements of rank n-1 in \mathcal{M}_n are \mathscr{R} -related if they lack the same point from their domains, and \mathscr{L} -related if they lack the same point from their codomains. Furthermore, every idempotent in $J_{n-1} \cap \mathcal{M}_n$ is a projection: an element of rank n-1 in \mathcal{M}_n is idempotent if and only if its domain and codomain are equal, and so every idempotent in $J_{n-1} \cap \mathcal{M}_n$ is fixed by the * operation. Therefore, we may use Corollary 4.82 to describe the maximal subsemigroups that arise from this \mathscr{J} -class.

We use Lemma 4.8 to describe the maximal subsemigroups of \mathcal{M}_n that arise from the \mathcal{J} class $J_{n-2} \cap \mathcal{M}_n$. To apply this lemma, we require [33, Lemma 4.11]. In the case that r = n-1, this lemma states that one generating set for \mathcal{M}_n consists of id_n , the elements of rank n-1, and the projections of rank n-2 that contain no singleton blocks. We also require [33, Lemmas 4.3 and 4.4], which states that the elements of rank n-1 in \mathcal{M}_n generate those elements of rank n-2 whose non-transverse blocks are singletons.

Theorem 5.60. Let $n \in \mathbb{N}$, $n \geq 2$, be arbitrary. The maximal subsemigroups of \mathcal{M}_n are:

- (a) $\mathcal{M}_n \setminus {\mathrm{id}_n}$ (type (M5));
- (b) The union

$$\left(\mathcal{M}_n \setminus J_{n-1}\right) \cup \bigcup_{i \in A} \left\{ \alpha \in \mathcal{M}_n : \operatorname{rank}(\alpha) = n-1 \text{ and } \{i\} \text{ is a block of } \alpha \right\}$$
$$\cup \bigcup_{i \notin A} \left\{ \alpha \in \mathcal{M}_n : \operatorname{rank}(\alpha) = n-1 \text{ and } \{i'\} \text{ is a block of } \alpha \right\},$$

where A is any non-empty proper subset of $\{1, \ldots, n\}$ (type (M2));

- (c) $\mathcal{M}_n \setminus \{ \alpha \in J_{n-2} : \{i', (i+1)'\} \text{ is a block of } \alpha \} \text{ for } i \in \{1, \dots, n-1\} \text{ (type (M3))}; \text{ and } i \in \{1, \dots, n-1\} \text{ (type (M3))}; \text{ (type (M3)$
- (d) $\mathcal{M}_n \setminus \{ \alpha \in J_{n-2} : \{i, i+1\} \text{ is a block of } \alpha \} \text{ for } i \in \{1, ..., n-1\} \text{ (type (M4))}.$

In particular, for $n \geq 2$, there are $2^n + 2n - 3$ maximal subsemigroups of \mathcal{M}_n .

	1	2	3	4	5	6	7	8	9	10	11	12	n
\mathcal{M}_n	2	5	11	21	39	73	139	269	527	1041	2067	4117	$2^n + 2n - 3$

Table 5.61: The number of maximal subsemigroups of the Motzkin monoid \mathcal{M}_n , for $n = 1, \ldots, 12$, along with the general formula. See Theorem 5.60 for a description of the maximal subsemigroups of \mathcal{M}_n .

Proof. The group of units of \mathcal{M}_n is the trivial group $\{\mathrm{id}_n\}$, and so by Lemma 4.73, $\mathcal{M}_n \setminus \{\mathrm{id}_n\}$ is the unique maximal subsemigroup to arise from the group of units.

Since the Motzkin monoid consists of planar partitions, it is \mathscr{H} -trivial. By Lemma 4.49(a), there are no maximal subsemigroups of \mathcal{M}_n of type (M1). Given the earlier description of the \mathscr{L} - and \mathscr{R} -classes in $J_{n-1} \cap \mathcal{M}_n$, it follows by Corollary 4.82 that the maximal subsemigroups that arise from the \mathscr{J} -class of rank n-1 elements are those described in the theorem of type (M2), and that there are $2^n - 2$ of them.

It remains to describe the maximal subsemigroups that arise from the \mathcal{J} -class that consists of rank n-2 elements. For $i \in \{1, \ldots, n-1\}$, define the subsets

$$X_i = \{ \alpha \in \mathcal{M}_n : \operatorname{rank}(\alpha) = n - 2 \text{ and } \{i, i+1\} \text{ is a block of } \alpha \}, \text{ and}$$
$$X_i^* = \{ \alpha^* : \alpha \in X_i \} = \{ \alpha \in \mathcal{M}_n : \operatorname{rank}(\alpha) = n - 2 \text{ and } \{i', (i+1)'\} \text{ is a block of } \alpha \}$$

of the \mathscr{J} -class $J_{n-2} \cap \mathcal{M}_n$. Note that X_i is an \mathscr{R} -class of \mathcal{M}_n , and X_i^* is an \mathscr{L} -class of \mathcal{M}_n .

Let A be any subset of $J_{n-2} \cap \mathcal{M}_n$ such that $(\mathcal{M}_n \setminus J_{n-2}) \cup A$ generates \mathcal{M}_n , and let $i \in \{1, \ldots, n-1\}$ and $\alpha \in X_i$ be arbitrary. Then α can be written as a product $\alpha = \beta_1 \cdots \beta_k$ of the generators in $(\mathcal{M}_n \setminus J_{n-2}) \cup A$; moreover, we may assume that $\operatorname{rank}(\beta_j) \in \{n-1, n-2\}$ for each β_j ; see Lemma 1.12. If $\operatorname{rank}(\beta_1) = n-1$, then β_1 , and therefore each of its right multiples, contains the singleton block $\{j\}$ for some $j \in \{1, \ldots, n\}$. However, α is a right multiple of β_1 , and α contains no such block. Hence $\operatorname{rank}(\beta_1) = n-2 = \operatorname{rank}(\alpha)$. Lemmas 1.10 and 5.43 imply that $\operatorname{ker}(\alpha) = \operatorname{ker}(\beta_1)$, i.e. $\beta_1 \in A \cap X_i$. A dual argument shows that $A \cap X_i^* \neq \emptyset$.

Conversely, let A be any subset of $J_{n-2} \cap \mathcal{M}_n$ that intersects X_i and X_i^* non-trivially for all $i \in \{1, \ldots, n-1\}$. Let $i \in \{1, \ldots, n-1\}$ be arbitrary, and define ξ_i to be the projection of rank n-2 in \mathcal{M}_n whose non-transverse blocks are $\{i, i+1\}$ and $\{i', (i+1)'\}$. We aim to show that $\xi_i \in \langle \mathcal{M}_n \setminus J_{n-2}, A \rangle$. Fix $\alpha \in X_i \cap A$ and $\beta \in X_i^* \cap A$. Since α and β have rank n-2, codom(α) = $\{j'_1 > j'_2 > \cdots > j'_{n-2}\}$ and dom(β) = $\{k_1 < k_2 < \cdots < k_{n-2}\}$, for some points j_1, \ldots, j_{n-2} and $k_1, \ldots, k_{n-2} \in \{1, \ldots, n\}$, and using the canonical ordering defined in (5.40). Define γ to be the partition of degree n whose transverse blocks are $\{j_1, k'_1\}, \ldots, \{j_{n-2}, k'_{n-2}\}$, and whose non-transverse blocks are singletons. Then γ has rank n-2 and is planar. The non-transverse blocks of γ are singletons, and so by the earlier discussion, it follows that γ is a product of elements of rank n-1 in \mathcal{M}_n . In particular, $\gamma \in \langle \mathcal{M}_n \setminus J_{n-2}, A \rangle$, and since clearly $\xi_i = \alpha \gamma \beta$, it follows that $x_i \in \langle \mathcal{M}_n \setminus J_{n-2}, A \rangle$. Since $i \in \{1, \ldots, n-1\}$ was arbitrary, we deduce that $\langle \mathcal{M}_n \setminus J_{n-2}, A \rangle$ contains every projection in $J_{n-2} \cap \mathcal{M}_n$ that contains no singleton blocks. Since \mathcal{M}_n is generated by these projections and its elements of ranks n-1 and n, it follows that $\mathcal{M}_n = \langle \mathcal{M}_n \setminus J_{n-2}, A \rangle$. By Lemma 4.8, the maximal subsemigroups of \mathcal{M}_n arising from its \mathscr{J} -class of rank n-2 are the sets $\mathcal{M}_n \setminus X_i$ and $\mathcal{M}_n \setminus X_i^*$ for each $i \in \{1, \ldots, n-1\}$. \Box

Monoid	Group of units	Number of maximal subsemigroups	OEIS [120]	Result	Compare with
\mathcal{POI}_n	Trivial	$2^{n} - 1$	A000225	Theorem 5.29	[56 , Theorem 2]
\mathcal{PO}_n		$2^n + 2n - 2$	A131520	Theorem 5.12	[31, Theorem 1]
\mathcal{M}_n		$2^n + 2n - 3$	A131898	Theorem 5.60	
\mathcal{O}_n		$A_{2n-1} + 2n - 4$	A000931	Theorem 5.26	[25 , Theorem 2]
\mathcal{J}_n		$2F_{n-1} + 2n - 3$	A290140	Theorem 5.56	
\mathcal{PP}_n		$2F_{2n-1} + 4n - 3$	A290140	Theorem 5.56	
\mathcal{PODI}_{2n}	Order 2	$3 \cdot 2^{n-1} - 1$	A052955	Theorem 5.30	[29, Theorem 4]
\mathcal{PODI}_{2n-}	1	$2^n - 1$	A052955	Theorem 5.30	[29 , Theorem 4]
\mathcal{POD}_n		$2^{\lceil n/2\rceil} + n - 1$	A016116	Theorem 5.13	_
\mathcal{OD}_n		$A_n + n - 3$	A000931	Theorem 5.27	[68, Theorem 2]
\mathcal{POPI}_n	\mathcal{C}_n	$ \mathbb{P}_n + \mathbb{P}_{n-1} $	A059957	Theorem 5.35	
\mathcal{POP}_n		$ \mathbb{P}_n + 2$	A083399	Theorem 5.32	
\mathcal{AJ}_n		$ \mathbb{P}_n + 1$	A083399	Theorem 5.58	
\mathcal{OP}_n		$ \mathbb{P}_n + 1$	A083399	Theorem 5.33	[28, Theorem 1.6]
\mathcal{PORI}_n	\mathcal{D}_n	$1 + \mathbb{P}_{n-1} + \sum_{p \in \mathbb{P}_n} p$	A290289	Theorem 5.36	
\mathcal{POR}_n		$3 + \sum_{p \in \mathbb{P}_n} p$	A008472	Theorem 5.32	
\mathcal{OR}_n		$2 + \sum_{p \in \mathbb{P}_n}^{r \in \mathbb{P}_n} p$	A008472	Theorem 5.33	[28, Theorem 2.6]
\mathcal{T}_n	\mathcal{S}_n	$s_n + 1$	A290138	Theorem 5.7	
\mathcal{I}_n		$s_n + 1$	A290138	Theorem 5.7	
\mathcal{I}_n^*		$s_n + 1$	A290138	Theorem 5.50	[94, Theorem 19]
\mathfrak{F}_n		$s_n + 1$	A290138	Theorem 5.48	
\mathcal{B}_n		$s_n + 1$	A290138	Theorem 5.47	
\mathcal{PT}_n		$s_n + 2$	A290138	Theorem 5.7	
\mathcal{PB}_n		$s_n + 3$	A290138	Theorem 5.49	_

5.4 Table of results

 \mathcal{P}_n

Table 5.62: Information about the maximal subsemigroups of the monoids considered in this chapter, where $n \in \mathbb{N}$ is sufficiently large (usually $n \geq 2$ or $n \geq 3$; see the relevant theorem for the precise value). The maximal subsemigroups themeselves are described in the referenced theorems. For $k \in \mathbb{N}$, \mathbb{P}_k is the set of primes that divide k; A_k is the k^{th} term of the sequence defined in (5.25); F_k is the k^{th} Fibonacci number, defined in (5.54) with $F_1 = F_2 = 1$; and s_k is the number of maximal subgroups of S_k , the symmetric group of degree k.

A290138

Theorem 5.45

For small values of n, these numbers of maximal subsemigroups may be verified by applying the function NrMaximalSubsemigroups (which is described in Section 4.6.1) from the SEMI-GROUPS package [101] for GAP [58] to the appropriate monoid. For example, by executing

List([1 .. 10], n -> NrMaximalSubsemigroups(PORI(n)));

 $s_n + 4$

List([1 .. 10], n -> NrMaximalSubsemigroups(JonesMonoid(n))); or

List([1 .. 10], n -> NrMaximalSubsemigroups(PartialTransformationMonoid(n)));

one can produce a list of the number of non-empty maximal subsemigroups of \mathcal{PORI}_n , \mathcal{J}_n , or \mathcal{PT}_n , respectively, for n = 1, ..., 10.
Appendix A

Open problems

In this appendix, we collate the open problems that we posed elsewhere in the text.

Open Problem 2.2 (Page 40). Develop techniques for computing the Green's structure of a direct product of finite semigroups in terms of the Green's structures and other semigroup-theoretic properties of the factors.

Open Problem 2.3 (Page 40). Investigate how the maximal subsemigroups of a direct product of finite semigroups relate to the maximal subsemigroups of the factors.

Open Problem 2.34 (Page 58). Generalise Theorem 2.31 and Corollary 2.33 for direct products of an arbitrary number of finitely generated surjective semigroups. In particular, develop an upper bound for the rank of such a direct product that involves the ranks and the numbers of maximal \mathscr{L} - and \mathscr{R} -classes of the factors.

Open Problem 2.35 (Page 58). Let S be an arbitrary finite semigroup. Develop practical methods for computing whether there exist non-trivial semigroups T and U such that $S \cong T \times U$; given this, develop methods for finding such semigroups T and U when they exist.

Open Problem 3.13 (Page 68). Let S be a finite Rees 0-matrix semigroup over an arbitrary finite semigroup T, and assume that any necessary semigroup-theoretic properties of T are known *a priori*. Give algorithms for counting and listing the idempotents of S, using the properties of T and the Graham-Houghton graph of S.

Open Problem 3.14 (Page 68). Let S be a finite Rees 0-matrix semigroup over an arbitrary finite semigroup T, and assume that any necessary semigroup-theoretic properties of T are known a priori. Give an algorithm for constructing a generating set for F(S) that contains at most a constant multiple of rank(F(S)) elements.

Open Problem 3.29 (Page 73). Let S be a finite semigroup to which the techniques of [37] apply. Building on the results of [37], develop methods for computing the Green's structure of an arbitrary left or right ideal of S that do not necessarily exhaustively enumerate the ideal.

Open Problem 3.30 (Page 73). Let S be a finite semigroup to which the techniques of [37] apply. Building on the results of [37], develop methods for computing the partial orders of \mathscr{L} -and \mathscr{R} -classes of S without necessarily exhaustively enumerating S.

Open Problem 3.46 (Page 80). Let $S = \mathscr{M}^0[T; I, \Lambda; P]$ be an arbitrary finite Rees 0-matrix semigroup where T is a monoid and P contains a unit. Give a formula that describes rank(S)

in terms of $\operatorname{rank}(T)$, the semigroup-theoretic properties of T, and the matrix P.

Open Problem 4.51 (Page 118). If A_L and A_R are arbitrary finite acyclic digraphs, does there exist a finite semigroup S with a regular \mathscr{J} -class J such that $\Gamma_{\mathscr{L}}(S, J) \cong A_L$ and $\Gamma_{\mathscr{R}}(S, J) \cong A_R$?

Open Problem 4.87 (Page 136). Develop tools for computing subsemigroups that are maximal with respect to some property, such as maximal commutative or maximal regular subsemigroups of a finite semigroup; or maximal inverse subsemigroups of a finite inverse semigroup.

Bibliography

- A. Ja. Aizenštat. The defining relations of the endomorphism semigroup of a finite linearly ordered set. Sibirsk. Mat. Ž., 3:161–169, 1962.
- [2] Isabel M. Araújo. Presentations for semigroup constructions and related computational methods. PhD thesis, University of St Andrews, 2000.
- [3] M. Aschbacher and L. Scott. Maximal subgroups of finite groups. J. Algebra, 92(1):44–80, 1985.
- [4] Karl Auinger. Krohn-Rhodes complexity of Brauer type semigroups. Port. Math., 69(4):341–360, 2012.
- [5] H. Ayik and N. Ruškuc. Generators and relations of Rees matrix semigroups. Proc. Edinburgh Math. Soc. (2), 42(3):481–495, 1999.
- [6] Ralph W. Ball. Maximal subgroups of symmetric groups. Trans. Amer. Math. Soc., 121:393–407, 1966.
- [7] Ralph W. Ball. Indices of maximal subgroups of infinite symmetric groups. Proc. Amer. Math. Soc., 19:948–950, 1968.
- [8] James E. Baumgartner, Saharon Shelah, and Simon Thomas. Maximal subgroups of infinite symmetric groups. Notre Dame J. Formal Logic, 34(1):1–11, 1993.
- [9] Georgia Benkart and Tom Halverson. Motzkin algebras. European J. Combin., 36:473– 502, 2014.
- [10] Jan De Beule, Julius Jonušas, James D. Mitchell, Michael Torpey, and Wilf A. Wilson. Digraphs – GAP package, Version 0.13.0, September 2018.
- [11] Peter Biryukov and Valery Mishkin. Set ideals with maximal symmetry group and minimal dynamical systems. Bull. London Math. Soc., 32(1):39–46, 2000.
- [12] Mirjana Borisavljević, Kosta Došen, and Zoran Petrić. Kauffman monoids. J. Knot Theory Ramifications, 11(2):127–143, 2002.
- [13] Wieb Bosma, John Cannon, and Catherine Playoust. The Magma algebra system. I. The user language. J. Symbolic Comput., 24(3-4):235–265, 1997. Computational algebra and number theory (London, 1993).
- [14] Marcus Brazil, Jacinta Covington, Tim Penttila, Cheryl E. Praeger, and Alan R. Woods. Maximal subgroups of infinite symmetric groups. Proc. London Math. Soc. (3), 68(1):77– 111, 1994.
- [15] Coen Bron and Joep Kerbosch. Algorithm 457: finding all cliques of an undirected graph. Comm. ACM, 16(9):575–577, 1973.

- [16] Peter J. Cameron, Maximilien Gadouleau, James D. Mitchell, and Yann Péresse. Chains of subsemigroups. *Israel J. Math.*, 220(1):479–508, 2017.
- [17] John Cannon and Derek F. Holt. Computing maximal subgroups of finite groups. J. Symbolic Comput., 37(5):589–609, 2004.
- [18] John J. Cannon. Computation in finite algebraic structures. PhD thesis, University of Sydney, 1969.
- [19] John J. Cannon. Computing the ideal structure of finite semigroups. Numer. Math., 18:254–266, 1971/72.
- [20] John J. Cannon and Derek F. Holt. Automorphism group computation and isomorphism testing in finite groups. J. Symbolic Comput., 35(3):241–267, 2003.
- [21] J.-M. Champarnaud and G. Hansel. AUTOMATE, a computing package for automata and finite semigroups. J. Symbolic Comput., 12(2):197–220, 1991.
- [22] A. H. Clifford and G. B. Preston. The algebraic theory of semigroups. Vol. I. Mathematical Surveys, No. 7. American Mathematical Society, Providence, R.I., 1961.
- [23] Jacinta Covington, Dugald Macpherson, and Alan Mekler. Some maximal subgroups of infinite symmetric groups. Quart. J. Math. Oxford Ser. (2), 47(187):297–311, 1996.
- [24] Manuel Delgado, Pedro A. García-Sánchez, and José João Morais. NumericalSgps GAP package, Version 1.1.8, June 2018.
- [25] I. Dimitrova and J. Koppitz. On the maximal subsemigroups of some transformation semigroups. Asian-Eur. J. Math., 1(2):189–202, 2008.
- [26] Ilinka Dimitrova. The maximal subsemigroups of the semigroup of all orientationpreserving partial injections. In *Proceedings of the Fourth International Scientific Conference—FMNS2011. Volume 1*, Math. Nat. Sci., pages 97–104. South-West Univ. "Neofit Rilsky", Blagoevgrad, 2011.
- [27] Ilinka Dimitrova. The maximal subsemigroups of the semigroup of all partial orderpreserving isometries. In *Proceedings of the Fifth International Scientific Conference— FMNS2013. Vol. 1. Mathematics and informatics*, Math. Nat. Sci., pages 95–101. South-West Univ. "Neofit Rilsky", Blagoevgrad, 2013.
- [28] Ilinka Dimitrova, Vítor H. Fernandes, and Jörg Koppitz. The maximal subsemigroups of semigroups of transformations preserving or reversing the orientation on a finite chain. *Publ. Math. Debrecen*, 81(1-2):11–29, 2012.
- [29] Ilinka Dimitrova and Jörg Koppitz. The maximal subsemigroups of the ideals of some semigroups of partial injections. Discuss. Math. Gen. Algebra Appl., 29(2):153–167, 2009.
- [30] Ilinka Dimitrova and Jörg Koppitz. The maximal subsemigroups of the ideals of the semigroup of all isotone partial injections. In *Proceedings of the International Scientific Conference. Volume 1*, Math. Nat. Sci., pages 45–49. South-West Univ. "Neofit Rilsky", Blagoevgrad, 2009.
- [31] Ilinka Dimitrova and Tsvetelina Mladenova. Classification of the maximal subsemigroups of the semigroup of all partial order-preserving transformations. Union of Bulgarian Mathematicians, 41(1):158–162, 2012.

- [32] Andreas Distler and James D. Mitchell. Smallsemi GAP package, Version 0.6.11, April 2017.
- [33] Igor Dolinka, James East, and Robert D. Gray. Motzkin monoids and partial Brauer monoids. J. Algebra, 471:251–298, 2017.
- [34] Igor Dolinka, James East, and James D. Mitchell. Idempotent rank in the endomorphism monoid of a nonuniform partition. Bull. Aust. Math. Soc., 93(1):73–91, 2016.
- [35] C. R. Donoven, J. D. Mitchell, and W. A. Wilson. Computing maximal subsemigroups of a finite semigroup. J. Algebra, 505:559–596, 2018.
- [36] Jacques Doyen. Équipotence et unicité de systèmes générateurs minimaux dans certains monoïdes. Semigroup Forum, 28(1-3):341–346, 1984.
- [37] J East, A Egri-Nagy, J. D. Mitchell, and Y Péresse. Computing finite semigroups. J. Symbolic Comput., to appear.
- [38] J. East, J. D. Mitchell, and Y. Péresse. Maximal subsemigroups of the semigroup of all mappings on an infinite set. Trans. Amer. Math. Soc., 367(3):1911–1944, 2015.
- [39] James East. Generators and relations for partition monoids and algebras. J. Algebra, 339:1–26, 2011.
- [40] James East. On the singular part of the partition monoid. Internat. J. Algebra Comput., 21(1-2):147–178, 2011.
- [41] James East, Attila Egri-Nagy, and James D. Mitchell. Enumerating transformation semigroups. Semigroup Forum, 95(1):109–125, 2017.
- [42] James East and D. G. FitzGerald. The semigroup generated by the idempotents of a partition monoid. J. Algebra, 372:108–133, 2012.
- [43] James East and Robert D. Gray. Diagram monoids and Graham-Houghton graphs: idempotents and generating sets of ideals. J. Combin. Theory Ser. A, 146:63–128, 2017.
- [44] James East, Jitender Kumar, James D. Mitchell, and Wilf A. Wilson. Maximal subsemigroups of finite transformation and diagram monoids. 2017. https://arxiv.org/abs/ 1706.04967.
- [45] James East, Jitender Kumar, James D. Mitchell, and Wilf A. Wilson. Maximal subsemigroups of finite transformation and diagram monoids. J. Algebra, 504:176–216, 2018.
- [46] James East, James D. Mitchell, Nik Ruškuc, and Michael Torpey. Congruence lattices of finite diagram monoids. Adv. Math., 333:931–1003, 2018.
- [47] Bettina Eick and Alexander Hulpke. Computing the maximal subgroups of a permutation group. I. In *Groups and computation*, III (Columbus, OH, 1999), volume 8 of Ohio State Univ. Math. Res. Inst. Publ., pages 155–168. de Gruyter, Berlin, 2001.
- [48] V. H. Fernandes. The monoid of all injective order preserving partial transformations on a finite chain. *Semigroup Forum*, 62(2):178–204, 2001.
- [49] Vítor H. Fernandes, Gracinda M. S. Gomes, and Manuel M. Jesus. Presentations for some monoids of partial transformations on a finite chain. Comm. Algebra, 33(2):587–604, 2005.

- [50] D. G. FitzGerald. A presentation for the monoid of uniform block permutations. Bull. Austral. Math. Soc., 68(2):317–324, 2003.
- [51] D. G. FitzGerald and Kwok Wai Lau. On the partition monoid and some related semigroups. Bull. Aust. Math. Soc., 83(2):273–288, 2011.
- [52] D. G. FitzGerald and Jonathan Leech. Dual symmetric inverse monoids and representation theory. J. Austral. Math. Soc. Ser. A, 64(3):345–367, 1998.
- [53] George E. Forsythe. SWAC computes 126 distinct semigroups of order 4. Proc. Amer. Math. Soc., 6:443–447, 1955.
- [54] Véronique Froidure and Jean-Eric Pin. Algorithms for computing finite semigroups. In Foundations of computational mathematics (Rio de Janeiro, 1997), pages 112–126. Springer, Berlin, 1997.
- [55] Harold N. Gabow. Path-based depth-first search for strong and biconnected components. Information Processing Letters, 74(3-4):107–114, 2000.
- [56] Olexandr Ganyushkin and Volodymyr Mazorchuk. On the structure of \mathscr{IO}_n . Semigroup Forum, 66(3):455–483, 2003.
- [57] Olexandr Ganyushkin and Volodymyr Mazorchuk. Classical finite transformation semigroups, volume 9 of Algebra and Applications. Springer-Verlag London, Ltd., London, 2009. An introduction.
- [58] The GAP Group. GAP Groups, Algorithms, and Programming, Version 4.10.0, November 2018.
- [59] Michael R. Garey and David S. Johnson. Computers and intractability. W. H. Freeman and Co., San Francisco, Calif., 1979. A guide to the theory of NP-completeness, A Series of Books in the Mathematical Sciences.
- [60] Gracinda M. S. Gomes and John M. Howie. On the ranks of certain semigroups of orderpreserving transformations. *Semigroup Forum*, 45(3):272–282, 1992.
- [61] N. Graham, R. Graham, and J. Rhodes. Maximal subsemigroups of finite semigroups. J. Combinatorial Theory, 4:203–209, 1968.
- [62] R. L. Graham. On finite 0-simple semigroups and graph theory. Math. Systems Theory, 2:325–339, 1968.
- [63] R. Gray and N. Ruškuc. Generating sets of completely 0-simple semigroups. Comm. Algebra, 33(12):4657–4678, 2005.
- [64] Robert D. Gray. A graph theoretic approach to combinatorial problems in semigroup theory. PhD thesis, University of St Andrews, 2006.
- [65] Robert D. Gray. The minimal number of generators of a finite semigroup. Semigroup Forum, 89(1):135–154, 2014.
- [66] J. A. Green. On the structure of semigroups. Ann. of Math. (2), 54:163–172, 1951.
- [67] P.-A. Grillet. Semigroups, volume 193 of Monographs and Textbooks in Pure and Applied Mathematics. Marcel Dekker, Inc., New York, 1995. An introduction to the structure theory.

- [68] I. Gyudzhenov and I. Dimitrova. Maximal subsemigroups of the semigroup of all isotone transformations with defect ≤ 2 . C. R. Acad. Bulgare Sci., 59(3):239–244, 2006.
- [69] Iliya Gyudzhenov and Ilinka Dimitrova. On the maximal subsemigroups of the semigroup of all monotone transformations. *Discuss. Math. Gen. Algebra Appl.*, 26(2):199–217, 2006.
- [70] Tom Halverson and Arun Ram. Partition algebras. European J. Combin., 26(6):869–921, 2005.
- [71] Peter M. Higgins. Techniques of semigroup theory. Oxford Science Publications. The Clarendon Press, Oxford University Press, New York, 1992. With a foreword by G. B. Preston.
- [72] Derek Holt. kbmag GAP package, Version 1.5.6, September 2018.
- [73] Derek F. Holt, Bettina Eick, and Eamonn A. O'Brien. Handbook of computational group theory. Discrete Mathematics and its Applications (Boca Raton). Chapman & Hall/CRC, Boca Raton, FL, 2005.
- [74] Eckehart Hotzel. Maximality properties of some subsemigroups of Baer-Levi semigroups. Semigroup Forum, 51(2):153–190, 1995.
- [75] C. H. Houghton. Completely 0-simple semigroups and their associated graphs and groups. Semigroup Forum, 14(1):41–67, 1977.
- [76] John M. Howie. Fundamentals of semigroup theory, volume 12 of London Mathematical Society Monographs. New Series. The Clarendon Press, Oxford University Press, New York, 1995. Oxford Science Publications.
- [77] Thomas W. Hungerford. Algebra, volume 73 of Graduate Texts in Mathematics. Springer-Verlag, New York-Berlin, 1980. Reprint of the 1974 original.
- [78] J. T. Hyde, N. J. Loughlin, M. Quick, N. Ruškuc, and A. R. Wallis. On the growth of generating sets for direct powers of semigroups. *Semigroup Forum*, 84(1):116–130, 2012.
- [79] J. Jonušas, J. D. Mitchell, and M. Pfeiffer. Two variants of the Froidure–Pin Algorithm for finite semigroups. *Port. Math.*, 74:173–200, 2017.
- [80] Andrei V. Kelarev and Cheryl E. Praeger. On transitive Cayley graphs of groups and semigroups. *European J. Combin.*, 24(1):59–72, 2003.
- [81] Janusz Konieczny. Green's equivalences in finite semigroups of binary relations. Semigroup Forum, 48(2):235–252, 1994.
- [82] Jörg Koppitz and Tiwadee Musunthia. Maximal subsemigroups containing a particular semigroup. Math. Slovaca, 64(6):1369–1380, 2014.
- [83] Gerard Lallement and Robert McFadden. On the determination of Green's relations in finite transformation semigroups. J. Symbolic Comput., 10(5):481–498, 1990.
- [84] Kwok Wai Lau and D. G. FitzGerald. Ideal structure of the Kauffman and related monoids. Comm. Algebra, 34(7):2617–2629, 2006.
- [85] Mark V. Lawson. *Inverse semigroups*. World Scientific Publishing Co., Inc., River Edge, NJ, 1998. The theory of partial symmetries.

- [86] Inessa Levi and G. R. Wood. On maximal subsemigroups of Baer-Levi semigroups. Semigroup Forum, 30(1):99–102, 1984.
- [87] Martin W. Liebeck, Cheryl E. Praeger, and Jan Saxl. A classification of the maximal subgroups of the finite alternating and symmetric groups. J. Algebra, 111(2):365–383, 1987.
- [88] Martin W. Liebeck and Aner Shalev. Maximal subgroups of symmetric groups. J. Combin. Theory Ser. A, 75(2):341–352, 1996.
- [89] S. A. Linton, G. Pfeiffer, E. F. Robertson, and N. Ruškuc. Monoid GAP 3 package, Version 2.4. University of St Andrews, September 1997.
- [90] S. A. Linton, G. Pfeiffer, E. F. Robertson, and N. Ruškuc. Groups and actions in transformation semigroups. *Math. Z.*, 228(3):435–450, 1998.
- [91] S. A. Linton, G. Pfeiffer, E. F. Robertson, and N. Ruškuc. Computing transformation semigroups. J. Symbolic Comput., 33(2):145–162, 2002.
- [92] Dugald Macpherson. Large subgroups of infinite symmetric groups. In Finite and infinite combinatorics in sets and logic (Banff, AB, 1991), volume 411 of NATO Adv. Sci. Inst. Ser. C Math. Phys. Sci., pages 249–278. Kluwer Acad. Publ., Dordrecht, 1993.
- [93] H. D. Macpherson and Peter M. Neumann. Subgroups of infinite symmetric groups. J. London Math. Soc. (2), 42(1):64–84, 1990.
- [94] Victor Maltcev. On a new approach to the dual symmetric inverse monoid \$\mathcal{I}_X^*\$. Internat. J. Algebra Comput., 17(3):567-591, 2007.
- [95] A. Markov. The impossibility of certain algorithms in the theory of associative systems. II. Doklady Akad. Nauk SSSR (N.S.), 58:353–356, 1947.
- [96] Paul Martin and Volodymyr Mazorchuk. Partitioned binary relations. Math. Scand., 113(1):30–52, 2013.
- [97] Volodymyr Mazorchuk. On the structure of Brauer semigroup and its partial analogue. Problems in Algebra, 13:29–45, 1998.
- [98] Volodymyr Mazorchuk. Endomorphisms of \mathfrak{B}_n , \mathscr{PB}_n , and \mathfrak{C}_n . Comm. Algebra, 30(7):3489-3513, 2002.
- [99] D. B. McAlister. Semigroup for Windows. Northern Illinois University, 2006.
- [100] V. Mishkin. Symmetry groups of ideals on ω. In Second International Conference on Algebra (Barnaul, 1991), volume 184 of Contemp. Math., pages 305–316. Amer. Math. Soc., Providence, RI, 1995.
- [101] J. D. Mitchell et al. Semigroups GAP package, Version 3.0.20, October 2018.
- [102] J. D. Mitchell and M. Torpey. libsemigroups C++ library for semigroups and monoids, Version 0.6.4, September 2018.
- [103] J. W. Moon and L. Moser. On cliques in graphs. Israel J. Math., 3:23-28, 1965.
- [104] A. Nagy. Special classes of semigroups. Advances in Mathematics Kluwer Academic Publishers. Springer, 2001.

- [105] T. E. Nordahl and H. E. Scheiblich. Regular *-semigroups. Semigroup Forum, 16(3):369– 377, 1978.
- [106] Jean-François Perrot. Contribution à l'étude des monoïdes syntactiques et de certains groupes associés aux automates finis. Université de Paris VI, Paris, 1972. Thèse présentée à l'Université Paris VI, Paris, pour l'obtention du Doctorat d'État ès Sciences, Spécialité: Mathématiques.
- [107] Jean-Eric Pin. Semigroupe 2.01: a software for computing finite semigroups. Laboratoire d'Informatique Algorithmique: Fondements et Applications (LIAFA), CNRS et Université Paris, 7(2.01), 2009.
- [108] Emil L. Post. Recursive unsolvability of a problem of Thue. J. Symbolic Logic, 12:1–11, 1947.
- [109] Cheryl E. Praeger. An O'Nan-Scott theorem for finite quasiprimitive permutation groups and an application to 2-arc transitive graphs. J. London Math. Soc. (2), 47(2):227–239, 1993.
- [110] John Rhodes and Benjamin Steinberg. The q-theory of finite semigroups. Springer Monographs in Mathematics. Springer, New York, 2009.
- [111] Fred Richman. Maximal subgroups of infinite symmetric groups. Canad. Math. Bull., 10:375–381, 1967.
- [112] E. F. Robertson, N. Ruškuc, and M. R. Thomson. On finite generation and other finiteness conditions for wreath products of semigroups. *Comm. Algebra*, 30(8):3851–3873, 2002.
- [113] E. F. Robertson, N. Ruškuc, and J. Wiegold. Generators and relations of direct products of semigroups. *Trans. Amer. Math. Soc.*, 350(7):2665–2685, 1998.
- [114] N. Ruškuc. On the rank of completely 0-simple semigroups. Math. Proc. Cambridge Philos. Soc., 116(2):325–338, 1994.
- [115] Nik Ruškuc. Semigroup presentations. PhD thesis, University of St Andrews, 1995.
- [116] Leonard L. Scott. Representations in characteristic p. In The Santa Cruz Conference on Finite Groups (Univ. California, Santa Cruz, Calif., 1979), volume 37 of Proc. Sympos. Pure Math., pages 319–331. Amer. Math. Soc., Providence, R.I., 1980.
- [117] Robert Sedgewick and Kevin Wayne. Algorithms, 4th Edition. Addison-Wesley, 2011.
- [118] Akos Seress. Permutation group algorithms, volume 152. Cambridge University Press, 2003.
- [119] Charles C Sims. Computation with finitely presented groups, volume 48. Cambridge University Press, 1994.
- [120] N. J. A. Sloane. The On-Line Encyclopedia of Integer Sequences. http://oeis.org.
- [121] Robert Tarjan. Depth-first search and linear graph algorithms. SIAM J. Comput., 1(2):146–160, 1972.
- [122] The Sage Developers. SageMath, the Sage Mathematics Software System (Version 8.4), 2018. http://www.sagemath.org.

- [123] Jacques Thévenaz. Maximal subgroups of direct products. J. Algebra, 198(2):352–361, 1997.
- [124] Kalcho Todorov and Ilinka Dimitrova. On some maximal subsemigroups of the symmetric semigroup. Math. Balkanica (N.S.), 19(3-4):427–434, 2005.
- [125] Etsuji Tomita, Akira Tanaka, and Haruhisa Takahashi. The worst-case time complexity for generating all maximal cliques and computational experiments. *Theoret. Comput. Sci.*, 363(1):28–42, 2006.
- [126] A. M. Turing. On Computable Numbers, with an Application to the Entscheidungsproblem. Proc. London Math. Soc. (2), 42(3):230–265, 1936.
- [127] Stewart Wilcox. Cellularity of diagram algebras as twisted semigroup algebras. J. Algebra, 309(1):10–31, 2007.
- [128] Herbert S. Wilf. Algorithms and complexity. A K Peters, Ltd., Natick, MA, second edition, 2002.
- [129] Hao Bo Yang and Xiu Liang Yang. Maximal subsemigroups of finite transformation semigroups K(n, r). Acta Math. Sin. (Engl. Ser.), 20(3):475–482, 2004.
- [130] Xiuliang Yang. A classification of maximal subsemigroups of finite order-preserving transformation semigroups. Comm. Algebra, 28(3):1503–1513, 2000.
- [131] Xiuliang Yang. Maximal subsemigroups of the finite singular transformation semigroup. Comm. Algebra, 29(3):1175–1182, 2001.

Table of notation

This table lists some of the mathematical notation that is used in this thesis. In any row, the first column gives an example of some notation, the second column gives a brief description of what the notation represents, and the third column gives the page number where the notation is defined or first used in the main text. In the original digital version of this document, many instances of such notation are hyperlinks to their corresponding entries in this table.

0_S	The multiplicative zero of the semigroup-with-zero S .	23
1_S	The identity element of the monoid S^1 .	24
Ø	The empty set.	20
X	The cardinality of the set X .	20
X^+	The free semigroup over the set X .	23
$\langle X \rangle$	The subsemigroup of a semigroup generated by its subset X .	24
$\langle X R \rangle$	The semigroup presentation with generators X and relations R .	23
$\lceil x \rceil$	The least integer greater than or equal to the real number x .	20
$\lfloor x \rfloor$	The greatest integer less than or equal to the real number x .	20
x^*	The image of x under the $*$ operation of a regular $*$ -semigroup.	29
x^{-1}	The unique inverse of an element x in a semigroup.	29
x^n	The element $\underline{xx} \cdots \underline{x}$.	23
	$\overline{n \text{ times}}$	
\mathcal{AJ}_n	The annular Jones monoid of degree n .	167
\mathcal{A}_n	The alternating group, consisting of all even permutations of degree n .	108
$(A_n)_{n\in\mathbb{N}}$	The sequence $A_1 = 1$, $A_2 = A_3 = 2$, and $A_n = A_{n-2} + A_{n-3}$ for $n > 3$.	157
B(G, I)	The $I \times I$ Brandt semigroup over the group G.	85
B(G, n)	The $n \times n$ Brandt semigroup $B(G, \{1, \ldots, n\})$.	85
\mathcal{B}_n	The Brauer monoid of degree n .	166
Cm	The cyclic group of order n generated by the <i>n</i> -cycle $(1, 2, \ldots, n)$.	148
$\operatorname{codom}(\alpha)$	The codomain of the partition α .	168
$\operatorname{coker}(\alpha)$	The cokernel of the partition α .	168
Ø	Croop's a relation	26
<i>ツ</i> D	The permutation group $/(1, 2, \dots, n) = (1, n)(2, n, -1) = (\lfloor n/2 \rfloor \lceil n/2 \rceil)$	20
\mathcal{D}_n	which is a dihedral group of order $2n$ when $n > 3$.	140
$\operatorname{dom}(\alpha)$	The domain of the partial transformation, or partition, α .	20
$\Delta = \Delta(S, J)$	The graph defined in §4.4.2, where S is a semigroup and $J \in S/\mathscr{J}$.	118
$\Delta(S)$	The graph $\Delta(S, J)$, where S is a monoid with a unique \mathcal{J} -class J covered by its group of units.	150
E(X)	The set of idempotents contained in a subset X of a semigroup.	23

$(F_n)_{n\in\mathbb{N}}$	The Fibonacci sequence [120, A000045] with $F_1 = F_2 = 1$.	174
\mathfrak{F}_n	The uniform block bijection monoid of degree n .	166
F(X)	The subsemigroup $\langle E(X) \rangle$ generated by the idempotents of X.	24
$\Gamma_{\mathscr{L}} = \Gamma_{\mathscr{L}}(S, J)$	A digraph representing the action of $S \setminus J$ on the \mathscr{L} -classes of J .	118
$\Gamma_{\mathscr{R}} = \Gamma_{\mathscr{R}}(S, J)$	A digraph representing the action of $S \setminus J$ on the \mathscr{R} -classes of J .	118
[G:V]	The index of the subgroup V in the group G .	29
${\mathcal H}$	Green's \mathscr{H} -relation.	26
id_n	The identity permutation of degree n .	30
$\operatorname{im}(\alpha)$	The image of the partial function α .	20
\mathcal{I}_n	The symmetric inverse monoid of degree n .	30
\mathcal{I}_n^*	The dual symmetric inverse monoid of degree n .	166
J^*	The principal factor of the Green's \mathcal{J} -class J .	28
J	Green's \mathcal{J} -relation.	26
J_k	The \mathscr{J} -class of \mathcal{PT}_n or \mathcal{P}_n consisting of all elements of rank k .	149
\mathcal{J}_n	The Jones monoid of degree n .	138
$\ker(\alpha)$	The kernel of the partial function, or partition, α .	20
\mathscr{K}^{S}	Green's \mathscr{K} -relation on the semigroup S , for $\mathscr{K} \in \{\mathscr{H}, \mathscr{L}, \mathscr{R}, \mathscr{D}, \mathscr{J}\}.$	27
K_x	The Green's \mathscr{K} -class of x in some semigroup.	27
K_x^S	The Green's \mathscr{K} -class of x in the semigroup S .	27
L	Green's \mathscr{L} -relation.	26
$\mathscr{M}[T; I, \Lambda; P]$	The $I \times \Lambda$ Rees matrix semigroup over T with matrix P.	25
$\mathscr{M}^0[T; I, \Lambda; P]$	The $I \times \Lambda$ Rees 0-matrix semigroup over T with matrix P.	26
\mathcal{M}_n	The Motzkin monoid of degree n .	166
\mathbb{N}	The natural numbers $\{1, 2, 3, \ldots\}$.	20
\mathbb{N}_0	The non-negative integers $\{0, 1, 2, 3, \ldots\}$.	20
$N_G(V)$	The normalizer of the subgroup V in the group G .	29
$O(n^2)$	Big O notation, used to describe the complexity of an algorithm.	32
\mathcal{OD}_n	All order-preserving or -reversing transformations of degree n .	148
\mathcal{O}_n	All order-preserving transformations of degree n .	148
\mathcal{OP}_n	All orientation-preserving transformations of degree n .	148
\mathcal{OR}_n	All orientation-preserving or -reversing transformations of degree n .	148
\mathcal{PB}_n	The partial Brauer monoid of degree n .	166
\mathcal{P}_n	The partition monoid of degree n .	165
\mathbb{P}_n	The set of primes that divide $n \in \mathbb{N}$.	162
\mathcal{PODI}_n	All order-preserving or -reversing partial permutations of degree n .	148
\mathcal{POD}_n	All order-preserving or -reversing partial transformations of degree n .	147
\mathcal{POI}_n	All order-preserving partial permutations of degree n .	148
\mathcal{PO}_n	All order-preserving partial transformations of degree n .	147

\mathcal{POPI}_n	All orientation-preserving partial permutations of degree n .	148
\mathcal{POP}_n	All orientation-preserving partial transformations of degree n .	148
\mathcal{PORI}_n	All orientation-preserving or -reversing partial permutations of degree n	. 138
\mathcal{POR}_n	All orientation-preserving or -reversing partial transformations in \mathcal{PT}_n .	148
\mathcal{PP}_n	The planar partition monoid of degree n .	166
\mathcal{PT}_n	The partial transformation monoid of degree n .	30
\mathscr{R}	Green's <i>R</i> -relation.	26
\mathbb{R}	The real numbers.	22
$\operatorname{rank}(\alpha)$	The rank of the semigroup element α .	20
$\operatorname{rank}(S)$	The least cardinality of a generating set for the semigroup S .	24
S^0	The semigroup formed by adjoining a zero to the semigroup S .	24
S^1	For a semigroup S, if S is a monoid, then $S^1 = S$, else S^1 is the	24
	monoid formed by adjoining the identity element 1_S to S .	
\mathcal{S}_n	The symmetric group of degree n .	30
s_n	The number of maximal subgroups of \mathcal{S}_n [120, A290138].	146
$\operatorname{Stab}_G(A)$	The setwise stabilizer $\{g \in G : Ag = A\}$ of A in the group G.	133
$\operatorname{SUB}(V, g_1, \ldots, g_n)$	The subset of a normalized finite 0-simple semigroup defined in (4.19) .	102
\mathcal{T}_n	The full transformation monoid of degree n .	30
$\Theta=\Theta(S,J)$	The graph defined in §4.4.2, where S is a semigroup and $J \in S/\mathscr{J}$.	118
\mathbb{Z}	The integers.	20

Index

0-simple semigroup, 25

acyclic digraph, 22 algorithm, 31 annular partition, 166 anti-homomorphism, 23 anti-isomorphism, 23 associativity, 23 band, 23big O notation, 32 bijection, 20 bipartite graph, 21 complete bipartite, 21 block, 165 block bijection, 165 Brandt semigroup, 85 clique, 21 maximal clique, 21 closed (under multiplication), 23 codomain, 168 cokernel, 168 colouring, 21 column-regular matrix, 71 commutative semigroup, 23 complement of a digraph, 22of a graph, 22completely 0-simple semigroup, 25 completely simple semigroup, 25 complexity, 31 composition of partial functions, 20 of relations, 19 computational group theory, 16 computational semigroup theory, 15 congruence, 23 conjugacy class of a subgroup, 29 conjugation, 29 connected component, 22graph, 22 covered (in a partial order), 19

cycle in a digraph, 22in a graph, 22 decomposable element, 42, 78, 94 degree of a partial transformation, 20 of a vertex, 21 digraph, 21 disjoint cycle notation, 20 domain of a function, 20of a partition, 168 edge of a digraph, 21 of a graph, 20embedding, 23 equivalence, 19 class, 19 relation, 19 factor (of a direct product), 39 factorization, 43 non-trivial, 43 Fibonacci sequence, 174 finitely generated semigroup, 24 finitely presented semigroup, 23 forest, 22 free semigroup, 23 function, 20bijective, 20 domain, 20image, 20injective, 20 kernel, 20partial, 20 surjective, 20generating set, 24Graham normal form, 62 graph, 20Green's class, 27

partial order of, 27

Green's relations, 26 Green's \mathcal{D} -relation, 26 Green's \mathscr{H} -relation, 26 Green's \mathcal{J} -relation, 26 Green's \mathscr{L} -relation, 26 Green's \mathcal{R} -relation, 26 group, 24, 29 group of units, 24 normal subgroup, 29 homomorphism, 23 ideal, 24left, 24 principal, 24 proper, 25 right, 24 two-sided, 24 idempotent, 23 idempotent generated subsemigroup, 24 identity, 23 left identity, 23 right identity, 23 identity partition, 165 identity permutation, 30 image of a function, 20of a point under a function, 20indecomposable element, 42, 78 independent subset, 21 maximal independent subset, 21 index of a subgroup in a group, 29 induced subdigraph, 22 subgraph, 22 inverse of a partial function, 20 of a relation, 19 of an element, 29semigroup, 29 isomorphism of digraphs, 22 of graphs, 22 of semigroups, 23 kernel of a function, 20of a partition, 168 left Cayley digraph, 35

left congruence, 23 left ideal, principal left ideal, 24 left identity, 23 left zero semigroup, 23 loop (in a digraph), 21 maximal in a partial order, 19 subgroup, 28, 91 subsemigroup, 91 arising from J, 94 minimal generating set, 24in a partial order, 19 monogenic semigroup, 24 monoid, 24 multiplication table, 33 multiplicative zero, 23 natural partial order, 88 neighbour, 21 in-neighbour, 21 out-neighbour, 21 non-regular Green's class, 28 normalization, 62normalized, 62normalizer, 107 orbit, 30order -preserving, 147 -reversing, 147 of a semigroup, 22 orientation -preserving, 147 -reversing, 147 partial function, 20permutation, 20 partial order, 19 greater than, 19 less than, 19 of Green's *J*-classes, 27 of Green's \mathscr{L} -classes, 27 of Green's \mathcal{R} -classes, 27 partially ordered set, 19 partition of a set, 19of degree n, 165

part of a partition, 19 path in a digraph, 21 in a graph, 21 path graph, 157 permutation, 20planar partition, 166 principal factor, 28 product Cartesian, 19 direct, 39 of semigroup elements, 23 set product, 24projection (in a regular *-semigroup), 29 proper, 25 pseudocode, 31 quotient of a digraph, 22 of a graph, 22 of a semigroup, 23rank of a partial transformation, 20 of a partition, 168 of a semigroup, 24reachable, 22 Rees 0-matrix semigroup, 26 Rees matrix semigroup, 25 regular *-semigroup, 29 Green's class, 28 matrix, 72 semigroup, 28 semigroup element, 28 relation, 19 antisymmetric, 19 composition of relations, 19 equivalence, 19 on a set, 19reflexive, 19 symmetric, 19 transitive, 19 relations (in a presentation), 23 relative left identity, 23 relative right identity, 24 right action of a group, 30 right Cayley digraph, 35 right congruence, 23

right coset, 29 right ideal, principal right ideal, 24 right identity, 23 right zero semigroup, 23 row-regular matrix, 71 semigroup, 22 semigroup presentation, 23 setwise stabilizer, 133 simple semigroup, 25sink (in a digraph), 21 source (in a digraph), 21spanning forest, 22tree, 22strongly connected component, 22digraph, 22 subgroup, 24 submonoid, 24 subsemigroup, 24proper subsemigroup, 25 surjective semigroup, 24, 41 time complexity, 31 transformation, 20partial transformation, 20 transitive group action, 30relation, 19 transversal, 19, 106 transverse block, 165 tree, 22two-line notation, 20 uniform block bijection, 165 unit, 24 vertex adjacent vertices, 20 isolated vertex, 21 of a digraph, 21 of a graph, $\underline{20}$ zero (of a semigroup), 23 zero semigroup, 23